

ANNALS OF THE FACULTY OF LAW IN BELGRADE BELGRADE LAW REVIEW

АНАЛИ правног факултета у београду

ЧАСОПИС ЗА ПРАВНЕ И ДРУШТВЕНЕ НАУКЕ

Journal of Legal and Social Sciences University of Belgrade Year LXVII, 2019, No. 4

ISSN 0003-2565 : UDC 34/35

International Editorial Board Members:

Yariv Brauner (University of Florida, USA), John Cerone (Tufts University, USA), Silvio Ferrari (University of Milan, Italy), Kenneth Einar Himma (University of Washington, USA), Christa Jessel Holst (Max Planck Institute for Comparative and International Private Law, Hamburg, Germany), Thomas Koenig (Northeastern University, USA), Peter Koller (University of Graz, Austria), Janez Kranjc (University of Ljubljana, Slovenia), Werner Krawietz (University of Münster, Germany), Rainer Kulms (Max Planck Institute for Comparative and International Private Law, Hamburg, Germany), Dzhenevra Igorevna Lukovskaya (St Petersburg State University, Russian Federation), Ingeborg Maus (Goethe University Frankfurt, Germany), Thomas Mertens (Radboud University, Netherlands), Slobodan Milačić (University of Bordeaux, France), Paul du Plessis (The University of Edinburgh, Scotland), Vesna Rijavec (University of Maribor, Slovenia), Željko Šević (Sohar University, Oman), Gerhard Thür (Institute for the Study of Ancient Culture, Austria)

University of Belgrade Editorial Board Members:

Zoran Tomić, Nebojša Jovanović, Goran Ilić, Bojan Milisavljević, Vojislav Stanimirović, Danilo Vuković

Editors-in-Chief: Marija Karanikić Mirić Svetislav Kostić Yariy Brauner

Deputy Editor-in-Chief:

Boris Begović

Managing Editors:

Miloš Vukotić Nikola Ilić Novak Vujičić

Proofreading:

Vuk Tošić

Publisher:

University of Belgrade Faculty of Law

Layout Manager:

Jovan Đorđević

Cover Design, Type Setting and Layout:

Dosije studio, Belgrade

Printed by:

Službeni glasnik, Belgrade



ANNALS

OF THE FACULTY OF LAW IN BELGRADE

BELGRADE LAW REVIEW

АНАЛИ

ПРАВНОГ ФАКУЛТЕТА У БЕОГРАДУ

ЧАСОПИС ЗА ПРАВНЕ И ДРУШТВЕНЕ НАУКЕ

Journal of Legal and Social Sciences University of Belgrade Year LXVII, 2019, No. 4

ISSN 0003-2565 : UDC 34/35

Year LXVII, 2019, No. 4, pp. 1-291

UDC 34/35 ISSN 0003-2565

TABLE OF CONTENTS

ARTICLES

Andrés Báez Moreno, Hugo López López, Brain Drain Tax vs. Brain Gain Benefits: General Thoughts from a Spanish Perspective ————————————————————————————————————	5
B. Bawono Kristiaji, Tax and Brain Drain: Justification, Policy Options and Prospect for Large Developing Economies	17
Giorgio Beretta, 'Fixing' the Social Contract: A Blueprint for Individual Tax Reform	68
Stjepan Gadžo, Using Tax Policy to Address Brain Drain and Depopulation: The Case of Croatia	116
Ricardo García Antón, Restoring the EU Citizenship for Tax Purposes	147
Iva Ivanov, Tax Incentives for Keeping and Attracting Highly Skilled Workers: The Case of Serbia	173
Nikolai Milogolov, Azamat Berberov, Brain Drain and Tax Competition: Do We Need Another BEPS?	200

Katerina Perrou, Katerina Savvaidou, Brain Drain: The Impact on Taxation and Measures to Combat the Brain	••
Drain	238
Fernando Souza de Man, Tax Measures to Combat Brain Drain: (In)Compatibility Issues with Double Tax Conventions	
and a Potential Way Forward	249
Guidelines for Authors	283

ARTICLES

UDC 336.22:314.151.3-054.73(460)

CERIF: S141

DOI: 10.5937/AnaliPFB1904005B

Andrés Báez Moreno, PhD*
Hugo López López, PhD**

BRAIN DRAIN TAX VS. BRAIN GAIN BENEFITS: GENERAL THOUGHTS FROM A SPANISH PERSPECTIVE

One might expect that tax benefits introduced by certain developed countries to attract foreign high-skilled workers would run contrary to taxes aimed at the alleviation or deterrence of so-called brain drain. This article shows, however, that this is not necessarily the case: brain gain benefits might be justified by the existence of home-state brain drain taxes and, at the same time, serve to alleviate international double taxation generated by this type of taxes.

Key words: Brain drain tax. – Brain gain tax benefits. – High-skilled workers taxation. – Highly mobile individuals taxation. – Inpatriates.

1. INTRODUCTION

The core of the proposal envisaged by Prof. Bhagwati more than 40 years ago is well known: "(...) the loss to the less developed countries of some of their best-trained citizens can be made up by a tax on emigrants, with the revenues channeled back home through the United Nations." Since then there has been no shortage of publications complimenting, criticizing and (more frequently) building new models based upon that original idea. However, the discussion on the very phenomenon of brain

^{*} Associate Professor, Universidad Carlos III de Madrid, abaez@der-pu.uc3m.es.

^{**} Associate Professor, Universidad Pública de Navarra, hugo.lopez@unavarra.es.

¹ Bhagwati (1976, 34–38).

² A good revision in Brauner (2010, 221–268).

drain taxation and the corresponding technical alternatives to articulate it have largely remained academic, inasmuch as no single jurisdiction has seriously implemented legislative measures in this direction. Quite the contrary: the policy of many developed states, and indeed a large part of EU countries, has been to design an entire arsenal of tax incentives to attract talent (*brain gain benefits*). Namely, these countries may accord a beneficial tax treatment for "new residents" postponing the full effect of ordinary residence, thereby limiting personal income taxation on domestic sources of income or even taxing these immigrants at reduced tax rates, all of this aimed at promoting the arrival of high-level professionals (sometimes including even sportspersons).³

We would therefore be facing two tax policies – "brain drain taxes" and "brain drain benefits" – which at first glance are radically opposed. However, and this is the suggestion of this contribution, the legal configuration of the Spanish brain gain benefits might well help to overcome some of the technical difficulties frequently perceived in certain versions of brain drain taxes.

The rest of this article is organized as follows: section 2 briefly exposes the main features of the Spanish tax treatment of highly mobile individuals placing the emphasis on the description of brain gain benefits. Section 3 analyzes the eventual coexistence of these benefits with different versions of brain gain taxes, with a view to detecting a clash or rather a synergy of (tax) policies. Section 4 contains a brief conclusion.

2. SPANISH TAX TREATMENT OF HIGHLY MOBILE INDIVIDUALS AND BRAIN GAIN BENEFITS

In a globalized world, immigration of skilled workers is far from being the matter of migration from developing to developed countries. It is much more a matter of attraction of high-qualified workers wherever they are and of retention of these types of workers, preventing their emigration.

In order to achieve these goals, states implement different kinds of measures. Among these measures, tax policy is one of the most relevant and Spain has set up a number of special tax regimes to deal with this phenomenon: tax rules (incentives and disincentives) to retain talent, on the one hand; and tax rules (incentives) to attract talent.

Regarding the tax measures to retain talent and more specifically, the tax disincentives, Spain has basically set up three exit taxes, which have been very problematic from a EU tax law perspective:⁴ 1) expanded

³ Falcón y Tella (2009).

⁴ On these and other Spanish exit taxes see CJEU, case C-269/09, European Commission v. Kingdom of Spain.

residence for taxpayers who move their residence to a tax haven;⁵ 2) a special timing rule, taxing unrealized income in case of loss of residence;⁶ 3) a specific rule taxing unrealized capital gains on shares with a market value of at least $4,000,000 \in$.⁷

The previously mentioned tax measures deal with international mobility of workers but they are by no means focused on the phenomenon of brain drain. These measures accent on mere "collection drain".

Regarding the tax measures to retain talent by granting tax incentives, Spain has set up a special regime for expatriates. However, the previous tax incentive focuses more on supporting Spanish companies in their process of internationalization than on providing tax incentives to retain talent.

The only tax measure actually related to incentivizing international mobility of (skilled) individuals is the one first introduced in 2004, in Article 93 of the PITA, whose relevance for this contribution recommends a closer examination. Any individual residing in Spain for personal income tax purposes would in principle be taxed on their worldwide income, at a maximum rate of around 45%, depending on the particular region in which the individual is a resident⁹. In very broad terms the special regime consists on an option for inpatriates fulfilling certain requirements to be taxes as non-residents: therefore these individuals may choose between being taxed as ordinary residents, at a progressive rate of up to 45%, on worldwide income, or, if deemed non-residents, at a proportional rate of 24% on Spanish income sources. ¹⁰ The regime has been heavily criticized since its inception and this explains to a large extent its successive amendments, the most noteworthy being: 1) the introduction of a maximum threshold of 600,000 euros in 2010 regarding the special tax rate; the limit, certainly explained by the critical situation of Spain since 2009, implied the existence of a dual track tax rate: i.e.

⁵ Art. 8.2 of the Law 35/2006, of 28 November 2006, del Impuesto sobre la Renta de las Personas Físicas y de modificación parcial de las leyes de los Impuestos sobre Sociedades, sobre la Renta de no Residentes y sobre el Patrimonio (Personal Income Tax Act, hereinafter PITA).

⁶ Art. 14.3 of the PITA. This rule is not without EU Law compatibility issues and it has been in fact revised in view of several decisions of the Court of Justice of the European Union considering similar rules of other Member States in breach of the EU freedom of establishment.

Art. 95 bis of the PITA.

⁸ Art. 7 p) of the PITA. For an analysis of this tax incentive, see López López (2015, 94–140).

⁹ The different rates for PITA purposes are derived from the particular Spanish partially decentralized personal income tax system.

This special regime also applies to the Spanish Impuesto sobre el Patrimonio (Capital Tax) whose high minimum exempt amount and low rates makes it practically irrelevant.

24% on source income up to 600,000 euros, and 45% above that figure. 2) in 2015 sportspersons were excluded to this tax regime; this change was the result of an obvious misalignment between the theoretical goal of the benefit – attracting highly-skilled professionals – and its effect in practice – the almost exclusive application to sportspersons, particularly football players. ¹¹

Beyond the fact that the special tax regime will in practice preclude the application of the Spanish double tax convention network to residents opting for it, 12 the rule has been harshly criticized from a strict constitutional perspective. Since its seminal Decision of 26 April 1990 (STC 76/1990) the Spanish Constitutional Court has repeatedly made clear that: a) the equality principle imposes the application of the same legal (tax) consequences to comparable factual situations; b) the principle of equality does not prohibit any differential legal treatment but only discriminations that may be considered artificial or unjustified for not being based on objective and reasonable criteria; c) for a different treatment to be in compliance with the constitutional principle of equality, it is not enough that it is objectively and reasonably justified, since the legal consequences of such treatment must be suitable and proportional to the pursued goal.

In a nutshell, according to the Spanish Constitutional Court, just the different (tax) treatment of comparable situations might be considered a violation of the Constitution if that different treatment is either not justified or justified but not proportional. Even if the Spanish Constitutional Court has not yet had the opportunity to rule on the special inpatriate regime, the application of the aforementioned case law should lead to a straightforward conclusion according to the following reasoning. Comparing the situation of an ordinary tax resident earning an income that is comparable (although not identical) to that of a resident inpatriate opting for the application of the special regime, it is obvious that the two taxpayers are treated differently. The former will be taxed on their worldwide income at a progressive tax rate of up to 45%. The latter will be taxed a 24% tax rate on Spanish-source income on the first 600,000 euros. The Spanish-source income exceeding 600,000 will be taxed at a

It is with good reason that the special tax regime received, at least before 2010, was called the Beckham Law, being this British football player was one of the first and most prominent beneficiaries of the regime.

¹² In fact, Article 120 of the Reglamento del Impuesto sobre la Renta de las Personas Físicas (PIT regulations hereinafter) allows for the issuing of a certificate of residence for beneficiaries of the special regime. This certificate, however, is not proof the residence in Spain for double taxation conventions purposes, according to the very regulations. Some scholars and practitioners in Spain have been extremely critical of this restriction, not only in regard to Spanish double taxation conventionss lacking Article 4(1)2 of the OECD Model Tax Convention. See Falcón y Tella (2010, 49–50).

rate of 45%. Even if the preamble to the law introducing the special inpatriate regime in Spain did not mention its purpose, there is a broad consensus among both tax authorities and scholars that the measure intends to attract high-skilled workers to Spain. 13 However, it is more than doubtful whether the special tax regime is actually adequate for such attraction, taking into account that the very scope of the measure does not differentiate the workers who might benefit from the special tax regime. The amendment introduced in 2015 – exclusion of sportspersons – merely corrected the initial roughest mistakes of the regime, which, however, remains incapable of objectively defining the criteria referring the highskilled workers that it theoretically aims to benefit. Finally, and most importantly, the differences in taxation between regular residents and inpatriates opting in for the special regime – which has additionally dragged on for six years - seem too great for them to be considered proportional. The amendment made in 2010 – dual rates for income below and above 600.000 € – does not seem to sufficiently alleviate this lack of proportion, particularly if taking into account that among the 2,000 persons who had enjoyed the regime before 2010, less than 10 percent had incomes above 600,000 €. ¹⁴ This would suggest that especially after 2015, with the exclusion of sportspersons, the regime in practice essentially consists of the application of the reduced rate (24%) on Spanish-source income.

At this point it can only be concluded that Spain lacks a clear policy regarding taxation of cross-border mobile individuals and that the only measure that seems at least to show a more or less clear intention is of questionable constitutionality.

3. BRAIN DRAIN TAXES AND BRAIN GAIN BENEFITS TO THE (MUTUAL) RESCUE

One might expect that brain gain benefits, such as those described for the Spanish tax system, are by definition the opposite of brain drain taxes. Ultimately, both sets of tax measures would serve different and rather contradictive policies, namely to deter the brain drain, for the benefit of developing countries, ¹⁵ and to promote it at their expense. However, as we will try to prove, this would only hold true regarding a particular and rather outdated version of brain drain taxes, whereas, in

¹³ See, among others, Álbarez Barbeito, Calderón Carrero (2010, 17–18), Sanz Clavijo (2013); Gorospe Oviedo (2010, 15–48); García Carretero (2015, 93).

Falcón v Tella (2009).

Explicitly recognized as one of the goals of the original Bhagwati tax proposal: Bhagwati, Dellalfar (1973, 95).

their evolved versions, brain gain benefits and brain drain taxes could be rather complementary.

3.1. The Original Bhagwati Proposal and Brain Gain Benefits

In brief, the original Bhagwati brain drain tax proposal implied the imposition of a tax on the income of immigrants, collected by the tax authorities of the developed host state and transferred to the developing home state. 16 It is highly evident that a host-state brain drain tax such as the original proposal by Bhagwati would be difficult to reconcile with the previously described brain gain benefits. Purely from a policy perspective, both measures seem to be rowing in completely different directions. But, beyond this, the coexistence of host-state brain drain taxes and brain gain benefits might well exacerbate the legal problems already suffered by both measures individually.¹⁷ Indeed, both measures are frequently blamed for discriminating regular residents of the host state and new residents after immigration, albeit for different reasons (better treatment of regular residents under host-state brain drain taxes, and better treatment of new residents after immigration under brain gain benefits). It seems apparent that the introduction of a measure (be it a host-state brain drain tax or a brain gain benefit) with a radically opposite purpose to that of the original one (be it again a host-state brain drain tax or a brain gain benefit) would be very unhelpful – actually rather detrimental – in the search of a proper justification or proportion of the described measures. Indeed, it is even harder to justify the different treatment of regular residents of the host state and new residents, whatever it may be, if the opposite policy is promoted at the same time. However, this conclusion proves to be of little value if one takes into account that a host-state brain drain tax has never been implemented, and even if it had been implemented, it is highly unlikely that the jurisdiction would at the same time have passed brain gain benefits. In any case, the proliferation of brain gain benefits is an additional indication of the absolute lack of interest on the part of developed countries in host-state brain drain taxes. However, it would be a huge mistake to extend this incompatibility to any shape of brain drain tax, as we try to demonstrate in the next section.

¹⁶ This was the version maintained in different papers at least until 1976. For a description of the evolution of the original proposal see Brauner (2010, 240–243).

¹⁷ For more on brain gain benefits, see Section 2 of this contribution; for more on the original Bhagwati tax proposal, see Brauner (2010, 242) and the literature referenced in footnote 134.

3.2. "Brain Drain Taxes 2.0" and Brain Gain Benefits

Even among those most committed to taxing the brain drain, it soon became apparent that a host-state brain drain tax was a dead end. This left no other alternative to a brain drain tax than being levied by the sending (home) state, normally in the form of extended or citizenship-based worldwide taxation. The coexistence of these "brain drain taxes 2.0" with brain gain benefits, as those previously described, must be considered in detail.

3.2.1. Brain drain taxes to the rescue of brain gain benefits

As indicated above, the enormous tax disparities between original residents and inpatriated residents, generated by brain gain benefits in the form of the Spanish special inpatriate regime, might well prove contrary to Constitutional Law constraints. On the other hand, the alleviation of that disparities might frustrate the very purpose of the measure. However, these benefits would not be in the search of justification and proportionality if they were shaped precisely as a way to alleviate excessive taxation generated by the overlapping of the home-state extended residence- or citizenship-based worldwide taxation and the regular personal income tax of the host state. 19 The degree of this alleviation depends of course on the very benefit and the intensity of the home-state brain drain tax, however, whatever this alleviation might mean in practice, it would provide a constitutional justification for brain gain benefits, and particularly a proportionality that they simply do not have in the absence of a homestate brain drain tax. On the other hand, unless all developed jurisdictions in search of skilled immigrants introduce identical brain gain benefits, the "incentive" would maintain its appeal.

3.2.2. Brain gain benefits to the rescue of brain drain taxes

It is quite clear that a home-state brain drain tax, in any of its possible modalities, might generate international juridical double taxation when combined with ordinary residence-based taxation in the host state, assuming that an immigrant will gain residence in the host state immediately upon emigration, according to its domestic law. Although brain gain benefits are not the only way to alleviate this double taxation, as we shall see, they might well serve the purpose of eliminating the most prominent phenomena of double taxation generated by the overlap of

¹⁸ In fact, Bhagwati himself acknowledged that legal restrictions mandated a tax levied by the developing (home) state rather than the developed (host) country. See Brauner (2010, 243).

¹⁹ For the sake of clarity and systematic presentation this excessive taxation will be analyzed in the next section.

national systems induced by brain drain taxes. Different scenarios can be presented regarding this particular issue.

1) In the absence of a double tax convention between the host and the home states.

In this situation the full freedom normally enjoyed by all iurisdictions to define residence criteria will not be limited by a superior rule of law²⁰ and therefore a situation of double tax residence will be consolidated, which involves, in principle, double taxation on worldwide income. A significant part of this double taxation may be alleviated by the unilateral mechanisms (credits) of the home and the host states,²¹ if it exists at all. However, we should not lose sight of the fact that: i) No unilateral mechanism will grant relief for taxes paid in third countries different from the home and the host states. Inasmuch as both countries will tax worldwide income of the migrant, this double taxation will not be relieved. ii) It is very probable that many host (developed) countries will not grant a credit for taxes paid in the home state on the wages gained in the host state. Indeed certain countries will not grant unilateral tax relief for income sourced in their territories, according to their domestic sourcing rules.²² It is obvious that the wages paid to the migrant in the host state will be sourced in that very state whatever its sourcing rules may be. iii) The frequent petty attitude of states in regard to the application of unilateral tax relief may jeopardize the correction of juridical double taxation in this scenario, as both the host and the home states may fear double credits and end up mutually denying it to the taxpayer.

Brain gain benefits, such as those contained in the Spanish system, will help to alleviate double taxation in the following scenarios: i) if the concerned jurisdictions – particularly the home state – do not have a unilateral mechanism to correct international juridical double taxation, or if it exists, it proves not to be applicable. ii) Inasmuch as the special tax regime implies being taxed just on Spanish-source income it prevents income sourced in third countries – different from the home and the host states – of also being taxed twice by the home and the host states, as states of residence (or the state taxing worldwide income based on citizenship). Of course juridical double taxation, generated by source taxation in a third state and residence or citizenship based taxation in the home state, will depend on the existence of treaties between those states

Unless the freedom to leave a country, envisaged in certain constitutions and international agreements on human rights, is considered a limit in this context.

In this regard see Pomp, Oldman (1979, 36–39).

Although this restriction to foreign tax credit has at times been presented as exceptional there are several countries where foreign tax credit is not recognized for items of income sourced in the residence state. It is a common statement, for example, that foreign taxes imposed on US-sourced income may not be credited (Choi, Rienstra).

and, perhaps more importantly, domestic mechanisms to correct international juridical double taxation.

2) If a double taxation convention exists between the host and the home states.

One might expect that in the presence of a double taxation convention a "brain drain tax 2.0" would not be possible. In fact, in the current international tax regime tax jurisdiction follows residence²³ and it seems certain that the immigrant would gain residence in the host state upon immigration. Additionally, although the home state may expand its residence criteria to also cover emigrated nationals or residents or even develop citizenship-based worldwide taxation, in the presence of an OECD or UN model patterned double taxation convention, the host state will always be the winner in an eventual double residence conflict between the two states. Indeed, the rules contained in Article 4(2) of both the OECD and UN models will normally lead to this result inasmuch as a) the first tie-break rule in the provision (permanent home available) is largely elective²⁴ – and the immigrant would easily avoid double residence and double taxation altogether; and b) the second tie-break rule – center of vital interests - would lead, at least in our opinion, to decide the residence conflict in favor of the host state ²⁵

In the previous context a home-state brain drain tax would be incompatible with the Convention and the brain gain benefits of the host state would lose their justification and therefore, allegedly, their constitutionality. However, two additional scenarios must be considered.

The first is double taxation conventions containing a provision patterned according to the "saving clause" of the U.S. Model Tax Convention. ²⁶ In these treaty circumstances it is obvious that citizenship-based worldwide taxation by the home (sending) state would be in accordance with the corresponding treaty; however, it is also obvious that the overlap of the home-state citizenship-based worldwide taxation and the host-state residence-based taxation might generate double taxation. It is true that – at least according to the current citizenship-based worldwide taxation in the Unites States – a citizen taxed abroad as a resident may

²³ Brauner (2010, 247).

²⁴ Brauner (2010, 250).

²⁵ For a different position see Brauner (2010, 250). We will later go back to this question.

According to this provision in the current Article 1(4) of the U.S. Model Tax Convention (2016) "Except to the extent provided in paragraph 5 of this Article, this Convention shall not affect the taxation by a Contracting State of its residents (as determined under Article 4 (Resident)) and its citizens. Notwithstanding the other provisions of this Convention, a former citizen or former long-term resident of a Contracting State may be taxed in accordance with the laws of that Contracting State."

claim foreign tax credits, in the U.S., for income taxes paid in the residence (host) state.²⁷ However even if rules similar to the U.S. Foreign Tax Credit were introduced together with citizenship-based worldwide taxation in developing home states, double taxation on income sourced in third countries would persist and brain gain benefits, such as those previously described, would help their elimination at least in regard to worldwide income taxation in both the home and the host states.

The second is double tax conventions with modified tie-break rules. Authors favoring home-state brain drain taxes have claimed a possible variation of tie-break rules contained in Article 4(2) of both the OECD and UN model tax conventions in order to avoid systematic defeats of home (sending) states in double residence conflicts, thereby enabling in practice a home-state brain drain tax based on extended residence of emigrants. The proposal is simply to put the center of vital interests as the tie-break rule before the "permanent home available" criterion, eliminating the elective use of the rule by taxpayers and assuming that during the first years of immigration the center of vital interests remains in the home state.²⁸ In our view this change would not guarantee the systematic triumph of the home state in possible double residence conflicts. Even if the current Commentaries to the 2017 OECD Model Tax Convention seem to give more weight to the personal aspect of the center of vital interests, ²⁹ it is important to bear in mind that: i) this interpretation has no basis in the wording of the OECD Model which refers to "...the State with which his personal and economic relations are closer (centre of vital interests)"; ii) the assumption that during the first years of immigration the center of vital interests remains in the home state is also questionable. Indeed, economic relations to the host state would be normally greater than those in the home state; on the other hand, it cannot be assumed lightly that all personal ties can be located in the home state. In any case, conflicts of double residence might persist, even under a change of the tie-break rules in the way suggested. Double taxation will arise in such cases, save for the unlikely event that one of the Contracting States relinquishes its claims, assuming the triumph of the other according to a different tie-break rule (normally habitual abode in the host state). This double taxation may proven to be more problematic than in non-treaty scenario, inasmuch as both contracting states might deny correction of double taxation assuming that the other country is not applying the treaty in a correct manner. In this context, the brain gain benefits would serve as a tool to grant partial – yet relevant – relief for this double taxation.

²⁷ Kirsch (2007, 504–505).

²⁸ Brauner (2010, 250).

²⁹ Article 4 para. 15 of the OECD Model Commentaries states "The circumstances must be examined as a whole, but it is nevertheless obvious that considerations based on the personal acts of the individual must receive special attention."

4. CONCLUSION

The Spanish brain gain benefits special regime, i.e. elective taxation of emigrants in Spain as non-residents during six years after immigration:

- 1) runs contrary to a traditional host-state brain drain tax increasing the many (legal) problems of the latter and exacerbating the constitutional concerns on the former.
- 2) may be a perfect complement to home-state brain drain taxes based on citizenship or extended residence inasmuch as: i) the existence of home-state brain drain taxes might well (constitutionality) justify brain gain benefits; ii) brain gain benefits may partially yet importantly correct international double taxation connatural to home state brain-drain taxes.

REFERENCES

- Álbarez Barbeito, Pilar, José Manuel Calderón Carrero. 2010. La tributación en el IRPF de los trabajadores expatriados e impatriados [Taxation of the personal income tax of expatriate and inpatriate workers]. La Coruña: Netbiblio.
- Bhagwati, Jagdish, William Dellalfar, 1973. The Brain Drain and Income Taxation. *World Development*, 1.
- Bhagwati, Jagdish N. 3/1976. Taxing the Brain Drain. *Challenge* 19 July/August.
- Brauner, Yariv. 2010. Brain Drain Taxation as Development Policy. St. Louis U. L.J. 55: 221.
- Choi, Wooje, John G. Rienstra. United States Corporate Taxation. *Country Tax Guides IBFD*, Sec. 6.1.4.2.
- Falcón y Tella, Ramón. 2009. Modificaciones en el régimen de los nuevos residentes: España (ley Beckham), Portugal y el Reino Unido [Modifications to the regime of new residents: Spain (Beckham Law), Portugal and the United Kingdom]. *Quincena Fiscal*, 21.
- Falcón y Tella, Ramón. 2010. *Derecho Fiscal Internacional* [International tax law]. Madrid: Marcial Pons.
- García Carretero, Belén. 2015. Régimen fiscal de expatriados e impatriados [Tax regime for expatriates and inpatriates]. Lisboa: Juruá.
- Gorospe Oviedo, Juan Ignacio. 2010. Los regímenes opcionales de "impatriados" y de no residentes en la imposición sobre la renta [Optional regimes of "inpatriates" and non-residents in income taxation]. *Quincena Fiscal*, 7.

- Kirsch, Michael S. Taxing Citizens in a Globalized Economy. *New York University Law Review*, 82.
- López López, Hugo. 2015. Régimen fiscal de los trabajadores desplazados al extranjero [Tax regime for workers displaced abroad]. Cizur: Aranzadi.
- Pomp, Richard D., Oliver Oldman. 1979. Tax Measures in Response to the Brain Drain. *Harvard International Law Journal*, 1.
- Sanz Clavijo, Alfonso. 2013. El régimen tributario especial en el IRPF de los trabajadores desplazados a España [The special tax regime in the personal income tax of workers displaced to Spain]. Madrid: Documentos del Instituto de Estudios Fiscales, 56 p.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(100)

CERIF: S141

DOI: 10.5937/AnaliPFB1904017B

B. Bawono Kristiaji*

TAX AND BRAIN DRAIN: JUSTIFICATION, POLICY OPTIONS AND PROSPECT FOR LARGE DEVELOPING ECONOMIES

International migration has continued to escalate over the last three decades, creating a risk of brain drain in developing countries. This paper reviews the extent to which the use of tax instruments to address brain drain can be justified in developing economies with large populations. Furthermore, it explores and assesses tax policy options that may be undertaken to prevent the emigration of high-skilled individual, namely the Bhagwati tax proposal, exit tax, revenue sharing and tax incentives.

Five things can be concluded from the assessment of several policy choices. First, there is no stand-alone tax policy that can optimally address brain drain, in the sense of reducing the number of high-skilled individuals who emigrate. Second, most policies put more focus on the element of fairness to compensate for the "loss" caused by the home country. Third, almost every available policy requires better coordination at the international level. Fourth, all policy options require closer collaboration with immigration agencies. Finally, each policy has the potential to produce unintended consequences.

Key words: Brain drain. – Large developing economies. – Bhagwati tax. – Exit tax. – Tax incentive.

1. INTRODUCTION

International migration has continued to escalate over the last three decades. Globalization, ease of immigration procedures, incentives to

^{*} Partner, Research and Training Services DDTC Indonesia, kristiaji@ddtc.co.id.

attract talented individuals, and wage inequality between countries act as factors compelling high rates of migration. Nevertheless, one crucial issue lies in international migration, namely the brain drain phenomenon. Brain drain is defined as the transfer of highly skilled human resources from one country to another which disadvantages the migrants' home countries (Gibson, Mckenzie 2011, 3–5).

The brain drain phenomenon is generally experienced by developing countries in which a large proportion of their high-skilled individuals emigrates to developed countries. Limited employment opportunities and the lack of certainty in conducting business have encouraged emigration – in particular, of tertiary-educated individuals – to countries with better wages and economic conditions. Although frequently criticized as one of the causes of the stagnation of economic development in developing countries, some parties are of the opinion that high-skilled migration will lead to benefits for the home country, for instance, the prospect of high remittances, technology transfer, and encouraging investment in education.

As such, how can taxes serve as one of the instruments to prevent the brain drain phenomenon? The role of Jagdish Bhagwati, who more than 40 years ago submitted a proposal considered quite 'advanced' for the time, is crucial to the study on this matter. Bhagwati argues that the home country of high-skilled migrants is expected to receive compensation from the country where the migrants receive income, through a tax scheme to guarantee fairness. Such an idea is then linked to the U.S. citizenship-based tax system. The notion of using tax instruments to prevent brain drain does not stop there. Furthermore, some literature has reviewed various other methods, such as tax incentives to keep high-skilled individuals in the country, exit tax, revenue sharing, and the development of Bhagwati's ideas.

From the standpoint of developing countries, especially those with a large population, the brain drain phenomenon is closely related to the testing of the government's commitment to providing employment opportunities and decent livelihoods for the high-skilled individuals. On the other hand, the movement of high-skilled individuals pertains to the tax base erosion that will, in turn, result in the reduced ability to finance development. In short, the policy dilemma faced by large developing economies is even more complex, and adopting the steps taken by other countries may not prove effective.

As such, to what extent are large developing economies justified in imposing taxes to prevent brain drain? What tax policy choices are ideal for them and what are the implications? This paper will attempt to address these issues.

This paper will review the extent to which apply taxation to address brain drain, in the case of developing economies with large populations, can be justified, i.e. have strong argumentation. There are four motives why we will chose large developing economies (LDEs) as the focus of this paper. First, from 2000 to 2010, the lower middle-income and low-income country groups saw the highest increase in tertiary-education migration, to nearly double. The greatest risk of brain drain occurs in middle-income countries, especially lower middle-income countries in which almost a third of the tertiary-educated population emigrates abroad. In contrast, in high-income countries, the emigration of the tertiary-educated population can be compensated by the immigration of the tertiary-educated persons from other high-income countries or middle—and low-income countries.

Second, largely populated countries generally face complexity in managing the quality of their human resources and ensuring job opportunities. Third, largely populated countries have a significant influence on the size of brain drain as they play an important role as labor-exporter countries.

Finally, Bhagwati himself states that in the context of developing economies, the impact of brain drain is heavily influenced by the size of the population of a country. For small developing economies, the impact of brain drain is greater. On the other hand, brain drain has no great impact on large developing economies given their large population bases. To address these questions, there must exist a legal standing and benefits for these large developing economies. Furthermore, several available policy options will be reviewed and contrasted with normative tax principles.

This paper will not provide any plenary policy recommendation, instead, will attempt to review the prospects and implications of the various policy choices from the context of large developing economies, among others, the links between exit tax and emigration, the implications of citizenship-based taxation on the principle of non-discrimination, the consistency of developing countries in maintaining the predisposition of the right to tax over source countries, global cooperation, prospects for the use of tax incentives, and so forth.

Within this paper, large developing economies refers to low-income and lower middle-income countries (based on the 2019 World Bank classification) with a large population. The research is limited to countries included in the 20 largely populated countries based on the World Population Database (2019). Of the two criteria, 10 countries are included in this research category, namely: Bangladesh, the Democratic Republic of Congo, Egypt, Ethiopia, India, Indonesia, Nigeria, Pakistan, the Philippines, and Vietnam.

This paper consists of six parts. The first part is the introduction. In the second part, the author reviews the concept, impact, magnitude of the migration of high-skilled individuals, and its relation to brain drain. An explanation of the economic situations, demographics, human development level, labor situations, and taxation challenges in 10 large developing economies are discussed in the third part, which also addresses the question of whether there exists any justification for large developing economies to impose taxes to prevent brain drain.

The fourth part of this article will examine the justifications for large developing economies to prevent brain drain through tax instruments. This chapter will also explore and assess four tax policy options that may be undertaken to prevent the emigration of high-skilled individuals. The four options are the so-called Bhagwati tax, exit tax, tax incentives to keep working in the country, and revenue sharing. In this article, we argue that by and large there is no optimal stand-alone tax policy. This is discussed in the fifth section, which covers the relation of such policies to tax competition, trends towards citizenship taxation, non-discrimination rule, increasing relevance of the jurisdiction to enforce taxes, and so forth. The sixth part provides a conclusion.

2. INTERNATIONAL HIGH-SKILLED MIGRATION AND BRAIN DRAIN

2.1. Understanding Brain Drain

According to the United Nations (2019), it is estimated that currently more than 270 million people worldwide reside in other countries as immigrants. In an increasingly integrated economy, migration will in due course follow the mobile acceleration of investment flows, trade, and information distribution. Such a trend has turned into a global phenomenon and every government seeks to continue to monitor and sustain its respective national interests. The rising trend of international migration is accompanied by the momentum of differences in demographic structure among countries and the decline in transportation and communication costs (Ozden, Schiff 2006, 2). As such, the fulfillment of labor supply and demands that differ between countries encourages migration. Consequently, preventing migration is increasingly difficult for any country.

Broadly speaking, the availability of the labor force in developed countries was highest in around 2010. Subsequent to the peak, the age dependency ratio of these countries continued to increase. Contrary to this trend, developing countries were heading towards a large labor surplus after 2010, with a declining value of the dependency ratio (Ozden, Schiff 2006, 2). This imbalance results in the demand and supply of labor from these two groups of countries. In normative terms, free mobility

among residents will generate economic efficiency. In addition to benefiting individuals who decide to migrate, there is also additional global productivity (Wamsley, Winters 2005, 690).

Individuals experience such positive impacts too. Since the decision to migrate is based on economic motives, the welfare of individuals also improves. The impact can even extend to the families or people who depend economically on these individuals in the home country, through remittances.

However, an aspect that sometimes escapes attention is the fact that a surplus of labor force availability may not necessarily be followed by a surplus of high-skilled labor. In developing countries with a large population, the need for workers with certain skills is even greater, thus labor has a positive externality to the development of quality and skills of other workforces in general (Grubel, Scott 1966, 273). In addition to aggregate and individual positive economic impacts, there are negative impacts arising from the economic loss of the home country due to the emigration. The absence of human resources that can replace the emigrants' contribution engenders a decrease in productivity in the home country.

In the context of developing countries, this phenomenon should be avoided, i.e. when human resources with certain skills, which may not necessarily experience a surplus, lose such potentials due to migration. As stated by Bhagwati (1976, 3), this is often referred to as brain drain or lack of highly-skilled individuals due to their migration to other countries, which are predominantly more developed.

2.2. Determinant Factors

Based on the perspective of an individual as a rational being, the motives underlying one's decision to migrate to another country are similar to the movement of capital. Given the wide range in wage rates among countries, a person has a different expected income between his home and the host country. Furthermore, these individuals deduct the expected income from the host country by the migration cost. If the result is greater than the current income, there exists a rational motive for the individual to migrate: to obtain economic gain.

Goldin, Cameron, Balajaran (2012, 41) argues that other than economic motives, political and social conditions serve as factors that encourage an individual to move to another country. These are push factors minimized by the home country whereas the pull factors are optimized by the host country.

2.2.1. Push Factors

As aforementioned, low welfare acts as a stimulus for a person to emigrate from their home country. Aspects resulting in such conditions serve as the push factors underlying the decision to change the situation (Elveren 2018, 45). When an individual perceives that the situation in the home country cannot change and thus causes non-optimal well-being, the urge to move abroad becomes stronger.

In general, these aspects cover economic, social, and political factors. According to Docquier (2014, 3–5), the economic factor is triggered by a variety of circumstances, such as inadequate income levels, limited opportunities in the labor market, as well as unstable economic situations or a recession. From a social perspective, possible push factors are cultural incompatibilities with fellow citizens, discrimination, and rejection by the community. On the other hand, possible political push factors are political instability, security, and unideal governance.

For home countries, in particular, developing countries, improving push factors is not an easy task and is time-consuming. Accordingly, regulations incentivized through taxes to discourage and prevent brain drain are applied as the short-term solution. Furthermore, Roudgar (2014, 3) argued some people tend to be impatient and frustrated by unfavorable political, social and economic conditions in the home country. With the expectation that there will be no immediate and significant change, the probability of such people leaving the country will be even greater.

Taxation of brain drain, i.e. by imposing taxes on income for citizens working in host countries in which the collection process is carried out by the host country, although not the most effective solution, is considered an alternative to reducing the pressure of inevitable push factors (Brauner 2010, 45). Nevertheless, as argued by Brauner (2010, 45), this method is deemed ineffective as it requires strong coordination between the home and host countries.

2.2.2. Pull Factors

Furthermore, the realities that act as the push factor in the home country will be rationalized based on the individual's expectations of the situation in the host country. Similar to push factors, better economic opportunities, more stable social and political conditions and compatibility with the culture of the host country will serve as pull factors. In addition to these aspects, tax instruments may also serve as an alternative in incentivizing highly skilled immigrants into the country, for instance, by creating a special tax treatment regime for expatriates with certain skills or working in certain sectors (Roudgar, Richards 2015, 80).

The push factors can be even more intense when the persons have a network of people who can introduce and facilitate them taking advantage of opportunities in the other country. Further, positive experiences from other people who succeed in other countries will incite a person's decision to migrate. Consequently, confidence in the ability to adapt increases.

In anticipation of this, the home country also applies a pull factor strategy targeted at its citizens to minimize brain drain. Improvement of governance, the supply of public goods, and efforts to increase political stability serve as a pull strategy that is generally carried out by the home country (see especially Huntington 1996). Moreover, a special tax regime is applied as a pull effort to invite expatriates back to the country.

2.3. Brain Drain Trend

Overall, almost every country has seen an increase in emigration over time, including in the percentage of the tertiary-educated population (see Table 1). However, from 2000 to 2010, the lower middle-income and low-income groups experienced the greatest increase, nearly doubling.

Table 1. Tertiary-Educated Emigration Based on Countries	s'
Income Group (% Total Emigration)	

In come Chave Coverties	Tertiary-Educated Emigration Rate				
Income Group Countries	2000	2010			
High-Income	6.3%	8.4%			
Upper Middle-Income	14.1%	23.6%			
Lower Middle-Income	16.3%	31.8%			
Low-Income	8.1%	14.7%			

Source: Brücker, Capuano, Marfouk (2013). Education, gender and international migration: insights from a panel-dataset 1980–2010, mimeo. Data is available online at https://www.iab.de/en/daten/iab-brain-drain-data.aspx (accessed 8 September 2019)

The greatest risk of brain drain occurs in middle-income countries, especially lower middle-income countries in which almost a third of the tertiary-educated population emigrated abroad. In contrast, in high-income countries, the emigration of the tertiary-educated population can be compensated by the immigration of the tertiary educated persons from other high-income countries or middle— and low-income countries. This is further confirmed by OECD findings, i.e. tertiary-educated persons

commonly emigrate to OECD countries rather than to non-OECD countries (see Table 2)

Table 2. Emigration Rate of Tertiary-Educated Person Based on Country Region and Destination in 2000

Pagion of Origin	Emigration Rates (%	of total emigration)
Region of Origin	to OECD countries	to Non-OECD countries
World	4.3	1.3
Africa	9.7	1.1
Asia	3.5	0.9
Europe	5.6	2.5
Latin America	7.8	1.1
North America	1.2	0.2
Oceania	7	0.2

Source: Dumont, Spielvogel, Widmaier (2010)

This may be associated with the low prospects of the labor market in developing countries. Various studies show that the increase in the level of education in developing countries may not be in line with better employment opportunities (Guarcello *et al.* 2008). In fact, in Sub-Saharan Africa countries, the highest unemployment rate is found in university graduates (Fan, Stark 2007, 76–87).

In general, as discovered by Ordine and Rose (2011, 582–97), unemployment occurs due to the faster rate of improvement of the education level compared to industrial improvements and the development of employment opportunities that require high skills. Consequently, an imbalance occurs between the availability and requirement of labor.

2.4. Implication

The brain drain experienced by the home country may in part lead to brain waste in the host country as a person with certain abilities from the home country works in the informal sector or performs a job that does not require special skills in the host country, such as a driver, janitor, waiter, and so forth. In other words, brain drain from the home country does not necessarily lead to brain gain in the host country

To sum up, brain drain is to be avoided and anticipated by developing countries. Nevertheless, the research on the implications of

brain drain increasingly shows that the negative impacts are not as severe as formerly expected.

Brain drain should be understood as an event that may not necessarily be permanent (Stark, Helmenstein, Prskawetz 1997, 227–34). The migration of highly skilled workers abroad may be temporary, ultimately returning to their home countries with higher skills. Secondment, training, and education may give rise to migration to a more developed country, leading to "brain investment", which has a positive long-term impact.

Furthermore, brain drain may also be followed by changes in the perception and behavior of a home country towards education and personal development. According to Beine, Docquier, Rapoport (2001, 275–289), with better opportunities abroad, the citizens of the home country will recognize that education has a high return and thus invest themselves and their family members in it. Thus, the brain drain phenomenon can trigger an implicit "brain gain" that would not be obtained without the opportunity to migrate abroad (Docquier, Rapoport 2007, 15–16). The positive impact on education and skills ultimately results in a multiplier effect, improving the overall benefits for the country.

Ultimately, the estimated impact of brain drain cannot be separated from the impact of brain gain due to the migration. Thus, the impact to be considered is the difference between the two, which may take the form of net brain drain or net brain gain.

3. LARGE DEVELOPING ECONOMIES: AN OVERVIEW

3.1. Context

In this article "developing economies" refers to the World Bank's classification as of June 2019, concerning low-income and lower middle-income countries. The countries in the two categories are classified as developing economies since they commonly emit migrants and yet are not preferred emigration destinations for residents of other countries. Thus, in net terms, these countries have higher emigration than immigration rates. Therefore, their interest in the issue of brain drain is significantly more relevant. Furthermore, the World Bank classifies low-income and lower middle-income countries as having a gross national income (GNI) per capita amounting to less than USD 3,996 per capita in 2018 (World Bank 2018).

On another note, the term "large" refers to countries with large populations. There are three underlying reasons why the term "large" is

used in this context. First, largely populated countries generally face complexity in managing the quality of their human resources and ensuring job opportunities. Second, largely populated countries have significant influence on the size of brain drain. Finally, as stated by Bhagwati, migration of high-skilled individuals should not have much impact on large developing economies.

This paper reviews 10 large developing economies as case studies. The population data used is from the World Population Prospect, published by the United Nations Population Division. Generally speaking, these 10 countries were selected to provide an overview of the issues and situations in large developing economies and not intended to specifically establish solutions for each country. The ten countries are as follows: Bangladesh, the Democratic Republic of Congo, Egypt, Ethiopia, India, Indonesia, Nigeria, Pakistan, the Philippines, and Vietnam. The population of these 10 countries stands at 2.68 billion, i.e. 35.3% of the world's population.

3.2. Economic Situation

This section discusses the economic structure of the 10 selected large developing economies (LDE). Some of the economic indicators discussed are the performance of Gross Domestic Product (GDP), demographics and labor conditions, and the quality of human development.

3.2.1. Gross Domestic Product¹

Based on its economic growth, Ethiopia is the country with the highest economic growth among the 10 sample countries. Based on the sectoral contributions to the GDP, this country relies heavily on agriculture, which accounts for 30% of the GDP. In 2014 its economic growth was more than 10%. However, this figure decreased to 6.8% in 2018, dropping approximately 3% compared to the previous year. One of the factors resulting in Ethiopia's high economic growth is the state's investment in the public sector, primarily in developing social and economic infrastructure. Further, the government intervenes in the rural economy, specifically in the agricultural sector (Seid, Alemanyehu, Seid 2016, 5).

Three other countries that show satisfying economic growth performance are Bangladesh, Vietnam, and India. Bangladesh is the world's second-largest textile exporter and is slowly reducing its dependence on imports and foreign aid. Bangladesh's GDP growth is quite satisfying, experiencing an upward trend, ranging from 6% to 8% in the past five years, i.e. from 2014 to 2018.

Data and information related to economic growth performance in this section (Gross Domestic Product (GDP) growth and sectoral contributions (agriculture, services, manufacturing, and mining) to GDP are sourced from the World Development Indicators – World Bank Group.

On the other hand, Vietnam's economic growth in 2018 reached 7.1%. This growth was driven by foreign investment, triggered by various policies favoring foreign investors. One of the policy mechanisms applied by the Vietnamese government is to completely open access to ownership of several domestic companies to foreign parties. The purpose of such a policy is to reduce the level of corruption and increase efficiency. However, ownership of shares in several sectors such as banking, telecommunications, aviation, and defense remains restricted (Jennings 2017). Additionally, the contribution of the manufacturing and service sectors to Vietnam's total GDP is sustainable despite the downward trend in the agriculture and mining sectors.

India, a neighboring country of Bangladesh, has an economic growth pattern that tends to be stable at around 7%. As a matter of fact, its economic growth in 2016 amounted to 8% but again declined in subsequent years. Slower post-2016 GDP growth may have stemmed from temporary disruptions in the economy. Two policies that resulted in the shock were the implementation of fiscal reform through the Goods and Services Tax (GST) and monetary reform through demonetization (World Bank 2018). Two other countries in the ASEAN Region, Indonesia and the Philippines, had stable economic growth in the range of 5% to 6% from 2014 to 2018. Pakistan also saw a similar economic growth rate even though the three countries have relatively different sectoral contribution patterns. Pakistan itself depends on the agricultural sector. In contrast, the majority of Indonesia's and the Philippines' GDP originates from the manufacturing sector.

Furthermore, the Democratic Republic of the Congo (DRC) and Nigeria tend to have a uniform pattern of economic growth. The two countries on the African continent managed to recover from the downward trend in GDP growth between 2014 and 2016. The DRC itself is a country that is highly dependent on the mining sector in its economic structure. The country's economic growth reached 6% in 2018, whereas in 2016 it stood at only 2.4%. Such a fact is inseparable from political and security conditions that have stabilized, which greatly affects economic activity (The Heritage Foundation 2019).

Nigeria managed to recover from a previously negative GDP growth, in 2016, to positive growth the following two years. A worldwide drop in oil prices, together with low foreign exchange revenue from the non-oil sector, resulted in low and decelerating economic growth in 2016, according to World Bank (2017). Egypt, another African country, has relatively low economic growth. It's GDP growth in 2014 only stood at approximately 3% and increasing to 5.3% in 2018. This is inextricably linked to the economic reform program carried out by the Egyptian government, relying on cooperation with the International Monetary Fund (IMF) since 2016 (IMF 2018).

3.2.2. Demographics and Labor

The identification of demographic and labor conditions with regard to brain drain can be traced through the age dependency ratio. In simple terms, the age dependency ratio can be defined as a comparison of the number of people who are of the non-productive age and those of the productive age.

World Bank data shows that the DRC is the country with the highest dependency ratio, 97%. This value shows that for every 100 productive age persons in the DRC, 97 residents depend on the productive age population. On the other hand, Vietnam places last, with a ratio of 44%, which that does not even amount to half of the DRC's. Nonetheless, the dependency ratio alone is insufficient to assess a country's economic conditions as it only indicates the size of the productive age population, regardless of whether it is employed. Thus, attention should be focused on other labor-related indicators.

Table 3. Age Dependency Ratio and Unemployment Rate in Selected Large Developing Economies (2018)

LDE Countries	Dependency Ratio (% of working age population)	Unemployment Rate (% of labor force)
Bangladesh	49	4.3
DRC	97	4.2
Egypt	64	11.4
Ethiopia	79	1.8
India	50	2.6
Indonesia	48	4.3
Nigeria	87	6.0
Pakistan	66	3.0
The Philippines	56	2.5
Vietnam	44	1.9

Source: World Bank (2019)

² The World Bank's version of the age dependency ratio is the total population under the age of 15 years and over 64 years compared to the population aged between 15 and 64 years, which is considered the working-age population. Source: https://data.worldbank.org/indicator/SP.POP.DPND.

Based on World Bank unemployment data, Egypt is the country with the largest unemployment rate, i.e. roughly 11.4% in 2018.³ This value is significantly higher than in the DRC, where the unemployment rate is only 4.2% in 2018, even though the dependency ratio was significantly higher than in Egypt. Thus, it can be implied that employment opportunities in Egypt are relatively low compared to its sizeable productive population. A low dependency ratio along with high unemployment may lead to emigration, particularly, for individuals of productive age, regardless of their level of education and skills. The comparison between the dependency ratio and the unemployment rate can be seen in the Table 3.

Table 3 shows the dependency ratios in the ten countries. Based on the information, India, Bangladesh, Indonesia, and Vietnam have demographic advantages compared to other LDE countries where the productive age dominates the population (demographic dividend).

3.2.3. Human Development Level

The factor that determines the economic development of a country is not only its economic growth but also the quality of human resources (HR), assessable through the Human Development Index (HDI). The HDI indicator itself is an assessment of the dimensions of human development which is subsequently normalized by a geometric index, which is estimated.

The dimensions of human development estimated in the HDI are health, knowledge, and economics. The health dimension contains indicators in the form of life expectancy for a country's population. In contrast, the dimension of knowledge is estimated through the length of education and the proportion of people attending school. Furthermore, the economic dimension that shows the quality of human resources is estimated using the Gross National Income (GNI) per capita.

Based on estimates conducted by the United Nations Development Program (UNDP), none of the LDE included in this study achieved the ranking of very high or high human development country in 2017. The Philippines ranked 113th, the highest in the HDI ranking, followed by Egypt (115), Indonesia and Vietnam (116), India (130), Bangladesh (136), and Pakistan (150) which were classified as medium human development countries. The other three countries are categorized as low human development countries, namely Nigeria, Ethiopia, and the DRC, ranked 157th, 173rd, and 176th, respectively.

³ The World Bank's version of unemployment data pertains to the number of unemployed people who are actively looking for work compared to the total workforce. Source: https://data.worldbank.org/indicator/SL.UEM.TOTL.ZS.

The HDI score is also supported by information pertaining to the portion of the population that has completed education up to the secondary level (equivalent to high school) and tertiary (equivalent to university). The Philippines has a relatively high school enrolment rate for secondary and tertiary education, around 89% and 35% respectively. In contrast, the DRC, the country with the lowest HDI, has the lowest number university graduates, only 7%.

3.3. Brain Drain: Magnitude of the Problem

This section identifies some patterns and trends causing emigration in LDE countries, with a view to establishing the right measures in formulating policy priorities related to the taxation of brain drain. These factors include migration patterns as well as economic contributions to the home country.

3.3.1. International Migration Pattern

It is recommended that developing countries with satisfactory economic development to observe the pattern of emigration by their citizens, to allow for the mapping in any country that has the potential for brain drain. One possible indicator is the classification of the level of education of emigrating citizens.

Having observed the role of human resources quality, which significantly determines the economic development of a country, we can now map patterns of population emigration from LDE countries to developed countries. This mapping can serve as an indicator of the extent of access that developed countries provide to emigrants from various developing countries, based on their level of education. Additionally, this mapping can also show the determinant factors of emigration in regard to the economic development potential of the home country.

Table 4. The Proportion of Emigrants Migrating to 20 OECD Countries by Education Level

Home country	2000			2010			
Home country	Low	Medium	High	Low	Medium	High	
Bangladesh	57.9%	11.5%	30.6%	43.8%	15.2%	40.9%	
Congo, Democratic Republic	38.1%	27.1%	34.8%	32.8%	28.2%	39.0%	

⁴ The data used is sourced from the World Bank relating to school enrolment per capita level, available at: https://data.worldbank.org/indicator/SE.SEC.ENRR for secondary education level, and https://data.worldbank.org/indicator/SE.TER.ENRR for tertiary education level.

Home country		2000		2010			
Home country	Low	Medium	High	Low	Medium	High	
Egypt	20.1%	23.4%	56.5%	14.6%	22.8%	62.6%	
Ethiopia	23.9%	32.7%	43.5%	16.6%	33.3%	50.1%	
India	30.1%	12.8%	57.1%	19.4%	12.0%	68.6%	
Indonesia	34.0%	25.4%	40.6%	27.3%	24.0%	48.7%	
Nigeria	19.9%	16.6%	63.4%	14.2%	15.4%	70.4%	
Pakistan	50.5%	14.5%	35.0%	37.2%	15.9%	46.9%	
The Philippines	13.4%	21.2%	65.4%	9.1%	18.8%	72.1%	
Vietnam	36.8%	27.9%	35.3%	30.3%	28.2%	41.5%	

Source: Brücker, Capuano, Marfouk (2013). Education, gender and international migration: insights from a panel-dataset 1980–2010, mimeo.⁵

Based on data on emigration to various developed countries as shown in Table 4, the proportion of tertiary-educated emigrants to developed countries has a growing trend. On the other hand, the number of secondary-educated emigrants migrating to developed countries has a decreasing trend. This indicates the potential for brain drain that actually benefits developed countries amid their slow population growth, supported by various types of pull factors that were formerly available.

In addition to the pattern of emigration to developed countries, we need to observe which are the citizens with high levels of education that emigrate the most. This may point to the country's push factors with the potential to cause brain drain. Table 5 shows the number of emigrants from 10 LDE countries throughout the world.

Table 5. Tertiary-Educated Emigration on Selected Large Developing Economies

Committee	1975 19	1980 1985 1990 1995 20	1975 1980	1980 1985 1990 1995 2000	1000	1005	2000	20	10
Country					2000	Male	Female		
Bangladesh	1.8%	1.7%	2.0%	1.8%	3.1%	4.0%	3.6%	3.6%	
Congo, D.R.	8.2%	7.7%	8.2%	8.0%	8.3%	7.6%	15.5%	7.1%	
Egypt	7.5%	5.9%	7.5%	5.4%	4.4%	4.1%	3.1%	3.9%	

⁵ Source: https://www.iab.de/en/daten/iab-brain-drain-data.aspx (accessed September 8, 2019).

Country	1975 19	1980	1985	1990	1995	2000	2010	
Country	19/3	1980	1983	1990	1993	2000	Male	Female
Ethiopia	1.5%	2.1%	4.9%	7.1%	8.3%	9.2%	N/A	N/A
India	2.9%	3.2%	2.7%	2.8%	3.1%	4.2%	4.1%	3.1%
Indonesia	3.3%	5.6%	8.3%	2.6%	1.9%	1.3%	2.8%	2.2%
Nigeria	2.2%	2.1%	4.3%	7.7%	9.3%	10.1%	N/A	N/A
Pakistan	2.7%	5.4%	6.7%	6.5%	9.5%	12.0%	6.9%	6.0%
The Philippines	9.9%	9.5%	10.9%	12.5%	12.1%	13.2%	9.6%	6.2%
Vietnam	16.1%	17.1%	26.2%	23.9%	25.2%	26.3%	10.9%	10.1%

Source: 1975–2000 data is from Cecily Defoort, Tendances de long terme en migrations internationales: analyse à partir de 6 pays receveurs, Manuscript in French, Université Catholique de Louvain. (2006); 2010 data is from Barro and Lee (2013) as quoted in Arslan (2016, 26–29).

On closer inspection, the potential for brain drain is greatest in countries in the South Asian region. Countries such as Bangladesh, India, and Pakistan have experienced a significant rise in the emigration of the population with a tertiary education level, compared to the total number of the countries' emigrants. Meanwhile, the emigration of people with a high level of knowledge decreased in the DRC and Indonesia. Other countries, such as Egypt, Ethiopia, Nigeria, the Philippines, and Vietnam have experienced moderate increases in emigration of this population. Finally, in the case of LDE countries, pull factors seem to be more significant than push factors in causing brain drain of individuals with high levels of education.

3.3.2. Remittances

Public debate generally infers that brain drain only benefits developed countries. In contrast, several parties suggest that this phenomenon could also contribute to the level of welfare of people in developing countries. One quantifiable consequence is the remittances from the diasporas.

Based on World Bank estimates in 2018, the ten LDE countries generate more than 36% of remittances from all over the world.⁶ India, the Philippines, and Egypt are three of the top countries receiving remittances, with revenues of US\$ 79 billion, US\$ 34 billion, and US\$ 29 billion respectively. However, the remittances received by India are relatively low compared to its GDP, as shown in Table 6.

⁶ Source: https://data.worldbank.org/indicator/BX.TRF.PWKR.CD.DT.

Table 6. Comparison of International Remittances to GDP

Country	Remittance (% GDP)
Bangladesh	5.7%
Congo, Dem. Rep.	3.8%
Egypt, Arab Rep.	10.2%
Ethiopia	0.5%
Indonesia	1.1%
India	2.9%
Nigeria	6.1%
Pakistan	6.8%
The Philippines	10.2%
Vietnam	6.5%
Total 10 Countries	4.0%

Source: World Bank (2019)

According to Kapur (2004, 16), remittances alone can be a relatively stable source of external financing, especially for developing countries. Moreover, co ceptually, remittances may have a positive impact on the economy of the recipient country. Remittances to recipient countries in the form of international remittances can contribute to the country's long-term savings and investment. According to Solimano (2013, 15), in the short run, this may lead to positive effects on aggregate demand and output through consumption by individual recipients in the home country.

3.4. Tax Situation

This section provides a review of the tax system, on a macroeconomic scale, relating to the contribution of tax revenue. Data on tax contributions to the economy, both in the form of the tax ratio and the tax revenue structure, is sourced from the World Bank, the OECD, and other

⁷ The World Bank data used in this article is available at https://data.worldbank.org/indicator/gc.tax.totl.gd.zs (tax ratio), https://data.worldbank.org/indicator/GC.TAX. GSRV.RV.ZS (VAT revenue against total tax revenue) and https://data.worldbank.org/indicator/GC.TAX.YPKG.RV.ZS (corporate and individual income tax revenues against total tax revenues).

⁸ The data on the tax ratio and revenue per type of tax against total tax revenue from OECD is sourced from the OECD Global Revenue Statistics Database, which can be

national sources. Furthermore, this section reviews tax regimes relating to the system of taxation of individuals in general, expatriates residing in these countries (inward expatriates), and the citizens of those countries who emigrate abroad.

3.4.1. Bangladesh

Bangladesh's tax ratio was recorded at 8.8% in 2016. According to World Bank (2019), in terms of the tax structure that same year, 25.2% of the total tax revenue was raised from income tax, while 32.3% of the total tax revenue was from VAT. Regarding taxation on individual income, the Bangladesh government taxes the worldwide income of residents, i.e. if a resident receives income outside the territory of Bangladesh, it will still be subject to taxation. On the other side, non-resident individuals are liable to tax on income received in Bangladesh regardless of where the income is generated. The tax rate ranges from 0% to 30%, with six income brackets.

A person is deemed a resident if residing in Bangladesh for 182 days or more in the fiscal year concerned. Furthermore, a person will also be considered a Bangladeshi resident if he resides for 90 days or more for the year in which the income concerned is generated if the person has previously stayed for more than 365 days in the span of four years prior to the fiscal year concerned.

Income from expatriates working on foreign aid projects established under an agreement between the Government of Bangladesh and a foreign government is exempt from taxation. Additionally, the government provides deduction for expatriates working in the field of technology. Following Ahmed (2019, 12), foreign technicians working in companies registered in Bangladesh and located in special economic zones, or the Bangladesh Hi-Tech Park area, involved in the procurement of goods and services, will receive 50% income tax relief for period of three years. No specific tax regime exists for non-resident Bangladeshi nationals.

3.4.2. Congo, Democratic Republic of (DRC)

The DRC's tax ratio in 2016 stood at 7.6%. This state tax revenue is supported by VAT, which reached 32.7% of the total tax revenue that same year. Furthermore, personal and corporate income tax revenues amounted to 15.8% and 14.5% of the total tax revenue, respectively (OECD 2018). For individual income tax, the DRC government taxes the "territorial" income of residents. In other words, if a DRC resident earns income outside the DRC territory, their income is not subject to tax. The

accessed via: https://stats.oecd.org/Index.aspx?DataSetCode=REV. The codes used are 1000 for Corporate Income Tax, 1100 for OP Income Tax, and 5110 for VAT.

tax rate ranges from 0% to 40%, with ten income brackets. A person is deemed as a resident if they reside in the DRC for more than six months during the given fiscal year.

Due to taxation on individuals' territorial income, expatriates are subject to the generally applicable employee income tax. Other remunerations paid to expatriates are subject to a special tax called the Exceptional Tax on Expatriate Remunerations. According to Kating (2019, 8), this special tax is imposed on employers. No specific tax regime exists for non-resident Congolese nationals.

3.4.3. Egypt

Egypt's tax ratio in 2016 was recorded at 15.2%. The personal income tax revenue accounted for 10.8% of the total tax revenue. Furthermore, VAT revenue raised 18.1% to the total tax revenue. The tax revenue was dominated by corporate income tax, which accounted for 31.9% of the total tax revenue (OECD 2018).

Regarding taxation on individual income, the Egyptian government taxes the "worldwide" income of residents. The tax rate ranges from 0% to 22.5%, with five income brackets. A person is classified as a resident if they reside in Egypt for more than 183 days during the given fiscal year and have permanent residence in Egypt. Furthermore, a person of Egyptian nationality who is domiciled abroad, but still earns income from Egypt, is also be considered a resident and subject to individual income tax by the Egyptian tax authority. There is no special tax treatment for expatriates. In other words, foreign nationals will receive the same treatment as Egyptian citizens. No specific tax regime exists for non-resident Egyptian nationals. (Hamzaoui 2019, 8).

3.4.4. Ethiopia

Ethiopia's tax ratio in 2017 stood at 7.6%. Corporate and individual income tax accounted for 29.7% of the tax revenue that year, with VAT contributing 33.3% (World Bank 2019).

In terms of taxation on individual income, the Ethiopian government taxes the "territorial" income of residents. The tax rate ranges from 0% to 35%, with seven income brackets. A person is classified as a resident if they reside in Ethiopia for more than 183 days during the given fiscal year. The income of foreign professionals recruited to transfer knowledge related to investment in exports is entitled to a tax exemption for a maximum of five years, under directives issued by the Minister (Lencho 2019, 10). No specific tax regime exists for non-resident Ethiopian nationals.

3.4.5. India

India's tax ratio in 2017 was 11.2%. In that year, India's corporate and personal income tax contributed 44.2% to the total tax revenue. Furthermore, the share of VAT revenue in the total tax revenues reached 31.5%. (World Bank 2019)

Regarding taxation on individual income, the Indian government taxes the "worldwide" income of residents. The tax rate ranges from 0% to 30%, with four income brackets. A person is classified as a resident if they reside in India for a minimum of 182 days during the fiscal year. Further, individuals residing in India for 60 days in the given fiscal year, with a record of staying at least 365 days within the four years prior to the given fiscal year will also be classified as residents.

Income paid to expatriates working in India is treated as income sourced in India and taxed according to the applicable provisions in India. Costs of living and travel expenses and remuneration may be granted tax breaks. Tax relief for remuneration given to foreign employees working in foreign companies is highly dependent on certain conditions, including not exceeding the domicile period within India and not making claims for tax deductions that may reduce the tax payable on income (Shah 2019, 10). No specific tax regime exists for non-residents of Indian nationals.

3.4.6. Indonesia

Indonesia's tax ratio in 2017 was 11.5%. According to the Ministry of Finance of Indonesia (2019), the share of tax revenue in the total revenue was 10.1% for personal income tax; 19.2% for corporate income tax; and 39.6% for VAT for the year.

In terms of taxation on individual income, the Indonesian government taxes the "worldwide" income of residents. The tax rate ranges from 5% to 30%, with four income brackets. A person will be classified as a resident if they reside, possess a work visa, a work contract, have a business and other activities in Indonesia for more than 183 days in the given fiscal year.

The tax authority can make adjustments to the amount of income earned by a foreign employee under the guidelines for salaries/wages of foreign nationals, if the income is not supported by proper documents. Also, expatriates' income is deemed as taxable income in Indonesia. This can occur if the expatriate is seconded to a local company by a foreign company where the local company subsequently relocates the expatriate's income in the form of payments (for example, management, technical, or other service costs) to a foreign company (Koo 2019, 8). No specific tax regime exists for non-resident Indonesian nationals.

3.4.7. *Nigeria*

Nigeria's tax ratio is classified as very low and based on the country's tax authority data, tax revenues only stood at 3.4% and 4.8%, in 2016 and 2017 respectively. However, these figures were up compared to the 2013 tax revenue of 1.5% of the total GDP. Furthermore, in 2013 the VAT only amounted to 0.1% of GDP, i.e. around 9.5% of the total tax revenue, while more than 80% of tax revenue was contributed by corporate income tax.⁹

As for individual income taxation, the Nigerian government taxes the "worldwide" income of residents. The tax rate ranges from 7% to 24%, with six income brackets. A person is classified as a resident if residing in Nigeria, staying for more than 183 days in a period of 12 months or serving as a Nigerian diplomatic agent outside Nigeria. There is no special expatriate tax regime nor special tax treatment for non-resident Nigerian nationals (Odimma 2019, 8).

3.4.8. Pakistan

Pakistan's tax ratio for the fiscal year 2017, namely from July 2016 to June 2017, was 12.5% (World Bank 2019, 1). Based on data from local tax authorities, the structure of tax revenue in the same fiscal year was supported by indirect tax revenue which contributed more than 60% of the total tax revenue. The revenue from VAT, which is an indirect tax, amounted to 39.9% of the total tax revenue. Furthermore, the direct tax revenue or income tax revenue amounted to 39.9% of the total tax revenue. ¹⁰

Regarding taxation on individual income, the Pakistani government taxes the "worldwide" income of residents. The tax rate ranges from 0% to 29%, with eight income brackets. A person is classified as a resident if they stay in the country at least 183 days in the given fiscal year. Furthermore, civil servants assigned abroad are considered residents.

The income of expatriates with resident status but sourced from outside Pakistan are entitled to a tax exemption if their domicile period in Pakistan does not exceed three years. Nonetheless, the tax relief does not apply if the expatriate's income is sourced from companies established in Pakistan or if the income from overseas is brought in or received by expatriates within Pakistan. As demonstrated elsewhere (Koo, Bukhari 2019, 12), foreign income from expatriates returning to their home countries is exempted for four years after the year they left Pakistan. No specific tax regime exists for non-resident Pakistani nationals.

⁹ Source: IMF Government Finance Statistics, available at: https://data.imf.org/?sk=FA66D646-6438-4A65-85E5-C6C4116C4416.

Source: Federal Board of Revenue of Pakistan available at: http://www.sbp.org.pk/ecodata/tax.pdf.

3.4.9. The Philippines

The Philippines' tax ratio in 2017 was 17.5%. In 2017, the structure of this state tax revenue was sustained by corporate income tax with its share amounting to 24.5% of the total tax revenue. Furthermore, the individual income tax and VAT contributed 14.1% and 13.2% of the total tax revenue, respectively (OECD 2019).

In connection with taxation on individual income, the Philippines government taxes the "worldwide" income of residents. The tax rate ranges from 0% to 35%, with six brackets based on income. A person is subject to individual income tax if they holds citizenship or are an alien individual. Additionally, all citizens are categorized as residents except in cases where they meet the criteria for non-residents.

Citizens and foreigners employed at regional headquarters, regional operations headquarters, foreign banking units, and oil service contractors or subcontractors located in the Philippines are subject to a 15% final tax on gross income. Foreigners who are considered as alien individuals are deemed equal to citizens. Furthermore, there is an immigration tax for individuals who enter the Philippines and stay for more than 60 days (Ocampoo 2019, 12). Currently, no specific tax regime exists for non-resident Philippine nationals, however, until the end of the 1980s the Philippines taxed its citizens on all of their income (see Pomp 1985).

3.4.10. Vietnam

Vietnam's tax ratio is relatively high. In 2015, tax revenues reached 18% of total GDP revenues. The highest share of tax revenue that year was from VAT, which accounted for 33.3% of the total tax revenue, i.e. approximately 6% of the GDP. Corporate income tax and the individual income tax revenues accounted for 25% and 7% of the total tax revenue, respectively (IMF 2018, 33).

As for individual income taxation, the Vietnamese government taxes the "worldwide" income of residents. The tax rate ranges from 5% to 35%, with seven income brackets. A person shall be deemed a resident if they stay for 183 days or more in the given fiscal year, starting from the date of arrival. Otherwise, a person whose residence is registered as a permanent home or a rental house with proof of a particular contract is also classified as a resident.

Non-residents are taxed at a flat rate of 20% on employment income sourced from the territory of Vietnam, without any tax deductions. As reported by Grunkorn, Do, Nguyen (2019, 7–8), however, a 50% tax deduction is granted to foreign experts working on the Official Development Assistance (ODA) projects. No specific tax regime exists for non-resident Vietnamese nationals.

4. JUSTIFYING TAX TO ADDRESS BRAIN DRAIN AND POLICY OPTIONS

4.1. Justification to Use Tax Instruments

Based on research related to the economic situation, demographics, human development levels, and the tax system in 10 LDEs, there exist at least four preliminary conclusions, in addition to the high-skilled migration patterns.

First, the 10 LDEs in this article have relatively varied economic developments. This is demonstrated by the growth and structure of their GDPs. However, other than having generally low per capita income, the contribution of the traditional sector is great and mostly from the agriculture sector. There are socio-political factors that distort the economy as well. Second, the level of human development generally features a fairly low human development index (2018). Specifically, for the education sector, variations in the level of education in the ten countries are still relatively low if observed based on the number of human resources with tertiary education level. Another interesting aspect is the tendency that the level of education correlates with the familial economic background. A person who comes from a wealthier family tends to have the ability to undertake tertiary education (Darvas, Gao, Bawany 2017, 25).

Third, given the large population, more significantly, the challenge faced by large developing economies is to ensure the availability of jobs. Interestingly, four out of the 10 countries examined in this article are in (or heading in the direction of) the demographic dividend phase, where the size of productive-age population will be greater than the non-productive-age population. The demographic dividend can certainly be utilized to increase the economic thrust if and only if employment is sufficiently available, otherwise unemployment is likely to escalate. Another noteworthy phenomenon is the rise of the educated unemployment – the labor force that does not have jobs but has tertiary education.

Fourth, the performance of tax revenue in these countries is for the most part relatively weak. This is indicated by the tax ratio, which ranges from less than 5% (Nigeria) to more than 15% (the Philippines and Vietnam). Gaspar, Jaramillo, Wingender (2016, 30) estimates that a tax ratio of 15% is the tipping point for growth stability.

The challenges faced by these countries generally stem from the informal economy, illicit financial flow, corruption in the tax sector, as well as tax revenues that are dominated by corporate income tax and certain sectors. Their performance is insufficient, especially for individual taxation. Individual income tax treatment for migrants generally refers to

the concept of residence, i.e. taxing income sourced from within and outside the country. Broadly speaking, citizens who are not categorized as residents (who reside and earn income abroad) are not taxed.

All the above-mentioned conditions are factors to be considered in discussing whether using tax instruments to prevent brain drain can be justified. Again, what we refer to as justification in this paper is the reasoning or argumentation, not merely the terms to have personal or economic attachment, in the context of international taxation.

4.1.1. Brain Drain or Brain Gain?

The debate about the costs and benefits of high-skilled migration is not something new. Although this phenomenon is often deemed as a loss for developing countries, because emigration makes it difficult for developing countries to achieve economic growth, there is also contrary opinion.

Emigration opportunities may serve as a solution to the availability of jobs. The economy of developing countries, which is quite dominated by the traditional sector, has resulted in employment for labor force with a higher education background. The data in Table 5 shows that emigrants from the 10 LDEs in this article are dominated by emigrants with tertiary education levels.

This strongly indicates the existence of labor opportunity and a relatively higher wages level for workers with tertiary education in developed countries, specifically the OECD. A similar pattern also exists in the case of unskilled labor. Employment opportunities abroad indirectly support developing countries in reducing unemployment, increasing foreign exchange, and reducing the possibility of social unrest. This seems to be the case in Indonesia which routinely sends low-skilled labor to Malaysia, Hong Kong, and Saudi Arabia.

Another strong argument pertains to emigration activities related to remittances. Remittances sent by high-skilled migrants are considered able to address liquidity problems, reduce poverty, catalyze technological adaptation, and stimulate investment in education. However, (Docquier, Rapoport 2011, 27) found that the effect of remittance is strongly influenced by the amount remitted by emigrants and the impact of its distribution in the home country. On a side note, the remittances received by 10 LDEs in this article amounted to USD 228.6 billion in 2018, which is far greater than the global value of official development assistance (ODA), which stood at only USD 162.8 billion.¹¹

¹¹ Global position in 2017. On a side note, net official development assistance (ODA) consists of disbursements of loans made on concessional terms (net of repayments of principal) and grants by official agencies of the members of the Development Assistance

In sum, why should high-skilled migration be taxed if it benefits the home country?

4.1.2. Motivation to Emigrate

Individual motivation to emigrate from developing countries is basically influenced by push and pull factors, as well as tax and non-tax motives. Empirical studies on the causes of emigration have shown various patterns, for instance, the high level of emigration of medical personnel from Africa is largely driven by pull factors, such as better salaries and livelihoods, while the dominant push factor is solely caused by the risk of contraction of HIV.

Interestingly, there is little argument that tax is one of the push factors for emigration decisions for high-skilled individuals. As reported in Kauppinen, Ropponen (2018), there are only a few empirical studies on this matter. On the one hand, this confirms that taxes are indifferent towards an individual's migration to another country. On the other hand, non-tax related matters are more likely to have the most significant impacts.

In terms of pull factors, we should be aware that developed countries strive – driven by the aging population problem and intended to boost the domestic economy – to attract new talents from around the world to migrate to their countries.

Avi-Yonah (2015, 45–56) presents a noteworthy argument: efforts to reduce tax burdens are presently possible if capital mobility is followed by the transfer of resident status, especially in the increasingly transparent tax landscape due to the automatic exchange of information cooperation. However, both elements – incentives and tax planning – are more relevant in non-tax compliance practices of high net-worth individuals (HNWI) and not in the context of brain drain.

The next question is can taxation be justified, if the tax factor is indeed a less-dominant factor in the decision to emigrate. Should the causal factors of the emigration be addressed instead?

With respect to public finance, fiscal instruments – taxes among others – have tasks that include allocation, distribution, and stabilization. In terms of their role in allocating the most efficient resources, taxes are

Committee (DAC), by multilateral institutions, and by non-DAC countries, to promote economic development and welfare in countries and territories on the DAC list of ODA recipients. It includes loans with a grant element of at least 25 percent (calculated at a rate of discount of 10 percent). Sources: Development Assistance Committee of the OECD, Geographical Distribution of Financial Flows to Developing Countries, Development Cooperation Report, and the International Development Statistics database, available at https://data.worldbank.org/indicator/DT.ODA.ODAT.CD.

intended to change the behavior of economic agents, among others, creating disincentives for high skilled individuals to emigrate.

4.1.3. Tax Burden and Redistribution

In a closed economic system, skilled and unskilled individuals are subject to domestic income taxes. In the context of ensuring income redistribution and preventing inequality, the individual income tax system will generally be designed progressively. This implies that a person with a higher income or a higher ability to pay will face a higher tax burden (vertical equity). It is worthy of note that the income received by a person is affected by skills and educational background, among other factors. The higher a person's education, the greater the possibility for them to obtain a position with relatively satisfactory returns/wages. Hence, the returns obtained by skilled individuals are in general far better than those by unskilled individuals.

On the other hand, in an open economy where individuals can migrate (particularly if perfect individual mobility exists), a skilled individual can choose a country where their income and welfare will be much better. In such cases, the tax system is ultimately unable to redistribute income fairly.

In turn, the tax system – which is intended to create fairness – will result in a higher tax burden for a high-skilled individual who remains in the home country. A higher tax burden and the opportunity to emigrate will encourage domestic high-skilled individuals to emigrate and result in a revenue loss (see, for example, Bhagwati and Hamada 1982). The state is ultimately pressured to restrict emigration by reducing the tax burden on high-skilled individuals (with a higher ability to pay). Consequently, this leads to a less egalitarian or unfair tax system. Further, the loss of the tax base (due to emigration) and the need for significant development funds will simultaneously increase the tax burden for individuals "left behind" in the home country.

It is true that individual mobility across-country can lead to a more efficient provision of public goods and services. However, this also limits the country's ability to distribute income fairly as hypothesized by Tiebout (1956, 417) in the context of sub-national taxes. Additionally, Wilson (2011, 75) also reports that immobile residents bear the burden of taxation due to the high emigration rate of high-skilled individuals. Hence, a tax on brain drain can ensure that the redistribution of income from high-skilled emigrants to lower-income residents, while providing the government with the ability to tax high-income residents. This justifies taxation to reduce the emigration of high-skilled individuals.

4.1.4. Political Perspective

Emigration could also have an impact on the disconnection between a citizen's political rights and obligations. Basically, taxation must always be accompanied by representation. This implies that the compulsory payment must be limited by the laws established by the people's representatives. The taxation with representation jargon, in this case, appears as a condition for political allegiance.

Whereas in the context of emigration, an individual who emigrates does not generally change their citizenship status, i.e. still intends to remain connected with the home country, such an individual commonly maintains their rights as citizens, for instance, obtaining services and protection from embassies, participating in general elections, etc. However, with his relatively long emigration and the possibility of becoming a tax resident in another country, the obligation to pay taxes in the home country no longer exists, but political rights from the home country still exist. Bhagwati (1987, 53) refers to this situation as "no taxation with representation."

This opinion should be a matter of concern by now, specifically with the current pattern of global migration. Conflicts in several regions and the rise of international refugees, demand for talented migrants from population-aging countries, and increasingly loose immigration regulations may lead to the majority of citizens of a country residing in other countries or a country accommodating a substantial number of immigrants. In such an event, the rights and obligations of the population in a country become increasingly asymmetrical. ¹³ Thus, the tax for emigrants is justifiable.

4.1.5. Efficiency and Revenue Adequacy

Free individual mobility encourages the government of a country to compete for residents and provide optimal budget allocations. Individual choices are assumed to be rational, i.e. choosing a country or jurisdiction considered to be the best in providing public goods. On the other hand, governments in various countries will adjust facilities according to public references (local public goods). In other words, the absence of instruments that limit the mobility of human resources ultimately engenders efficiency.

At the same time, high-skilled emigrants also experience taxation without representation in the host country, where paying taxes generally does not generate any political rights.

The situation can also be reviewed in the fiscal contract model. As the framework of reciprocal relations between the state and the people (the state provides public goods and services and the public pays taxes accordingly) the fiscal contract in the context of an open economy with individual migration has not been addressed by many scholars.

Nevertheless, one question is how a country can finance the provision of quality public goods when facing a brain drain, i.e. the loss of high-skilled individuals, which discourages efforts to boost the economy, as well as revenue mobilization? In the context of sub-national taxes, the measure would be revenue sharing or transfer allocation scheme. However, such a scheme is generally not available in the national or supra-national tax framework.

Hence, the revenue from taxing high-skilled emigrants incentivize developing countries to compete in welfare-improving tax and expenditure. These funds can be used by governments in developing countries to allocate spending to areas that may reduce the motivation to emigrate, e.g. education, industrial parks, safety. However, this argument needs to focus on the connection or link between the revenue from brain-drain tax and its use to prevent emigration.

Further, the poor performance of tax revenue in large developing economies indicates that developing economies require all available options to mobilize revenues and assess tax gaps. Specifically, individual income taxation has not played an important role in the structure of tax revenue, especially when compared to contributions from corporate income tax, VAT/GST, and revenue from extractive industries.

This is mainly influenced by the fact that the majority of individual taxpayers in developing countries have wages below the per capita income or the threshold for allowance. This figure is completely different from the structure of tax revenue in developed countries where individual income tax plays a significant role thus making the tax revenue more sustainable and not susceptible to business or sectoral conditions.

In short, to continuously improve the performance of their tax revenues, developing countries must increase the contribution of individual income tax. Considering that high-skilled emigrants generally have an income exceeding per capita income or the basic exemption threshold, the efforts to tax their income can be justifiable.

4.1.6. Who Provides the Benefits

One of the philosophical grounds for the state to collect taxes is the benefits principle. The next question is what is the role of the home country for skilled emigrants? Questions and criticisms pertaining to this matter have long been discussed. In general, this position departs from the fact that better income and life (and sometimes better tertiary education) are provided by the host country. If such is the case, what is the role of the emigrant country?

The debate must also be supported by the availability of data, but the criticism is not groundless. Allegations that developing countries are generally not able to provide optimal public goods for the welfare of the society are evidenced by various indicators, in particular, the low government spending on education, health, and infrastructure (Fan, Rao 2003).

However, as suggested by Darvas, Gao, Bawany (2017, 25), it is noteworthy that in general, the level of education of individuals in developing countries is closely related to the economic condition of their families. This implies that the level of income and wealth obtained by families from high-skilled migrants is, in essence, guaranteed by the home country. The guarantee of benefits originates from the protection of property rights, access to financial markets, and political stability. In this context, the home country also contributes positively. From the perspective of the benefit principle, the home and host country are equally justified to impose taxes.

4.1.7. Taxing Rents

According to Bhagwati (1979, 22), another strong argument in taxing skilled emigrants is the fact that the emigrant has windfall gains.

4.1.8. Conclusion

The seven aspects reviewed in the issue of the migration of educated workers to other countries, lead to the conclusion that taxation can be justified. Five of the seven aspects indicate stronger argumentation in favor of taxation, namely the issues of emigration motivation, tax burden and redistribution, political perspective, efficiency and revenue adequacy, and taxing windfall gain. On the other hand, two aspects show the weakness of the argument for taxation, specifically from the benefit theorem and the fact that the emigration of educated workers also gives rise to net gains.

4.2. Tax Policy Options

With (relatively) strong justifications for brain-drain taxation, what policy options are ideal for LDEs? Four policy options, namely the Bhagwati tax proposal, exit tax, tax incentives, and revenue sharing will be reviewed.

4.2.1. The Bhagwati Tax Proposal

The Bhagwati tax proposal refers to the contribution of renowned economist Jagdish Bhagwati, in reviewing the negative effects of brain drain and proposing the main ideas, along with various modifications, fiscal instruments considered ideal for addressing brain drain. This proposal was made in 1972 and has since evolved, but instead of changing, the basic idea continues to be supplemented, especially in the face of criticisms and information arising in academic debates. Some literature frequently refers to the proposals submitted by Bhagwati as the braindrain tax because the idea and model are specifically reconstructed to address brain drain.

Bhagwati initially proposed of a tax collected by the host country on immigrants from developing countries. The applicable rate was 15% (surtax) of the emigrant's income. The idea is that the tax collected by the home country's tax authority (in the context of the U.S., IRS) subsequently be transferred to the home country to compensate developing countries for the incurred losses.

This idea was further developed a year later, in cooperation with Dellalfar. With the support of data, they proposed a new rate which is considered more ideal, i.e. 10% for the adjusted taxable income of emigrants from less developed countries. This tax would also be collected for a maximum of 10 years after a person emigrates. The rate-based simulation showed that the potential tax revenue of the developing countries was substantial and far greater than the amount of foreign aid provided by the U.S. in 1971. In the paper, they put forward a more valid argument for brain-drain taxation, which is based on the principle of fairness. Through the fairness jargon, the brain-drain tax aims to compensate developing countries for the lagging and loss of human resources, and if possible, to decrease brain drain. 14 Bhagwati and Dellalfar (1973, 94–96) argue that the tax could be levied by the host country's tax authority or international organizations such as the United Nations. They also proposed to collect the tax with the assistance of the UN and the tax would be distributed to developing countries, with the exception of those that are corrupt and dictatorial.

Until the end of the 1970s, Bhagwati continued to complete his proposal through some scientific work ranging from emphasizing tax administration cooperation through bilateral agreements, underlining the differences between his proposal and revenue sharing schemes, reviewing political aspects, and strengthening the justification of brain-drain tax. Interestingly, Bhagwati (1979, 24–27) also argues that the adoption of the U.S. global tax system that adheres to citizenship-based taxation best enables the implementation of the proposal. In other words, the proposal refers to the U.S. method that deems its citizens as residents regardless of where they are. By using citizenship as a taxation nexus, the connection between the skilled migration and the home country is maintained until

Bhagwati's proposal is criticized as it is considered to add more burden to the emigrants. However, the emigrants are in a better situation as the increase in wages will be greater than the losses.

the change of citizenship status. Thus, a country that recognizes an individual's citizenship maintains its taxation rights.

In fact, almost no countries in the world tax their citizens on their worldwide income. Currently, the U.S. can be considered the only country that has succeeded in enforcing an extraterritorial tax system. Other countries that have attempted the same measure, such as Eritrea (see DSP-Groep BV 2017) and the Philippines (see Pomp 1985), have failed due to weak hegemony and the requirement of support from other countries. This implies that the success of the Bhagwati tax proposal is highly dependent on international cooperation (bilateral or multilateral). In short, taxation of income received by citizens abroad clearly requires assistance and support from the host country, both in terms of collection and exchange of information. Without coordination and exchange of information, the implementation of taxation on foreign-sourced income will be difficult (Keen, Lighthart 2004; Gadzo, Klemencic 2017).

Other criticisms are inseparable from the third-generation research on brain drain in the 1990s. With more accurate migration-related data, many academics have begun to doubt the existence of brain drain and instead showed the gain from high-skilled emigrants. Consequently, there is no moral argument regarding efforts to prevent migration including the absence of justification of brain-drain tax. However, according to Brauner (2010), the Bhagwati tax proposal is substantially driven by fairness, specifically from an economic standpoint and not from an ethical or moral argument. As such, this criticism can be considered not departing from the same perspective.

Although considered to reflect the principle of ability to pay, the Bhagwati tax proposal was also criticized for creating income inequality between skilled and unskilled individuals in the home country. This is caused by the impact of wage improvements for skilled individuals in the country. Consequently, the government's success in addressing unemployment and managing the availability of individual (labor) for certain sectors may be subject to disruptions (McCulloch, Yellen 1975, 249–64).

The Bhagwati tax proposal faces challenges in terms of administration as well. First, it creates compliance costs for individual taxpayers, as well as barriers to working overseas, as mentioned by Desai, Kapur, Mchale (2004, 681). As a matter of fact, developing countries still encounter challenges in taxing individual income. As an illustration, in Indonesia, the contribution from individual income tax other than withholding tax for employees only amounted to less than 1% of the total tax revenue during the 2013–2018 period. Second, the compliance of workers from developing countries working in host countries will be more difficult to ensure as it is far more difficult for them to return to the

home country due to their (financial) ability. This is different from workers from developed countries or at least upper-middle-income countries.

The assessment of the impact on revenue requires an estimation and complete data. In the absence of complete information, the Bhagwati tax proposal may not lead to definite revenues with the double-elimination tax mechanism through exemption and foreign tax credit. Interestingly, a study conducted by Desai, Kapur, and McHale (2004, 683) on the simulation of the application of this tax for India shows that the potential revenue from the Bhagwati tax is substantial. Overall, we have to consider its implications for citizenship changes, especially considering that the only way to be "free" of the tax burden from the home country is the change of citizenship.

4.2.2. Exit Tax

Unlike the Bhagwati tax, the exit tax aims to directly target the core issue of brain drain, which is to prevent losses from the migration of high-skilled individuals to other countries. The exit tax is a tax imposed to create disincentives for the decision of a person or a company to become a resident of another country. According to Larking (2005, 115–62), prior to the change into another country's resident, taxation is imposed on the taxpayer's assets deemed to be disposed of and resulting in a gain. Exit tax is frequently equated to departure tax (immediate exit tax) which is "... a prepayment of individual income tax levied on resident individuals leaving the country."

Nonetheless, the exit tax is broader than a departure tax scheme. According to de Broe (2002, 19–78), in addition to being imposed on a person or a company leaving a jurisdiction to become a resident of another country, the exit tax also includes extended tax liabilities as well as recaptures previously enjoyed benefits. However, the extended tax liability is, in reality, more inclined to the application of citizenship-based taxation, which is often discussed together with the Bhagwati tax proposal.

Several countries have implemented the exit tax in their domestic tax provisions. In terms of design, the exit taxes can be divided into two categories, general (all taxpayers' assets are considered) and limited (only a few assets are considered). For instance, Canada imposes a general immediate exit tax which is intended for long-term residents. The tax base is calculated on assets that do not continue to remain in the Canadian tax net and are deemed disposed of before the migration. On the other hand, Chand (2013) specifies that the Netherlands applies a limited exit tax for long-term residents who have substantial shareholdings in companies.

Although considered as one of the instruments to protect the taxation rights of a country as it provides disincentives for emigration, the exit tax is not free of criticism. The main criticism against the exit tax lies mainly in the nature of its imposition, which is applied before an emigrant becomes a resident of another country and earns income there (ex-ante). There can still be options to defer payments from deemed disposal assets. However, considering that an individual who will emigrate from developing economies only has limited income and assets, their decision is barely affected by the presence or absence of the exit tax.

Moreover, the exit tax assumes that emigrants will earn a far better income than what they are paid for from deemed disposal assets. However, the emigrants may or may not obtain good returns in the host country (for instance brain-waste cases). Furthermore, exit tax does not adhere to the principle of ability to pay, as the emigrant has yet to obtain additional economic capabilities. ¹⁵ The exit tax is thus considered an inefficient and inequitable policy (Bhagwati, Dellalfar 1973, 94–101).

Challenges also arise from non-economic aspects. The exit tax is considered an instrument that may violate human rights as it prevents a person's mobility to attain a decent living. ¹⁶ In the context of the European Union, the exit tax is also frequently debated, in particular, in relation to the Treaty on the Functioning of the European Union (TFEU).

Finally, the administrative feasibility of exit tax collection in developing economies also faces challenges, especially in terms of immigration control and asset appraisal. In developing countries, the obligation to obtain a tax identification number does not apply in general and has no connection to immigration documents. Presumably, this also explains why the exit tax instrument is rarely applied in developing countries.

4.2.3. Tax Incentives

Some countries currently take the opposite measure, i.e. they provide tax incentives, to stop talented and skilled individuals from remaining in a jurisdiction and becoming residents in other countries. This incentive is expected to prevent waves of brain drain. Every country offers varied trial and error programs to obtain an effective design (Agunias, Newland 2012).

For instance, in 2019 Poland plans to abolish taxes for young skilled workers, to prevent them from immigrating to other EU countries.

The ability to pay is itself one component of the equity principle. See Pistone et al. (2019).

¹⁶ See Article 13 (2) of the Universal Declaration of Human Rights: "Everyone has the right to leave any country, including his own, and to return to his country." (https://www.un.org/en/universal-declaration-human-rights/)

This income tax revocation incentive is available for residents under the age of 26 who earn less than USD 22,207 (which is above the average wage of Polish residents). One argument for granting this facility is inseparable from the fact that around 1.7 million Polish residents have left the country in the past 15 years.

In the Southeast Asian region, Malaysia has launched the Malaysian Returning Expert Program. The Malaysian Government provides benefits for Malaysian professionals working abroad for at least three years, namely the option of a flat tax rate of 15% on employment income for a period of five consecutive years, tax exemption for all personal effects brought into Malaysia, as well as tax/duties exemption for up to a maximum of MYR 150,000 when purchasing a car. This program is considered quite effective in targeting those who have the option of working abroad (Del Carpio *et al.* 2016). From 2011 to 2018 this incentive was given to approximately 5,024 individuals.

In addition to the incentives provided to citizens, developing countries are also working on a strategy known as reverse brain drain, i.e. the movement of high-skilled individuals from developed countries to developing countries (Gupte, Jadhav 2014, 83–87). As suggested by Cavallini *et al.* (2018, 5), these efforts may encourage competition among countries to attract high-skilled individuals to obtain positive economic and social impacts.

The idea of a tax incentive instrument is frequently discussed as a complementary policy for the implementation of the Bhagwati tax proposal. On a side note, the application of the Bhagwati tax is prone to non-compliance by emigrants abroad. The monitoring, incomplete data, and administrative weaknesses of tax authorities in developing countries are factors that influence such non-compliance, regardless of the penalty feature when the high-skilled migrants return to the home country. Penalties may lead to concerns as they encourage people to remain abroad. To avoid this, Wilson (2008, 2385–91) suggests that tax incentives, for example tax reduction, can in fact be given to compliant taxpayers when they return to the home country.

From the perspective of the tax administration, the provision of tax incentives as a method to address brain drain is clearly more feasible than the other three policy options that require cooperation and/or changes in the international tax system (revenue sharing and Bhagwati tax) as well as reliable assets wealth profiling data (exit tax). The degree of difficulty in applying these incentives is determined by the evidence or documentation by the applicant regarding their eligibility, to the terms

¹⁷ For details on the program visit: https://www.talentcorp.com.my/initiatives/returning-expert-programme

and criteria proposed in the regulations, for example, if the incentives are granted to emigrants with a certain income or who work in certain sectors.

The use of tax incentive instruments is very likely to undermine the equity principle. These incentives may compromise the sense of fairness for citizens domiciled in the country, in particular, high-skilled individuals. A progressive individual income tax will only target educated human resources who generally earn a high income and are "proven" loyal and do not have, or are yet to have, the intention to work abroad. Consequently, this may decrease the trust of loyal high-skilled individuals and encourage their non-compliance or "provoke" them to find ways to obtain the same incentives.

The main criticism against the use of incentives instruments to address brain drain lies in its effectiveness. First, according to Beretta (2017), the competition of providing expatriates with facilities has increased. Today, more and more countries are offering special regimes for expatriates with certain criteria, by mitigating the implementation of their worldwide system, flat tax, etc. Assuming economic rationale, tax incentives provided by developing countries must at least provide a better situation for high-skilled individuals compared to the expected return, plus the additional incentives offered in developed countries.

Second, it seems that the motive for migrating abroad for young workers is not only better income, but also lifestyle and experience, as stated by Heckert (2015) and in the World Youth Report. Thus, the incentives provided to prevent or re-invite emigrants are less efficient, especially for young professionals.

4.2.4. Revenue Sharing

Revenue sharing is one of the policy options proposed by Desai *et al.* (2004), in addition to the exit tax and the Bhagwati tax proposal (global tax system). Bhagwati (1979, 28) states that the scheme may take the form of compensations paid by a developed country to a developing country disadvantaged by the brain drain or brain gains by a developed country from a developing country, notwithstanding the presence or absence of losses in the developing country.

Broadly speaking, revenue sharing schemes can be found in literature on fiscal decentralization, where there is an allocation of revenue from the center to regional governments or between regional governments. Considering that there are currently no international (supranational) organizations responsible for the fiscal area, namely an International Tax Organization, the notion of revenue sharing seems to be more difficult to implement. However, with pressure from the competition for high-skilled individuals and restrictions on migration from developing countries, there

exists a "coercion" to engage in bilateral tax-sharing agreements, as stated by Desai, Kapur, McHale (2004, 684).

5. PROSPECT FOR LARGE DEVELOPING ECONOMIES: SOME COMMENTS

Instead of formulating a final form and practical guidance from various policy options — in particular, for example regarding the implementation of the Bhagwati tax proposal that resembles citizenship taxation — this section explores several points that can help address the prospects of taxes in reducing brain drain in LDE. The points in this section are intended to stimulate further research.

5.1. The Bhagwati Tax Proposal and Non-Discrimination Rule

Bhagwati's proposal that was developed towards citizenship-based taxation opens the possibility of violations of the non-discrimination principle. In taxation, non-discrimination emphasizes the need for the same tax treatments in the same situations, as well as the justifications for different tax treatments in different situations. In the context of taxation, according to Holmes (2007, 400), the term discrimination is defined as a less favorable tax treatment of a particular tax subject compared to other tax subjects under the same conditions.

In the international tax system, the non-discrimination principle also acts as the most prominent forewarning and is stipulated in Article 24 of the OECD Model, which stipulates the avoidance of discrimination in specific conditions.¹⁸

With regard to Article 24, it is necessary to distinguish acceptable different treatment (legitimate distinction) and unacceptable treatment (unjustified discrimination). Examples of different acceptable treatment regulated in the tax provisions of many countries are differences in the imposition of taxes that rely on the taxpayer's ability to pay (ability to pay principle), i.e. as reflected in progressive rates. Different treatments become unacceptable if the objectives are at least based on economic considerations. In short, as argued by Adonnino (1993, 22), such treatment is applied arbitrarily.

The question is: to what extent can differences in citizenship justify different tax treatment?

On further inspection, discrimination in the context of Article 24 of the OECD Model may be defined as: (i) unequal treatment for the same (comparable) cases, or (ii) the same treatment for dissimilar (incomparable)

OECD Commentary on Article 24, Paragraph 2.

cases.¹⁹ In this context, the OECD expressly states that every country that carries out any tax treaty is prohibited from discriminating against the resident status in another contracting state, based on the status of nationality, in applying the tax treaty.²⁰

This is stated in Article 24 paragraph (1) of the OECD Model: "Nationals of a Contracting State shall not be subjected in the other Contracting State to any taxation or any requirement connected therewith, which is other or more burdensome than the taxation and connected requirements to which nationals of that other State in the same circumstances, in particular with respect to residence, are or may be subjected. This provision shall, notwithstanding the provisions of Article 1, also apply to persons who are not residents of one or both of the Contracting States."

Through Bhagwati's tax proposal, the use of citizenship-based taxation has the potential to violate the principle of non-discrimination. As the scheme provides different (dissimilar) tax treatment under the same conditions, namely where non-resident citizens and non-residents are treated differently, i.e. one home country has the taxing rights while the other does not.

5.2. Exit Tax Is Only Appropriate for Emigration Driven by Tax Motives

The experiences of various countries related to exit taxes provide an important lesson, i.e. even though the exit tax prevents the transfer of resident status for individuals, its application serves as an anti-avoidance provision (Kubicova 2016). Put differently, it acts as an instrument to prevent changes in resident status triggered by tax motives, either in the context of avoiding capital gains tax or an effort to seek lower tax burdens in other jurisdictions.

Such a statement can be proven by the implementation of the exit tax provision as one of the six measures initiated by the European Union in the Anti-Tax Avoidance Directive (ATAD). Unfortunately, none of the ten LDEs reviewed as case studies in this article apply an exit tax. However, lessons from similar developing countries, such as South Africa, have shown that the exit tax is intended to prevent emigration encouraged by tax motivation (Mazansky 2010).

From the perspective of large developing economies, the application of an exit tax would be more relevant if associated with high-net-worth individuals. This proposal is further driven by the notion that with the

According to IBFD (2005, 124), discrimination is defined as "In international tax context discriminations most often takes of the form of different treatment of taxpayers whose situations are comparable except in respect of characteristics such as nationality."

Nationality is defined as citizenship status for individuals or, for companies, the place where it is established.

non-optimal tax system and governance, changes in HNWI's resident status in developing countries may be motivated not only by tax factors but also related to efforts to cover up illicit financial flow, corruption in the political sector, and transnational crime (Buchanan, McLaughiln 2017, 8–9).

5.3. Bhagwati's Tax Proposal Without Earmarked Budget is Ineffective

The issue of brain drain alone cannot be completely resolved with citizenship-based taxation rights, embodied in the Bhagwati tax proposal. Citizenship tax can only address the prevention of potential revenue forgone from the tax base (citizens) that emigrate, but it is not necessarily effective in preventing emigration (loss of human resources). In essence, citizenship tax does not create a disincentive for high-skilled individual to emigrate, since the decision to emigrate may not be compelled by the tax factor in the home country as a push factor, implying that they can enjoy a high income in the host country while still contributing to the home country through taxes

Moreover, considering that brain drain is a loss for the home country, due to the loss of skilled human resources beneficial to economic development, revenue from citizens who become residents of other countries can only reduce the impact of brain drain if it is directly dedicated to improving the labor market, education, and R&D in the home country. Without an earmarked budget scheme, home countries can find themselves in a situation that resembles the "flypaper effect" (Crowley, Hoffer 2018). In the absence of an earmarked budget, the Bhagwati tax proposal cannot restore the pre-conditions of brain drain, but only serves as a "tool" to increase individual tax income revenue at the global level.

In non-benevolent or corrupt and authoritarian governments, revenue without an earmarked budget can also encourage inappropriate behavior. In reality, this discourages the government to invest in the provision of quality public goods, while concurrently "transfering" the government's responsibility to another country and encouraging emigration to transfer the "burden" of public goods provision, by allowing a maximum flow of emigration, thus (prospective) high-skilled citizens may (attend school and) earn income. In return, the government obtains tax revenue from the emigrants.

5.4. Prospects of Global Acceptability of the Bhagwati Tax Proposal

There are at least three things to consider regarding the prospects of implementing the Bhagwati tax. First, with the increasingly relevant concept of citizenship-based taxation, there will be potential for asymmetrical taxation rights in the future. Disputes and debates on international tax for related to personal and economic connecting factors may re-emerge.²¹

The fiscal preferences and tax sovereignty of each country seems to be too strong to simply "succumb" in order to address the brain drain issue. As such, middle-ground solutions are required, for instance, the abandonment of the principles of residence and citizenship, which would be replaced by time-tests that better reflect increased individual mobility (Beretta 2019, 107–10).

Second, there is a concern that with the transition to citizenship-based taxation, each country will compete to discourage the change of citizenship status (for home countries) or offer the change of citizenship status (for the host country). The former presumption is most likely true, while the latter is not necessarily the case. The migration policy and citizenship status of a country will be increasingly relevant in regard to a culture of openness and will be influenced by national security issues (Adamson 2006, 165–99). Thus, any matter that may "disrupt" the national security agenda is subject to long and careful consideration.

Third, the prospect of successful implementation of the Bhagwati tax will depend on how the proposal is linked to the world's main concerns. Accordingly, the brain drain issue and the Bhagwati tax proposal must be linked to a new development agenda (Brauner 2010), in which case, large developing economies, along with BRICS countries (Brazil, Russia, India, China, and South Africa), may play an important role in advocacy at the global level (Pistone, Brauner 2015, 385–92).

5.5. Citizenship-based Taxation Tests the Consistency of Developing Countries Concerning Favor Towards the Source Country

The Bhagwati tax proposal is heading towards taxation in favor of the citizen's country (highlighting personal attachment) whereas to date,

This primarily relates to dual residents. Article 4 paragraph (1) of the OECD and UN Model does not define resident taxpayers. Provisions concerning such a matter are stipulated in the domestic provisions of the two countries establishing the tax treaty. As such, what determines whether a tax subject is a resident taxpayer in the countries that enters into the treaties is based on the domestic provisions of the two countries. If the tax subject is a resident taxpayer in both countries (dual resident), Article 4 paragraph (2) and (3) provide guidance to address the dual resident issue through a tie-breaker rule that aims to prevent double taxation, hence the tax subject may only be a resident taxpayer in one country. Subsequently, the tie-breaker rule determines the residency status of an individual through the tests of a permanent home, vital interest, habitual abode, nationality, and through Mutual Agreement Procedure (MAP). See Schwarzenhofer (2005, 20) for further reference.

Presently, many countries have offered citizenship by investment, as practiced by Cyprus, Malta, Moldova in Europe, and Dominica, Grenada and St. Lucia in the Caribbean.

most of the capital importing countries – which are developing countries – tend to be proponents of the source country (highlighting economic attachment). On various occasions, developing countries often voice their demands for a fairer (greater) allocation of taxation rights to the source country as well as "accusations" against the OECD Model (Pistone, Brauner 2015, 480). The siding results from the differences between the OECD Model and the UN Model – as a representation of developing countries.

The demand of developing countries, as importers of capital, for greater taxing rights for the source country stands on the argument that active economic activities are, in essence, carried out in the source country (sometimes referred to as the market jurisdiction). Conversely, the taxing rights of the resident country, as the location of the capital owners, should be limited.

In the context of the Bhagwati tax, the position of developing countries (labor exporters) may differ. Are the arguments for granting taxation rights to developing countries also valid and in favor of the same principles when developed countries (capital exporters) claim their rights? This question is worth exploring and can lead us to other intense discussions, such as whether the host country (a developed country) will demand a withholding tax mechanism or not.

5.6. Revenue Sharing and Demand for an International Tax Organization

Revenue sharing is essentially made possible through the presence of global organizations in the tax sector. In 2015, at the UN Third International Conference on Financing for Development, held in Addis Ababa, Ethiopia, there was a discussion and plan to establish an International Tax Organization (ITO). ²³ G77 developing countries were initially eager to permanently transform the UN Committee of Experts on International Cooperation in Tax Matters (UN Tax Committee) into the ITO, as a global tax system formulation mechanism that no longer requires that the OECD play a role. This idea was challenged by developed countries. In the end, the forum only agreed to strengthen the UN Tax Committee's capacity, and not to its transformation.

In some literature, the ITO is expected to perform several functions, for instance, monitoring trends and statistics concerning the tax situation on a regular basis, acting as an international tax forum, providing advice and solutions to global tax problems, and supervising information exchange cooperation (Tanzi 2016). Although interesting, the notion of

²³ In this conference, 193 countries agreed to improve the performance of state revenue mobilization. These efforts are called for to finance 17 Sustainable Development Goals (SDG's) 2016–2030 agenda, as a further commitment of MDG's.

the ITO conflicts with tax sovereignty. The tax sovereignty is intended to maximize the welfare of the population, guarantee income redistribution, oriented towards national interests, side with the community, and guarantee democratic values (Dagan 2013).

Clearly, ITO would reduce the freedom of each country to design its tax system in accordance with its national orientation. Such an opinion is not fully acceptable. As a matter of fact, the ITO is believed to be able to guarantee tax sovereignty (Dietsch 2015). After all, the tax sovereignty of every country has been eroded without the ITO. The sovereignty of countries in designing corporate income tax policies has diminished. As pointed by IMF (2014, 13), governments are now unable to formulate tax policies in a "closed" environment, but consider the measures currently undertaken by other countries and how they may impact the economy. The ITO guarantees tax sovereignty to the same degree in all countries.

The idea of the ITO is increasingly relevant to the fact that tax non-compliance and fair allocation of taxation, caused by increased labor and capital mobility amid various tax systems of different countries, has become a global issue. Securing the tax base from erosion can now be categorized as one of the global public goods, not unlike environmental sustainability, the stability of international financial markets, global security, and others (Kaul *et al.* 2016). The ITO is an expected solution to the tragedy of commons, which in this case refers to fair share tax (Tanzi 2016, 256–59).

5.7. The Relevance of Substantive and Enforcement Jurisdiction

When Bhagwati submitted his proposal more than 40 years ago, the idea of supporting tax collection by the host country seemed utopian. In the course of time, the discussions regarding the development of international taxation, specifically in the context of the digital economy, underline the increasing relevance that the role of jurisdictions in collecting taxes that they are not entitled to.

Hellerstein (2003) proposes a new concept in terms of tax jurisdiction, with two jurisdiction categories, based on their power to tax, namely the substantive jurisdiction, related to the power of a state to impose a tax on the subject matter of an exaction; and the enforcement jurisdiction, related to the power of a state to compel collection of the tax over which it has substantive tax jurisdiction.²⁴

²⁴ As quoted by Hellerstein (2003): "Substantive jurisdiction to tax includes such questions as whether a state has the power to impose a tax on the income that a non-resident earns from sources within the state, or to impose a tax on goods or services purchased outside but consumed within a state. ... Enforcement jurisdiction includes such questions as whether a state has power to enforce the collection of a tax on income earned by a non-resident from sources within the state, or whether a state has power to enforce

This classification can be applied both in terms of income tax and consumption/value added tax and has four possible scenarios, namely (i) the substantive and enforcement jurisdictions are both available; (ii) the substantive jurisdiction is available but the enforcement jurisdiction is not; (iii) the substantive jurisdiction is not available, but the enforcement jurisdiction is; (iv) neither the substantive nor the enforcement jurisdiction is available. Problems arise if the combination does not occur symmetrically (both are available/not available).

In other words, the policy design of substantive and enforcement jurisdiction allocation should be one of the points to be formulated, especially in the context of the Bhagwati tax proposal. Furthermore, we should be aware that the principle of sovereignty prevents a country from claiming taxes in areas outside the country without strong taxation rights. Fortunately, at the international level, assistance in tax collection has been made possible by the 2003 revision of the OECD Model.²⁵

5.8. Promoting Tax Incentives as a Quick Response

Within the framework of tax competition, the tax incentive instrument is the best and most rational way for developing countries to help ensure their involvement in the global arena. Compared to Bhagwati's proposal, tax incentives are a relatively risk-free domestic instrument as opposed to the international tax system (for example, treaty override potentials).

Moreover, concerns about the massive development of tax incentives, which may lead to harmful tax practices frequently mentioned in the Base Erosion and Profit Shifting (BEPS) Project Action Plan 5, are groundless. Tax incentives related to the migration of residents are connected to substantial economic activity (OECD 2019). Notably, in the context of brain drain, the generally debated income is the salaries of employees that are part of active economic activities.

As such, to what extent can tax incentives be effective in addressing the issue of brain drain? The answer is unclear, given that in the context of brain drain, the motive is not always the tax factor in the home country. Nonetheless, efforts to design incentives that can exceed quantified returns and non-economic factors (lifestyle, ease of bureaucracy, etc.) are worth trying. To be more effective, as suggested by Del Caprio *et al.* (2016), it would be best if the tax incentives were embodied and combined with other non-tax incentives.

the collection of a tax on goods or services purchased by an in-state consumer from a remote vendor."

²⁵ See Article 27 and Commentary of the OECD Model Tax Convention regarding Assistance in the Collection of Taxes.

Tax incentives can also be designed to create an ecosystem that keeps high-skilled individuals in the home country, for example, cost-based tax incentives (e.g. for R&D activities, training costs for certain skills) or profit-based incentives (tax holiday for labor-intensive sectors). These incentives would encourage technological development, improve the quality of human resources, and ensure employment for certain skills.

6. CONCLUSION

Considering the situation in large developing economies regarding international migration, the use of tax instruments in addressing the brain drain, although weak, is justifiable, especially considering the fact that they also enjoy benefits from high-skilled emigration, ranging from high remittance rates, reduced unemployment, prevention of social unrest, and a large tertiary-educated population in their countries. There are at least five things that can be concluded from the assessment of the four policy choices.

First, there is no stand-alone tax policy that can optimally address brain drain, in the sense of reducing the number of high-skilled individuals who emigrate. An exit tax may serve as the best possible policy, however, considering that the majority of individuals from large developing economies do not yet have sufficient wealth and income, the imposition of an exit tax shortly before departure abroad will not have much effect. Moreover, the exit tax is more appropriate if associated with the issue of preventing tax noncompliance, such as tax avoidance and tax evasion.

Second, most policies focus more on the element of fairness to compensate for the "loss" caused by the host country. This is found in Bhagwati's tax proposal and revenue sharing, which prioritizes a "guarantee" of revenue for the home country. For large developing economies, this guarantee of revenue is certainly useful, but without an earmarked budget scheme to improve the economic situation and job opportunity, such a guarantee may encourage misallocation, and therefore the root causes of brain drain would remain.

Third, almost every available policy requires better coordination at the international level. Potential non-discrimination principle and dual resident violations (the Bhagwati tax proposal) and dependence on the existence of international tax organizations (revenue sharing) undermine human rights since they discourage migration (exit tax). In this context, tax incentives seem to be the most rational policy. One thing is certain, the role and voice of large developing economies are called for (though they may not necessarily be influential) in order to raise the issue of resolving brain drain in international forums.

Fourth, all policy options require closer collaboration with immigration agencies. All taxable events, as well as the enforcement of

these policies, require clearer information on the immigrants' home country, duration, time of return, migrants' economic capacity, and background. At the present, the capacity of the tax administration, especially the cooperation in information access among tax authorities and immigration authorities in developing countries may be suboptimal.

Fifth, each policy has the potential to produce unintended consequences. For instance, there is competition for nationality status and a wave of tax incentive competitions for highly— talented individuals. These two things will ultimately be unable to constrain migration rates.

In conclusion, notwithstanding the fact that the available options are quite promising, there is no ideal policy. In the end, the problems raised in this paper are expected to stimulate future research.

REFERENCES

- Adonnino, Pietro. 1993. General Report. 22 in *Non-discrimination rules* in international taxation. IFA Cahier Vol. 78b.
- Adamson, Fiona B. 1/2006. Crossing Borders: International Migration and National Security. *The MIT Press* 31: 165–199.
- Agunias, Dovelyn R., Kathleen Newland. 2012. Developing a Road Map for Engaging Diasporas in Development: A Handbook for Policymakers and Practitioners in Home and Host Countries. Geneva: International Organization for Migration.
- Ahmed, M. Mushtaque. 2019. Bangladesh Individual Taxation. *IBFD Country Surveys*: 12.
- Arslan, Cansin, Jean-Christophe Dumont, Zovanga L. Kone, Çağlar Özden, Christopher R. Parsons, Theodora Xenogiani. 2016. International Migration to the OECD in the 21st Century. KNOMAD Working Paper 16: 26–29.
- Avi-Yonah, Reuven S. 2015. And Yet It Moves: Taxation and Labor Mobility in the Twenty-First Century. 45–56 in *Taxation and Migration*, edited by Reuven S. Avi-Yonah and Joel Slemrod. The Netherlands: Wolters Kluwer Law & Business.
- BBC News. 2019. Nigeria: Why Is It Struggling to Meet Its Tax Targets? https://www.bbc.com/news/world-africa-49566927 (last visited 3 October 2019).
- Beine, Michel, Frederic Docquier, Caglar Ozden. 2009. Diasporas. Policy Research Working Paper 4984.
- Beine, Michel, Frederic Docquier, Hillel Rapoport. 2001. Brain Drain and Economic Growth: Theory and Evidence. *Journal of Development Economics*: 275–289.

- Beine, Michel, Frederic Docquier, Hillel Rapoport. 2008. Brain Drain and Human Capital Formation in Developing Countries: Winners and Losers. *The Economic Journal* 118: 631–652.
- Beretta, Giorgio. 8/2017. Italy From Worldwide to Territorial Taxation: Is Italy Now an Attractive Destination for Migrating Individuals?. *Bulletin for International Taxation* 71.
- Beretta, Giorgio. 1/2019. Cross-Border Mobility of Individuals and the Lack of Fiscal Policy Among Jurisdictions (Even) After the BEPS Project. *Intertax* 47: 107–110.
- Bhagwati, Jagdish N., Koichi Hamada. 1/1982. Tax Policy in the Presence of Emigration. *Journal of Public Economics* 18: 291–317.
- Bhagwati, Jagdish N. 1976. The International Brain Drain and Taxation: A Survey of the Issues. 34–38 in *The Brain Drain and Taxation: Theory and Empirical Analysis*, edited by Jagdish N. Bhagwati. Amsterdam: North Holland.
- Bhagwati, Jagdish N. 3/1979. International Migration of the Highly Skilled: Economics, Ethics, and Taxes. *Third World Quarterly* 1: 28.
- Bhagwati, Jagdish N., William Dellalfar. 1–2/1973. The Brain Drain and Income Taxation. *World Development* 1: 94–101.
- Brauner, Yariv. 1/2010. Brain Drain Taxation as Development Policy. Saints Louis University Law Journal 55: 45.
- Brücker, Capuano, Marfouk (2013). Education, gender and international migration: insights from a panel-dataset 1980–2010, mimeo.
- Buchanan, John, Lucilla McLaughlin. 2017. Revenue Administration: Implementing a High-Wealth Individual Compliance Program. *Technical Notes and Manuals*: 8–9.
- Cavallini, Simona, Rosella Soldi, Lorenzo Di Matteo, Beatrice Errico. 2018. *Addressing Brain Drain: The Local and Regional Dimension*. European Union.
- Cejie, Katia. 6/2012. Emigration Taxes Several Questions, Few Answers: From Lasteyrie to National Grid Indus and Beyond. *Intertax* 40.
- Chand, Vikram. 2013. Exit Charges for Migrating Individuals and Companies: Comparative and Tax Treaty Analysis. *Bulletin for International Taxation* 67.
- Christians, Allison. 2/2017. A Global Perspective on Citizenship-Based Taxation. *Michigan Journal of International Law* 38.
- Clemens, Michael A. 2014. A Case Against Taxes and Quotas on High-Skill Emigration. Working Paper 363.
- Colombo, John D. 23/1998. A Proposal for an Exit Tax on Nonprofit Conversion Transactions. *J. Corporation Law* 779.

- Crowley, George R., Adam J. Hoffer. 2018. Earmarking Tax Revenues: Leviathan's Secret Weapon?. in For Your Own Good: Taxes, Paternalism, and Fiscal Discrimination in the Twenty-First Century, edited by Adam J. Hoffer and Todd Nesbit. Arlington, VA: Mercantus Center.
- Dagan, Tsilly. 2018. *International Tax Policy: Between Competition and Cooperation*. United Kingdom: Cambridge University Press.
- Dagan, Tsilly. 2013. The Tragic Choices of Tax Policy in a Globalized Economy. in *Tax, Law and Development*, edited by Yariv Brauner and Miranda Stewart. Northampton: Edward Elgar Publishing.
- Darvas, Peter, Shang Gao, Bilal Bawany. 2017. Sharing Higher Education's Beyond the Few in Sub-Saharan Africa. Washington, D.C.: The World Bank.
- De Broe, Luc. 2002. The Tax Treatment of Transfer of Residence by Individuals. *International Fiscal Association*. De Hague: Kluwer Law International 2002.
- Desai, Mihir A., Davesh Kapur, John McHale. 2004. Sharing the Spoils: Taxing the International Human Capital Flows. *International Tax and Public Finance* 11: 681–684.
- De Man, Fernando, Tiiu Albin. 12/2011. Contradicting Views of Exit Taxation Under OECD MC and TFEU: Are Exit Taxes Still Allowed in Europe? *Intertax* 39.
- Del Caprio, Ximena V., Caglar Ozden, Mauro Testaverde, Mathis Wagner. 2016. Global Migration of Talent and Tax Incentives: Evidence from Malaysia's Returning Expert Program. *Policy Research Working Paper* 7875.
- Dietsch, Peter. 2015. *Catching Capital: The Ethics of Tax Competition*. Oxford: Oxford University Press.
- Docquier, Frederic. The Brain Drain from Developing Countries. 2014. *IZA World of Labor* 31: 3–10.
- Docquier, Frederic, Hillel Rapoport. 2011. Globalization, Brain Drain, and Development. *IZA Discussion Paper Series* 5590.
- Docquier, Frederic, Hillel Rapoport. 2007. Skilled Migration: The Perspective of Developing Countries. *IZA Discussion Paper Series* 2873: 15–16.
- Docquier, Frederic. 2006. Brain Drain and Inequality Across Nations. *IZA Discussion Paper Series* 2440.
- Dos Santos, Manon D., Fabien Postel-Vinay. 1/2003. Migration as a Source of Growth. *Journal of Population Economics* 16: 161–175.
- DSP-Groep. Study on the Levying and Collection of The Tax in Seven European Countries. https://www.dsp-groep.eu/projecten/the-2-pct-tax-for-eritreans-in-the-diaspora/ (last visited 2 October 2019).

- Dumont, Jean-Christophe. Gilles Spielvogel, Sarah Widmaier. 2010. International Migrants in Developed, Emerging and Developing Countries: An Extended Profile. OECD Social, Employment and Migration Working Papers No.114. Paris.
- Dumont, Jean-Christophe, Gilles Spielvogel. 2010. International Migrants in Developed, Emerging and Developing Countries: An Extended Profile. *OECD Social, Employment and Migration Working Papers* 114: 27.
- Elveren, Adem Y. 2018. *Brain Drain and Gender Inequality in Turkey*. Switzerland: Springer International.
- Egger, Hartmut, Josef Falkinger, Volker Grossman. 2007. Brain Drain, Fiscal Competition, and Public Education Expenditure. *IZA Discussion Paper Series* 2747.
- European Commission. https://ec.europa.eu/taxation_customs/business/ company-tax/anti-tax-avoidance-package/anti-tax-avoidancedirective en (last visited 4 October 2019).
- Fan, Simon C., Oded Stark. 2007. International Migration and Educated Employment. *Journal of Development Economics* 83: 76–87.
- Gadzo, Stjepan, Irena Klemencic. 2/2017. Effective international information exchange as a key element of modern tax systems: Promises and pitfalls of the OECD's common reporting standard. *Public Sector Economics* 41.
- Gaspar, Vitor, Laura Jaramillo, Philippe Wingender. 2016. Tax Capacity and Growth: Is there a Tipping Point? IMF Working Paper WP/16/234. International Monetary Fund Department of Fiscal Affairs, Washington, D.C., USA.
- Georgaki, Vera. 2010. Emigration Tax Consequences in Greece. *European Tax Studies*.
- Gibson, John, David McKenzie. 2011. Eight Questions about Brain Drain. *Policy Research Working Paper* 5668: 3–5.
- Goldin, Ian, Geoffrey Cameron, Meera Balarajan. 2012. Exceptional People: How Migration Shaped Our World and Will Define Our Future. New Jersey: Princeton University Press.
- Gordon, Stephen. 2016. Bargaining Power and the Incidence of Income Taxes on High Earners in Canada, SSRN Electronic Journal
- Grubel, Herbert B., Anthony D. Scott. 1966. The International Flow of Human Capital. *American Economic Review* 56: 273.
- Grunkorn, Wolfram, H.T. Thi Do, Quynh N.T. Nguyen. 2019. Vietnam Individual Taxation. *IBFD Country Surveys*: 7–8.
- Guarcello, Lorenzo, Jean Fares, Marco Manacorda, Furio C. Rosati, Scott Lyon, Christina A. Valdivia. 2008. School-to-Work Transitions: Regional Overview. 109–148 in *Youth in Africa's Labor Market*, edited by Marito Garcia and Jean Fares. World Bank Publications.

- Gupte, Medha, Kiran Jadhav. 2014. The Concept of Reverse Brain Drain and Its Relevance to India. 1/2014. *Journal of Research in Management & Technology* 3: 83–87.
- Hamzaoui, Ridha. 2019. Egypt Individual Taxation. *IBFD Country Surveys*: 8.
- Heckert, Jessica. 27/2015. New perspective on youth migration: Motives and family investment patterns. *Demographic Research* 33: 765–800.
- Hellerstein, Walter. 1/2003. Jurisdiction to Tax Income and Consumption in the New Economy: A Theoretical and Comparative Perspective. *Georgia Law Review* 38: 36.
- Holmes, Kevin. 2007. *International Tax Policy and Double Tax Treaties*. Amsterdam: IBFD.
- Huntington, Samuel P. 2/1996. Democracy for the Long Haul. *Journal of Democracy* 7: 3–13.
- IBFD. 2005. International Tax Glossary. Amsterdam: IBFD.
- IMF. 2014. Spillovers in International Corporate Taxation. *IMF Policy Paper*.
- IMF. 2018. Egypt: Time to Entrench Growth and Make it More Inclusive. *IMF News*. 23 January.
- IMF. 18/2018. Article IV Consultation Press Release and Staff Report. *IMF Country Report* 215: 33.
- Jennings, Ralph. 2017. Vietnam's Corruption Crackdown Is All about Protecting Its Economic Miracle From Its SOEs. *Forbes*. 19 December.
- Kane, Mitchell A. 2/2015. A Defense of Source Rules in International Taxation. *Yale Journal on Regulation* 32.
- Kapur, Devesh. 2004. Remittances: The New Development Mantra? *G-24 Discussion Paper Series* 29: 16.
- Kating, Philippe-Alexandre S. M. 2019. Congo (Dem.Rep.) Individual Taxation. *IBFD Country Surveys*: 8.
- Kaul, Inge. 2016. *Global Public Goods*. Northampton: Edward Elgar Publishing.
- Kauppinen, Ilpo, Olli Ropponen. 2018. Taxation, Migration and Location of Innovative Activity. Background Report for the Economic Policy Council.
- Keen, Michael, Jenny E. Ligthart. 117/2004. Information Sharing and International Taxation. *CentER Discussion Paper*.
- Koo, Angela. 2019. Indonesia Individual Taxation. *IBFD Country Surveys*: 8.
- Koo, Angela, H. Bukhari. 2019. Pakistan Individual Taxation. *IBFD Country Surveys*: 12.

- Kubicova, Jana. 2017. Exit Tax in the World of International Migration of Companies and Individuals. Conference Proceedings. 16th International Scientific Conference Globalization and Its Socio-Economic Consequences.
- Larking. Barry. 2005. *IBFD International Tax Glossary*, 5th ed.: 115–162. Lencho, Taddese. 2019. Ethiopia Individual Taxation. *IBFD Country Surveys*: 10.
- Lister, Matthew. 2016. A Tax-Credit Approach to Addressing Brain Drain. Saint University Law Journal 62.
- Martin, George. 2019. Poland scraps income tax for an entire generation to reverse brain drain and entice its workers back home. *Mail Online*, 1 August.
- Mason, Ruth. 2016. Citizenship Taxation. Southern California Law Review 89: 170–240.
- Mazansky, Ernest. 5/2010. South Africa South Africa's Exit Charges (And Something on Immigration Too). *Bulleting for International Taxation* 64.
- McCulloch, Rachel, Janet L. Yellen. 1975. Consequences of a Tax on the Brain Drain for Unemployment and Income Inequality in Less Developed Countries. *Journal of Development Economics* 2.
- Ocampo, Jude. 2019. Philippines Individual Taxation. *IBFD Country Surveys*: 12.
- Odimma, Sebastine. 2019. Nigeria Individual Taxation. *IBFD Country Surveys*: 8.
- OECD. 2003. *Model Tax Convention on Income and on Capital*. Paris: OECD Publishing.
- OECD. 2019. Harmful Tax Practices 2018 Progress Report on Preferential Regimes. Paris: OECD Publishing.
- OECD. 2018. Resumption of Application of Substantial Activities for No or Nominal Tax Jurisdictions BEPS Action 5. Paris: OECD Publishing.
- Ordine, Patrizia, Giuseppe Rose. 2011. Inefficient Self-Selection into Education and Wage Inequality. *Economics of Education Review* 30: 582–597.
- Ozden, Caglar, Maurice Schiff. 2006. International Migration, Remittances, and the Brain Drain. Washington DC: Palgrave Macmillan.
- Pistone, Pasquale, Jennifer Roeleveld, Johann Hattingh, Joao Felix Pinto N., Craig West. 2019. 10 in the component of the equity principle. Fundamentals of Taxation: An Introduction to Tax Policy, Tax Law and Tax Administration. The Netherlands: IBFD.

- Pistone, Pasquale, Yariv Braune. 2015. BRICS and the Emergence of International Tax Coordination. The Netherlands: IBFD.
- Pinto, Dale. 2007. Exclusive Source or Residence-Based Taxation Is a New and Simpler World Tax Order Possible?. *Bulletin for International Taxation*. 61: 277–291
- Pomp, Richard D. 1985. The Experience of the Philippines in Taxing Its Non-resident Citizens. *Faculty Articles and Papers* 77: 245–286.
- Rannveig, Dovelyn, Kathleen Newland. 2012. *Developing A Road Map for Engaging Diasporas in Development*. Washington DC: International Organization for Migration.
- Roudgar, Iraj. 2014. *The Global Brain Drain: Theory and Evidence*. Saarbrucken: Lambert Academic Publishing.
- Roudgar, Iraj, Cameron K. Richards. 2015. The Policy Challenges of the Global Brain Drain: Addressing Dilemmas of Contributing Push-Pull Factors. *International Journal of Public Policy* 11: 73 85.
- Schiff, Maurice. 2018. Brain Drain-Induced Brain Gain and the Bhagwati Tax: Are Early and Recent Paradigms Compatible? *IZA Discussion Paper Series* 11551.
- Schwarzenhofer, Renate M. 2005. Dual Resident Companies in International Tax Law. 20 in *Source versus Residence in International Tax Law*, edited by Hans-Jorgen Aigner and Walter Loukota. Vienna: Linde Verlag.
- Seid, Yared, Ali S. Alemayehu, Nuru Seid. 2016. Ethiopia: An Agrarian Economy in Transition, in *Africa's Lions: Growth Traps and Opportunities for Six African Economies*. Washington D.C: Brookings Institution Press.
- Shah, Shreyas. 2019. India Individual Taxation. *IBFD Country Surveys*: 10.
- Solimano, Andrés. 2013. Remittances by Emigrants: Issues and Evidence. CEPAL – SERIE Macroeconomía del desarrollo 26: 15.
- Stark, Oded, Christian Helmenstein, Alexia Prskawetz. 2/1997. A Brain Grain with a Brain Drain. *Economic Letters* 55: 227–234.
- Stewart, Aeris. 2019. Poland's Youth Tax Cut Creates Dueling Incentives. *Tax Foundation*.
- Tanzi, Vito. 2016. Lakes, Oceans, and Taxes: Why the World Needs a World Tax Authority. 256–259 in *Global Tax Fairness*, edited by Thomas Pogge and Krishen Mehta. Oxford: Oxford University Press.
- TalentCorp. *Returning Expert Programme* for Malaysians Who Are Away From Home.
- https://www.talentcorp.com.my/initiatives/returning-expert-programme (last visited 3 October 2019).

- Thani, Rabi. 2006. Brain Drain in IDB Member Countries: Trends and Development Impact. *IDB Occasional Paper* 12.
- The Heritage Foundation. 2019. Index of Economic Freedom: Democratic Republic of Congo. https://www.heritage.org/index/pdf/2019/countries/democraticrepubliccongo.pdf (last visited 4 October 2019).
- Tiebout, Charles M. 10/1956. A Pure Theory of Local Expenditures. *Journal of Political Economy* Vol. 64 No. 5: 416–424.
- United Nations. 2019. International Migrant Stock 2019. https://www.un.org/en/development/desa/population/migration/data/estimates2/estimates19.asp. (last visited 2 October 2019).
- United Nations. 2013. World Youth Report on Youth & Migration. New York: United Nations Publication.
- United Nations. 1948. Universal Declaration of Human Rights
- Wagner, Don. 2000. Do Tax Differences Cause the Brain Drain. *Policy Options*.
- Wamsley, Terrie L., Alan L. Winters. 2005. Relaxing the Restrictions on the Temporary Movement of Natural Person. *Journal of Economic Integration* 20: 688–726.
- Wilson, John D. 2011. Brain-Drain Taxes for Non-benevolent Governments. *Journal of Development Economics* 95: 68–76.
- Wilson, John D. 2008. A Voluntary Brain-Drain Tax. *Journal Public of Economics* 92: 2385–2391.
- World Bank. Age Dependency Ratio. 2019. World Development Indicators. https://data.worldbank.org/indicator/ny.gdp.mktp.kd.zg /last visited 3 October 2019).
- World Bank. GDP Growth. 2019. World Development Indicators. https://data.worldbank.org/indicator/SP.POP.DPND (last visited 3 October 2019).
- World Bank. 2017. Publication: Nigeria Economic Update: Beyond Oil, Key Drivers for Sustainable Growth. https://www.worldbank.org/en/country/nigeria/publication/nigeria-economic-update-beyond-oil-key-drivers-for-sustainable-growth (last visited 2 October 2019).
- World Bank. 2018. *India Development Update: India's Growth Story*. New Delhi: The World Bank Publishing.
- World Bank. 2019. Pakistan Customs: Mobilizing Domestic Revenues for Economic Development. *Government Notes*: 1.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019. UDC 336.22:314.15(100); 330.341:005.952

CERIF: S141

DOI: 10.5937/AnaliPFB1904068B

Giorgio Beretta, PhD*

'FIXING' THE SOCIAL CONTRACT: A BLUEPRINT FOR INDIVIDUAL TAX REFORM**

In the face of population ageing and demographic decline, nowadays all countries compete for an increasingly valuable asset: human capital. Indeed, the drain of human capital from one country to another concerns not only highly-skilled individuals seeking job opportunities abroad, but also pensioners relocating to sunnier and more tax-friendly jurisdictions. Absent global action, the risk of uncoordinated and unilateral measures taken by countries to increase and protect their own tax base, with adverse effects both from the inter-nation and the intranation equity perspective, is very concrete. So far, however, neither the OECD nor the European Union have developed specific policies or measures in the domain of individual taxation. Arguing that scope of reform exists also in this field, the article explores various policies and measures as a blueprint for individual taxation reform, with the double aim to curb tax competition among countries and fix the crumbling social contract.

Key words: Human capital. – Individual taxation. – Migration. – Social contract. – Tax policy.

^{*} Researcher, gberetta@liuc.it; giorgio.beretta90@gmail.com.

^{**} An early version of this article was presented at the Tax Aspects of the Brain Drain' conference, hosted by the Faculty of Law at the University of Belgrade, on 11 October 2019. The author would like to acknowledge the faculty members for organizing the conference and all the participants at that event for their valuable comments and remarks. The usual disclaimers apply.

'Society is indeed a contract ... As the ends of such a partnership cannot be obtained in many generations, it becomes a partnership not only between those who are living, but between those who are living, those who are dead, and those who are to be born.'

(Burke 1790)

'It is no longer the call to 'Give me your tired, your poor, your huddled masses'; now we ask for your alert, your privileged, your brainy, your talented. Our machines can do the menial work. Today the emphasis is on technical skill, sophisticated training and adaptability to modern society.'

(Perkins 1966, 617)

'A place in the sun and a tax-free pension'

(Somerset Webb 2015)

1. INTRODUCTION

The launch and subsequent delivery of the broad and multifaceted Base Erosion and Profit Sharing (BEPS) 15 Action Points in 2013–2015 undoubtedly marked a turning point for international taxation (on BEPS, see Christians, Shay 2017; Brauner 2014). After BEPS, in fact, no one can seriously hold the traditional view of a completely sovereign autonomy of countries in tax matters (for a discussion, see Rocha, Christians 2017). As a matter of fact, major theoretical developments in tax policies are now achieved not only through political and legal processes undertaken at the national level, but also in a multilateral setting and with the increased participation of non-governmental actors. In the past the OECD has been and, certainly still is, the major organization to act as a central hub for shaping international tax policies (see, especially, Cockfield 2005).

The action spearheaded by the OECD and undertaken by all countries participating to the Inclusive Framework on BEPS,² however, has narrowly focused on closing tax loopholes exploited by multinational

¹ An example of the increasing intervention of non-governmental actors in a global tax governance is given by the Platform for Collaboration on Tax, launched in April 2016 by the International Monetary Fund (IMF), the Organisation for Economic Co-operation and Development (OECD), the United Nations (UN), and the World Bank Group (WBG).

² The OECD/G20 Inclusive Framework on BEPS was established in 2016 as a means to ensure interested countries and jurisdictions, including developing economies, can participate on an equal footing in the development of standards on BEPS-related issues, while reviewing and monitoring the implementation of the OECD/G20 BEPS

enterprises (MNEs) and has sought to establish a new international tax order in the field of corporate taxation only (see, especially, Christians 2016).³ Remarkably, no action has so far been taken at the international level in the realm of individual taxation. The same has indeed occurred in the European Union (EU), where the fight against Harmful Tax Competition (HTC), since the establishment of the Code of Conduct Group in 1997, has only revolved around the identification and elimination of preferential tax regimes designed for companies and other legal entities.⁴

Such dearth of action is rather surprising given that, although revenue losses for national governments due to international tax evasion and avoidance are far greater in the corporate sector, the number and extent of threats arising in the field of individual taxation are by no means negligible.⁵

The author indeed posits that three distinct challenges — each of which is somehow referred to in the three passages quoted in the epigraph — deserve, in particular, closer attention. The first challenge is related to the threat posed to the social contract by the combined effects of population ageing and demographic decline in nearly all developed countries, which contribute to a widening divide across generations and urge governments all around the world to search for additional sources of revenue. The second challenge is related to the phenomenon of the brain drain, which sees countries fiercely competing among themselves for increasingly valuable assets such as human capital and poaching one another's pool of talented individuals. The third challenge is related to the increasingly large wave of pensioners who migrate from one country to another in search of a milder climate and often a more tax-friendly environment, which causes a revenue drain in the country where pension income was built up and/or from which it is paid out.

Project. As to October 2019, over 130 countries and jurisdictions are collaborating on the implementation of the BEPS 15 Action Points.

³ Notably, in pursuit of the BEPS goals, countries have committed to implementing four minimum standards, respectively concerning measures on Harmful Tax Practices (HTPs) (Action 5), on Tax Treaty Abuse (Action 6), on Country-by-Country (CbC) Reporting (Action 13), and on a Mutual Agreement Procedure (MAP) (Action 14), all of which, however, relate only to corporate taxation (see OECD/G20, 2019a).

⁴ This in spite of the fact that the Preamble of the Resolution on a Code of Conduct, of 1 December 1997, explicitly contemplated the possibility to tackle HTC practices also with regard to 'special tax arrangements for employees' (see European Commission 1998, 1). A similar plea was then reiterated by the Commission in its 2012 Communication titled 'An Action Plan to Strengthen the Fight against Tax Fraud and Tax Evasion', but it did not actually lead to the enactment of any measure in this field (see European Commission 2012a, 7).

⁵ Notably, revenue losses for governments due to BEPS practices by MNEs are conservatively estimated by the OECD at around 4–10% of global corporate income tax revenues or USD 100–240 billion annually (see OECD 2019a).

While all these three challenges indeed point to the need to undertake global action in the field of individual taxation, with a discussion of each of them provided in the following, in terms of proposals, the article mostly focuses on migration of pensioners and cross-border taxation of pension income. Despite such a narrow context, the author submits that the proposed policies and measures may offer valuable suggestions for rethinking individual taxation on a more general scale.

The article is organized as follows. Section 2 provides background information on the increasing challenges faced by the implicit social contract, which underpins the Welfare State currently adopted by nearly all developed countries. In particular, the discussion centres around the threats posed by a widening divide across different generations. Section 3 traces the main causes and consequences of the brain drain and the battle for human capital which is fiercely being waged by countries worldwide. Section 4 describes the phenomenon of migrating pensioners as well as the main features of the different pension taxation regimes. Section 5 deals with taxation of pension income on an international plane, with focus on the treatment currently provided under the OECD Model. Exploration of the tax treatment of pension income at the international level is used for individualizing possible policies and measures to be enacted in the field of individual taxation. This task is undertaken in Section 6, where a blueprint for individual tax reform is laid down, and pros and cons of each proposed measure are closely compared. Section 7 concludes.

2. THE SOCIAL CONTRACT UNDER THREAT AND THE WIDENING INTERGENERATIONAL DIVIDE

We are arguably entering an age of increasing global instability and social disillusion, both of which may be seen as prominent hallmarks of the end of the globalization thrust and the beginning of an opposite 'deglobalization' era (see, especially, van Bergeijk 2019; James 2017). The symptoms of a growing instability and disillusion are variably expressed in politics, society and the economy, in so far as all these areas are experiencing a surge of nationalist and protectionist movements, fuelled by popular grievance and general distrust of elites (see, in this regard, Lagarde 2019). In the aftermath of the financial crisis of 2008, it has in fact become common for people, especially the middle-class in developed countries, to have declining perceptions of well-being and trust in the future, 6 whereas the global wealthiest one percent has gained

⁶ According to OECD (2019b, 13), due to nearly stagnating wages, growing lifestyle costs and housing prices, rising job insecurity in the middle of fast-transforming labour markets, 'today the middle class looks increasingly like a boat in rocky waters'.

enormously throughout the past decades (for different perspectives in this regard, see Smith et al. 2019; Piketty 2013).

Rising inequality – both at the national and international level – is certainly a major source of government and individual concerns (see, for example, Wilkinson, Pickett 2019; Stiglitz 2015), as indeed those worries are further exacerbated by gloomy forecasts of employment conditions in the near future due to the rapid pace at which epochal phenomena such as automation (see, for a discussion, Baldwin 2019; Ford 2015) and population ageing 7 are occurring.

The Welfare State, adopted after World War II by nearly all developed countries, since it is seen as a valuable weapon against inequalities in society, is currently under tight scrutiny. This is largely due to the social contract implicitly agreed upon between generations, which underpins the Welfare State and, arguably, contributes to holding a society together (for a perspective on the situation in this regard in the United Kingdom, see House of Commons 2016, 8–23).

The intuitive idea of such an intergenerational social contract is that the redistributive mechanism underpinning the Welfare State justifies the obligation of the current productive generations to finance the health, pension and care services of the older generations, by arguing that future generations will provide the same kinds of benefits once the current generations retire (see Hammer, Istenič, Vargha 2018, 22). In this way, the Welfare State facilitates solidarity across different generations or age cohorts, via financial transfers to the old, mainly in the form of pensions, and to the young, mainly in the form of education, both of which are funded principally by taxing the current working-age population (see Resolution Foundation 2018, 25–27).

But there is a catch. In principle, everyone is to pay in during their working life, drawing down in early years and retirement, for a broadly neutral lifetime result. However, the amount of transfers and benefits provided in return may well change over time, as indeed do tax rates and the size of generations that are contributing or withdrawing. As a result, over their lifetime span, different generations can end up with net gains or net losses, a circumstance that is very much capable of skewing the redistributive mechanism underpinning the Welfare State (see Gardiner 2016, 7).

⁷ Tellingly, by 2020, for the first time in history, there will be more people on the planet over the age of 65 than under five (see He, Goodkind, Kowal 2016, 3).

⁸ As early as 2000, Avi-Yonah (2000, 1578) warned that 'globalization leads to a more pressing need for revenues at the same time that it limits governments' ability to collect those revenues. This dilemma threatens to undercut the social consensus about the value of the Welfare State that underlies modern industrialized societies and to create a backlash against the globalization that produces too many overall benefits'.

Presently, there is in fact a widespread consensus that the social contract is not being honoured for today's younger generations and that, in particular, the Baby Boom generation, commonly identified as individuals born between 1945 and 1965, are receiving a net gain over later coming generations, such as those of the Generation X, i.e. individuals born between 1965 and 1980, and the Millennials, composed of those born between 1980 and 2000.

Worries on this matter concentrate, in particular, on this latter age cohort. Tellingly, the Resolution Foundation (see Gardiner 2016, 5) has revealed gloomy economic forecasts for those belonging to that generation, signalling that Millennials are 'the first generation that has so far earned less than the one before at every age' and warning that, if productivity growth remains as low as now, 'Millennials are at risk of becoming the first ever generation to record lower lifetime earnings than their predecessors'. ¹⁰ On a similar strain, European Commission (2017, 12) has flagged increased concerns that today's young people in the EU and their children may actually end up worse off than their parents. Concerns also surround the future pensions of current workers, whose social sustainability is indeed put under a severe test, in so far as it is not clear whether the amount of the present contributions will provide adequate living conditions for tomorrow's retirees (see, especially, Scarpetta, Blundell-Wignall 2015). On a broader perspective, there is also a risk that a growing intergenerational divide would widen inequalities and wealth gaps existing in society, therefore the overall importance of inheritances and private transfers between generations is expected to grow (see Resolution Foundation 2018, 114–117). 11

Such dire prospects for today's younger generations are indeed the ultimate fruit of various ongoing trends in society and the economy. The

⁹ This classification of generations follows Willetts 2010. One should caution, however, that defining different generations inevitably entails an element of arbitrary choice, as long as, for instance, those individuals born immediately before a generational dividing line may well dispute their implicit association with those born20 years earlier, for example, but not with those born only one year later.

¹⁰ Such gloomy prospects, however, are contested by others (see, for example, Ganesch 2016), who point out, for instance, that economists indeed 'cannot account for the dazzling consumer gains that come with technology and competition multiplied by the passage of time', perhaps embodied at best by 'all the facilities now inherent to a smartphone' which 'would have cost a teenager in 1980 a king's ransom in separate, clunky machines'.

¹¹ See also Bangham (2018, 3–6), pleading for the elimination of the UK current inheritance tax and its replacement with a lifetime receipts tax to be levied on recipients with fewer exemptions, a lower tax-free allowance and lower tax rates, whose revenues are to support a GBP 10,000 'citizen's inheritance' – a restricted-use asset endowment for all young adults, from the age of 25, to sustain skills, entrepreneurship, housing and pension savings.

first challenge is related to population ageing, due to a combination of an increased life expectancy and a decreasing trend in birth rates in nearly all developed countries, ¹² both phenomena that indeed are expected to intensify in the coming decades, so that a growing demand for health, pension and care services will have to be sustained by the fiscal revenues extracted by a shrinking working-age population, thus increasing the socalled 'dependency ratio', measuring the number of pensioners per working age person (see Resolution Foundation 2018, 87–89). 13 Next, it comes the inequality challenge, with an increasing share of wealth globally owned by older generations, who have managed to shield their income and assets from the financial crisis of 2008 better than the younger generations (see Gardiner 2016, 23-25). The third challenge is related to poor job prospects for the young, who, mainly due to fast-paced automation, experiences increasing challenges in finding an employment, at a time when overall job quality, particularly in terms of work stability and benefits provided, has been reduced dramatically (see, especially, OECD 2019c).

As a result, a new divide is ripping society apart and it is based on age, in so far as when a person was born increasingly matters in determining their present and future living standards. ¹⁴ This situation, of course, generates significant backlashes – often depicted even in terms of 'intergenerational warfare' (see, most notably, Willetts 2019; Pickard 2019) – across generations and in society, further fuelled by a misleading propaganda on both sides (see, especially, Sternberg 2019; Bristow 2019). Older generations are thus depicted as a gerontocracy of the early-retiring and asset-rich, in contrast to precariously housed and insecurely employed younger generations, ¹⁵ whereas the latter are accused of living frivolously and have even been caricatured for consuming avocado toast (!) and

¹² Indeed, as revealed by He, Goodkind, Kowal (2016, 15), birth rates in all countries, with the exception of African ones, are already below the so-called 'population replacement level', which is the number of children per woman needed to sustain population replacement.

Against this backdrop, it may be contended (see, for example, European Commission 2018a) that an ageing population eventually favours private expenditure on a whole new set of goods and services, from connected health devices to age-friendly universities, all of which contributing to the flourishing of the so-called 'silver economy'.

Evidence of such growing divide between the old and the young became apparent with the Brexit referendum, which indeed showed that British politics is deeply polarized by age, with a substantial majority of older people voting for leaving the EU, while a large majority of younger generations voting for remaining in the Union (see Norris 2018). It should also be noted that, as their own population grows older, the political weight in all developed countries becomes increasingly tipped in favour of older generations.

For instance, the New York Times columnist Thomas Friedman (Friedman 2010) has gloriously railed against 'a Grasshopper Generation', one that 'has eaten through all that abundance like hungry locusts', whereas David Willetts (Willetts 2010),

priced coffee, instead of working and saving for the future as, supposedly, former generations did (see Levin 2017).

3. BRAIN DRAIN, TALENT AND THE INTERNATIONAL BATTLE FOR HUMAN CAPITAL

New kinds of wars are being waged by many countries all around the world for hoarding an increasingly valuable asset: human capital. Human capital can broadly be described as all the wealth of knowledge, skills, competences and attributes – which, overall, might be labelled as 'super talent' – that a few of individuals are endowed with and that facilitate the creation of personal, social and economic prosperity, being all of these preconditions for the flourishing of the 21st century 'knowledge society' (for a conceptualization, see Drucker 1993). As a proxy for all these endowments, educational attainments of those individuals are generally used. ¹⁸

Amid those international wars and battles (see, in this regard, Brücker *et al.* 2012), countries' victories and losses against one another are measured by means of inbound and outbound flows, i.e. by looking at the overall number and quality of the endowments of individuals permanently moving in or out the territory of the given country. Indeed, this two-way flow is neither necessarily nor under all circumstances well-balanced. Quite the contrary, such flow can be one-way. If this 'human capital' exchange is overall positive, i.e. more highly-skilled individuals are moving in rather than out, the country has a 'profit' or, more appropriately, a 'brain gain'. On the other hand, if for a given country the

chair of the UK Resolution Foundation, has claimed that 'the Baby Boomers took their children's future'.

Although the origins of the expression can be traced as back as to Adam Smith (1723–1790), the modern usage of the term 'human capital' is generally attributed to Gary S. Becker (1930–2014), especially in regard to his influential book *Human Capital: A Theoretical and Empirical Analysis, with Special Reference to Education*, first published in 1964.

According to Shachar, Hirschl (2013, 72), '[t]he desire to be great, to make a lasting mark, is as old as civilization itself. Today, it is no longer measured exclusively by the size of a nation's armed forces, the height of its pyramids, the luxury of its palaces, or even the wealth of its natural resources. Governments in high-income countries and emerging economies alike have come to subscribe to the view that something else is required in order to secure a position in the pantheon of excellence: it is the ability to draw human capital, to become an "IQ magnet", that counts'.

¹⁸ In this connection, however, it should be noted that the category that is used to qualify an individual as highly-skilled is related to the possession of tertiary education, an element that by itself is very crude, in so far as it includes in this category even individuals with (only) practical and technical education degrees. For an overview about education classification at the international level, see UNESCO 2012.

said balance is overall negative, i.e. more highly-skilled individuals are moving out rather than in, it faces a 'loss' or, more appropriately, a 'brain drain' (see Boeri 2012, 1).

The expression 'brain drain' was first coined by the British Royal Society (see Royal Society 1963) to narrowly describe the outflow of scientists and technologists from the United Kingdom to both the United States and Canada in the 1950s and early 1960s. 19 However, presently, the term is more broadly used to illustrate the departure of highly-skilled individuals – thus, not necessarily scientists or technologists – from their own countries to others where usually wages and life conditions are more favourable overall, or are at least perceived as such. 20 As a break-down of this compound expression suggests, the word 'brain' refers to the wealth of knowledge, skills, competences and attributes with which the emigrating individuals are believed to be endowed. The word 'drain' implies that the rate of those leaving a country is far greater than the normal or desirable level of departures from a country. The link between these two words means that the departure of the most talented and highly-skilled individuals from a country actually occurs at an appreciable rate (see Giannoccolo 2009, 2).

Brain drain is indeed the source of major concerns for governments and policy makers in the countries of origin (see, for example, *The Italian Insider* 2019; Filipovic 2019), which especially complain about efficiency losses to their economy or, even, about the shortage of talented people in specific economic sectors (e.g. in the healthcare or education sector), in so far as those nations blame the country of arrival for poaching their own base of talented individuals, whose education and training were often financed by means of fiscal revenues, so that an export of its 'human capital' effectively becomes a sunk investment for the country of origin.²¹

¹⁹ If the emigrant is an unskilled individual, then one could perhaps speak about 'muscle drain' rather than 'brain drain' (see Pomp 1985, 250, 260 and 286).

Compare the definition of 'brain drain' contained in the Cambridge English Dictionary ('the situation in which large numbers of educated and very skilled people leave their own country to live and work in another one where pay and conditions are better') with the narrower one included in the Collins English Dictionary ('the movement of a large number of scientists or academics away from their own country to other countries where the conditions and salaries are better'). For their own account, EU institutions (European Commission 2019) define 'brain drain' as 'the loss suffered by a country as a result of the emigration of a (highly) qualified person'.

²¹ Tellingly, a 2019 report prepared by the Westminster Foundation for Democracy (2019, 23) for the UK government estimates that the 'sunk' cost of education of emigrants from a country such as Serbia in a single year is more than the total annual earnings from the IT services exported by that country. On the other hand, it could be contended (see Boeri 2012, 9) that 'selective immigration policies increase individual incentives to invest in human capital in the sending countries, so that the impact of migration on human capital formation in the country of origin may not be so strong'.

Although the general thrust of the conventional view is that the emigration of human capital is detrimental to a country, the actual validity of such a statement is open to discussion, as it is related to an empirical question whose answer varies from case to case (see, especially, Kapur, McHale 2005; Commander, Kangasniemi, Winters 2004). Moreover, literature also points out that, to the extent that the brain drain allocates human capital resources more efficiently, such phenomenon is likely to benefit more people globally (see Sykes 1992, 1). From another perspective, it is also contended that the brain drain is nothing more than the free exchange occurring across country borders, in as much as goods and services flow in and out a country (see, in this regard, Carens 1987),²² which states professing a liberal creed certainly cannot obstruct, at least if they have committed to respect fundamental human rights such as freedom of movement, which is even enshrined in several international charters and declarations. ²³ Lastly, there are additional phenomena related to brain drain, such as remittance, diasporas and returns, whose net effects on the country of origin are difficult to assess (for a discussion, see Faini 2017; Wei, Balasubramanyam, 2006; Dustmann, Fadlon, Weiss, 2011).

Various reasons can be traced at the roots of the brain drain phenomenon. The main determinant of the brain drain is generally recognized as being the wage differentials existing between countries, which may function as either a push or pull factor for both inbound and outbound migration patterns (see, especially, Borjas 2001). Another traditional factor encouraging migration is related to cross-country unemployment differentials (see, especially, Piracha, Vadean 2009). The quality of public institutions and standards of living may also help explain the decision of an individual to migrate from one country to another (see, especially, Cooray, Schneider 2016). Other non-financial benefits could equally motivate talented and highly-skilled individuals to move from a country, such as the existence of centres of excellence in a specific economic sector in the country of arrival: in a sense, 'brains' go where other 'brains' are (see Tesón 2008, 902).²⁴ Intended as such, the brain drain – like any other economic phenomenon – is governed by the law of supply and demand and by the law of comparative advantages (see Tesón 2008, 902).

Many authors (see, for example, Freeman 206; Pritchett 2006), however, criticize that the current wave of globalization includes 'everything but labour.

²³ See e.g. Universal Declaration of Human Rights (10 December 1948), Art. 13 (2); International Covenant on Civil and Political Rights (16 December 1966), Art. 12 (2); European Convention on Human Rights (4 November 1950), Art. 2 (2) Prot. No. 4. An alternative, although nowadays minoritarian, view instead regards emigration as a privilege to be granted by the country of origin, rather than a right to which each individual is entitled (for a discussion, see Risse 2012, 152–166).

²⁴ For a discussion about 'brain hubs in the United States, i.e. innovation clusters where the average GDP and patents for new technologies are higher, see Moretti (2012, 82–88).

The patterns of 'brain' migration also vary. Brain drain may affect developing countries in favour of developed countries, such as non-OECD countries in favour of OECD countries (see, especially, Docquier, Lohest, Marfouk 2007). However, the phenomenon does also occur among OECD countries, as the experience of outward individual movements in developed nations like Italy and New Zealand conspicuously demonstrates (see Brücker *et al.* 2012, 43–47). Further, brain drain can be caused by a reversal of social and economic conditions or unexpected political decisions occurring within a country, which, apparently, is the case of the brain drain that is greatly feared after the Brexit vote in the United Kingdom (see Fazackerley 2018). Lastly, it should be duly considered that migration patterns are likely to change over time, as demonstrated by the history of Europe during the 20th century, when it went from an emigration to an immigration continent, (see Hatton, Williamson 1994, 533–539).

The brain drain phenomenon is a tangible reality also within the EU, where the free movement of workers is one of the four economic freedoms to which Union citizens are entailed and it is a right guaranteed by Article 45 of the Treaty on the Functioning of the European Union (TFEU).²⁵ In the EU, reasons at the roots of the brain drain relate, mostly, to wage and employment differentials across the Member States as well as different EU regions. While migration patterns mainly followed an East to West route, from countries of the former Soviet bloc – all joining the EU in 2004 and in 2007 – to Western EU-founding Member States during the first decades of the 2000s, the past few years have instead signalled a clear increase of emigration rates from the South to the North of the Old Continent, especially those involving highly-skilled individuals.²⁶

Quite intuitively, individual migration patterns also have a significant impact on fiscal revenues. Emigration of individuals, especially

Notably, Article 45 TFEU stipulates that 'freedom of movement for workers shall be secured within the Union', which entails, *inter alia*, the right 'to move freely within the territory of Member States'. EU law, in fact, guarantees both the right of an individual to leave his Member State of origin and the right to enter and live in another Member State. Therefore, freedom of movement of workers is related to the emigration country as well as to the immigration country, both of which are indeed precluded from hindering cross-border movements and discriminating workers based on their different nationality. However, in so far as tax systems and economic rights arising from the Welfare State of the various Member States differ, the economic consequences of an individual's decision to move from one country to another may well be discouraging, which is an issue that the Commission has long committed to tackling but has failed to address so far (see European Commission 2010).

 $^{^{26}\,}$ For a more detailed description of past, present, and future trends concerning migration of highly-skilled individuals within the EU, see European Commission 2018b. On labour migration from Eastern to Western Europe in the past decade, see Atoyan et~al.~2016.

those talented and highly-skilled, who presumably earn an above average salary, erodes the tax base and dampens fiscal revenue in the country of origin. The situation is exactly opposite for the country of arrival, as it later sees an increase in its own tax base and fiscal revenues (with specific regard to individuals moving from India to the United States, see the economics analysis by Desai *et al.* 2009). It is no wonder, therefore, that some measures of control – particularly, in the form of taxes to compensate or promote development in the 'losing' country, i.e. the country of the 'brain' departure²⁷ – have long been proposed as a way to restore global or inter-country 'fairness' (see, especially, Bhagwati 1976; Brauner 2010).

Similar considerations apply to the current situation within the EU, where Member States should arguably endeavour to harmonise their own fiscal policies rather than fiercely competing against each other as they actually are (see, in this regard, Alcini, Gros 2019), as clearly shown by the increased number of special tax regimes for incoming individuals enacted by Member States in recent years (for a discussion of these regimes, see Beretta 2019a; Beretta 2019b; Beretta 2017; Arginelli, Avella 2017; Ribes Ribes 2017; Bader, Seiler 2015; Cassiano Neves 2010; van Zantbeek 2010; Roxburgh 2006).

From a tax policy perspective, what is particularly worrying of this growing trend is that countries seems to design these special tax regimes before even understanding the real nature of their own social and economic troubles, thus ending up granting tax benefits to individuals based on rather objectionable – if not constitutionally flawed – criteria (see Kostić 2019a).²⁸

4. PENSIONERS ON THE MOVE: RETIRING ACROSS BORDERS

If there is a word that perhaps should be retired nowadays, it would be 'retirement' (see Ezra 2019). Just as individual working lives have changed dramatically over the past several decades, so has the conventional wisdom about retirement. Notably, time and again experts advise to

Measures in this regard may be taken by the country of origin, the country of arrival or, even, adopted as the result of international cooperation (for a discussion, see Kapur, McHale 2005).

Reportedly (see Tax Foundation 2019), as from 1 August 2019, Poland introduced a blunt exemption from income tax for all Poles aged below 26 and earning less than a given annual salary (approximately EUR 22,500) as a measure to induce Polish youth to remain in its territory. For its own account, starting in 2019, Portugal (see República Portuguesa 2019) introduced a special tax regime (called 'Programa Regressar'), providing a 50% reduction of employment income tax, which is specifically designed to encourage the return of former residents who have fled the country in the last years. Indeed, in this as in other cases, one may well question the differentia specifica that may justify providing a special tax treatment based solely on the odd criteria such the age or the former residence of an individual (see Kostic 2019a).

prepare for the 100-year life (see, in this regard, Gratton, Scott 2016), in which the three traditional stages of life – education, work and leisure – are going to be subverted.²⁹ And indeed, anecdotal evidence indicates that droves of people are already 'unretiring' and going back to work (see Cavendish 2019, 71–99; Harding 2018; Span 2018), being that such a decision is favoured by the shrinking of the working-age population, due to declining fertility rates in nearly all developed countries (see He, Goodkind, Kowal 2016, 15).

The circumstances that such that pensioners are, generally, not only healthier but also wealthier; as a matter of fact, the two major sources of private wealth, i.e. illiquid and liquid assets such as houses and pensions, are steadily in their hands – has also brought emigration within the financial reach of many of them (see Gardiner 2016, 33–39). Moving, therefore, is no longer necessarily a young person's game. Indeed, statistics show that an increasing number of pensioners are retiring in countries other than the ones in which they spent their entire or a substantial part of their working life, staying there for at least a considerable part of the year (see, for example, Cruccu 2018; KeepTalkingGreece 2018; Tilbrook 2018; Gehring 2017; ONS 2017; The Economist 2017).

Although there is very little research into migration patterns of the elderly population and, indeed, the exercising of the right to free movement across the EU by 'economically inactive' citizens, who have reached their retirement age has received scant attention so far,³¹ for those individuals the decision to migrate seems to be favoured by a general loosening of occupational and social ties that normally bind an individual to a certain place of residence during their entire working life (see, in this regard, Pyte, Rahmonov 2019). In the EU, cross-border mobility of pensioners is further encouraged by the obligation imposed by EU law upon Member States to eliminate national restrictions that

Notably, the three stages of life, i.e. education, work and leisure, were first laid down by Harold Entwistle in *Education, Work and Leisure* (Routledge 1970).

Against this background, Young (2017, 3, 16, 40) contends that 'people moving across state lines are young', since 'people move not because they are cold and calculating but because of where their opportunities lie', which, according to that author, is more likely to materialize when a person is still relatively young and is trying to establish a career. Conversely, the propensity to move supposedly decreases when a person reaches the peak of their career, due to a variety of factors, such as growing family responsibilities and the accumulation of human, social and cultural capital in the place where the person has settled.

Nevertheless, one recent groundwork study (see Gehring 2019) has pinpointed three main reasons for a retiree to cross country borders: (1) increased free time and the absence of work obligations, (2) availability of budget flights for most destinations as well as the possibility to rely on distance-shortening technologies such as videocalls, and (3) in the EU, the right to free movement across Member States.

impede or discourage the provision of pension portability without objective justification or that are not proportionate to their own aims.³²

Although, in principle, the brain drain phenomenon only pertains to highly-skilled individuals of working age, the outbound flow of pensioners – indeed, a 'drain', and hence a parallel with the brain drain phenomenon may be established – is also a source of concern for governments and policy makers, in so far as it generates a loss of fiscal revenues for the country of origin and a corresponding gain for the country of arrival, a circumstance that induces countries to tightly compete in offering those individuals the most favourable tax and non-tax conditions (in general, with regard to the fiscal effects of migration by an individual from one country to another, see Beretta 2019a; Betten 1998).

Furthermore, even from a purely intra-country perspective, emigration of pensioners undermines the effectiveness of deferred taxation of pension income and leads those countries to shift the fiscal burden on the young, thus impairing intergenerational fairness (see Redonda *et al.* 2019; Xu 2015, 75–77). Given that private pensions are among the most significant financial assets currently held in the household sector, the importance of pensions as a source of revenue for countries is quite obvious and, indeed, it is expected to also remain significant in the near future (see Gardiner 2016, 33–39).

The background of this discourse is that pay-as-you-go (PAYG) regimes, in the form of compulsory contributions, are still a relevant part of pension regimes, in many countries, as well as in most EU Member States, the most common among those schemes being the EET system (Exemption for the individual contributions, Exemption of the savings and capital market returns accumulated in the pension fund, and Taxation upon disbursement of pension wealth once an individual retires).³³ Under

³² Indeed, the Commission issued a communication on the elimination of tax obstacles to cross-border provision of occupational pensions in 2001, followed by an update in 2003, and launched several infringement proceedings against a number of Member States in the subsequent years (see European Commission 2001a; European Commission 2001b; European Commission 2003). Among the infringement proceeding launched by the Commission over the years, it is worth pointing out the case against Denmark, which resulted in a decision rendered against that Member State by the Court of Justice of the European Union in 2007 (CJEU, case C-150/04, Commission of the European Communities ν. Kingdom of Denmark, ECLI:EU:C:2007:69). The lack of pension portability and double taxation of cross-border pension have long been identified as a significant obstacle to cross-border movements and a factor of lost income for EU citizens (for a discussion, see Williams 2001; Gutmann 2001). More recently, a regulation on a pan-European Personal Pension Product (PEPP) was passed by the European Parliament in 2019 (see European Parliament 2019).

³³ There are indeed various types of old-age pensions and all are generally underpinned by three tiers of retirement income, i.e. public, occupational and private, whose quantitative significance however varies markedly across countries as well as

such scheme, pensions become taxable for the first time when benefits start being paid out. Alternatives to the EET system are the ETT system (Exempt contributions, Taxed investment income and capital gains of the pension fund, Taxed benefits) and the TEE system (Taxed contributions, Exempt investment income and capital gains of the pension fund, Exempt benefits), although other combinations are also possible.³⁴

While in a closed economy setting the aforementioned pension taxation regime works quite smoothly, the migration of a retired person from one country to another instead creates havoc in such a scheme, in so far as the emigrating person pays no taxes in the country of origin, despite the employment activity and the income thereof to which pension contributions can be traced having generally been made in that country (see Starink 2016, 6–13).

As such, the cross-border aspects of private pensions is characterized primarily by a potential conflict between two distinct elements: (1) the ability of an individual to accrue a pension without impediments during the contribution and accumulation phases, regardless of where one person works or lives, and (2) the tax claim by the country of origin over payments made from pensions accrued under favourable tax provisions upon disbursement (see Kavelaars 2007).

The quasi-contractual argument that lies behind such a claim is evident: the emigrating pensioner has received a tax benefit from his country of origin and, therefore, has a duty – a moral one, at least – to pay it back to the country from which he departs (see Brokelind, Axmin 2017, 261). As a matter of fact, the flow of pensioners and, accordingly, of pension income between two countries could very well not be reciprocal and, in some cases, may represent a relatively substantial net outflow for the country of origin of these elderly migrants (see Staats 2015).

Indeed, this quasi-contractual argument gains further traction if the pension income goes untaxed not only in the country of origin but also in the country of arrival, effectively achieving international double non-taxation. Notably, this situation occurs where the emigrating pensioner

between individuals within a country. Notably, in order to render their pension systems more sustainable over time, countries generally motivate employers and employees to support occupational and private retirements savings with various forms of tax preferences or direct subsidies. For an overview of the current and prospective pension systems at the international level, see OECD 2017a.

According to Dilnot, Johnson (1993, 2), 'three main transactions constitute most private pension schemes and it is these transactions which are the possible occasions for taxation: (1) contributions into the scheme, from employer or employee, (2) income derived from the investment of contributions, and (3) payment of retirement benefits from the accumulated fund'. For a discussion of the various pension taxation regimes in the EU, see Brokelind 2014, which concludes that 'cross-border workers may have a lot to lose compared to non-migrant workers, just because of a lack of simplicity in mixing the systems'.

moves from an EET country to a TEE country, in so far as the differences between the pension taxation regimes that are in place in the two countries in question ultimately lead to double non-taxation of the particular income. Indeed, double non-taxation of pensions may also occur if a tax treaty is in place between the country of origin and the country of arrival and such a treaty follows the OECD Model, but the latter country provides for an exemption or simply does not actually tax the relevant pension income (see Beretta 2019b). This situation can be best understood by reviewing the current regime for taxation of pension income under double tax treaties, which is done in the next section.

5. TAXATION OF PENSION INCOME UNDER TAX TREATIES

Under the current version of the OECD Model Convention on Income and on Capital (2017), pension income from past private employment is addressed in Article 18. This article provides for a single – for some, indeed, 'deceptively simple' (see Brown 2019, para. 1.1.1.)³⁶ – taxation rule, stipulating that pension and similar remuneration, paid in consideration of past private employment, are taxable only in the state of the individual recipient.³⁷

Notably, double taxation and non-taxation as a result of an individual moving across state borders were dealt with at a seminar during the 2008 IFA Congress in Brussels (see De Broe, Neyt 2009). For an analysis of similar issues in the EU, see European Commission 2016.

Notably, according to Brown (2019, para. 1.1.1.), such 'deceptively simplicity' is related to the fact that Article 18 of the OECD Model 'provides no definition and, of course, no source rule. In fact, unlike most of the other distributive rules in tax treaties, the provision is not limited to pensions that arise in one state and are paid to a resident of the other state'. Lacking a tax treaty definition, pursuant to Article 3 (2) of the OECD Model, the term 'pension' must be interpreted in accordance with the domestic law of the jurisdiction imposing the tax, unless the context requires differently. Furthermore, as long as the OECD Model does not include a specific provision regarding social security benefits or annuities, it might be doubtful whether, in a concrete situation, those items of income fall under Article 18 or not (see, most recently: CJEU, case C-372/18, Ministre de l'Action et des Comptes publics v. Mr and Mrs Raymond Dreyer, ECLI:EU:C:2019:206, concerning the actual characterization of contributions paid by an individual resident in France to a Swiss social security scheme). Moreover, since Article 18 of the OECD Model provides for no taxation by the source state, it also does not contain any source rule. Accordingly, the allocation rule contained in Article 18 is not limited geographically, which means that all payments that fall within the definitional scope of Article 18 are governed by such rule, without any regard to where those payments actually 'arise'.

³⁷ Article 18 of the UN Model indeed contains two alternative provisions, i.e. (A) and (B), for taxation of pension income from past private employment. Notably, these two alternatives reflect very distant tax policies. The first alternative (A) includes a general rule that follows the corresponding OECD Model provision. The second alternative (B), instead, ensures taxation by the state of residence of the recipient and the state of which

As an allocation rule, Article 18 closely follows the residence principle. The taxing rights of the source state are therefore completely disregarded. On the other hand, pursuant to Article 19 (2) of the OECD Model, pension income from past government employment is taxable only in the source country, which is identified as the country where the government services were in fact rendered.³⁸ Importantly, the OECD Model and double tax treaties in general focus only on the actual disbursement of pension income, disregarding the contribution and accumulation phases.

Historically, taxing rights over private pension income shifted from the source country to the residence country at time of the drafting of the 1946 London Model, under the sponsorship of the League of Nations.³⁹ The main reason for the overhaul is related to the fact that the same shift occurred for taxation of income from movable capital and that private pensions were ultimately regarded as just a form of income from capital.⁴⁰ The new allocation of taxing rights among the source and residence countries indeed gained further confirmation in all subsequent updates of the OECD Model⁴¹ and, eventually, the rule was upheld by ensuing discussions which took place inside the various Working Party Committees through the years.⁴²

the payer is a resident. It is worth noting that both alternatives provide for exclusive taxation of social security payments by the source state.

³⁸ Blank, Ismer (2015, 252–253) indeed suggest bluntly deleting this provision from the OECD Model, in so far as they argue that 'the paying state principle', on which this provision is based, creates a lot of complexities as well as opportunities for tax arbitrage and that, furthermore, a great deal of simplification could be achieved by providing a single rule that applies to all pensions, from both private and government past employment. Along the same lines, see Lang (2007).

Although it is not entirely clear what was the reason taxing rights over private pension income were allocated to the source state instead of the residence state prior to 1946, it should be noted that the 1927 League of Nations Draft Convention also proposed to extend the treatment that had applied only to public pensions – i.e. taxation by the state from which payment was made – to also include private pensions. The Commentaries to Article 8 of the 1927 Draft Convention (League of Nations 1927, 16 [4130]), in fact, explained this decision by stating that 'it appeared both right and practical that all pensions should be made subject to the same rules'. As it happened, the treatment of private pensions provided under the 1927 Draft Convention had little effect on the drafting of actual tax treaties between countries (see Brown 2019, para. 1.2.1.1.).

⁴⁰ See League of Nations (1946, 28 [4348]) reasoning that '[i]n the London Draft, private pensions and life annuities are made taxable in the country of fiscal domicile of the creditor, as in the case of interest from debts'. For a discussion, see Starink (2016, 8).

⁴¹ As recalled by Brown (2019, para. 1.2.2.), in truth, the United Kingdom made an attempt to add a subject-to-tax test to the provision that would have been then included in the 1963 OECD Draft, but it only gained the support of the United States.

⁴² See OECD (1973, 6) pointing out that 'the article as it stands does not seem to have given rise to difficulties'.

Despite, as a rule, exclusive source-based taxation displays a number of strengths, 43 four broad justifications are usually found for providing exclusive residence-based taxation of pension income from past private employment. Notably, those reasons relate to:

- (1) the ability-to-pay principle, as its concrete assessment depends on the worldwide income of the individual taxpayer and it is assumed that personal and family circumstances of the pensioner are better evaluated by the residence state, which, therefore, is also able to ensure personal income taxation of the individual taxpayer on a net basis. ⁴⁴ On the other hand, taxation of pensions at source is likely to result in excessive taxation, especially if the source state imposes a final withholding tax on the gross amount of pension payments; ⁴⁵
- (2) the need to fund expenses associated with an aging population, especially for health, pension and care services available to pensioners, whose costs are to be borne by the residence state (see Kavelaars 2007; Blum 1999, 656–657);⁴⁶
- (3) easiness of tax administration by the competent authorities, as long as significant hurdles might arise in the case of individuals who have worked in more than one state, changed residence during their career, or

⁴³ Notably, the main advantage of exclusive source-based taxation is related to the existence of a clear causal link between pension and private employment income, which implies that it is reasonable to tax pensions, as a manifestation of income subject to a suspensive condition, in the very same country where employment income is also taxed. See UN Model Tax Commentary on Article 18 (2017), para. 11. Noteworthy, exclusive source-based taxation is provided under the multilateral Nordic Convention. See Denmark-Faroe Islands-Finland-Iceland-Norway-Sweden Income and Capital Tax Convention (Nordic Convention) (1996) (as amended through 2008), Art. 18 (1).

OECD Model Commentary on Article 18 (2017), para. 17. The Commentaries on Article 18 were amended in the 2005 Update of the OECD Model, following discussions among representatives of Member States at the OECD level (see OECD, 2003). Notably, similar considerations can be found also in the case-law of the CJEU (see e.g. CJEU, case C-279/93, Finanzamt Köln-Altstadt ν Roland Schumacker, ECLI:EU:C:1995:31, para. 32).

⁴⁵ It should be noted, however, that in *Hirvonen* (CJEU, case C-632/13, Skatteverket v. Hilkka Hirvonen, ECLI:EU:C:2015:765, para. 49) the CJEU ruled that the refusal by the source state to grant non-resident taxpayers, who obtain the majority of their income from the source state and who have opted for the taxation at source regime, the same personal deductions as those granted to resident taxpayers under the ordinary taxation regime, does not constitute, by itself, a discrimination contrary to EU law, in particular where the non-resident taxpayers are not subject to an overall tax burden greater than that placed on resident taxpayers.

⁴⁶ In this connection, Kemmeren (2001, 32) draws a distinction between the production of income and its consumption, arguing that payment of consumption taxes provide sufficient compensation for the public services offered to emigrated taxpayers in the new country of residence. This argument, however, is rejected by other scholars (see, especially, Starink 2016, 12).

derived pensions from funds established in a state other than the one in which they worked;⁴⁷

(4) simplification of tax compliance obligations for individual taxpayers, since exclusive residence-based taxation enables emigrated individuals to deal with income tax rules and tax authorities of only one country.⁴⁸

Although exclusive residence-based taxation, as the relevant taxing rule, is mandated by Article 18 of the OECD Model and, as seen, a series of justifications for its adoption can be found, actual tax treaty practice shows that allocating taxing rights to the source state is equally possible.

Notably, a study conducted by the IBFD in 2014 (see Wijnen, de Goede 2014) highlighted that, up until 2013, out of 1,811 tax treaties included in the survey, seven tax treaties concluded between two OECD countries provided for exclusive source-based taxation, whereas 25 of them allocated non-exclusive taxing rights to the source state, limited to a certain percentage, ranging between 10% and 25%. As for tax treaties concluded between an OECD and UN country, 44 tax treaties provided for exclusive source-based taxation, whereas 31 of them allocated non-exclusive taxing rights to the source state, limited to a certain percentage, ranging between 10% and 25%.

The tendency to attribute at least some private pension income taxing rights to the source state is indeed growing, in particular among pension-exporting nations like the northern countries in the EU. Denmark, for instance, terminated its tax treaties with France and Spain in 2009, after repetitive failures to negotiate some form of source-based taxation of private pension income with those countries.⁴⁹

Along the same lines, recent tax treaties concluded between the Netherlands, on the one hand, and respectively, Ireland and Germany, on the other, the latter of which came into effect in 2016 (in contrast, the Dutch-Irish income tax treaty is not yet in force), provide for source state taxation of private pensions exceeding, respectively, EUR 25,000 and EUR 15,000 per annum.⁵⁰

OECD Model Commentary on Article 18 (2017), paras. 19–19.2.

⁴⁸ OECD Model Commentary on Article 18 (2017), para. 20.

⁴⁹ Notably, the income tax treaties with Spain and France were terminated by Denmark, effective 1 January 2009. See Dyppel 2011, reporting that 'from a Danish perspective, it is crucial that future treaties contain provisions resulting in a more balanced allocation of rights to tax pension income ... As neither France nor Spain seems to show consideration for the Danish taxation of pensions as a whole and conclude a new treaty with a provision in line with this view, the Minister does not expect new treaties to be entered into in the near future'.

⁵⁰ Ireland-Netherlands Income and Capital Tax Treaty (signed on 13 June 2019, not yet in force), Art. 17 (2); Germany-Netherlands Income Tax Treaty (1 Jan. 2016), Art. 17 (2).

Exclusive residence-based taxation, compounded with the adoption of an EET taxation system of private pensions by most countries, ultimately leads to a 'fairness dilemma'. On the one hand, by bilaterally agreeing to such a regime, the country of origin in fact forgoes all its potential fiscal revenues. On the other hand, the emigrating pensioner is effectively double-taxed if the country of origin tries to close the tax income gap by, for instance, taxing pension contributions, whereas the country of arrival, following the treaty, also taxes the pension benefits upon receipt by the individual (see Genser, Holzmann 2016, 10–15). Indeed, in the EU, this situation is further complicated by the encroachment of the freedom of movement across different Member States to which all EU citizens – including 'economically inactive' ones such as pensioners – are entitled.⁵¹

Nevertheless, as a result of the growing willingness and capacity of pensioners to move across country borders, maintaining an exclusive residence-based taxation for income from private pensions in double tax treaties has become increasingly problematic.⁵² Indeed, if, at time when exclusive residence-based taxation was conceived, the amounts of pensions paid cross-border were relatively small in relation to other types of cross-border payments such as dividends, interest and royalties, so that the costs for the source state of giving up its own taxing rights were not seen that great, this is no longer the case in the current political, social and economic landscape (see Brown 2019, para. 1.1.1).

To add to this problem, in a few cases the residence state provides for a blunt exemption or simply does not tax the relevant pension income.⁵³ This situation occurs in Portugal, which has a special tax

Indeed, as clarified by the CJEU, first, in *Pusa* (CJEU, case C-224/02, Heikki Antero Pusa v. Osuuspankkien Keskinäinen Vakuutusyhtiö, ECLI:EU:C:2004:273, para. 18) and then in *Turpeinen* (CJEU, case C-520/04, Pirkko Marjatta Turpeinen, ECLI:EU:C:2006:703, paras. 13–23), the exercising of an economic activity is no longer a requirement for an emigrant to have treaty standing, as the combination of Union citizenship and the right of residence avails the 'economically inactive' citizen of a right to national treatment in the state of destination and of a right of non-restriction in the state of origin. See also CJEU, case C-300/15, Charles Kohll and Sylvie Kohll-Schlesser v. Directeur de l'administration des contributions directes, ECLI:EU:C:2016:361, para. 28.

⁵² See OECD Model Commentary on Article 18 (2017), stipulating that '[t]he globalisation of the economy and the development of international communications and transportation have considerably increased the international mobility of individuals, both for work-related and personal reasons. This has significantly increased the importance of cross-border issues arising from the interaction of the different pension arrangements which exist in various States and which were primarily designed on the basis of purely domestic policy considerations. As these issues often affect large numbers of individuals, it is desirable to address them in tax conventions so as to remove obstacles to the international movement of persons, and employees in particular'. For a discussion of movements of pensioners across country borders inside the EU, see Del Sol, Rocca 2017.

Granting an exemption to foreign-source pensions does not necessarily imply the complete forfeiture of fiscal revenues, in so far as a country may well expect an

regime providing for a 10-year exemption for foreign-source pension income.⁵⁴ Repeated failures to negotiate a new tax treatment for private pensions by Finland with the Portuguese tax authorities led the Scandinavian state to terminate the income tax treaty with Portugal as from 1 January 2019 (see Ambagtsheer-Pakarinen 2018).

Indeed, the number of variations on and deviations from any of the standard models, or even the alternatives included in the Commentaries to the OECD and UN Models, as well as the circumstance that countries are normally prone to negotiate 'bespoke' provisions combining multiple provisions from the Commentaries on Article 18, or ignore them altogether, indicate the existence of scope for reforming the current tax treatment of pension income under double tax treaties (see Brown 2019, para. 1.1.1).

6. A BLUEPRINT FOR INDIVIDUAL TAX REFORM

6.1. Rethinking individual taxation for the 21st century challenges

There are indeed good reasons to believe that international wars and battles for human capital will intensify in the next few decades. Fast-paced automation combined with the increasing specialization of developed countries in human capital-intensive activities are, in fact, expected to spur the general demand for labour by highly-skilled individuals and, thus, also the extent of the brain drain phenomenon. Also, population ageing along with the growing willingness and capability of pensioners to move across borders are predicted to impose tight budget constraints and, thus, put additional pressure on the Welfare State of most developed countries. Ultimately, the aforementioned two phenomena might be in correlation, in so far as challenges related to an ageing population spur the general demand for workers, especially highly-skilled individuals, from abroad.

Uncontrolled flows of people across borders, being either highly-skilled or elderly individuals, could well increase the extent of strategic tax competition among countries, thus draining the brain and fiscal resources of many nations (see Dagan 2018, 59; Rixen 2011, 449). This

increase in collected revenues through indirect taxation. This is indeed the case of Portugal which, reportedly, experienced a sharp increase of new residents in the last years, largely due to its preferential tax regime for foreign-source pensions (see Wise 2019). As stated, Kemmeren (2001, 32) takes the view that payment of consumption taxes by the emigrated individual in the country of arrival offers sufficient compensation for the public services provided by that country to those individual.

⁵⁴ Código do Imposto sobre o Rendimento das Pessoas Singulares (CIRPS) [Portuguese Individual Income Tax Code], Arts. 16 (8–12), 72 (6) and 81 (4–6). For a discussion of the Portuguese special tax regime, see Cassiano Neves 2010.

is even truer inside the EU, given the freedom of movement that workers and Union citizens are entitled to under EU law. As a result of such cross-border movements, wealth gaps between those who leave and those who remain – the former not necessarily being the younger, the latter not necessarily being the older – are also likely to widen.

Against this background, the author submits that a coordination strategy to address the current disarray existing in the realm of individual taxation at the international level is highly desirable and that the allocation rules as provided under current double tax treaties, not only for corporate but also for individual taxpayers, should be duly reconsidered.⁵⁵ Accordingly, in the following, various policies and measures that might constitute a blueprint for individual tax reform are analysed and their respective pros and cons are in turn evaluated. Importantly, the ensuing discussion mostly focuses on Articles 18 of the OECD Model and taxation of cross-border pension income from past private employment, the author arguing that such an examination might offer valuable suggestions for rethinking individual taxation on a more general scale. Also worth noting is that the following sections only deal with how the taxing rights between the source and the residence state, i.e. the country of origin and the country of arrival in case of migration of an individual from one country to another, could be allocated, without further discussing how the proceeds resulting from such allocation should be used by the countries concerned. As a further word of caution, given that each of the proposed policies and measures warrants an article of its own, only the main elements and arguments of each are hereinafter delineated.

6.2. Extended residence-based taxation

A first measure to address the current challenges encountered in the field of individual taxation may consist in granting taxing rights to the country of origin of the emigrants, being either highly-skilled or elderly individuals, over income received by those persons while abroad. Notably, the possible strategies that the country of origin may implement in order to protect its own tax base against tax-induced migration of individuals can essentially be divided into three broad categories: (1) exit taxes, (2) extended tax liabilities, and (3) recaptures of previously enjoyed benefits, deductions or deferrals (see De Broe 2002, 23).

For a thoughtful examination and some reconsideration of allocation rules for employment income under tax treaties, in particular with regard to Article 15 of the OECD and UN Models, see Kostíć 2019b.

⁵⁶ As a matter of international law (see Norr 1961, 432), countries are free to assert jurisdiction over the worldwide income of an individual abroad, provided that a 'minimum connection' or 'nexus' exists between the country and the individual or the income concerned.

'Exit taxes' or 'departure taxes' can be summarily described as taxes that the country of origin levies upon a person when they cease to be its resident. It is worth nothing that becoming a resident of the other Contracting State under a tax treaty's tie-breaker rule is, in most circumstances, equated to an expatriation. The primary purpose of an exit tax is to ensure that, following the change of residence by a taxpayer, the income accrued while that person was a resident does not escape taxation altogether because of the excluded or limited taxing rights permitted to the source state (i.e. the country of arrival) under its domestic law or by virtue of tax treaty obligations.

As regards their theoretical design, two main types of exit taxes can be distinguished: namely 'general' and 'limited' exit taxes. General exit taxes are fiscal liabilities imposed on all accrued-but-not-yet-realized income (e.g. capital gains) of the emigrated individual. Limited exit taxes are instead imposed on accrued-but-not-yet-realized items of income from certain types of property, such as income from the alienation of a substantial shareholding.

Exit taxes are quite problematic. By imposing an exit tax, a state might in fact be found in breach of its tax treaty obligations. Indeed, an exit tax in regard to pension rights imposed by the Netherlands was found inconsistent with its tax treaty obligations, which, pursuant to Article 18 of the OECD Model, attributed taxing rights on pension income exclusively to the state of residence of the individual recipient. Exit taxes might also be troublesome in relation to obligations deriving from EU law, in so far as those measures amount to illegitimate restrictions on one or more of the four freedoms (for an introduction to this topic, see Helminen 2019, Chapter 2). Since they are immediately charged to the emigrated individual, exit taxes also present complications in cases of temporary migrations, i.e. where an individual moves from one country to another and remains therein only for a few years, to the extent that the individual taxpayer, once returned, is not able to recover the tax paid to the country of origin upon emigration.

The second type of defensive measures is related to 'extended' tax liabilities or 'trailing' taxes. These are taxes that are levied on income that is not otherwise subject to the country of origin's source rules, accrued to an individual within a given period following his change of residence

⁵⁷ See: Hoge Raad, BNB 2009/263, 19 June 2009. However, in a more recent decision (see Hoge Raad, BNB 17/186, 14 July 2017), the Dutch Supreme Court held to be compliant with the country's treaty obligations the law enacted by the Dutch government in response to the 2009 Supreme Court decision, prescribing a 'conservatory assessment' limited to the tax-exempt pension contributions accrued to an individual until emigration. For a comment, see Pötgens, Kool (2018).

⁵⁸ See Helminen (2002, 234), submitting that 'a mere temporary emigration of a Finnish national should not trigger limited tax liability in Finland. Only emigration, which may be regarded as final, should trigger limited tax liability'.

(generally, five to 10 years).⁵⁹ Following the imposition of a trailing tax, based on an idiosyncratic definition of residence (see Oldman, Pomp 1979, 31), the emigrated individual remains liable for tax on their worldwide income in the country of origin, both on income derived from assets owned at the time of departure and on income accrued to them thereinafter. In contrast to an exit tax, a trailing tax is not assessed at the time of the transfer of residence, but only subsequently, i.e. when the individual actually receives the income thereof.

Indeed, the scholarly proposal to change the order of the tie-breaker rules for individual residence purposes currently used in the OECD Model, by primarily assigning residence to the country where the individual taxpayer has their 'centre of vital interests' rather than 'a permanent home available to him', as is presently the case, can be seen as a sort of extended tax liability or trailing tax also (see Brauner 2010, 250). Further, the use, by a country, of citizenship as the main personal connecting factor for income tax purposes, to the extent that by doing so such country succeeds in taxing its expatriated citizens, leads to the same effects. ⁶⁰ Ultimately, citizenship may also be used, even if not as the main personal connecting factor, in the context of extended liability provisions, by countries having a residence-based tax system (this is the case of Finland, Hungary and Sweden).⁶¹ While these kinds of constraints to taxdriven expatriation are usually unilateral, nothing prevents a specific provision allowing citizenship-based taxation to be inserted in a double tax treaty. France has followed this route in its double tax treaties with Andorra and Monaco.⁶²

Notably, a Dutch ten-years trailing tax, although in the field of inheritance tax, was at stake in *van Hilten* (CJEU, case C-513/03, Heirs of M.E.A. van Hilten-van der Heijden v. Inspecteur van de Belastingdienst/Particulieren/Ondernemingen buitenland te Heerlen, ECLI:EU:C:2006:131).

However, it should be noted that at the present, the United States is one of the few countries that still uses citizenship as the main personal connecting factor (for an overview, see Holm 2014). The only other country that uses citizenship as the main personal connection factor, Eritrea, was in fact condemned by both the UN and the EU for the practice of imposing a 2% levy, named 'Diaspora Tax' or 'Recovery and Rehabilitation Tax', on its citizens permanently living abroad. See: United Nations, Resolution 2023, UN Doc. S/RES/2023, 5 December 2011; European Parliament, Resolution on the Situation in Eritrea, 2016/2568(RSP), 10 March 2016. Past practices by other states (most notably, Mexico and Philippines) to levy income tax based on citizenship were, indeed, largely unsuccessful, mainly due to the difficulties encountered by those countries in enforcing tax obligations on their expatriated citizens (see Pomp 2015).

⁶¹ Tuloverolaki 1992 [Finnish Income Tax Act], Sec. 11; Inkomstskattelag 1999 [Swedish Income Tax Act], Sec. 7; Személyi jövedelemadóról szóló 1995. évi CXVII. törvén 1995 [Hungarian Law on Individual Income Tax], Sec. 3 (2) (a).

⁶² France-Monaco Income Tax Treaty (18 May 1963), Art. 7; Andorra-France Income Tax Treaty (2 April 2013), Art. 25 (1) (d). For a discussion of the provisions contained in these two treaties, see Kallergis (2015).

As a potential alternative or in addition to the aforementioned measures, the country of origin may decide to recapture or 'claw-back' benefits, deductions or deferrals previously granted to an individual upon emigration. In this way, the country of origin essentially aims to safeguard its latent taxing rights over an emigrant's income. ⁶³ However, claw-back provisions imposed on income such as pensions are highly problematic, in so far as those measures frequently generate a liquidity shortage for the emigrated individual, who might not have readily or entirely available cash needed to pay the tax assessment concerned. Arguably, such kinds of income recaptures should therefore at least contemplate payment in instalments. Notably, with respect to pension income, a proportionate method of tax remittance might take the form of a withholding on monthly pension payments. ⁶⁴

Whether any of the measures discussed above is included in a blueprint for a given tax reform, the establishment of some procedural rules would also be needed. In particular, it would be useful to provide for an effective exchange of information and adequate tax collection mechanism between countries. While imposing a tax on its emigrated individuals, a country is in fact confronted with two kinds of hurdles. First, it must obtain accurate information about the emigrated individual's income in order to assess their tax liabilities and, second, it must collect the amount of tax owed. Indeed, an exchange of tax information can also be useful for the country of arrival, as long as a specific obligation is imposed upon such country to take into account the tax charged by the other state while levying its own taxes on the individual taxpayer.

6.3. Subject-to-tax rule(s)

As it is known, under international tax law states are under no obligation to prevent either double taxation or non-taxation, unless specific provisions to that effect are inserted in a double tax treaty. Subject-to-tax rules fulfil precisely this function, by ensuring that income

⁶³ In a sense, previously enjoyed deductions represent a sort of 'tax loan', which must be recouped at a later date. See: Opinion of Advocate General Stix-Hackl, case C-150/04, Commission of the European Communities ν. Kingdom of Denmark, ECLI:EU:C:2006:357, para. 68.

⁶⁴ Interestingly, in their proposal for a 'brain drain tax', Bhagwati, Dellalfar (1973, 96) suggested the tax be collected for 10 years following migration or, preferably, through lifetime payments.

 $^{^{65}\,\,}$ Those kinds of procedural rules are set forth, respectively, in Articles 26 and 27 of the OECD and UN Models.

Noteworthy, such an obligation exists in the EU for exit taxes levied on emigrated corporate taxpayers after the first Anti-Tax Avoidance Directive (ATAD) entered into force in July 2016. See: Council Directive (EU) 2016/1164 of 12 July 2016 laying down Rules against Tax Avoidance Practices that Directly Affect the Functioning of the Internal Market, Art. 5 (5), OJ L 193/1 of 19/6/2016.

is taxed at least by one of the two Contracting States (see Rust 2015, 1624 paras. 34). Seen from this perspective, subject-to-tax rules provide a concrete example of how the single tax principle, i.e. the principle stipulating that the same income is to be taxed once and only once, can act as a coordination mechanism to turn the international tax regime into a more comprehensive one.⁶⁷

The idea underlying subject-to-tax rules is anything but new (for a discussion, see Burgstaller, Schilcher 2004; Lampe 1999). Although not generally recommending that states include subject-to-tax rules in their double tax treaties, ⁶⁸ the Commentaries to the OECD and UN Models in fact mention time and again the possibility for countries of bilaterally agreeing on a rule according to which the tax relief to be granted by one Contracting State is contingent upon the income being subject to tax in the other Contracting State. ⁶⁹ It is worth noting that a subject-to-tax rule is also included in the Global anti-Base Erosion (GloBE) proposal unveiled by the OECD in early 2019, which essentially aims to ensure that internationally operating businesses pay a minimum level or 'fair share' of taxes (see OECD/G20 2019b; OECD/G20 2019a).

⁶⁷ For a theoretical concept of the single tax principle as a cornerstone of the international tax regime, see Avi-Yonah (2007, 8–10). Gil García (2019) argues that 'single taxation is not pursued by tax treaties but is, rather, a consequence when specific provisions are implemented', such as subject-to-tax rules, whereas Shaviro (2015, 6) points out that the single tax principle can be seen as 'an often useful coordinating device'.

Until 2014, the Commentaries to Article 1 on the OECD Model in fact stipulated that '[g]eneral subject-to-tax provisions provide that treaty benefits in the State of source are granted only if the income in question is subject to tax in the State of residence. This corresponds basically to the aim of tax treaties, namely, to avoid double taxation. For a number of reasons, however, the Model Convention does not recommend such a general provision'. OECD Model Tax Convention Commentary on Article 1 (2014), para. 15. The quoted passages were deleted during the 2017 Update of the OECD Model (see OECD 2017b, 47). It is also worth recalling that the BEPS Action 6 Final Report proposed to add new provisions to Article 11 (Interest), Article 12 (Royalties) and Article 21 (Other Income) of the OECD Model, stipulating that interest, royalties or other income arising in a Contracting State and beneficially owned by a resident of the other Contracting State 'may be taxed in the first-mentioned Contracting State in accordance with domestic law if such resident is subject to a special tax regime' (see OECD/G20 2015, 98). The provisions in question would essentially allow taxation by the source country when there is a preferential tax regime in the residence state and this is defined in the relevant tax treaty. However, the proposed new provisions were not included in any of the aforementioned articles during the 2017 Update of the OECD Model.

⁶⁹ See e.g. OECD Model Tax Convention Commentary on Article 13 (2017), para. 21, stipulating that '[a]s capital gains are not taxed by all States, it may be considered reasonable to avoid only actual double taxation of capital gains. Therefore, Contracting States are free to supplement their bilateral convention in such a way that a State has to forego its right to tax conferred on it by the domestic laws only if the other State on which the right to tax is conferred by the Convention makes use thereof'. See also UN Model Tax Convention Commentary on Article 13 (2017), para. 4.

The forms and wordings of subject-to-tax rules contained in the various double tax treaties concluded by countries are indeed manifold. According to relevant literature, one criterion for categorizing such rules is whether the subject-to-tax rule only applies to a certain item of income – thus, resulting in a 'specific' subject-to-tax rule – or whether it applies to all categories of income covered by a double tax treaty – thus, resulting in a 'general' subject-to-tax rule (see Burgstaller, Schilcher 2004).

A specific subject-to-tax rule is envisaged in the Commentaries to Article 18 of the OECD Model (reproduced in the Commentaries to Article 18 of the UN Model), allowing source taxation of pension payments where the residence state does not subject to tax these payments 'under the ordinary rules of its tax law'. The adoption of a general subject-to-tax rule by EU Member States in their double tax treaties was instead proposed by the European Commission in its 2012 Recommendation on Aggressive Tax Planning. Moreover, a general subject-to-tax rule is laid down in Article 26 (2) of the multilateral Nordic Convention. A mechanism ultimately resulting in a similar effect to that of a general subject-to-tax rule – commonly called a 'switch-over clause' — is also envisaged in paragraph 4 of Article 23 (A) of the OECD Model and is related to the exemption method used by the residence state, which is prevented from exempting items of income from tax whether those incomes have not been taxed in the source state.

OECD Model Tax Convention Commentary on Article 18 (2017), para. 15. Notably, the subject-to-tax rule on pension income was added to the OECD Commentaries following the 2003 Discussion Draft on Tax Treaty Issues Arising from Cross-Border Pensions (see OECD 2003, 6).

The See European Commission 2012b. Dourado (2015, 50–51) submits that 'in the current EU context of tax competition and lack of will to harmonize, it is very unlikely that EU Member States would adopt such a subject-to-tax clause, especially regarding intended gaps, aimed at promoting investment abroad or investment in developing countries. Moreover, EU Member States may also be resistant to adopting a general subject-to-tax clause geographically limited to the EU territory. Taking into account free movement of capital, subject-to-tax clauses should ideally be adopted universally or at least in the OECD context, in order to avoid diversion of investment to those States that do not adopt those rules'. Remarkably, thus far, all these predictions have been fulfilled. For a critical analysis of the subject-to-tax rule recommended by the European Commission in 2012, see Marchgraber 2014.

⁷² Convention between the Nordic Countries for the Avoidance of Double Taxation with respect to Taxes on Income and on Capital (23 September 1996, as amended through 2018), Art. 26 (2).

 $^{^{73}}$ See e.g. van Horzen, De Groot (2018), discussing the switch-over clauses included in the EU anti-BEPS rules.

⁷⁴ See also OECD Model Tax Convention Commentary on Article 18 (2017), para.
35. By contrast, Rust (2015, 1655 para. 102) considers that the relevant provision 'does not constitute a subject-to-tax clause'.

In actual tax treaty practice, general subject-to-tax rules can be found in several bilateral treaties, such as those signed by Italy with France and Germany or by Austria with Malta and the United Kingdom. Specific subject-to-tax rules concerning pension income from past private employment can also be found in many double tax treaties, for instance those between Cyprus and Switzerland, Denmark and the United Kingdom, Estonia and Serbia, or France and Switzerland. ⁷⁶

Although, as stated, subject-to-tax rules are nothing new under the sun and, indeed, can be found in various double tax treaties, no internationally agreed standard has evolved yet. A blueprint for individual tax reform including such measures could thus offer a valuable framework for harmonizing their interpretation and application. It is worth noting that subject-to-tax rules might be particularly useful to address in situations where pension income from past private employment is not taxed in the resident state of the emigrated retiree due to the operation of a preferential tax regime.⁷⁷

However, it should be noted that a subject-to-tax rule, by itself, is not able to tackle situations in which pension income is actually taxed by the residence state, but a preferential tax rate applies. In fact, even the exact meaning of the term 'subject-to-tax' is far from clear and, thus, the answer to this question is very much open to different interpretations, especially in borderline situations. What if, for instance, no preferential regime exists for pension income in the residence state, but such country is a TEE state and therefore it simply does not levy any tax upon

⁷⁵ 1989 Protocol of the France – Italy Income and Capital Tax Treaty (5 October 1989), Point 15; Protocol of the Germany – Italy Income and Capital Tax Treaty (18 Oct. 1989), Point 18 (b); Austria – United Kingdom Income Tax Treaty (30 April 1969, as amended through 2009), Art. 2 (2); Austria – Malta Income and Capital Tax Treaty (29 May 1978), Art. 2 (5).

⁷⁶ 2014 Protocol Cyprus – Switzerland Income and Capital Tax Treaty (25 July 2014), Point 4 (b); Denmark – United Kingdom Income Tax Treaty (11 November 1990, as amended through 1996), Art. 18 (1); Estonia – Serbia Income Tax Treaty (24 September 2009), Art. 18 (2); France – Switzerland Income and Capital Tax Treaty (9 September 1966, as amended through 2014), Art. 20 (2).

 $^{^{77}}$ As recalled in section 5 above, this is the case of foreign-source pensions in Portugal.

⁷⁸ For instance, as from 2019, Italy has introduced a special tax regime for incoming pensioners to which a 7% substitute tax of the income tax apply (see Beretta 2019b).

⁷⁹ See Lang (2004, 111), also noting that 'in some languages, the term 'subject to tax' means the same as 'liable to tax', thus adding further confusion to the interpretation and application of the expression in question'. The general subject-to-tax rule included in the recommendation issued by the European Commission in 2012 was surprisingly short. It only stipulated that 'an item of income should be considered to be subject to tax where it is treated as taxable by the jurisdiction concerned and is not exempt from tax, nor benefits from a full tax credit or zero-rate taxation' (see European Commission 2012b).

disbursement of pension income? Or, even, what if the amount of pension income is below the minimum taxable amount in the residence state so that no actual tax liability arises? Or, further, what if a substitute tax rather than the statutory income tax applies to pension income, so that such levy might be excluded from the scope of a double tax treaty pursuant to Article 2 of the OECD Model? Shall the subject-to-tax rule operate also in those situations? Moreover, in addition to the specific case of an emigrated pensioner, a subject-to-tax rule fails to entirely address the brain drain issue, since no financial compensation is provided to the country of origin of the highly-skilled emigrant if the income that he receives once in the country of arrival is subject to tax therein.

6.4. Exclusive source-based taxation

One may imagine addressing the challenges arising in the field of individual taxation by means of changes to the relevant allocation rules currently provided under double tax treaties. Notably, with regard to pension income from past private employment, this would imply abandoning exclusive residence-based taxation in favour of exclusive source-based taxation.⁸¹

A proposal to that effect was recently advanced by Genser and Holzmann. Notably, the two authors implore for a coordinated shift from EET to TEE (or TTE) taxation of pension income, since they regard the latter taxation system of pension income better suited for a world of increasingly mobile individuals, than the EET, adopted by most countries (see Genser, Holzmann 2016, 16–23). In their opinion, universal or widespread adoption of TEE (or TTE) taxation of pension income – in their word 'front-load taxation' instead of 'back-loaded taxation' – would prevent revenue losses for the country of origin when the individual taxpayer emigrates, as the income will have already been taxed at the time it was earned, and would also avoid double taxation, as the residence state would be required to exempt the income in question. ⁸³ As an

⁸⁰ As it might occur in the case of the substitute tax that applies to incoming pensioners in Italy as of 1 January 2019.

A model provision to that effect is indeed included in the Commentaries to the OECD Model. See OECD Model Tax Convention Commentary on Article 18 (2017), para. 15. It is worth noting that exclusive source-based taxation is also provided under the multilateral Nordic Convention. See the Convention between the Nordic Countries for the Avoidance of Double Taxation with respect to Taxes on Income and on Capital (23 September 1996, as amended through 2018), Art. 18 (1).

⁸² See also Schindel, Atchabahian (2005, 40), noting that 'from the point of view of inter-nation equity and efficiency, exclusive or predominant taxation at source is shaping up as the most reasonable basis of taxation'.

⁸³ A tax levied by the country of origin over the income of the emigrated individual also emerged from international discussion as the most feasible version of the Bhagwati brain drain tax (see Oldman, Pomp 1979, 246–247).

alternative, they also propose that pension taxation by the source state be deferred until the relevant income is effectively disbursed, so that the tax becomes due only at time of disbursement of the monthly pension benefits (see Genser, Holzmann 2016, 20–21).

The most important advantage of applying the TEE (or TTE) rather than the EET system is that cross-border movements of pensioners from one country to another no longer distort inter-country equity. Pension income is, in fact, taxed already at time when contributions to pension systems are not deductible from employment income in the country of origin, so that no recouping of income tax relief is required to restore equity between different jurisdictions once the individual taxpayer leaves their country of origin. A second advantage is related to the administration of the TEE (or TTE) system in contrast to the EET one, in so far as the former method requires no control of correct deductions for pension savings and since, if the TTE system is applied, old-age pension contributions and pension savings do not reduce the income tax base in the country where the relevant income is built up. The third advantage is related to the fact that, since old-age pension benefits to pensioners are tax-free, for the emigrated taxpayers filing income tax returns in the country of origin is not a requirement, even if pension income is received from several sources, possibly located in different countries. Accordingly, there is no need to establish any source rule either.⁸⁴

Exclusive source-based taxation, however, will only work if countries universally adopt the TEE (or TTE) system. If this is not the case, bilateral tax treaty negotiations will be complicated furthermore by the fact that resident pensioners will receive pension benefits from different source countries, so situations may arise where a country that suffered a tax revenue loss from preferential treatment accorded during the contribution and accumulation phases of pension income, is not the source country paying out the pension income and, is therefore, not part of the negotiation process with the residence state. In fact, a consistent solution to this dilemma would require establishing some form of multilateral consent. In this regard, however, the claim made by Genser, Holzmann (2016, 24) that a pan-European decision to move from a EET to a TEE (or TTE) system of taxation of pension income would put pressure on non-European countries to replicate such an approach, so as to avoid revenue shortfalls and double taxation, does not seem sufficiently grounded.

There are indeed three different possible source rules for pensions, i.e. their source may be located: (1) where the fund paying pension income is established, (2) in the state in which employment services were rendered, or (3) in the state in which deductions in respect of the pension have been claimed. See OECD Model Tax Convention Commentary on Article 18 (2017), para. 19.1; UN Model Tax Convention Commentary on Article 18 (2017), para. 13.

It should also be taken into account that exclusive, rather than shared, taxing rights attributed to the source state are likely to bring additional pressure on national governments during tax treaty negotiations (see Starink 2016, 11). S A solution to this shortcoming might then be found in granting the source state shared – as opposed to exclusive – taxing rights with the residence state, regarding pension income. This practically implies limited source-based taxation, meaning that the source tax cannot exceed a specified rate, while the residence state is obliged to credit the tax levied by the source state, as similarly provided for dividends and interest, respectively, in Articles 10 and 11 of the OECD and UN Models. And yet, even this solution presents some hurdles, since limited source-based taxation might not actually be sufficient to fully compensate the country of origin for the fiscal revenues forgone as a consequence of the emigration of an individual taxpayer.

6.5. Compensation tax

An alternative to the aforementioned measures may be to leave the current allocation rule (taxing rights over pension income vested solely to the residence state) unchanged, but to provide at least some fiscal compensation to the source state, which should be identified as the country from which pension income payments are made. The ground idea is that the country of arrival is to levy a tax on pension income to fully or partially compensate the country of origin for the revenues forgone following the expatriation of the individual taxpayer, being either a highly-skilled or an elderly individual. Indeed, although abandoned in later versions of the proposal, as it was found difficult to actually enforce, the original Bhagwati tax proposal envisioned a surtax imposed by the country of arrival (see Bhagwati 1972, 44).

A proposal featuring a sort of compensation tax to address the brain drain phenomenon was also advanced more recently in Lister (2017). The key feature of the proposal contained therein is to resort to a tax credit – roughly akin to the foreign tax credit currently available to US citizens living and working abroad⁸⁷ – as a means to compensate the countries of origin experiencing a revenue loss following the departure of highly-skilled individuals from their own territory.⁸⁸ Specifically, it is proposed

⁸⁵ However, Brauner (2010, 163) contends that a 'brain drain tax' can also be implemented by countries bilaterally, through purposive changes to existing double tax treaties.

A model provision to that effect is included in the Commentaries to the OECD Model. See OECD Model Tax Convention Commentary on Article 18 (2017), para. 15.

⁸⁷ Internal Revenue Code (IRC), Title 26, Sub. A, Ch. 1, Subch N, Part III, Subpart A, Sections 901–909.

Lister (2017, 75) defines highly-skilled individuals as those who, cumulatively: (1) have received higher education or skills training, (2) whose training or education was

that the country of origin levy an income tax over employment income earned abroad by its emigrated highly-skilled individuals and that the resulting fiscal proceeds, collected by the country of arrival, are credited against employment taxes due in that country, whereas the remainder is returned to the countries of origin of the emigrated individuals, thereby compensating – at least to a degree – those latter countries for the sunk investment made in human capital that has left its territory. Lister (2017, 76) further stipulates that the levying of the compensation tax is to last long enough to fully compensate the country of origin for the lost investment in the highly-skilled individual.

Cases in point can be found in actual tax treaty practice by countries. It is worth noting that under Article 9 of the 2015 Protocol to the double tax treaty concluded between France and Germany, the resident state of the individual recipient of the pension paid out under the statutory social insurance schemes is entailed to tax the income in question, but it must pay back to the state in which the payments arise a 'compensation amount' corresponding to the tax which that state would have charged under its tax laws.⁸⁹

The main advantage of the aforementioned proposal is that it aligns the interests of both the countries concerned, since it provides for shared allocation of taxing rights between the residence and source states and, therefore it also allows shared allocation of tax revenues between the country of origin and the country of arrival. This is consistent with the fact that, arguably, both countries have a legitimate claim to tax the income of the emigrated individual. Another, related advantage is that countries no longer have to strive for exclusive source-based taxation as a means to tackle tax-induced emigration of individual taxpayers from their own territory. Indeed, this very circumstance is likely to significantly smooth tax treaty negotiations between countries. In the context of the brain drain from a developing to a developed country, the further advantage of this proposal is related to the fact that the compensation tax builds upon the administrative capabilities of the developed country (see Lister 2017, 82).

The major concern related to the proposal under discussion are the nature and characteristics of such a 'compensation tax'. If, in fact, the proposed compensation tax is designed to apply separately and on top of income taxes levied by the residence state (i.e. the country of arrival), it might not actually be considered as an income tax covered by a double tax treaty, pursuant to Article 2 of the OECD and UN Models, with the

largely or completely publicly funded, and (3) have left their own country to work in another within a defined number of years after completing their education or training.

^{89 2015} Protocol of the France – Germany Income and Capital Tax Treaty (21 July 1959, as amended through 2015), Art. 9, introducing a new Art. 13c in the text of the Convention.

consequence that the double taxation relief mechanisms provided in the relevant double tax treaty would not apply. Such a compensation tax might be considered as a sort of 'extraordinary tax', i.e. a levy imposed for a limited period – particularly, until the country of origin is fully compensated for the lost investment in the highly-skilled individual – and for certain reasons, provided various circumstances are also met. 90 Another important disadvantage is related to the fact that the implementation of the proposal requires quite a smooth system through which the collected tax is passed on by the emigrated individual's country of arrival to their country of origin. A further drawback is related to the circumstance that an emigrated individual will be at a disadvantage vis-àvis an individual resident in the country in the same personal and economic circumstances. As such, the compensation tax seems to run contrary to the general obligation of non-discrimination, which is enshrined both in tax treaties pursuant to Article 24 (1) of the OECD Model and at the EU level in Article 18 TFEU, to the extent that taxation of incoming individuals equates to taxation of foreigners by the country of arrival. 91 However, probably the major source of concern is that the actual implementation of the proposal seems utopian at best, since the country of arrival would not only miss out on fiscal revenues, but it would also have to help collect those proceeds, all for the sole benefit of the emigrated individual's country of origin. 92

6.6. Global minimum tax

The GloBE proposal unveiled by the OECD in early 2019 envisages an international tax regime where MNEs are required to pay, at least, a minimum level of taxes. This practically ensures that a 'global minimum

⁹⁰ 'Extraordinary taxes' are also considered in the Commentaries to the OECD and UN Models. Notably, it is stipulated therein that Article 2 'does not mention 'ordinary taxes' or 'extraordinary taxes'. Normally, it might be considered justifiable to include extraordinary taxes in a Model Convention, but experience has shown that such taxes are generally imposed in very special circumstances. In addition, it would be difficult to define them. They may be extraordinary for various reasons; their imposition, the manner in which they are levied, their rates, their objects, etc. This being so, it seems preferable not to include extraordinary taxes in the Article. But, as it is not intended to exclude extraordinary taxes from all conventions, ordinary taxes have not been mentioned either. The Contracting States are thus free to restrict the convention's field of application to ordinary taxes, to extend it to extraordinary taxes, or even to establish special provisions'. See OECD Model Tax Convention Commentary on Article 2 (2017), para. 4. For a discussion, see Ismer, Blank (2015, 15 para. 28).

⁹¹ Based on paragraph 6 of Article 24 of the OECD Model, the prohibition of discrimination, 'notwithstanding the provisions of Article 2', applies to 'taxes of every kind and description', consequently, in principle, also to a 'compensation levy' that is not covered by a double tax treaty.

 $^{^{92}}$ As admitted by the same proponent of the 'compensation tax' against brain drain illustrated in this section (see Lister 2017, 83).

tax' is ultimately paid by MNEs. A 'global minimum tax' in the field individual taxation may eventually be introduced, mimicking in a way developments occurring in the corporate sector at the international tax level.

A 'global minimum tax' in the individual sector would in fact display a number of strengths. Probably the most important advantage is related to the fact that its worldwide implementation by countries would provide a unique opportunity for meaningful multilateralism, although it does not seem equally feasible to entrust the administration and collection of such global minimum tax to a 'World Tax Organization'⁹³, as it was notably stipulated under a version of the Bhagwati tax proposal, which assigned such a task to the UN.⁹⁴

A first concern with regard to a global minimum tax is related to its nature and characteristics. In this sense, similar considerations to those presented above with regard to a compensation tax apply. In addition to the issue of devising a robust enough effective tax rate, there is also the issue of establishing the category of persons to be subject to the tax as well as the rules and principles governing the calculation of the tax base. To put it into a perspective: should a global minimum tax be imposed only on highly-skilled and/or elderly individuals or, also on all/other categories of emigrated taxpayers? Notably, what about emigration of individuals from one country to another for a short period – say, two or three years – such as it may occur in the case of academics and students? In fact, the individual motives for a person to reside abroad could also change over time. Furthermore, as regards the calculation of the tax base, should the income tax rules of the country of arrival or those of the country of origin apply? An autonomous set of rules for calculating the tax base could also ultimately be laid down. Another set of concerns is related to the actual implementation of such a global minimum tax. Even if an adequate consensus is built around it and a multilateral framework is then established, implementing such a tax is by no means straightforward, to the extent that this would require a change of bilateral tax treaties. In this sense, it seems far more practical to amend current tax treaties through a multilateral convention. The experience of the BEPS Multilateral Convention (MLI) can provide useful insights in this regard (see OECD

⁹³ A first plea for a supervising 'World Tax Organization' was famously made by Tanzi (1999). Questioning the actual feasibility of a 'World Tax Organization' as a means to achieve international tax coordination, see Schön (2009), who considers that 'a realistic outlook will be the ongoing use of bilateral treaties, including some regional multilateral conventions which would extend the number of participants but would not change the traditional character of this instrument as such'.

⁹⁴ Oldman, Pomp (1979, 44–58), however, also suggest that the United Nations might only promulgate a set of guidelines for the imposition of an 'international brain drain tax', or 'IBDT', by individual countries.

2017c). However, this would again involve a demanding process, involving modification thousands of existing bilateral treaties on the basis of a complex set of options to accommodate many different possible combinations of treaty partner preferences, as a quick glance at the OECD MLI Matching Database clearly shows (see OECD 2019d).

7. CONCLUSIONS

After BEPS, the idea of tax sovereignty, i.e. that national governments have a non-exclusive right to shape their own tax policies completely independently of one another, seems a distant memory at best. In the post-BEPS world, the unconditional sovereign autonomy of countries over tax matters is, in fact, no longer conceivable.

It is unclear, however, whether the new international tax order that the OECD has long envisaged will ultimately lead to more cooperation or, rather, it will bring more competition among countries. A meaningful cooperation would indeed require building-up a global consensus, based on which a tax level playing field would be established among countries. Without such a global consensus, an international tax order would be difficult to shape, since countries would compete against each other in a global strategic game, based on volatile preferences reflecting their political and economic bargaining power rather than on a sound framework of jointly established principles and rules.

If the ultimate outcome of the action undertaken by the OECD is hard to predict, it is clear that the consequences of non-action at the international level are quite dire, in so far as an increasing number of countries would likely introduce unilateral measures to preserve their own tax base. Indeed, several decentralized actions by countries might ultimately produce the dissolution of any sort of international tax regime. Specifically, with regard to migration of individual taxpayers from one country to another, in the absence of any form of cooperation at the international level, bilateral negotiations would likely be stalled and, perhaps, even rolled back by the intrinsic antagonism of the countries concerned, as a result of their opposing budgetary interests.

This would certainly be detrimental not only from an inter-country perspective, but also from an intra-country point of view. As a matter of fact, in the current political, social and economic landscape, welfare-enhancing objectives can only be achieved if the international and national level are considered simultaneously and, possibly, aligned. The author therefore posits that if a new and fairer social contract is to be established at the national level, the terms and the course of the international tax

order should also be more clearly articulated among countries. In this sense, the various policies and measures explored in this article might eventually kick-off discussions on establishing such an international tax order.

REFERENCES

- Alcini, Cinzia, Daniel Gros. 2019. EU Mobile Workers: A Challenge to Public Finances. Contribution for Informal ECOFIN, Bucharest, 5–6 April 2019, CEPS Special Report.
- Ambagtsheer-Pakarinen, Laura. 2018. Finland; Portugal Treaty between Finland and Portugal Terminated by Finland. *News IBFD*, 13 June.
- Arginelli, Paolo, Francesco Avella. 2–3/2017. New Italian Tax Regime to Attract High Net Worth Individuals. *European Taxation* 57.
- Atoyan, Ruben, Lone Engbo Christiansen, Allan Dizioli, Christian Ebeke, Nadeem Ilahi, Anna Ilyina, Gil Mehrez, Haonan Qu, Faezeh Raei. 2016. *Emigration and Its Economic Impact on Eastern Europe*. Washington D.C.: International Monetary Fund.
- Avi-Yonah, Reuven S. 7/2000. Globalization, Tax Competition, and the Fiscal Crisis of the Welfare State. *Harvard Law Review* 113: 1573–1676.
- Avi-Yonah, Reuven S. 2007. *International Tax as International Law*. Cambridge: Cambridge University Press.
- Bader Daniel, Corinna Seiler. 4–5/2015. The Taxation of High-Net-Worth Individuals in Switzerland. *Bulletin for International Taxation* 69.
- Baldwin, Richard. 2019. *The Globotics Upheaval: Globalization, Robotics, and the Future of Work*. Oxford: Oxford University Press.
- Bangham, George (Resolution Foundation). 2018. The New Wealth of Our Nation: The Case for a Citizen's Inheritance. https://www.resolutionfoundation.org/app/uploads/2018/05/The-new-wealth-of-our-nation.pdf (last visited 31 October 2019).
- Beretta, Giorgio. 8/2017. From Worldwide to Territorial Taxation: Is Italy Now an Attractive Destination for Migrating Individuals?. *Bulletin for International Taxation* 71.
- Beretta, Giorgio. 1/2019a. Cross-Border Mobility of Individuals and the Lack of Fiscal Policy Coordination Among Jurisdictions (Even) After the BEPS Project. *Intertax* 47: 91–112.
- Beretta, Giorgio. 4/2019b. The New Special Tax Regime for Inbound Pensioners. *European Taxation* 59.

- Betten, Rijkele. 1998. *Income Tax Aspects of Emigration and Immigration of Individuals*. Amsterdam: International Bureau of Fiscal Documentation (IBFD).
- Bhagwati, Jagdish N. 1972. The United States in the Nixon Era: The End of Innocence. *Daedalus*: 25–47.
- Bhagwati, Jagdish N., William Dellalfar. 1–2/1973. The Brain Drain and Income Taxation. *World Development* 1: 94–101
- Bhagwati, Jagdish N. 3/1976. Taxing the Brain Drain. *Challenge* 19: 34–38.
- Blank, Michael, Roland Ismer. 3/2015. Rethinking Article 19 OECD MC (Government Service): A Provision in Search of a Rationale?. *Intertax* 43: 252–253.
- Blum, Cynthia. 3/1999. Should the Government That Pays Social Security Benefits across Borders also Tax Such Benefits?. *Virginia Tax Review* 18: 621–664.
- Boeri, Tito. 2012. Introduction. 1–14 in *Brain Drain and Brain Gain. The Global Competition to Attract High-Skilled Migrants*, edited by Tito Boeri *et al.*. Oxford: Oxford University Press.
- Borjas, George J. 2001. *Heaven's Door: Immigration Policy and the American Economy*. Princeton: Princeton University Press.
- Brauner, Yariv. 1/2010. Brain Drain Taxation as Development Policy. *Saint Louis University Law Journal* 55: 221–268.
- Brauner, Yariv. 2/2014. What the BEPS. Florida Tax Review 16: 55–116.
- Bristow, Jennie. 2019. Stop Mugging Grandma: The 'Generation Wars' and Why Boomer Blaming Won't Solve Anything. New Haven: Yale University Press.
- Brokelind, Cécile. 2014. Workers Mobility, Social Security Issues and Taxation of Frontier Workers: The Case of Pensions. 329–352 in *Movement of Persons and Tax Mobility in the EU: Changing Winds*, edited by Ana P. Dourado. Amsterdam: International Bureau of Fiscal Documentation (IBFD).
- Brokelind, Cécile, Martina Axmin. 2017. Migrant Pensioners Taxation and Healthcare Issues in the EU. 257–286 in *Elder Law*, edited by Ann Numhauser-Henning. Cheltenham: Edward Elgar Publishing.
- Brown, Patricia A. 2019. Articles 18 and 19 (2): Pensions Global Tax Treaty Commentaries. in *Global Tax Treaty Commentaries IBFD*. Amsterdam: International Bureau of Fiscal Documentation (IBFD).
- Brücker, Herbert, Giovanni Facchini, Anna Maria Mayda, Giovanni Peri. 2012. Understanding Highly Skilled Migration in Developed Countries: The Upcoming Battle for Brains. 15–208 in *Brain Drain and Brain Gain. The Global Competition to Attract High-Skilled Migrants*, edited by Tito Boeri *et al.* Oxford: Oxford University Press.

- Burgstaller, Eva, Michael Schilcher. 6/2004. Subject-To-Tax Clauses in Tax Treaties, *European Taxation* 44.
- Burke, Edmund. 1790. Reflections on the Revolution in France. London
- Carens, Joseph H. 2/1987. Aliens and Citizens: The Case for Open Borders. *The Review of Politics* 49: 251–273.
- Cassiano Neves, Thiago. 5/2010. Portuguese Taxation of Inward Expatriates and Pensioners: A Sunny Welcome. *European Taxation* 59.
- Cavendish, Camilla. 2019. *Extra Time: Ten Lessons for an Ageing World.* London: HarperCollins Publishers.
- Christians, Allison, Stephan Shay. 2017. General Report. 17–63 in *Assessing BEPS: Origins, Standards, and Responses*. vol. 102A. IFA Cahiers.
- Christians, Allison. 4/2016. BEPS and the New International Tax Order. Brigham Young University Law Review 6: 1603–1648.
- Cockfield, Arthur J. 1/2005. The Rise of the OECD as Informal "World Tax Organization' through National Responses to e-Commerce Tax Challenges. *Yale Journal of Law & Technology* 8: 136–187.
- Commander, Simon, Mari Kangasniemi, L. Alan Winters. 2004. The Brain Drain: Curse or Boon? A Survey of the Literature. 235–278 in *Challenges to Globalization: Analysing the Economics*, edited by Robert E. Baldwin, L. Alan Winters. Chicago: University of Chicago Press.
- Cooray, Arusha, Friedrich Schneider. 1/2016. Does Corruption Promote Emigration? An Empirical Examination. *Journal of Population Economics* 29: 293–310.
- Cruccu, Matteo. 2018. Italian Pensioners Flee to Portugal to Avoid Tax. *Corriere della Sera*, September 13.
- Dagan, Tsilly. 2018. *International Tax Policy. Between Competition and Cooperation*. Cambridge: Cambridge Tax Law Series.
- De Broe, Luc. 2002. General Report. 19–78 in The Tax Treatment of Transfer of Residence by Individuals. vol. 87B. IFA Cahiers.
- De Broe, Luc, Robert Neyt. 3/2009. Tax Treatment of Cross-Border Pensions under the OECD Model and EU Law. *Bulletin for International Taxation* 69.
- Del Sol, Marion, Marco Rocca. 2/2017. Free Movement of Workers in the EU and Occupational Pensions: Conflicting Priorities? Between Case Law and Legislative Interventions. *European Journal Social Security* 19: 141–157.
- Desai, Mihir A., Devesh Kapur, John McHale, Keith Rogers. 1/2009. The Fiscal Impact of High-Skilled Emigration: Flows of Indians to the US. *Journal of Development Economics* 88: 32–44.

- Dilnot, Andrew, Paul Johnson. 1993. *The Taxation of Private Pensions*. London: Institute for Fiscal Studies.
- Docquier, Frédéric, Olivier Lohest, Abdeslam Marfouk. 2/2007. Brain Drain in Developing Countries. *The World Bank Economic Review* 21: 193–218.
- Dourado, Ana P. 1/2015. Aggressive Tax Planning in EU Law and in the Light of BEPS: The EC Recommendation on Aggressive Tax Planning and BEPS Action 2 and 6. *Intertax* 43: 42–57.
- Drucker, Peter F. 2/1993. The Rise of the Knowledge Society. *The Wilson Quarterly* 17: 52–72.
- Dustmann, Christian, Itzhak Fadlon, Yoram Weiss. 1/2011. Return Migration, Human Capital Accumulation and the Brain Drain. *Journal of Development Economics* 95: 58–66.
- Dyppel, Katjia J. 2001. Danish Tax Minister: Following Termination of Tax Treaties with France and Spain, No Plans to Conclude New Treaties. *News IBFD*, 19 October.
- Entwistle, Harold. 1970. Education, Work and Leisure. Abingdon: Routledge.
- European Commission. 1998. Conclusions of the ECOFIN Council Meeting on 1 December 1997 concerning Taxation Policy Resolution of the Council and the Representatives of the Governments of the Member States, meeting within the Council of 1 December 1997 on a Code of Conduct for Business Taxation Taxation of Saving, OJ C 2/1 of 6 January 1998.
- European Commission. 2001a. Communication from the Commission to the Council, the European Parliament and the Economic and Social Committee The Elimination of Tax Obstacles to the Cross-Border Provision of Occupational Pensions, COM(2001)214 final.
- European Commission. 2001b. Commission Strategy to Remove Tax Obstacles to Cross-Border Pension Provision. MEMO/01/142, April 19.
- European Commission. 2003. Pension Taxation: Commission Tackles Discrimination against Foreign Pension Funds in Six Member States. IP/03/179, February 5.
- European Commission. 2010. Communication from the Commission to the Council, the European Parliament and the European Economic and Social Committee. Removing Cross-Border Tax Obstacles for EU Citizens, COM(2010) 769 final.
- European Commission. 2012a. Communication from the Commission to the Council, the European Parliament and the Economic and Social Committee. An Action Plan to Strengthen the Fight against Tax Fraud and Tax Evasion, COM(2012) 722 final.

- European Commission. 2012b. Commission Recommendation of 6 December 2012 on Aggressive Tax Planning, OJ L 338/41 of 12 December 2012.
- European Commission. Report of Expert Group. 2016. Ways to Tackle Cross-Border Tax Obstacles Facing Individuals within the EU. https://publications.europa.eu/en/publication-detail/-/publication/4bfee942-ca41-11e5-a4b5-01aa75ed71a1 (last visited 31 October 2019).
- European Commission. 2017. Employment and Social Developments in Europe 2017. https://ec.europa.eu/social/main.jsp?catId=738&lan gId=en&pubId=8030&furtherPubs=yes (last visited 31 October 2019).
- European Commission. Directorate-General of Communications Networks, Content & Technology. 2018a. *The Silver Economy*. Brussels: European Union Official Publications.
- European Commission. Directorate-General for Employment Social Affairs and Inclusion. 2018b. Study on the Movement of Skilled Labour. Final Report. https://op.europa.eu/en/publication-detail/-/publication/05079c0e-fc2e-11e8-a96d-01aa75ed71a1/language-en (last visited 31 October 2019).
- European Commission. Migration and Home Affairs. 2019. Brain Drain. https://ec.europa.eu/home-affairs/what-we-do/networks/european_migration_network/glossary_search/brain-drain_en (last visited 31 October 2019).
- European Parliament. 2019. European Parliament Legislative Resolution of 4 April 2019 on the Proposal for a Regulation of the European Parliament and of the Council on a Pan-European Personal Pension Product (PEPP). COM(2017)0343—C8—0219/2017—2017/0143(COD).
- Ezra, Don. 2019. Life Two: What We Used to Call 'Retirement'. *Financial Times*, 19 July.
- Faini, Riccardo. 2/2007. Remittances and the Brain Drain: Do More Skilled Migrants Remit More?. *The World Bank Economic Review* 21: 177–191.
- Fazackerley, Anna. 2018. Brexit Brain Drain: Elite Universities Say They Are Losing Future Research Stars. *The Guardian*, 6 March.
- Filipovic, Gordana. 2019. Serbia Tackles the Plunging Population Plaguing the Balkans. *Bloomberg*, 26 February.
- Ford, Martin. 2015. Rise of the Robots: Technology and the Threat of a Jobless Future. New York: Basic Books.
- Freeman, Richard B. 2/2006. People Flows in Globalization. *Journal of Economic Perspectives* 20: 145–170.

- Friedman, Thomas L. 2010. Root Canal Politics. *The New York Times*, 13 May.
- Ganesch, Janan. 2016. Millennials Do Not Know How Lucky They Are. *Financial Times*, 18 March.
- Gardiner, Laura (Resolution Foundation). 2016. Stagnation Generation. https://www.resolutionfoundation.org/app/uploads/2016/06/ Intergenerational-commission-launch-report.pdf (last visited 31 October 2019).
- Gehring, Anoeshka. 5/2017. Pensioners on the Move: A 'Legal Gate' Perspective on Retirement Migration to Spain. *Population, Space and Place* 23: e2007.
- Gehring, Anoeshka. 2/2019. Mobile Pensioners: Retirement Migrants' Perspective of EU Citizenship and Free Movement. *European Journal of Social Science Research* 32: 254–269.
- Genser Bernd, Robert Holzmann. 2016. The Taxation of Internationally Portable Pensions: Fiscal Issues and Policy Options. Working Paper No. 5702. Munich: Center for Economic Studies and ifo Institute (CESifo).
- Giannoccolo, Pierpaolo. 2009. The Brain Drain: A Survey of the Literature. Working Paper. University of Milano-Bicocca, Department of Statistics, Milan, Italy.
- Gil García, Elizabeth. 3/2019. The Single Tax Principle: Fiction or Reality in a Non-Comprehensive International Tax Regime?. *World Tax Journal* 11.
- Gratton, Lynda, Andrew Scott. 2016. *The 100-Year Life: Living and Working in an Age of Longevity*. London: Bloomsbury Publishing.
- Gutmann, Daniel. 13/2001. Tax Treatment of Pensions: A Comparative Perspective. *European Taxation* 41 (special issue).
- Hammer, Bernhard, Tanja Istenič, Lili Vargha. 1/2018. The Broken Generational Contract in Europe: Generous Transfers to the Elderly Population, Low Investments in Children. *Intergenerational Justice Review* 4: 21–31.
- Harding, Jillian. 2018. More Older Americans Are "Unretiring". *CBS News*, 19 September.
- Hatton, Timothy J., Jeffrey G. Williamson. 3/1994. What Drove the Mass Migrations from Europe in the Late Nineteenth Century?. *Population and Development Review* 20: 533–59.
- He, Wan, Daniel Goodkind, Paul R. Kowal (US Census Bureau). 2016. An Aging World: 2015. Washington D.C.: US Government Publishing Office.
- Helminen, Marjaana. 2002. Finland. 231–247 in The Tax Treatment of Transfer of Residence by Individuals. vol. 87B. IFA Cahiers.

- Helminen, Marjaana. 2019. EU Tax Law. Direct Taxation 2019. Amsterdam: International Bureau of Fiscal Documentation (IBFD).
- Holm, Ashley R. 2014. Extraterritorial Taxation in the Design of United States Tax Policy: Citizenship-Based Taxation and the Foreign Account Tax Compliance Act. 447–470 in *Tax Policy Challenges in the 21st Century*, edited by Raffaele Petruzzi, Karoline Spies. Vienna: Linde.
- House of Commons. Work and Pensions Committee. 2016. Intergenerational Fairness. Third Report of Session 2016–17. https://publications.parliament.uk/pa/cm201617/cmselect/cmworpen/59/59.pdf (last visited 31 October 2019).
- Ismer, Roland, Alexander Blank. 2015. Art. 2. Taxes Covered. in *Klaus Vogel on Double Tax Conventions 4th Ed.*, edited by Johannes Becker, Ekkehart Reimer and Alexander Rust. Alphen and den Rijn: Kluwer Law International.
- James, Harold. 2017. *Deglobalization as a Global Challenge*. Waterloo: Centre for International Governance Innovation.
- Kallergis, Andreas. 5/2015. Citizenship and Taxation in France. *European Taxation* 55.
- Kapur, Devesh, John McHale. 2005. Give Us Your Best and Brightest: The Global Hunt for Talent and Its Impact on the Developing World. Washington D.C.: Center for Global Development.
- Kavelaars, Peter. 2007/1. Cross-Border Aspects of Pensions in the Netherlands, including Tax Treaty and EU Law. *Bulletin for International Taxation* 61.
- *KeepTalkingGreece*. 2018. "Elders Drain": Greece's Pensioners Migrate Seeking Affordable Living. 1 October.
- Kemmeren, Eric C.C.M. 2001. *Principle of Origin in Tax Conventions, a Rethinking of Models*. Tilburg: Katholieke Universiteit Brabant.
- Kostíć, Svetislav V. 2019a. Tax Aspects of Migrations. 29 June. http://kluwertaxblog.com/2019/10/29/tax-aspects-of-migrations/ (last visited 31 October 2019).
- Kostíć, Svetislav V. 2/2019b. In Search of the Digital Nomad Rethinking the Taxation of Employment Income under Tax Treaties. *World Tax Journal* 11.
- Lagarde, Christine. 2019. Forging a Stronger Social Contract The IMF's Approach to Social Spending. 14 June. https://www.imf.org/en/News/Articles/2019/06/14/sp061419-md-social-spending (last visited 31 October 2019).
- Lampe, Marc. 1999. General Subject-To-Tax Clauses in Recent Tax Treaties. *European Taxation* 39.
- Lang, Michael. 2004. General Report. 73–119 in *Double Non-Taxation*. vol. 89A. IFA Cahiers.

- Lang, Michael. 1/2007. Article 19 (2): The Complexity of the OECD Model Can Be Reduced. *Bulletin for International Taxation* 61.
- League of Nations. 1927. Double Taxation and Tax Evasion, Report presented by the Committee of Technical Experts on Double Taxation and Tax Evasion. http://taxtreatieshistory.org/data/html/league of nations 1926–04.html (last visited 31 October 2019).
- League of Nations. 1946. London and Mexico Model Tax Conventions Commentary and Text. http://adc.library.usyd.edu.au/view?docId=split/law/xml-main-texts/brulegi-source-bibl-15.xml;chunk.id=item-15;toc.depth=1;toc.id=item-15;database=;collection=;brand=default (last visited 31 October 2019).
- Levin, Sam. 2017. Millionaire Tells Millennials: If You Want a House, Stop Buying Avocado Toast. *The Guardian*, 15 May.
- Lister, Matthew. 1/2017. A Tax-Credit Approach to Addressing Brain Drain. *Louis University Law Journal* 62: 73–84.
- Marchgraber, Christoph. 5/2014. The Avoidance of Double Non-Taxation in Double Tax Treaty Law: A Critical Analysis of the Subject-To-Tax Clause Recommended by the European Commission. *EC Tax Review* 23: 293–302.
- Moretti, Enrico. 2012. *The New Geography of Jobs*. Boston: Houghton Mifflin Harcourt.
- Norr, Martin. 3/1961. Jurisdiction to Tax and International Income. *Tax Law Review* 17: 431–464.
- Norris, Pippa. 2018. Generation Wars over Brexit and Beyond: How Young and Old Are Divided over Social Values. 15 August. https://blogs.lse.ac.uk/politicsandpolicy/generation-wars-over-brexit/ (last visited 31 October 2019).
- OECD. Working Party No. 1 on Double Taxation of the Committee on Fiscal Affairs. Working Group No. 28 (Denmark-Ireland-Sweden). 1973. First Report on Articles 14, 15, 18 and 20 of the 1963 OECD Draft Convention (Note by the Secretariat) (14 November 1973), CFA/WP1(73)13.
- OECD. 2003. Tax Treaty Issues Arising from Cross-Border Pensions. Public Discussion Draft. Paris: OECD Publishing.
- OECD. 2017a. *Pensions at a Glance 2017: OECD and G20 Indicators*. Paris: OECD Publishing.
- OECD. 2017b. 2017 Update to the OECD Model Tax Convention. http://www.oecd.org/ctp/treaties/2017-update-model-tax-convention.pdf (last visited 31 October 2019).
- OECD. 2017c. Multilateral Convention to Implement Tax Treaty Related Measures to Prevent Base Erosion and Profit Shifting. Paris: OECD Publishing.

- OECD. 2018. The Tax Treatment of Retirement Savings in Private Pension Plans. https://www.oecd.org/daf/fin/private-pensions/Tax-treatment-of-retirement-savings-Policy-Brief-1.pdf (last visited 31 October 2019).
- OECD. 2019a. What Is BEPS?. https://www.oecd.org/tax/beps/about/ (last visited 31 October 2019).
- OECD. 2019b. *Under Pressure: The Squeezed Middle Class*. Paris: OECD Publishing.
- OECD. 2019c. *OECD Employment Outlook 2019*. Paris: OECD Publishing.
- OECD. 2019d. MLI Matching Database. https://www.oecd.org/ctp/treaties/mli-matching-database.htm (last visited 31 October 2019).
- OECD/G20. 2015. Action 6 2015 Final Report. Preventing the Granting of Treaty Benefits in Inappropriate Circumstances. Paris: OECD Publishing.
- OECD/G20. 2019a. OECD/G20 Inclusive Framework on BEPS Progress Report July 2018 May 2019. http://www.oecd.org/tax/beps/inclusive-framework-on-beps-progress-report-july-2018-may-2019. pdf (last visited 31 October 2019).
- OECD/G20. 2019b. Base Erosion and Profit Shifting. Addressing the Tax Challenges of the Digitalisation of the Economy. Public Consultation Document 13 February 6 March 2019. https://www.oecd.org/tax/beps/public-consultation-document-addressing-the-tax-challenges-of-the-digitalisation-of-the-economy.pdf (last visited 31 October 2019).
- Oldman, Oliver, Richard D. Pomp. 1/1979. *Tax Measures in Response to the Brain Drain*. Harvard International Law Journal 20: 1–60.
- Perkins, James A. 4/1966. Foreign Aid and the Brain Drain. Foreign Affairs 44: 608–619.
- Pickard, Susan. 2/2019. Age War as the New Class War? Contemporary Representations of Intergenerational Inequity. *Journal of Social Policy* 48: 369–386.
- Piketty, Thomas. 2013. *Capital in the Twenty-First Century*. Cambridge: Harvard University Press.
- Piracha, Matloob, Florin Vadean. 2009. Return Migration and Occupational Choice. IZA Discussion Paper No. 3922. Institute for the Study of Labor (IZA), Bonn.
- Pomp, Richard D. 2/1985. Experience of the Philippines in Taxing Its Non-Resident Citizens. New York University Journal of International Law and Politics 17: 245–286.
- Pötgens, Frank P.G., Elena M.L. Kool. 9/2018. The Relationship Between Tax Treaties and Protective Tax Assessments Issued in Respect of

- Pensions and Annuities upon Emigration: Pacta Sunt Servanda and Compartmentalization. *European Taxation* 58.
- Pritchett, Lant. 2006. Let Their People Come: Breaking the Gridlock on Global Labor Mobility. Washington, D.C.: Brookings Institution Press.
- Pytel, Sławomir, Oimahmad Rahmonov. 3/2019. Migration Processes and the Underlying Reasons: A Study on Pensioner Migrants in Poland. *Population, Space and Place* 25: e2197.
- Redonda, Agustin, Vincenzo Galasso, Mark Mazur, Miranda Stewart, Matthew Whittaker. 2019. G20 Japan 2019. Taxation in Aging Societies: Increasing the Effectiveness and Fairness of Pension Systems. https://t20japan.org/wp-content/uploads/2019/03/t20-japan-tf10-3-taxation-in-aging-societies.pdf (last visited 31 October 2019).
- República Portuguesa. 2019. Programa Regressar. https://www.programaregressar.gov.pt (last visited 31 October 2019).
- Resolution Foundation. 2018. A New Generational Contract: The Final Report of the Intergenerational Commission. https://www.resolutionfoundation.org/app/uploads/2018/05/A-New-Gene rational-Contract-Full-PDF.pdf (last visited 31 October 2019).
- Ribes Ribes, Aurora. 6–7/2017. Expatriate Taxation in Spain: Some Reflection for Debate. *Intertax* 43: 460–473.
- Risse, Mathias. 2012. On Global Justice. Princeton: Princeton University Press.
- Rixen, Thomas. 4/2011. Tax Competition and Inequality: The Case for Global Tax Governance. Global Governance: A Review of Multilateralism and International Organizations 17: 447–467.
- Rocha, Sergio André, Allison Christians (eds.). 2017. *Tax Sovereignty in the BEPS Era*. Alphen aan den Rijn: Kluwer Law International.
- Roxburgh, Douglas. 10/2006. Domicile and the Remittance Basis in UK Taxation. *European Taxation* 46.
- Royal Society. Committee on the Emigration of Scientists from the United Kingdom. 1963. *Emigration of Scientists from the United Kingdom:* Report of a Committee Appointed by the Council of the Royal Society. London: Royal Society.
- Rust, Alexander. 2015. Article 23A. Exemption Method. in *Klaus Vogel on Double Tax Conventions* 4th ed., edited by Johannes Becker, Ekkehart Reimer and Alexander Rust. Alphen and den Rijn: Kluwer Law International.
- Scarpetta, Stefano, Adrian Blundell-Wignall (OECD). 2015. Editorial The Next Frontier for Pension Policy: Focusing More on Social Sustainability. in *Pensions at a Glance 2015: OECD and G20 Indicators*. Paris: OECD Publishing.

- Schindel, Angel, Adolfo Atchabahian. 2005. General Report. 21–99 in *Source and Residence*. vol. 90A. IFA Cahiers.
- Schön, Wolfgang. 1/2009. International Tax Coordination for a Second-Best World (Part I). *World Tax Journal* 1.
- Shachar, Ayelet, Ran Hirschl. 1/2013. Recruiting Super Talent: The New World of Selective Migration Regimes. *Indiana Journal of Global Legal Studies* 20: 71–107.
- Shaviro, Daniel. 1/2015. The Crossroads versus the Seesaw: Getting a 'Fix' on Recent International Tax Policy Developments. *Tax Law Review* 69: 1–42.
- Smith, Matthew, Danny Yagan, Owen M. Zidar, Eric Zwick. 2019. *Capitalists in the Twenty-First Century*. Washington D.C.: National Bureau of Economic Research.
- Somerset Webb, Merryn. 2015. A Place in the Sun and a Tax-free Pension. The Financial Times, March 13
- Span, Paula. 2018. Many Americans Try Retirement, Then Change Their Minds. *The New York Times*, 30 March.
- Staats, Gerard. 3/2015. Personal Pensions in the EU. EC Tax Review 24: 144–157.
- Starink, Bastiaan. 1/2016. Source versus Residence State Taxation of Cross-Border Pension Payments: Trouble Shared Is Trouble Halved. *Intertax* 44: 6–13.
- Sternberg, Joseph C. 2019. *The Theft of a Decade: How the Baby Boomers Stole the Millennials' Economic Future*. London: Hachette UK.
- Stiglitz, Joseph E. 2015. *The Great Divide: Unequal Societies and What We Can Do About Them.* New York: Norton & Co.
- Sykes, Alan O. 1992. The Welfare Economics of Immigration Law: A Theoretical Survey with an Analysis of US Policy. Working Paper No. 10. Harvard Law School. J.M. Olin Law & Economics, Cambridge.
- Tanzi, Vito. 1999. Does the World Need a *World Tax Organization*?. 173–186 in *The Economics of Globalization*, edited by Assaf Razin and Efraim Sadka. Cambridge: Cambridge University Press.
- Tax Foundation. 2019. Poland's Youth Tax Cut Creates Dueling Incentives. 14 August. https://taxfoundation.org/poland-income-tax-cut-for-young-workers-creates-dueling-incentives/ (last visited 31 October 2019).
- Tesón, Fernando R. 2008. Brain Drain. San Diego Law Review 45: 899–932.
- *The Economist.* 2017. The Sun Sets on British Pensioners' Migration to Europe. 19 December.

- The Italian Insider. 2019. Tria, Brain Drain Costs Italian Economy 14 Billion. 17 July.
- Tilbrook, Andy. 2018. Retiring to Europe: The Brexit Race Is On. *Financial Times*, 14 June.
- UK Office for National Statistics (ONS). 2017. Pensioners in the EU and UK. https://www.ons.gov.uk/economy/investmentspensionsandtrusts/articles/pensionersintheeuanduk/2017–09–05 (last visited 31 October 2019).
- UNESCO. Institute for Statistics. 2012. *International Standard Classification of Education: ISCED 2011*. Montreal: UNESCO Institute for Statistics.
- van Bergeijk, Peter AG. 2019. *Deglobalization 2.0: Trade and Openness During the Great Depression and the Great Recession*. Cheltenham: Edward Elgar Publishing.
- van Horzen, Fred, Isabella M. de Groot. 2018. Chapter 9: The Switch-Over Clause: To Exempt or Not to Exempt, That Is the Question. in *The Implementation of Anti-BEPS Rules in the EU: A Comprehensive Study*, edited by Pasquale Pistone and Dennis Weber. Amsterdam: International Bureau of Fiscal Documentation (IBFD).
- van Zantbeek, Anton. 5/2010. Tax-Driven Relocation of High Net Worth Individuals: Where to Run to?. *European Taxation* 50.
- Wei, Yingqi, Vudayagiri N. Balasubramanyam. 11/2006. Diaspora and Development. *World Economy* 29: 1599–1609.
- Westminster Foundation for Democracy. 2019. Cost of Youth Emigration. https://www.wfd.org/wp-content/uploads/2019/05/Cost-of-yoth-emigration-Serbia.pdf (last visited 31 October 2019).
- Wijnen, Wim F.G., Jan J.P. de Goede. 3/2014. The UN Model in Practice 1997–2013. *Bulletin for International Taxation* 68.
- Wilkinson, Richard, Kate Pickett. 2019. The Inner Level: How More Equal Societies Reduce Stress, Restore Sanity and Improve Everyone's Well-Being. London: Penguin Press.
- Willetts, David. 2010. The Pinch: How the Baby Boomers Took Their Children's Future and Why They Should Give It Back. London: Atlantic Books.
- Willetts, David. 2019. Intergenerational Warfare: Who Stole the Millennials' Future?. *Financial Times*, 3 July.
- Williams, David. 13/2001. The Taxation of Cross-Border Pension Provision. *European Taxation* 41 (special issue).
- Wise, Peter. 2019. Portugal Lures Foreigners with Tax Breaks and Anti-Populist Stance. *Financial Times*, 7 August.

Xu, Angela. 3/2015. Pension Taxation in the EU: A Concern for Mobile Pensioners?. *CESifo DICE Report* 13: 75–77.

Young, Charles. 2017. The Myth of Millionaire Tax Flight: How Place Still Matters for the Rich. Palo Alto: Stanford University Press.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019. UDC 336.22:314.15(497.5); 330.341:005.952

CERIF: S141

DOI: 10.5937/AnaliPFB1904116G

Stjepan Gadžo, PhD*

USING TAX POLICY TO ADDRESS BRAIN DRAIN AND DEPOPULATION: THE CASE OF CROATIA

As the end of the Twenty-Tens approaches, there is a growing public consensus in Croatia that the key challenge facing the country is of demographic nature. Put simply, the accession to the European Union (EU) in July 2013 only exacerbated the negative trends regarding the emigration of mostly young and high-skilled workers to other, more developed countries. However, policymakers have hitherto failed to offer a comprehensive set of countermeasures, with tax policy being no exception. Accordingly, it is the aim of this paper to explore possible tax measures the Croatian legislator may employ in tackling the brain drain phenomenon, with special emphasis on highly skilled workers. More specifically, starting from the assumption that policymakers want to assume a more proactive role in addressing brain drain, the main contribution of the paper is in drawing the contours of a coherent tax-related response to this issue.

Key words: Brain drain. – Tax policy. – Personal income tax. – Preferential tax regimes. – Exit taxation.

1. INTRODUCTION

"Demographic disaster", "Croatian exodus", "Massive immigration worse than in the times of war" – these and similar headlines have appeared frequently in the Croatian media in recent years, painting a dire picture of the demographic trends and related socio-economic challenges the country is faced with at the end of the current decade. While a

^{*} Assistant Professor, University of Rijeka Faculty of Law, sgadzo@pravri.hr.

¹ See, for example, https://www.slobodnadalmacija.hr/novosti/hrvatska/clanak/id/580359/prava-demografska-katastrofa-iz-hrvatske-je-iselilo-cak-26-puta-vise-ljudi-nego-

number of forces underlie these developments, it is beyond doubt that the Croatian accession to the European Union (EU) in July 2013 only exacerbated the problem. Put simply, troves of Croatian citizens have taken advantage of the EU freedom of movement and emigrated to other, more developed Member States, such as Germany and Ireland. While the sheer number of émigrés is staggering – one study puts it at 230.000 in the 2013–2016 period alone (Draženović, Kunovac, Pripužić 2018, 436) – their structure causes even more concerns. Namely, in the post-EU accession period there is a notable increase in the emigration of both younger and highly-skilled people (Knezović, Grošinić 2017, 34). Special concerns relate to the flight of healthcare professionals and experts in other propulsive sectors of the economy, e.g. in the information and communication technology (ICT) sector.

Accordingly, the phenomenon of 'brain drain' – defined as the emigration of skilled and professional workers from a country (Wong 2009, 131) – is a genuine problem, albeit not a completely new one, that Croatian policymakers have to grope with. While the socio-economic implications of brain drain are undeniably deep and rather daunting, there is a general public consensus that hitherto no comprehensive set of policy countermeasures has been offered to this effect. This also applies to the more limited sphere of tax policy, even if cross-country experience confirms that tax instruments may play an important role in addressing international mobility of high-skilled labour (OECD 2011, 124).

In this respect, it has to be noted that the body of economic research confirms that cross-country differentials in individual income taxation play a role in people's location decisions. Moreover, such responsiveness seems to be higher for specific categories of workers, such as high-income earners and people whose human capital is not location-specific (e.g. inventors) (Kleven *et al.* 2019; Muñoz 2019). In any case, contemporary migration literature acknowledges that individual countries often use tax policy to address both outbound and inbound cases of highly-skilled migration. Regarding the former, one may speak of 'protective' or 'defensive' tax instruments, such as an exit tax imposed on the emigrant,

sto-kaze-nasa-sluzbena-statistika-najvise-se-odlazi-u-tri-europske-drzave; https://www.vecer nji.hr/vijesti/demografska-katastrofa-iz-hrvatske-iselilo-vise-od-20-000-djece-1243509; https://www.slobodnadalmacija.hr/novosti/hrvatska/clanak/id/580359/prava-demografskakatastrofa-iz-hrvatske-je-iselilo-cak-26-puta-vise-ljudi-nego-sto-kaze-nasa-sluzbenastatistika-najvise-se-odlazi-u-tri-europske-drzave; https://www.vecernji.hr/vijesti/demografs ka-katastrofa-iz-hrvatske-iselilo-vise-od-20-000-djece-1243509; https://www.index.hr/vijesti /clanak/becki-institut-masovno-iseljavanje-iz-hrvatske-kakvog-nije-bilo-ni-u-ratovi ma/2075476.aspx (all last visited 31 October 2019)

² For an overview see, for example, OECD (2011, 128–129); Kleven *et al.* (2019).

while the latter may be designated as 'offensive' measures, e.g. a preferential tax regime offered to immigrants.³

Against this broad backdrop, it is the aim of the present paper to explore possible tax policy measures the Croatian legislator may employ in tackling the brain drain, with the special emphasis on highly skilled workers (HSWs), i.e. individuals with at least a tertiary level of education.⁴ More specifically, in Section 2 the paper provides a depiction of migration and demographic trends in Croatia, serving as an illustration of why urgent policy action is warranted. In Section 3 it subsequently provides a general overview of the tax instruments that may be used on the domestic level to tackle the brain drain, allowing for lessons be drawn from crosscountry experiences. Presuming that Croatian policymakers want to assume a more proactive role in addressing brain drain, Section 4 proceeds with the analysis of pertinent developments hitherto and proposes future course of action. In doing so, particular attention is paid to the newly proposed preferential tax scheme for 'young workers', which is expected to come into force in 2020. While this tax scheme suffers from serious shortcomings, some building blocks of what the author believes should be a coherent tax policy response to the brain drain are expounded, with the aim to influence future debate. The main outcomes of the analysis are summarized in the concluding section of the paper.

Conversely, the design of a global or multilateral solution to the brain drain phenomenon, which is inextricably tied to cosmopolitan perspectives to tax justice, is beyond the scope of the present paper. While the author shares the view that such an approach is indeed desirable and may offer long-term answers to the most pertinent problems, it is hugely debatable whether it constitutes a truly realistic option under the current framework of international tax governance.

2. CROATIAN BRAIN DRAIN: IS THERE THE NEED FOR SERIOUS ACTION?

While it is beyond the scope of the present paper to analyse the specificities of the Croatian brain drain in great detail, it is undeniable that any "anti-brain drain policy" – including tax measures – has to be informed by at least a basic understanding of the phenomenon. Put simply,

³ Both sets of measures will be explored in detail below, in Section 3. In doing so, the paper departs from the analytical framework laid out in Berretta (2018).

⁴ While there is no ubiquitous definition of HSWs, the present paper departs from the assumption that "highly skilled" actually means "highly educated". See the definitions used by the OECD (2011, 124) and the European Committee of the Regions (2018, 7).

Such is the perspective taken, e.g. by Brock (2015) and Lister (2017).

domestic policymakers have to assess the magnitude of the brain drain, evaluate the main factors that influence the location decisions of Croatian migrants, and estimate the overall socio-economic effects of these dynamics. Therefore, this section proceeds with a brief description of migration and demographic trends in Croatia, serving as an illustration of why urgent policy action is warranted.

At the outset it is vital to note that Croatia has historically – since the 15th century – held the status of an emigration country, due to a combination of economic and political factors (Župarić-Iljić 2016, 16). While the country's favourable geographic position and overall standard of living have traditionally also attracted troves of immigrants, particularly from other countries of South-East Europe (Knezović, Grošinić 2017, 16), the negative net migration balance during the entire 20th century has been estimated at 1.2 million (Gelo, Akrap, Čipin 2005, 70). It is a well-known fact that, among countries of comparable size, Croatia has one of the largest diaspora communities, with more than 3 million Croatian citizens living abroad, compared to a domestic population of around 4.2 million (Knezović, Grošinić 2017, 26–27).

The most recent emigration wave of Croats could be traced down to the beginning of the economic downturn in the country in 2009. The ensuing recession lasted for six years (2009–2014), making Croatia as one of the worst economic performers among EU member states. Unsurprisingly, faced with such dire economic conditions – especially a lack of employment opportunities – a number of citizens opted to leave the country. In fact, official government data compiled by the Croatian Bureau of Statistics (CBS) suggests that the onset of the crisis reversed the trend of positive net migration from the beginning of the 21st century (Draženović, Kunovac, Pripužić 2018, 420). The relatively low and stable rates of negative net migration post-2009 have taken a visible turn for the worse since the Croatian accession to the EU in July 2013, demonstrating clearly that the access to the EU labour market constitutes one of the major drivers of emigration (Župarić Iljić 2016, 16–17).

There has been a lively public debate in Croatia on the exact magnitude of these most recent migration flows. Due to a number of factors, mainly of methodological nature, the official migration statistics issued annually by the CBS have been rejected as unreliable in literature, with a shared view that the real numbers of emigrants are significantly higher. As an illustration, while official data puts the negative migration balance – including the relations with non-EU countries – in the 2013–2018 period at around 100,000 (Croatian Bureau of Statistics 2019a), a recent study based on official data compiled by the destination countries

⁶ In this regard, we can refer to a comprehensive account provided in Draženović, Kunovac, Pripužić (2018).

estimates that 230,000 people left for other EU countries in the 2013–2016 period alone (Draženović, Kunovac, Pripužić 2018, 436). The numbers seem ever more worrying when viewed together with the negative trends regarding natural population decrease, with the total number of deaths outnumbering the total number of live births by 106,000 in the 2009–2018 period (Croatian Bureau of Statistics 2019b). In fact, Croatia is one of only ten countries in the world that have experienced both negative natural increase and negative net migration in the current decade (2010–2020) (United Nations 2019, 35), thus fuelling concerns of population decline and ageing. According to the UN's World Population Prospects 2019, in the year 2100 Croatia is expected to number less than 2.2 million inhabitants. Another recent study estimates that by the year 2050 around 45% of the domestic population will be aged 55 years or more (Eurostat 2019a, 15).

Beyond the sheer number of people who have left the country, the recent migration flows raise even more concerns when one takes a look at the structure of émigrés. First, it seems that the average age of emigrants has decreased sharply in the post-EU accession period (Draženović, Kunovac, Pripužić 2018, 420–421). In 2018 around 45% of the emigrants were ages 20–39 (Croatian Bureau of Statistics 2019a). Accordingly, a particular source of concern that most emigrants are in their prime age regarding fertility and work abilities (Župarić Iljić 2016, 23). Second, statistical shortcoming aside, it may be reasonably assumed that the EU accession intensified emigration of highly-skilled labour (Knezović, Grošinić 2017, 34; Jurić 2017, 349). According to the report published by the European Committee of the Regions (2018, 12), in the 2014–2017 period Croatia recorded the second largest increase in the number of highly-educated movers (+46%), i.e. migrants with a tertiary level of education, among the EU member states.

While comprehensive sectoral analyses have been largely absent, anecdotal evidence suggests that the brain drain has had significant effects in the healthcare and medical sector (Župarić Iljić 2016, 23–24). In the 2013–2018 period, 525 doctors aged between 25 and 46 have left the country (Vračić 2018, 7). A recent survey conducted among doctors and other healthcare professionals revealed that more than 50% of the doctors aged 45 or younger are thinking about leaving Croatia. Further concerns relate to the flight of experts in the field of ICT, with the number of these highly-sought professionals leaving the country far exceeding the number

These detailed projections are available at https://population.un.org/wpp/Down load/Probabilistic/Population/ (last visited 31 October 2019)

⁸ The survey was conducted by the Croatian Medical Chamber in September 2019. The data is available at https://www.hlk.hr/istrazivanje-hlk-cak-60-posto-lijecnika-spremno-povuci-suglasnost-za-prekovremeni-rad.aspx (last visited 31 October 2019)

of newly graduated ICT experts entering the domestic labour market. Put simply, Croatia is at the losing side of intra-EU migrations linked with the so-called knowledge economy (European Committee of the Regions 2018, 9–14).

Regarding the main drivers of emigration from Croatia, there is an amalgam of different 'push' and 'pull' factors at work (Župarić Iljić 2016, 2–3). A recent empirical analysis confirms that access to the EU internal market has indeed been the main driver of emigration since 2013 (Draženović, Kunovac, Pripužić 2018, 435). However, other important factors include the differentials in short-term economic conditions between origin and destination countries (including, e.g. labour market indicators) as well as the degree of corruption in a country (Draženović, Kunovac, Pripužić 2018, 432–436). The importance of non-economic drivers of emigration is confirmed in a number of other studies. For example, in an analysis of the motives for emigration to Germany – by far the most popular country of destination for Croatian emigrants post-2013 – Jurić (2017) highlights the importance of push factors such as corruption, immorality of political elites and legal uncertainty. This is an important lesson for policymakers, since it calls for a holistic approach to the brain drain phenomenon, beyond pure economics.

In any case, these new migratory trends are usually the topic of public discussion in Croatia in terms of their negative socio-economic effects. Biggest concerns relate to the loss of human capital – particularly young, highly skilled professionals – distortions in the labour market, a decrease in the overall productivity of the economy, and the ensuing pressures on the social safety net (Župarić Iljić 2016, 23). On the other hand, policymakers have to be aware that emigration may also have some positive effects, both in the short and long term. Notably, it is quite possible that emigrants may in due time return to their country of origin, armed with newly acquired knowledge and skills. Thus, the related concepts of 'brain regain' and 'brain circulation' have been long acknowledged in migration literature as potential benefits for the sending countries or regions. Further alleviation of the brain drain problem may come in the form of remittances sent to the country of origin by citizens working abroad. In the case of Croatia, these transfer of money from abroad amounted to almost 2.2 billion USD – or 4.3% of the domestic GDP in 2017 (United Nations 2017, 30).

Finally, it has to be noted that the intra-EU mobility of Croatian workers since the accession in 2013 has been subject to some important limitations, since a total of 13 EU Member States employed transitional restrictions on access of workers from Croatia to their respective labour

⁹ For an overview see Brauner (2010, 228–237).

markets.¹⁰ As of July 2018 such restrictions are applied only by Austria, which is traditionally – due to geographic proximity and a variety of historic, cultural and socio-economic causes – an important country of destination for Croatian citizens (Župarić Iljić 2016, 18). This last transitory measure will expire by July 2020, adding to the urgency for domestic policymakers to devise a comprehensive brain drain strategy.

3. TAX POLICY RESPONSES: AN OVERVIEW

While the primary purpose of taxation is raising revenues necessary for the financing of public goods, taxes - or, more precisely, certain elements of a particular type of tax – may also have other purposes. 11 According to Avi Yonah (2006, 22-25), aside from a pure fiscal or revenue-raising goal, modern day taxation has two other main functions - redistribution of income and regulation. The latter entails the usage of taxes with the aim to affect the behaviour of citizens, corporations, and other private sector actors, either by incentivizing some activities or by disincetivizing – or rather penalizing – others. This theoretical framework offers a good backdrop for considering the role of taxation vis-à-vis the international mobility of individuals. Namely, countries may employ various tax schemes with a regulatory aim of influencing location decisions of prospective migrants. These measures should obviously be tailored to the local idiosyncrasies regarding migration flows and the overall socio-economic context, with some countries – like Croatia (see Section 2 above) – being under pressure to act to reverse the negative migration trends, while others – usually developed countries – may use tax measures to utilize the ever-growing global mobility of individuals to their advantage, by attracting international talent with, inter alia, a competitive economic and tax climate. Further justification for some of these schemes may be found in the traditional legal and economic principles of taxation, such as the ability to pay principle or the benefits principle.

Tax policy responses to the new reality of mobile highly-skilled workers may be divided into two main groups (Beretta 2018, 7). First, there are 'protective' or 'defensive' tax measures that may be used by a country of emigration (the 'sending country') and that, in the jargon of migration literature, have the effect of a 'pull' factor. Put simply, here the

¹⁰ A brief overview of these transitory measures, with their date of expiry, is available at https://ec.europa.eu/social/main.jsp?catId=1067&langId=en (last visited 31 October 2019).

This is sometimes referred to in the literature as the 'instrumentalism' of tax law. See, for example, Gribnau (2003, 25).

sending country reacts to prospective migration by protecting its tax claims on emigrants' income, encompassing the income accrued but not realized before the relocation and the income emigrants may derive in the future. Accordingly, it is possible to differentiate three main types of defensive measures (Beretta 2018, 11): 1) exit taxes; 2) trailing taxes; and 3) claw-back provisions regarding the tax benefits granted in the previous period.

The second group of tax measures may be labelled as 'offensive', in that they are used to induce immigration of HSWs – which may include the return of previous emigrants – into the country. Such schemes may be labelled as 'preferential tax regimes' or 'tax concessions', since here the country grants some kind of beneficial tax treatment (e.g. reduced tax rate) to a targeted group of mobile individuals (Beretta 2018, 19; OECD 2011, 137–141).

The rest of this section proceeds with an analysis of specific defensive and offensive tax measures targeting brain drain, on the basis of selected comparative examples. Since the goal of this exercise is to draw lessons for particular case of Croatia, of special importance is to explore the legal and economic underpinnings of these schemes.

3.1. Defensive measures

From a purely international tax perspective, the event of an individual's emigration from a country is important since it generally results in the termination of the link (nexus) between that country and the individual providing the legal basis for the imposition of income tax. ¹² In other words, the individual ceases to be a tax resident of the origin country and her income is thus placed outside of the ambit of that state's tax jurisdiction. The ensuing revenue loss for the coffers of the state of emigration is yet another concern related to global migration flows of young professionals. On this point, one needs to distinguish between at least two parts of the income that is relocated beyond the jurisdictional reach of the emigration state (De Broe 2002, 23; Beretta 2018, 10): (i) gains accrued but not yet realized before the emigration, including the appreciations of the migrant's assets; (ii) income accrued to the taxpayer post-migration, e.g. income from future employment or investment.

Accordingly, a number of countries have made a sovereign decision to impose some sort of a 'departure tax' (or 'emigration tax'), i.e. a tax triggered by the individual's departure from the country. The main

For a general discussion on the 'income tax nexus' in international tax law see Gadžo (2018).

For terminological nuances and the differentiation between the terms 'exit tax', 'emigration tax' and 'departure tax', see, for example, De Broe (2002, 23–25). Of course,

justification provided in tax literature for such measures relates to the legal fundamentals of taxing cross-border income: the state of departure is free to protect its latent taxing claim over the income that is accrued in its territory as well as extending taxing claim on future revenue streams of ex-residents (De Broe 2002, 23). In the well-known jargon of EU tax law, emigration taxes are *prima faciae* reasonable since they ensure fiscal coherence, at least from the viewpoint of the state of emigration (Terra, Wattel 2012, 955–956). However, for the purpose of a meaningful analysis of the normative merits of such measures, it is useful to clearly separate three main types of departure taxes (De Broe 2002, 23): (i) a general or limited exit tax on the accrued gains; (ii) extended tax liabilities or trailing taxes; and (iii) recaptures of the tax benefits enjoyed pre-departure.

Regarding 'classical' exit taxes, it is self-evident that any taxation of accrued unrealized income – whether it extends to all assets belonging to the expatriate (e.g. in Canada or Australia), or is limited only to specific types of property (e.g. the Dutch regime of taxing 'substantial shareholdings')¹⁴ – acts as a disincentive for migration (Arsenault 2009, 59). However, it is extremely doubtful whether this extra cost of departure will offset the expected benefits of the move and thus influence the location decision (Brauner 2010, 265). Further justification for an exit tax regime may be found in the so-called 'benefits principle', which is one of the two main benchmarks of equity or fairness in the distribution of tax burden among individual taxpayers. ¹⁵ Put simply, the imposition of a tax burden on the act of emigration may be justified by the benefits the taxpayer in question has previously enjoyed in the state of departure, including the legal protection of her assets. 16 Indeed, it has been acknowledged in tax scholarship that economic cooperation between society members gives rise to certain mutual benefits; in turn, society members need to accept some distributive obligations, including tax obligations (Gadžo 2018, 208). Such benefits provided by the government and other entities belonging to the public sector include, inter alia, a functioning judicial system, protection of property rights, public infrastructure, etc. According to Dietsch (2015, 80-89) this may be labelled as a 'membership principle', demanding that individuals should be liable to tax in a country of which they are member, i.e. countries where they benefit from public services and infrastructure. Against this backdrop, it seems beyond doubt that an emigrant who lived in a country

emigration taxes may be also imposed on corporations and other legal entities, but this is of no relevance for the analysis in the present paper.

¹⁴ For an analysis of Canadian and Dutch exit tax schemes, see Chand (2013).

¹⁵ For a more detailed discussion on the content of tax equity see Gadžo (2018, 200–205), and the sources referred to therein.

This argument has been raised in the U.S. within the debate on the desirability of introducing the tax on expatriates in 2008. See Arsenault (2009, 59).

for a considerable period of time had access to public services and public infrastructure to a reasonable extent, thus making them a member of the society with ensuing distributive obligations. This point is linked to the design of an exit tax scheme in countries that follow the 'look-back' approach (e.g. Denmark, Spain, the Republic of Korea), in that they impose the exit only if the expatriate has lived in the country for a substantial number of years before emigrating (Beretta 2018, 13).

Trailing taxes or extended tax liabilities may be more controversial, since they involve the imposition of tax by the origin state in the taxable years following the taxpayer's departure. Accordingly, while these measures are usually discussed in literature as burdening assets previously connected with the territory of the origin state (De Broe 2002, 29–30; Chand 2013, sec. 2.3), they may also be imposed on the streams of income resulting from future employment or entrepreneurial activity in the state of destination.¹⁷ This is in line with some suggestions in the migration literature – epitomized by the so-called 'Bhagwati tax' proposal presented in 1972 – that developing countries, as traditional countries of emigration, should be entitled to a share of the income tax collected on the future income derived by their émigrés (Brock 2015, 52–53). It seems that here the above-discussed benefits argument plays an even more important role than in the case of a classical exit tax. Namely, starting from the assumption that the majority of emigrants are younger, highly skilled individuals 18, one can identify a trove of public benefits provided to them by the emigration country, including the costs of education and training (Brock 2015, 62-63). This may be perceived as a 'sunk investment' that emigration countries may legitimately seek to recuperate, at least partly, by imposing a trailing tax on the former members of their community, i.e. on ex-residents (Brauner 2010, 229). It has to be noted that this type of trailing tax - imposed e.g. on emigrant's future employment income – raises far more concerns from a public international law perspective, since the legal link between the origin state and the taxpayer is less evident. Accordingly, in terms of design features, such schemes often take the form of a 'deemed residence rule', in that the emigrant continues to qualify as a resident of her state of origin postdeparture (De Broe 2002, 29-30; Beretta 2018, 15).

Finally, the state of origin may employ the so-called 'recapture' or 'claw-back' rules, i.e. the rules that allow it to recoup, upon the act of emigration, deductions, deferrals and other tax benefits previously granted to the taxpayer in question (Beretta 2018, 18). While in comparative tax systems recapture rules are not that common and are usually linked to a

¹⁷ See Beretta (2018, 15).

¹⁸ This, of course, entails an analysis of the individual countries of emigration. For the case of Croatia see the statistics presented in Section 2 above.

deferred tax schemes (e.g. vis-à-vis pension income), they may be justified, similarly to a classical exit tax, by achieving the overall coherence of a tax system. For the purpose of the present paper it is particularly interesting to note how these measures may be used as a backup for a preferential tax regime granted to highly skilled individuals (see below, Section 3.2).¹⁹

While the preceding discussion in this Section emphasized the reasons why departure taxes may be justified from a policy perspective, the majority of countries around the globe – including Croatia (see Section 4 below) - have so far abstained from introducing such instruments in their tax systems. Indeed, it is rather easy to identify main policy shortcomings linked to the introduction of an emigration tax. First, it has to be acknowledged that any sort of tax imposed on the event of individual's migration prima faciae runs against the basic economic tenet of tax efficiency, in that people's decisions to move across national borders should not be influenced by a barrier in the form of extra tax burden.²⁰ As pointed out at the beginning of this section, however, tax legislators often put pure economic logic aside having a legitimate regulatory goal in mind. Second, one may find departure taxes questionable from the perspective of tax equity (fairness), since emigrants may feel that their distributive obligations to the country of origin should cease at the moment of departure, with any ensuing increase in the ability to pay belonging to the tax ambit of the destination country. Of even more concern from the fairness point of view is the potential double taxation that may ensue, depending on the interaction between the tax rules of the state of origin and the state of destination (Beretta 2018, 42). Third, departure taxes may involve significant compliance costs for the taxpayer, as well as administrative costs for the tax authorities of the state of origin, particularly in the absence of relevant international agreements in the area of mutual assistance. Fourth, one also has to have in mind the potential advantages flowing to emigration countries from the global mobility of individuals, including the benefits of "brain circulation" (see Section 2, above).

3.2. Offensive measures

Driven by the regulatory goal of inducing highly-skilled individuals to live and work in their own territory, a number of countries employ a specific preferential tax regime provided to this category of (potential) migrants (Kleven *et al.* 2019, 6). This fits well with the competition-based paradigm of international taxation, whereby tax measures are used

¹⁹ For an Italian example in this regard, see Beretta (2018, 18).

On this point in the context of free movement within the EU internal market, see Terra, Wattel (2012, 955).

by states to compete for mobile tax bases (e.g. capital). Put simply, starting from the assumption that wage differentials – which are also affected by the overall tax and social security burden imposed on labour income, i.e. the 'tax wedge'²¹ – are one of key drivers of HSWs' location decisions²², individual countries may offer certain tax incentives for immigration. From a tax-technical perspective, these incentives may take the form of lower tax rates, tax exemptions, tax allowances, deductions, etc. For example, in the year 2010 such incentives were found in tax systems of 16 OECD member states (OECD 2011, 131).

It would be wrong, however, to view preferential tax regimes for HSWs as a policy tool reserved for developed countries. Developing countries may also employ such schemes, either to incentivize return migration, or to increase their overall competitiveness for global talent (Brauner 2010, 266; Del Carpio et al. 2016, 2). The latter point is of particular importance today, since no country can ignore the role of human capital in propping up their knowledge-based economy. A good example in this regard is provided by the Malaysian Returning Expert Program (REP). Introduced in 2011, this scheme provides several benefits to Malaysian citizens that have been residing and employed abroad continuously for at least three years prior to application. The benefits granted upon return include a 15% flat tax on employment income, for a period of five years post-return, tax exemptions related to any personal assets the returnee brings into Malaysia, as well as exemption from duties with regard to the purchase or import of a personal vehicle (Del Carpio et al. 2016, 7–8). According to data published by the administrative agency in charge of the programme, more than 5,000 individuals have used its benefits in the 2011–2018 period.²³

The tax concessions provided to HSWs in domestic tax laws may differ considerably in terms of their design. First, it is vital to delimit the subjective scope of the scheme, having in mind the targeted population. While a small group of countries (e.g. Australia, Israel, Spain, United Kingdom) do not demand any skill requirements in order for a migrant to be eligible for the preferential regime, the majority of countries use the so-called targeting provisions, in that the benefits are provided only to migrants possessing certain skill type or level (OECD 2011, 137). Whereas from the administrative perspective it may be simpler to rely on the level of formal education in this regard – e.g. by making the scheme

²¹ For its statistical purposes, the OECD defines tax wedge as "(T)he sum of personal income tax, employee and employer social security contributions plus any payroll tax less cash transfers expressed as a percentage of labour costs." (OECD 2018, 576).

See note 2 above, and the accompanying text.

²³ See https://www.talentcorp.com.my/initiatives/returning-expert-programme (last visited 31 October 2019).

available to everyone holding a tertiary degree – some countries narrow the subjective scope by targeting specific types of prospective workers. For example, the famous Dutch 'tax ruling scheme', first introduced in 1985, addresses non-residents with professional expertise and skills that are scarce in the Netherlands, including scientific researchers. In a similar vein, Denmark makes its preferential regime available only to scientists and other employees with a salary above the prescribed high threshold (Casarico, Übelmesser 2018, 31–32). It seems that a sort of a hybrid approach is followed in Italy, offering preferential tax treatment both to migrant researchers and other persons with a certain level of education.²⁴ Conversely, some taxpayers may be explicitly declared ineligible for the scheme, as is the case with professional athletes in Spain (Beretta 2018, 19).

Countries may further rely on criteria such as nationality or (previous) residence to further limit the subjective scope of the preferential regime. Accordingly, in some cases, such as in Israel or in Malaysia (see above in this Section), the scheme is open only to non-resident citizens. A completely opposite approach is followed in other countries (e.g. Belgium, Korea, Netherlands) targeting only foreign nationals (OECD 2011, 143–144). Quite obviously, in light of freedom of movement within the internal market, EU Member States are prohibited from discriminating in this regard against nationals of other EU countries.

Second, regarding the objective scope of the scheme, in most cases only employment income (i.e. a salary) is covered (Beretta 2018, 19–20). However, a broader, more generous approach may be followed, for example extending the benefits to self-employment income and pension income, as is the case in Portugal (Beretta 2018, 20). Due to administrative complexities associated with migration, there are countries that extend preferential treatment also with regard to immigrant's foreign-sourced income (e.g. Australia, New Zealand, Portugal) (OECD 2011, 141; Beretta 2018, 20).

Third, the standard approach is to set the time limitations for preferential tax treatment, thus making it a temporary concession (OECD 2011, 142–143). Time thresholds vary across countries, from two years in Finland to 10 years in Portugal (Kleven *et al.* 2019, 27). In this regard it is worthy to take a look at the historic development of the Dutch scheme, which originally relied on a five-year time threshold. At a later point, the threshold was extended to 10 years, while currently it is stands at eight years. In any case, time restrictions are closely related with potential

Detailed information on the Italian scheme is available at https://www.agen ziaentrate.gov.it/portale/documents/20143/233483/Tax+incentives+for+attracting+huma n+capital+in+Italy_Tax_incentives_for_attracting_human_capital_in_Italy.pdf/f4a91a80-8ed0-92a5-0186-424a9013bfc3 (last visited 31 October 2019).

policy shortcomings of preferential regimes for HSWs, which is discussed in more detail below.

As already mentioned above, every preferential tax regime for high-skilled workers is primarily a regulatory instrument, in that its main aim is to attract, or retain, targeted individuals in the territory of the given country. This aim is based on the view that there are numerous benefits flowing to the country from an increase in the number of HSWs within its territory. These include, *inter alia*, knowledge-related spillovers increasing the overall level of productivity in the country, positive effects on the labour market in cases of shortages of specific skills, etc. (OECD 2011, 133–134; Casarico, Übelmesser 2018, 31). Put simply, preferential regimes for HSWs constitute a worthy state intervention in order to reap such positive externalities. In this regard it is vital to have in mind the subjective scope of the scheme (see above in this Section), with a number of countries targeting exclusively professionals involved in research and development (R&D) activities.

Likewise, these schemes are a tool of international tax competition, a concept that captures the realpolitik of taxing internationally mobile tax bases, including the labour income of HSWs. 25 In this setting, no country - particularly a small, open economy - can afford to ignore the effective tax burdens in other jurisdiction, and may be forced to reply with its own measures. This seems particularly relevant for high-tax countries and may explicate the introduction of preferential regimes for HSWs in Belgium, Denmark, Finland and Sweden (OECD 2011, 132). Accordingly, some preferential tax schemes may be viewed as having compensatory rather than 'offensive' nature in incentivizing the behaviour of global migrants. Similar logic ostensibly underlies granting beneficial tax treatment to various items that may be categorized as 'costs of migration', e.g. travel costs, additional costs of relocating other family members, etc. For example, one of the justifications provided by the Dutch government in favour of the preferential regime, whereby 30% of qualified employee's gross salary may be paid out as a tax free allowance, is that it provides compensation to foreign professionals for the additional costs related to migration.²⁶

Proponents of preferential regimes for HSWs will be also quick in pointing out that their overall fiscal effect for the state of destination tends to be positive, even if the government foregoes some revenue in the

²⁵ It is beyond the ambit of the present paper to discuss the fundamentals of international tax competition. For an overview see, for example, Faulhaber (2018).

²⁶ See https://www.belastingdienst.nl/wps/wcm/connect/bldcontenten/belastingdienst/individuals/living_and_working/working_in_another_country_temporarily/you_are_coming_to_work_in_the_netherlands/30_facility_for_incoming_employees/30_facility_for_incoming_employees (last visited 31 October 2019).

first step by granting tax concessions (OECD 2011, 134). Put simply, most migrants – particularly HSWs and workers below a certain age – will be net contributors to the tax and social security system of the immigration country.

On the other hand, there is also a number of policy shortcomings linked with the introduction of preferential regimes for mobile individuals. The primary objection comes from the perspectives of equality and equity, which constitute basic legal principles of taxation, usually embodied in national constitutions.²⁷ It is abundantly clear that giving HSWs preferential tax treatment violates these legal precepts, since one class of taxpayers is put in a privileged position vis-à-vis other members of society (Li 2009, 54). It should be noted that this fundamental criticism may be raised also with regard to other tax expenditures (Gribnau 2003, 25–27). Moreover, the principle of ability to pay dictates in general that the better-off participate more to the financing of public goods, in relative terms. Since the individuals qualifying for preferential regime tend to possess highly sought skills and thus earn above-average income, it is objectionable on the face of it to require these individuals to make lower tax payments than what is mandated by their respective economic faculties (Beretta 2018, 24–25). On the other hand, it is well-understood among tax scholars that legislators have to balance the general normative guidance provided by the principles of tax fairness with other desired objectives of the particular tax measure (e.g. its regulatory goal), meaning that the former may constitute justification to deviate from the latter (Gribnau 2003, 29–30).

Further to this point, critics of preferential schemes often emphasize that their actual behavioural effects are uncertain, meaning that the effectiveness of attracting mobile migrants, as an overarching justification for the introduction of the scheme, is questionable and may be difficult to prove empirically (Li 2009, 54). In other words, due to a number of other factors that affect location decisions, mobile individuals may not respond to the preferential regime as expected by the authorities in the state of destination. Even individuals who decide to relocate to the country may leave when the time limitations of the scheme are reached. In any case, there may be better, less distortive alternatives that pursue similar migration-related goals, including those outside of the ambit of the tax system, e.g. more targeted public investments in education or training (OECD 2011, 136–137).

Finally, it must be accepted that any introduction of a special regime that deviates from the ordinary system of taxation in the country raises the overall level of tax complexity and may therefore bring about

²⁷ For a general discussion on the importance of these principles in the context of taxation, see, for example, Gribnau (2003).

²⁸ For an empirical study see Kleven *et al.* (2019).

new administrative and compliance costs. This point is related to the above-discussed design features of a particular preferential regime for HSWs, since it would be difficult to legislate a well-targeted facility in simple legal language (Li 2009, 55).

4. CROATIAN EXPERIENCE AND POTENTIAL POLICY DIRECTIONS

The preceding section provided a general overview of the tax instruments that may be used on the domestic level to tackle brain drain. Against this backdrop, and taking into account relevant tax policy objectives, some lessons from cross-country experiences may be drawn and applied to the particular case of Croatian brain drain, which has been described in Section 2. Since the main aim of the present paper is to explore potential tax policy measures the Croatian legislator may employ in addressing present migration and demographic trends, with a particular emphasis on HSWs, this section proceeds with the analysis of pertinent developments hitherto and proposes building blocks for a new, coherent approach.

As to the last point, it is presumed that Croatian policymakers want to assume a more proactive role in alleviating socio-economic pressures resulting from the troubling rates of immigration of younger, highlyskilled workforce.²⁹ The following discussion is fruitless and may be dismissed as academic daydreaming if the governing elites continue with a largely passive approach to global migration flows. It must be noted in this regard that Croatia is yet to adopt a new, comprehensive demographic strategy on the national level, even if some of the measures discussed at the level of official expert groups have been leaked to the general public. Admittedly, negative demographic prospects have been explicitly acknowledged and highlighted as one of the major pressure areas in several strategic documents issued recently by the central government (Government of the Republic of Croatia 2019a, 48-49; Government of the Republic of Croatia 2019b, 38–47). However, the envisaged measures for addressing these concerns have been limited to areas such as the institutional support for infants and pre-school children, maternal leave and parental benefits, etc., with no explicit reference being made to the role of taxation (Government of the Republic of Croatia 2019a, 48–49).

Accordingly, it comes as no surprise that none of the measures discussed in Section 3 as tools of anti-brain drain tax policy have been featured in the Croatian tax system since its inception at the beginning of

²⁹ On the necessity of a more proactive demographic policy in Croatia see, for example, Jurun, Ratković, Ujević (2017).

the 1990s. Even if income is traditionally taxed on a worldwide basis, there has been no serious attempt to employ a departure tax regime for individuals, with the only major tax concern for permanent emigrants being the regulation of their residence status. Such an idea has not come to the fore even though an exit tax will be introduced in the area of corporate taxation, starting 1 January 2020, as a result of the necessity to implement the rules from the EU Anti-Tax Avoidance Directive. Therefore, the Croatian tax system is largely neutral to the act of individuals' emigration from the country – including HSWs – with no special provisions other than the general framework on taxation of income, aimed at disincentivizing such location choices.

Likewise, there are no preferential tax schemes in the Croatian tax system provided to highly-skilled immigrants, including potential returnees. Any immigrant has to take into account the general rules on income taxation in order to determine her tax burden post-migration, including the worldwide-based taxation of resident taxpayers.

4.1. The New Preferential Regime for 'Young Workers': A Well-Intended, but Misdirected Tax Instrument?

Quite interestingly, it was only in 2019 that the capacity of tax policy to respond to migration concerns was explicitly acknowledged. Namely, at the time of writing, a new, migration-related tax measure was in the legislative pipeline: the proposed amendments of the Personal Income Tax Act (PITA)³², expected to enter into force on 1 January 2020, envisage introduction of the completely new preferential tax regime targeting young workers. More precisely, the proposed benefit takes the legal form of a tax credit in that a fixed percentage is to be subtracted from the tax liability on employment income, calculated under the general rules of personal income taxation. Taxpayers younger than 25 years are to be granted a 100% tax credit, meaning that they will effectively pay no tax on employment income. The second category of beneficiaries are

While the domestic rules on tax residence may be considered similar in nature to 'trailing taxes' (Beretta 2018, 17–18), they are not left out of the analysis in the present paper, since it is difficult to see a genuine connection with the brain drain phenomenon here.

³¹ See Council Directive (EU) 2016/1164 of 12 July 2016 laying down rules against tax avoidance practices that directly affect the functioning of the internal market, OJ L 193 of 19 July 2016. It is interesting to note that Poland made the choice to extend the exit tax provisions of the ATAD also vis-à-vis individuals. See https://assets.kpmg/content/dam/kpmg/pl/pdf/2018/09/pl-en-tax-alert-KPMG-2018-09-12-taxation-of-unrealized-capital-gains-exit-tax.pdf (last visited 31 October 2019).

³² Official Gazette of the Republic of Croatia, 115/2016 and 106/2018. The proposed legislative amendments, together with public comments, are available at https://esavjetovanja.gov.hr/ECon/MainScreen?entityId=12039_(last visited 31 October 2019).

individuals aged 25–30, who are granted a 50% credit in regard to the tax liability on employment income.

In the explanatory text accompanying the bill, the Government explicates the policy underlying this new preferential scheme. In short, by reduction of the tax wedge in regard to young workers' wages, and the corresponding increase in their net disposable income, young and highly-skilled people are given an incentive to stay in Croatia, instead of moving to other countries. Furthermore, the Government explicitly acknowledges the demographic aim of the proposed scheme, without providing any detailed impact assessment or explaining the expected causality.

Evidently, the introduction of this regime is based on the assumption that cross-country wage differentials indeed have a big impact on the location decisions of younger migrants.³³ In terms of the dichotomy between defensive and offensive anti-brain drain tax measures, discussed in Section 3 above, the new scheme displays mostly defensive or protective characteristics. Namely, while it does not impose a new tax burden on individuals' departure from Croatia, as is the case with an exit tax regime, it acts as a tax disincentive to emigration at the expense of the public coffers. In doing so, it is aimed at primarily preserving the existing tax base within Croatia's taxing powers. Conversely, it is difficult to imagine how the new regime may induce immigration to Croatia, particularly due to the relatively low age thresholds limiting its subjective scope. Since the scheme will apply to all workers below the specified age limit, it does not target immigration, including the return of expatriates to Croatia. Quite interestingly, the only other country in Europe that currently applies a similar preferential tax regime for young workers is Poland. Whether the Polish regime – targeting workers below the age of 26, and in effect since 1 August 2019 – inspired the legislative developments in Croatia, remains unanswered at the present time.

The proposed scheme for young workers has been subject to heavy public criticism in Croatia ever since its announcement. First, the predicted behavioural effects of the scheme have been contested from several standpoints. For example, while the Government is apparently convinced of its immediate effects on the level of net salaries, the possibility remains that the reduced tax wedge will be simply soaked up by the employer, thus leaving the net amounts flowing to the employees intact. Even if net salaries do indeed increase as a result of these legislative changes, it is debatable whether its effect will be substantial enough to affect young people's decision on whether to migrate. Namely, it may be reasonably assumed that most young workers, particularly those under the age of 25, have relatively low earnings and are thus not heavily burdened by the personal income tax. Furthermore, a simple calculation shows that an

³³ See above, note 2 and the accompanying text.

individual without children, aged 27 and earning the average salary in Croatia, will experience an increase in the monthly disposable income of around 55 EUR. Such benefit does very little to compensate for huge wage differentials between Croatia and the EU Member States that are most popular destination countries for Croatian migrants (i.e. Germany, Austria, Ireland, Sweden). Moreover, the new regime does not provide any answers to the vexing question of whether beneficiaries will leave the country once the preferential regime expires, i.e. when a person turns 31 years of age. Whether some type of recapture or claw-back rules are warranted in this respect is discussed below (Section 4.2).

The second line of criticism is related to the target of the measure. Due to relatively low age thresholds, the regime does not target the population in the 30–39 age group, even though this segment of the population, in prime age regarding fertility and work, constitutes a large share of the emigrants in recent years (Croatian Bureau of Statistics 2019a). Moreover, it must be noted that, unlike most preferential regimes for immigrants discussed above (Section 3.2), there are no skill requirements for the beneficiaries. If the aim of the scheme is to influence migration decisions of HSWs, then the age thresholds are set rather poorly, since individuals with a tertiary education enter the labour market at the age of 23–24 at the earliest.

Third, is has been suggested that the scheme should be deemed unconstitutional, particularly in light of the constitutional prohibition of discrimination (Article 14 of the Constitution of Croatia)³⁴ and the principles of equality and equity in taxation, embodied in Article 51 of the Constitution. Regarding the discrimination objection, the envisaged scheme targeted at 'young employees' undeniably causes unequal tax treatment on the basis of age, which is one of the protected grounds of discrimination, even though not explicitly listed in Art. 14(1) of the Constitution.³⁵ However, on its own, this may not lead to the conclusion on constitutionally prohibited discrimination, since one has to take into account the justifications and policy objectives underlying the rules in question. Namely, the case law of the Croatian Constitutional Court (Ustavni sud Republike Hrvatske) has affirmed that the legislator has a wide margin of appreciation in setting economic and social policy, including tax policy.³⁶ Accordingly, unequal treatment produced by tax

³⁴ Official Gazette of the Republic of Croatia, 56/90, 135/97, 8/98, 113/00, 124/00, 28/01, 41/01, 55/01, 76/10, 85/10 and 05/14.

³⁵ However, age is directly referred to among discriminatory grounds in Article 1(1) of the Anti-Discrimination Act (*Official Gazette of the Republic of Croatia*, 85/2008 and 112/2012). More on this, see Omejec (2009, 883–886).

³⁶ See, for example, decision of the Constitutional Court of the Republic of Croatia, U-IP/3820/2009, 17 November 2009, para. 10. For a general discussion see Bagić (2016, 324–325).

rules may be justified by some legitimate objectives that these rules aim to achieve. As regards various tax benefits – such as the preferential regime discussed here – substantial legislative leeway has been confirmed in the jurisprudence of the Constitutional Court, provided that such provisions are underpinned by reasonable justifications.³⁷ In other words, tax benefits lav outside of the scope of the constitutional review, as long as they are not arbitrary and unsubstantiated by legitimate policy reasons. Turning attention back to the 'young employees' scheme', it seems clear that while the choice of the age criterion, along with the exact thresholds prescribed in the law. may be derived from the policy perspective, it does not render the scheme entirely arbitrary and thus unconstitutional per se. Namely, the claims of the Government that the scheme serves useful social and economic objectives, particularly from the demographic viewpoint, will be probably adjudged by the Constitutional Court as rational considerations of a sufficient degree to pass the prospective constitutional review. A similar conclusion may be reached from the perspective of Article 51 of the Constitution. 38 Here the case-law of the Constitutional Court again confirms a wide margin of appreciation left to the legislator in deciding which facts are relevant in regulating the distribution of the tax burden between taxpayers: as long as a rational ground is provided for the tax classification or the differentiation enshrined in the law, corresponding at least partly to the abstract notion of justice, there is no arbitrariness leading to the conclusion that the constitutional principles of equality and equity are infringed.³⁹ As previously noted, the policy justifications provided in the Government's bill, amending the Personal Income Tax Act regarding the preferential treatment of young workers, are not to be easily dismissed from this perspective inherent to Article 51 of the Constitution.

Finally, there is a familiar concern that the introduction of the special scheme will add yet another layer of administrative complexity to the Croatian tax system, with the greatest burden falling on the small and medium enterprises, acting as employers. As noted in the above discussion on preferential schemes for HSWs (Section 3.2), there is no real way around this argument, since any deviation from the ordinary system of taxation brings forth additional administrative and compliance costs. The tough question for the policymakers then is whether these costs can be justified by the underlying policy objectives of the scheme at hand.

Decision of the Constitutional Court of the Republic of Croatia, U-IP/3820/2009,
 November 2009, para. 10.

³⁸ For a general discussion on the Article 51 of the Croatian Constitution, see Arbutina (2012, 1285–1296)

Decision of the Constitutional Court of the Republic of Croatia, U-IP/3820/2009,
 November 2009, para. 15.4.

To sum up, it is difficult to see how the proposed regime for young employees may be a success story. Its shortcomings from the policy perspective outnumber its potential gains and the Croatian legislator would be well-advised to opt for alternative instruments to tackle brain drain. Most importantly, these measures should form part of a coherent policy, as deliberated below (Section 4.2). On the other hand, one may have sympathy for the policymakers' desire to come up with a quick fix to a conundrum that reaches far outside the ambit of purely tax domain, or even the economic, for that matter. This is an important point to make, since it signals that there is finally an understanding within the governing elites that the time has come to take a more proactive position vis-à-vis migration flows.

4.2. A Look at the Future: The Basic Tenets of a Coherent nti-brain Drain Tax Policy

In conceptualizing potential solutions to the brain drain problem, and proposing a particular path of action to Croatian policymakers, one must have a good grip on the following facts informing the policy debate:

i. The socio-economic underpinnings of emigration and brain drain: In essence, every regulatory tax measure – such as those discussed in the present paper (see above, Section 3) - has to be based on a fundamental understanding of the situation that requires state intervention. In this regard, the main determinants, magnitude, and effects of the Croatian brain drain have been laid out in Section 2 above. It is evident that the problem mainly boils down to younger (i.e. under the age of 40) and highly-skilled people permanently leaving the country, particularly in the post-EU accession period. Emigration of people with tertiary education seems particularly worrying if one takes into account the education and training costs heavily subsidized from the public purse. For example, according to one estimate, 2 billion EUR has been spent by the government on educating people who have emigrated in the 2013–2017 period. 40 Furthermore, emigration concerns seem to be more pressing as regards certain professions, such as healthcare workers, IT experts, and other employees educated in the wider field of Science, Technology, Engineering and Mathematics (STEM). This mostly has to do with the fact that these professionals face lower barriers when entering labour markets in other countries and are especially coveted in those EU Member States that are traditional destinations for Croatian emigrants (e.g. Germany). Special attention has to be paid to the category of so-called knowledge workers. While there is no ubiquitous definition of this term, they generally include

⁴⁰ See https://prviplan.hr/aktualno/koliko-nas-kosta-odljev-mozgova/ (last visited November 1st 2019).

all individuals with "high degrees of expertise, education, or experience, and the primary purpose of their jobs involves the creation, distribution, or application of knowledge" (Davenport 2005, 10). Accordingly, the term encompasses scientists, researchers, engineers, programmers, and other workers whose knowledge is essential in bringing about innovation, having positive spillovers on the economy as a whole. In this respect, it has to be emphasized that Croatia lags behind other EU countries in terms of innovation, being labelled as a "moderate innovator", leaving behind only Bulgaria and Romania (European Commission 2019a, 7). Even more worryingly, Croatia is the worst performing Member State when it comes to exports of knowledge-intensive services and is well below the EU average regarding human resources, i.e. the availability of a high-skilled and educated workforce (European Commission 2019a, 53). The latter finding has been confirmed in the latest Global Competitiveness Report, noting that the skill set of graduates, the percentage of the active population apt in digital skills, and the ease of finding skilled employees all amount to weaknesses of the Croatian economy (World Economic Forum 2019, 176). 41 The orientation to improve the innovation climate in the near future times ahead has been acknowledged by the domestic policymakers, with an explicit reference to the role of relevant human resources (Government of the Republic of Croatia 2019a, 58-62). The attainment of this goal is undoubtedly under threat if the current trends in emigration of knowledge workers continue, with no real policy on how to possibly induce previous emigrants to return to the country.

ii. The fundamentals of the Croatian tax system: It must be acknowledged that any tax measure targeting brain drain has to fit within the domestic framework of taxing income. Accordingly, a deep understanding of the current rules regarding taxable income, computation of the tax base, tax rate schedule, tax exemptions and privileges, etc., is necessary to choose a particular path of action, if any. All of these elements determine the effective tax burden on labour, which is assumed to influence the individuals' location decision. In this respect, it must be noted that the tax wedge on low-wage earners in Croatia stands at 33.7% and is below the EU average (38.2%), but still much higher than in Ireland (24.2%) or United Kingdom (26.1%) (Eurostat 2019b). Even more importantly for the globally mobile knowledge workers, the top marginal tax rate, including statutory personal income tax rate and surcharges, stands at 42.5%, which is higher than the EU average (39.4%) and seems

⁴¹ Compare also the findings on Croatia in Cornell University, INSEAD, WIPO (2019).

⁴² It must be taken into account that this indicator measures the tax wedge for an individual without children, earning 67% of the average salary in the country's business sector.

particularly high compared to the top rates in the new Member States, such as Hungary, Poland, Romania, Latvia, and Slovakia (European Commission 2019b, 26). Put simply, while Croatia is definitely not a high-tax country⁴³, its ordinary system of taxing employment income is not competitive vis-à-vis high earners, with a significant portion of high-skilled workers falling in this category (e.g. IT experts).

iii. Comparative trends with regard to personal income tax and migration-related measures: In the international competition-based setting, domestic legislators must to pay attention to the global developments in order to improve their competitive position for mobile tax bases. Global trends show that a number of countries have introduced changes to their systems of taxing personal income by either cutting nominal rates or narrowing the tax base (OECD 2019). The latter point is particularly important, since base-narrowing reforms are usually underpinned by specific redistribution and regulatory objectives. This logic may also apply to tax measures targeting migrating knowledge workers, such as the preferential regimes discussed earlier in the paper (Section 3.2).

iv. The EU law dimension: Any prospective tax measure introduced at the domestic level should comply with EU law requirements. While the regulation of individual income tax still remains firmly in the hands of national legislators, with a severely limited role of the EU institutions, domestic rules should not run afoul of the primary EU law, i.e. the provisions enshrined in the EU treaties, such as the rights to free movement within the internal market.⁴⁴ In this respect, imposition of a departure tax regime (e.g. a strictu sensu exit tax) raises more concerns, since it directly impedes the free movement of labour. However, the jurisprudence of the European Court of Justice (ECJ) has confirmed that the imposition of a tax on the act of individual's emigration may be a justified restriction, provided that the design of the scheme passes the proportionality test. 45 On the other hand, the compatibility of preferential tax regimes granted to mobile individuals with EU law, even if objectionable from the internal market perspective, has hitherto not been under serious scrutiny and thus remains a domaine réservé of the individual Member States.46

⁴³ According to the latest data compiled by the European Commission (2019b, 15), Croatia ranks 13th in the EU with regard to the overall tax burden (including social security contributions), relative to the domestic GDP. What is particularly striking is that the only new Member State with a higher tax burden than Croatia is Hungary.

For a general overview see Beretta (2018, 40–43).

Further analysis of the ECJ's case-law remains outside of the scope of this paper. See Terra, Wattel (2012, 957–962).

Compare the discussion in Beretta (2018, 47–51).

Against this backdrop, the remainder of this section presents the main policy recommendations reflecting the views of the author on the particular case of Croatia:

- i. Introduction of defensive measures, taking the form of either an exit tax *strictu sensu* or a 'trailing tax', is not advisable, since the normative disadvantages of such tax schemes outweigh the potential behavioural impacts and other potential justifications (e.g. the benefits principle). Most importantly, the effect of the exit tax regime on the emigration decision would be greatly diminished due to EU law limitations regarding its exact design, rendering the whole exercise an unnecessary waste of tax authorities' and taxpayers' resources.
- ii. Conversely, there is some merit to introducing special provisions on 'recapture' or 'claw-back' regarding previously enjoyed tax benefits. This point is particularly important if policymakers decide to employ preferential regimes for young and/or highly-skilled individuals. As seen from the preceding discussion in this section, one such scheme targeting 'young employees' is currently in the legislative procedure in Croatia and is expected to come into force in 2020. Since one of the main objections to this scheme is that it fails to address the migration decision of the beneficiaries once they turn 31 years of age, the legislator may decide to back up the scheme with a protective claw-back provision. For example, one could prescribe that the beneficiaries of the scheme who decide to emigrate must repay the amount of benefit previously enjoyed. It has to be noted that in drawing up such a rule, the EU law requirements – particularly ECJ's case law on exit taxation – must be taken into account. Accordingly, recapture should not in all likeliness entail an order for an intra-EU migrant to immediately repay the full amount of benefits received.
- iii. Whether Croatia needs a special, preferential regime for highly-skilled immigrants, including returning expatriates, remains the most difficult question to answer here. While the author believes that other, general tax measures aimed at strengthening Croatian tax competitiveness are more desirable from a purely theoretical perspective, one has to acknowledge that the current state of affairs regarding demography and migration may call for some quicker, even if less than perfect solutions. In the author's opinion, a well-designed preferential scheme for immigrants and returnee citizens has in any case numerous advantages over the proposed 'young employees' scheme. As stated by Brauner (2010, 266), "(...) The basic idea here is to design tax incentives that will encourage behavior beneficial for development in the context of the brain drain. Tax incentives can and should target specific, well-defined, and isolated

behavior." Indeed, if one connects this argument with the Croatian specificities laid out above, it seems that the prospective preferential tax scheme should have a very limited subjective scope. Namely, against the backdrop of the preceding discussion on the importance of innovations for the economic development of the country and the role of knowledge workers in this regard, preferential regime should target only those professionals possessing crucial skills and expertise in increasing the overall level of innovation. Moreover, the situation in the domestic labour market should play a part in defining the targeted population and making tax-relevant classifications. Some comparative practices may lend a hand in this respect, e.g. the Korean scheme targeted exclusively employees in the high-tech sector (OECD 2011, 144). Regarding the objective scope of the preferential scheme, it would be desirable to extend the benefit to both employment and self-employment income earned in Croatia postimmigration, including foreign-sourced capital income that Croatia may have the right to tax, to further increase the attractiveness of the programme. In any case, preferential tax treatment should be temporary, with a fiveyear limitation seeming reasonable. This is important to mitigate the impact of the scheme on the overall equity of the tax system. As already pointed out above, any new preferential regime should be accompanied by a claw-back rule vis-à-vis the benefits granted to the migrant.⁴⁷

iv. If one stays within the ambit of personal income tax, there are definitely other options, with a more general scope of application, that seem viable in alleviating the migration-related pressures. In this respect it appears that Croatian policymakers are aware of the global trends and recent years have seen a gradual decrease of the effective tax burden on employment income. However, the tax wedge still remains comparably high, particularly for highly-skilled and highly-mobile individuals earning above-average salaries. Accordingly, a reform of the tax rate schedule should be considered in the mid-term. Furthermore, the system of taxing fringe benefits has also been recently reformed, with an increase of taxexempt amounts. 48 It still, however, remains relatively rigid in comparison with other EU Member States. One legislative change in the right direction, albeit with a limited scope, has come into force on 1 January 2019, with the introduction of a new, more tax-efficient way for companies to provide stock options to their employees. 49 Since stock option models are a standard way of remunerating workers in the most innovative business sectors, e.g. the IT industry, this change may be adjudged as well-targeted for the category of workers generally highly responsive to cross-country wage differentials.⁵⁰

⁴⁷ For an Italian example regarding the claw-back rule see Beretta (2018, 18).

⁴⁸ For an overview see Pezo (2019).

⁴⁹ See Božina, Wagner (2019, 31).

⁵⁰ Compare also the discussion in OECD (2011, 142).

v. Furthermore, it must be acknowledged that there are other areas of the tax system with at least indirect effects on migration decisions of the HSWs, mainly by creating a more competitive tax environment for enterprises engaged in the knowledge-based economy. In this respect is has to be noted that one of the measures envisaged in the Government's tax package, currently in the legislative pipeline and expected to come into force next year, relates to the extension of a lower corporate income tax rate (12%) to all enterprises with an annual turnover up to 1 million EUR. It is thus expected that more than 90% of small corporate taxpavers pay this rather competitive tax rate, which also provided an additional stimulus for the development of the knowledge-intensive start-up sector. While the recent changes in the Croatian tax system are aligned with the global trends of steadily reducing tax burdens on labour and capital, there are certainly some specific instruments the legislator may additionally employ in order to improve the investment climate in innovation-based industries. A number of countries have recently expanded the use of R&D tax incentives (OECD 2019). In this respect an interesting example relevant for knowledge workers comes from Italy, which offers tax credits for corporate taxpavers related to the costs of employee training on 'Industry 4.0' topics. Taken at face value, this seems like a good example of how to simultaneously provide tax benefits to enterprises engaged in the new economy and incentivize improvements of the domestic workforce's relevant skills. One can also contemplate whether Croatia needs a 'patent box' or 'intellectual property box scheme', i.e. a preferential corporate tax regime offered to companies engaged in the development of relevant intangible assets.

vi. Finally, one has to understand that taxation has a severely limited role regarding brain drain. A number of studies have shown that young, highly-skilled people emigrate from Croatia driven by various non-economic factors, including, inter alia, the perception of corruption within the society (see Section 2 above). This necessarily calls for a further strengthening of the overall institutional framework in the country. In this respect the tax system – however complex it may seem – is only a small piece of a complex socio-economic mosaic that must aim to improve the overall well-being of the population. For example, if one focuses again on the young and highly-skilled, it is pretty evident that reforms in the education system – more aligned with the conditions in the domestic labour market - may have a more far-reaching impact on the reversal of migration patterns in the long run than piecemeal state interventions regarding net disposable income. Perhaps even more importantly, if Croatia wants to not only prevent further outflow of its top talent, but also to reap the benefits of the 'brain circulation' concept, a coherent set of measures aimed at returnee expatriates and their reintegration into society has to be devised (Vračić 2018, 11–14).

5. CONCLUSION

At the end of the Twenty-Tens, Croatian society is at a crossroads: the governing elites have to decide whether to resign themselves to the role of passive onlookers of the current adverse demographic and migration trends – threatening to tear apart the very socio-economic fabric of the country in the long run – or to adopt a more proactive approach and formulate a set of appropriate policy responses, ranging across different pressure areas. The analysis in this paper shows that, unfortunately, the role of taxes and tax policy in this regard has not been seriously acknowledged hitherto, even if many other jurisdictions have reacted to the global migration of individuals by introducing special tax rules at the domestic level. Admittedly, one may note that a turning point was reached in the autumn of 2019, i.e. at the time of writing this paper, when the proposal to introduce a preferential tax scheme for 'young workers' was brought to the legislative process.

Against this background, the present paper tried to lay out potential tax policy responses to the actual brain drain situation in Croatia, with particular emphasis on highly-skilled workers (or knowledge workers). Its main contribution lies in identifying the main building blocks of a coherent anti-brain drain tax policy, on the basis of both cross-country experiences and relevant economic and legal principles of taxation. What emerges from the paper is that while targeted measures, such as a preferential regime for immigrants and/or returnees, may be problematic from a theoretical standpoint – with horizontal measures providing a better long-term alternative – the magnitude and the structure of emigration from Croatia may warrant some sort of a quick-fix solution. Accordingly, it has been suggested that policymakers may opt for the introduction of a preferential scheme for HSWs, with the main design conundrum being how to draw up proper targeting rules. In any case, the author shares the view that such a migration-related tax instrument has numerous advantages over the proposed 'young workers' scheme, mainly due to the extremely uncertain behavioural effects of the latter. Hope remains that the future debate will refine this or some alternative, and possibly more appropriate, policy approach.

REFERENCES

Arbutina, Hrvoje. 2012. Ustavna i nadnacionalna ograničenja ovlaštenja na oporezivanje. *Zbornik Pravnog fakulteta u Zagrebu* 62: 1283–1322.

- Arsenault, Steven J. 2009. Surviving a Heart Attack: Expatriation and the Tax Policy Implications of the New Exit Tax. *Akron Tax Journal* 24: 37–67.
- Avi-Yonah, Reuven S. 2006. The Three Goals of Taxation, *Tax Law Review* 60: 1–28.
- Bagić, Snježana. 2016. *Načelo razmjernosti u praksi europskih sudova i hrvatskog ustavnog suda (s posebnim osvrtom na vlasništvo)*. Zagreb: Pravni fakultet Sveučilišta u Zagrebu.
- Beretta, Giorgio. 2018. Mobility of Individuals and Taxation, *IFA Research Paper 2017/2018*.
- Božina, Anja, Anamarija Wagner. 2019. Novosti u obračunu doprinosa i poreza na dohodak u 2019. godini. *Pravo i porezi* 21: 30–37.
- Brauner, Yariv. 2010. Brain Drain Taxation as Development Policy. St. Louis University Law Journal 55: 221–268.
- Brock, Gillian. 2015. What Does Global, Justice Require?. in *Debating Brain Drain. May Governments Restrict Emigration?*, edited by Gillian Brock, Michael Blake. New York: Oxford University Press.
- Casarico, Alessandra, Silke Übelmesser. 2018. Migration Policies and the Highly Skilled. *ifo DICE Report* 16: 28–33.
- Chand, Vikram. 2013. Exit Charges for Migrating Individuals and Companies: Comparative and Tax Treaty Analysis. *Bulletin for International Taxation* 67.
- Cornell University, INSEAD, WIPO. 2019. The Global Innovation Index 2019: Creating Healthy Lives The Future of Medical Innovation. Ithaca Fontainebleau Geneva.
- Croatian Bureau of Statistics. 2019a. Migration of Population of Republic of Croatia, 2018. https://www.dzs.hr/Hrv_Eng/publication/2019/07-01-02_01_2019.htm (last visited 1 November 2019).
- Croatian Bureau of Statistics. 2019b. Natural change in population in the Republic of Croatia. https://www.dzs.hr/Hrv_Eng/publication/2019/07-01-01_01_2019.htm (last visited 1 November 2019).
- Davenport, Thomas H. 2005. Thinking for a Living: How to Get Better Performance and Results from Knowledge Workers. Boston: Harvard Business School Press.
- De Broe, Luc. 2002. General Report. in *The Tax Treatment of Transfer of Residence by Individuals*. IFA Cahiers 2002 87B.
- Del Carpio, Ximena V., Caglar Ozden, Mauro Testaverde, Mathis Wagner. 2016. Global migration of talent and tax incentives: Evidence from Malaysia's returning expert program. Washington, D.C.: World Bank Group, Development Research Group, Trade and International Integration Team.

- Dietsch, Peter. 2015. *Catching Capital: The Ethics of Tax Competition*. Oxford: Oxford University Press.
- Draženović, Ivana, Marina Kunovac, Dominik Pripužić. 2018. Dynamics and determinants of emigration: The case of Croatia and the experience of new EU member states. *Public Sector Economics* 42: 415–447.
- European Commission. 2019a. *European Innovation Scoreboard 2019*. Luxembourg: Publications Office of the European Union.
- European Commission. 2019b. *Taxation Trends in the European Union:*Data for the EU Member States and Norway. Luxembourg: Publications Office of the European Union.
- European Committee of the Regions. 2018. Addressing brain drain: the local and regional dimension. https://cor.europa.eu/en/engage/stu dies/Documents/ex-sum-addressing-brain-drain/ex-sum-addressing -brain-drain-EN.pdf (last visited 30 October 2019).
- Eurostat. 2019a. *Ageing Europe 2019 edition*. Luxembourg: Publications Office of the European Union.
- Eurostat. 2019b. Wages and labour costs. https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Wages_and_labour_costs#Net earnings and tax burden (last visited 2 November 2019).
- Faulhaber, Lilian V. 2018. The Trouble with Tax Competition: From Practice to Theory. *Tax Law Review* 71: 311–366.
- Gadžo, Stjepan. 2018. Nexus Requirements for Taxation of Non-Residents' Business Income: A Normative Evaluation in the Context of the Global Economy. Amsterdam: IBFD.
- Gelo, Jakov, Akrap, Anđelko, Čipin, Ivan. 2005. *Temeljne značajke demografskog razvoja Hrvatske Bilanca 20. stoljeća*. Zagreb: Ministry of Family Affairs, War Veterans and Intergenerational Solidarity.
- Government of the Republic of Croatia. 2019a. National Reform Programme 2019. https://vlada.gov.hr/UserDocsImages/Vijesti/2019/04%20Travanj/30%20travnja/NRP%202019%20i%20Odluka%20Vlade/NRP%202019%20english/National%20reform%20programme%202019.pdf (last visited 1 November 2019).
- Government of the Republic of Croatia. 2019b. Convergence programme of the Republic of Croatia for the period 2019 2022. http://www.mfin.hr/adminmax/docs/Convergence%20programme%20of%20the%20Republic%20of%20Croatia%20for%20the%20period%202019%20-%202022.pdf (last visited 1 November 2019).
- Gribnau, Hans. 2003. Equality, Consistency, and Impartiality in Tax Legislation. Legal Protection against Discriminatory Tax Legislation: The Struggle for Equality in European Tax Law, edited by Hans Gribnau. The Hague: Kluwer Law International.

- Jurić, Tado. 2017. Suvremeno iseljavanje Hrvata u Njemačku: Karakteristike i motivi. *Migracijske i etničke teme* 33: 337–371.
- Jurun, Elza, Nada Ratković, Ivana Ujević. 2017. A cluster analysis of Croatian counties as the base for an active demographic policy. *Croatian Operational Research Review* 8: 221–236.
- Kleven, Henrik, Camille Landais, Mathilde Munoz, Stefanie Stantcheva. 2019. Taxation and Migration: Evidence and Policy Implications. NBER Working Paper No. w25740. https://ssrn.com/abstract=3372036 (last visited 25 October 2019).
- Knezović, Sandro, Maja Grošinić. 2017. *Migration Trends in Croatia*. Zagreb: Hans Seidel Stiftung.
- Li, Jinyan. 2009. International Mobility of Highly Skilled Workers in the Canadian Context: Tax Barriers and Reform Options. Commissioned Reports, Studies and Public Policy Documents. Paper 55.
- Lister, Matthew J. 2017. A Tax-Credit Approach to Addressing Brain Drain (2017). Saint Louis University Law Journal 62: 73–84.
- Muñoz, Mathilde. 2019. Do European top earners react to labor taxation through mobility? WID.world Working Paper 2019/12. https://wid.world/wp-content/uploads/2019/09/WID_WORKING_PAPER_2019_12_MigrationMobility.pdf (last visited 20 October 2019).
- OECD. 2011. Taxation and Employment, OECD Tax Policy Studies, No. 21. Paris: OECD Publishing. https://doi.org/10.1787/9789264120808 -en (last visited 2 November 2019).
- OECD. 2019. Tax Policy Reforms 2019: OECD and Selected Partner Economies. Paris: OECD Publishing.
- Omejec, Jasna 2009. Zabrana diskriminacije u praksi Europskog suda za ljudska prava. *Zbornik Pravnog fakulteta u Zagrebu* 59: 873–979.
- Pezo, Sandra. 2019. Topli obrok, naknada za smještaj i ostali (ne)oporezivi primitci radnika. *RRIF* 22: 11–16.
- Terra, Ben, Peter J. Wattel. 2012. *European Tax Law*. Alphen aan den Rijn: Kluwer Law International.
- United Nations, Department of Economic and Social Affairs, Population Division. 2017. *International Migration Policies: Data Booklet* (ST/ESA/ SER.A/395).
- United Nations, Department of Economic and Social Affairs, Population Division. 2019. *World Population Prospects 2019: Highlights* (ST/ESA/SER.A/423).
- Vračić, Alida. 2018. The Way Back: Brain Drain and Prosperity in the Western Balkans. https://www.ecfr.eu/publications/summary/the_way_back_brain_drain_and_prosperiy_in_the_western_balkans (last visited 20 October 2019).

- Wong, Kar Yiu. 2009. Brain Drain. in *The Princeton Encyclopedia of the World Economy*, edited by Kenneth A. Reinert, Ramkishen S. Rajan. Princeton: Princeton University Press.
- World Economic Forum. 2019. *The Global Competitiveness Report 2019*. Geneva.
- Župarić-Iljić, Drago. 2016. *Iseljavanje iz Republike Hrvatske nakon ulaska u Europsku uniju*. Zagreb: Friedrich-Ebert-Stiftung.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(4-672EU)

CERIF: S141

DOI: 10.5937/AnaliPFB1904147G

Ricardo García Antón*

RESTORING THE EU CITIZENSHIP FOR TAX PURPOSES

Citizenship-based taxation has become insignificant due to high mobility of individuals, which are completely detached from their polities. The lack of political, social and cultural bonds of the individual with the states has shifted the focus to the concept of tax residence. This contribution sheds light on the concept of supranational citizenship in clear opposition to the nation-state citizenship, for the purposes of legitimizing levying a tax on EU citizens. The ongoing concept of EU citizenship anchored firstly in the principle of mutual recognition and secondly in the emergence of democratic and pluralist values under the so-called "European way of life," yields certain imbalances and asymmetries derived from a steep distinction between economically active and economically inactive EU citizens. In the author's view, levying a tax upon EU citizens would enhance the demos and solidarity within the current withered EU integration project.

Key words: Citizenship. – EU law. – Residence-taxation.

1. INTRODUCTION

In the globalized economy still suffering the post-traumatic effects of the financial crisis, the European Union free movement of persons has allowed individuals to benefit from better job opportunities in different EU Member States. The migration of highly skilled workers from Southern Member States to Northern EU Members looking for a better life is now a common reality in the EU polity.

Such mobility of individuals benefiting from the EU freedom of circulation of persons has provoked a spillover effect in the EU countries.

 $^{^{\}ast}$ Assistant Professor, Tilburg University, School of Economics and Management, r.garciaanton@uvt.nl.

On one side, the Member States create incentives to lure individuals and on the other side, they create incentives to "recapture" those who have already left. The EU mobility of workers mirrors the mobility of capital and urge us to rethink in the nexus to allocate taxing rights. Whilst in the past, taxes were physically constrained to the boundaries of the State, the base erosion and profit shifting (BEPS) landscape completely turned the picture upside-down, thereby presenting multinationals the opportunity to cherry pick the most convenient tax regime to channel their investments. The same dynamics concern individuals who are no longer constrained by the physical boundaries of the State to pay their taxes.

Such cross-border mobility and cherry picking has challenged the concept of citizenship, which is traditionally anchored in the strong nexus between the individual and nation-state. Citizenship strongly pleads for membership, for a common status, namely for being accepted and engaged as fully-fledged member of a polity. In the democratic founding of the modern State, the citizens were identified as the taxpavers. Those who are member of the polity are the exclusive ones deciding the levies and taxes to support the public expenditure and to benefit from the taxes collected. Nevertheless, as Schön points out, taxation and representation undergoes serious conflicts between "those who vote on the tax, those who pay the tax, and those who enjoy the spending of the tax" (Schön 2018).² Several questions exemplify such conflicts: Why do the States implement redistribution polices to the detriment of certain taxpayers? Is it necessary to finance public goods if the taxpayer is not interested? How does the State defend the taxpayer against excessive or "expropriation" taxation? How does the well-known "race-to-the-bottom", fostered by the States to lure individuals within a tax competition environment, threaten the redistribution policies of the State in favor of poor citizens?

In the field of taxation, the concept of citizenship linked to the nation-state is in clear decay. Citizenship has been largely replaced by the concept of tax residence (Schön 2018, 41; Beretta 2019, 227–260). Only the US and Eritrea still apply citizenship-based taxation. The consolidation of the tax residence to the detriment of citizenship shakes the groundings of the democratic binomial taxation and representation: "It starts from the fact that citizens living abroad are by and large free of tax burdens in their home country but can retain voting rights while resident foreigners have to pay taxes on their worldwide income without enjoying formal participation in the political process.[...] Should voting rights be made

¹ See regimes for High-Net-Worth Individual regime in Italy, the "Sunny Welcome" for EU pensioners in Portugal, the 30% Dutch tax ruling, *rientro dei cervelli* in Italy etc. Beretta, Giorgio (2018) offers a good overview of these policies. The impact of these domestic policies in a EU competition environment should not be disregarded, as noted by Schön, Wolfgang (2003).

² On the democracy and taxation conflicts, see Schön, Wolfgang (2018).

dependent on being subject to domestic taxation? Why are foreign resident taxable at all? Should their personal liability to tax be complemented by voting rights or at least by a constitutional principle of non-discrimination vis-à-vis taxpaying citizens?" (Schön 2018, 41). While citizenship is irrelevant in the domestic context, this contribution wonders whether the concept of citizenship, and particularly restoring the binomial citizentaxpayer, can still play a decisive role in the current EU integration process.

Aside from the EU own resources (VAT and customs duties), the national contributions from the Member States are the largest source for the EU budget.³ Since the EU lacks of a direct tax on the EU citizens, this contribution poses the following research question: is the "EU citizenship" concept resilient enough to support the levy of an EU tax? Why is a tax needed and not direct contributions from the Member States? Section 2 of this contribution sketches the features of the concept of supranational citizenship in antagonism towards the "nationalistic citizenship". Such a concept of supranational citizenship serves us as a benchmark to measure the current development of the EU citizenship in Section 3. In Section 4, the author supports the claim that levying a tax on EU citizens is needed to enforce not only the democratic channels but also the solidarity principle within the EU polity. Section 5 provides a conclusion.

2. SUPRANATIONAL CITIZENSHIP VERSUS NATIONALISTIC CITIZENSHIP

Despite the disputes and controversies on the content and meaning of citizenship – which go beyond the scope of this contribution – the extensive literature on political theory dealing with the citizenship traditionally boils it down to the relationship between the individual and a *locus* of politics (Dobson 2006, 20; Bauböck 2006 19; Clarke *et al.* 2014, 10). Such a sense of belonging between the individual and the political community comprises rights and duties within the borders of the nation state (Dobson 2006, 21). Citizenship imbued within a nationalistic spirit reflects sentiments of attachment and common identity to a particular ethnic, political or historic group, but at the same time, it has unfortunately fed the politics of exclusion against the non-citizens (Kochenov 2019).

Under this narrative of citizenship, constrained within the boundaries of the nation-state, Beretta conceives citizenship as a jurisdictional tax nexus (Beretta 2019). The fact that there is a genuine or sufficient link between the individual and its community, namely the

³ See revenue figures in https://ec.europa.eu/budget/graphs/revenue_expediture. html (last visited 26 November 2019).

State, justified the emergence of the well-known "benefit principle" and the "ability to pay principle": "taxes are nothing less than the price that individuals must pay for the various benefits that they receive from the state. Alternatively, citizenship as a criterion for a state to impose its jurisdiction to tax can be premised on 'ability-to-pay theory' considerations, suggesting that individuals are bound to pay taxes, as members of a polity, according to a criterion of distributional equity" (Beretta 2019, Section 2, online version). Nowadays, citizenship-based taxation, which is only visible now in the US and Eritrea, has given way to residence-based tax systems. However, in the latter systems, Beretta stills identifies "citizenship footprints", for example in provisions that extend the state's taxing rights over citizens transferring their residence to low-tax jurisdictions or tax havens (Beretta 2019, Section 4, online version), or in the nationality test in tie-breaker rule in article 4 (2) of the OECD MC to determine the tax residence under a treaty. Beretta argues in favor of disentangling citizenship from playing any role in tax matters, and therefore eliminating such "footprints" in allocating taxing rights. The rationale supporting this claim is derived from the current cross-border mobility of individuals, which weakens the sentiment of belonging or membership to a particular community. The use of citizenship by the States becomes simply instrumental in obtaining more revenue, thereby extending their taxing rights over individuals who are no longer active member of the polity.

The cross-border mobility of individuals, together with the technological advancements, not only must deprive citizenship from any tax meaning, as Beretta previously defends, but also the tax residence concept itself has been recently challenged by Kostic as a nexus to allocate taxing rights under article 15 OECD/UN MC (Kostic 2019). The fact that work can be easily exercised "from any place that allows an internet connection" triggers the decay of the current understanding of how employment is exercised and the categories of employer/employee (Kostic 2019, Section 4 Online version). Accordingly, international tax rules must provide for solutions to the so-called *digital nomads* wherein there is no longer a deep personal link with a certain country (permanent home, family).

The above-mentioned recent diagnosis by tax scholars enhances the mobility of individuals as the rationale to get rid of "old categories", such as citizenship or tax residence. However, both proposals are trapped within the borders of the nation-state. Indeed, the cross-border mobility of individuals has revealed the lack of effective political participation of the individual with the state coupled with the lack of historical, cultural or ethnical bonds to such particular polity. At the outset of the 21st century, however, is the individual only member of the state as a polity? Is it

possible to build up a different meaning of "membership" of the individual with a broader polity than the state? The extensive literature in political theory and philosophy has quite profusely put forward new theories of citizenship deprived from the nationalistic spirit (Bellamy 2008; Christodoulidis 1998). A first attempt to overcome the nation-state boundaries can be found in the cosmopolitan understanding of citizenship, enshrined in the works of philosophers such as Nussbaum and Linklater (Nussbaum 1996; Linklater 1999). Whilst in Nussbaum, cosmopolitan citizenship transpires an "allegiance to a moral community made by the humanity of all human beings" (Nussbaum 1996, 5), in Linklater's Habermasian view, cosmopolitan citizenship aims to create "universal frameworks of communication" (Linklater 1999, 37), in which the excluded, vulnerable and dispossessed can find channels to participate and contest in global governance.

Nevertheless, cosmopolitan citizenship faces up severe criticisms as Dobson convincingly points out. Rather than territorial boundaries, the cosmopolitan citizenship still relies on an "extensive membership" beyond the boundaries of the nation-state to cover the inhabitants of the planet as a whole. Such extensive membership lacks of institutional boundaries, which are needed to deliver political input (information, taxation, etc.) and political output (laws, policies, allocation of tasks and resources) (Dobson 2006, 37). In other words, the world – under the cosmopolitan perception – is too big to be singled out as a "polity" in which the citizen participates in the common life of the community. Cosmopolitan citizenship detached from a legal and administrative institutional system simply becomes "a universal ethic", a sort of "generalized disposition to benevolence exercised within discursive civil communities: a mode of sociability" (Dobson 2006, 38).

In rejecting the cosmopolitan citizenship postulates, Dobson inevitably attaches citizenship to a political institutionalization process. Her concept of supranational citizenship conveys the self-definition of the individual in a complex political order. Supranational citizenship refers to a complex set of institutions consisting of organizational bodies, roles and rules in which the individuals have political rights to interact with each other. In this sense, citizenship can no longer be understood as a kind of personal identity derived from membership of an already-existing social group which gives you an privileged status (i.e. exclusive access to

⁴ As noted by Dobson (2006, 170),: "[...] conception of supranational citizenship as the institutional embodiment of the active and collective agency of reasonable composite selves in a community of rights, shaping their common and separate destinies under conditions of political equality and mutual recognition and respect. Whatever its territorial scope, insofar as that citizenship consists in effective powers and constitutes a political order conducing to the wellbeing and freedom of individuals, it authorises and justifies the framework of political authority."

a range of "club goods") compared to non-members (Dobson 2006, 44). Supranational citizenship represents a community of rights providing individuals the capacities to shape the context of their lives and promote the freedom and well-being of others. Supranational citizenship requires the existence of relational bonds among the individuals of a polity beyond the nation-state borders, regardless their own identity. In other words, supranational citizenship becomes a status and its substance (activities, tasks, purposes, dispositions, rights, and duties) is derived from the relationship with other individuals within a complex set of institutions forming a polity beyond the nation state (Dobson 2006, 43).

Strumia also reaffirms the political dimension of supranational citizenship, thereby abandoning any reference to an exclusive identity or ethno-cultural affinity (Strumia 2017). She sketches three prongs of the concept of supranational citizenship: "projection of citizenship beyond the state in the context of a non-hegemonic project; articulation of this beyond-state citizenship within the boundaries of a supranational entity pursuing a collective purpose; and reconfiguration of citizenship beyond nationality through a dynamic of mutual recognition of national citizenships" (Strumia 2017, 672). In principle, the EU regional integration project corresponds to the above-mentioned prongs. First, it is a nonhegemonic project under the constitutional pluralism premises;⁵ second, it pursues shared collective values and political goals (articles 1–3 Treaty of the European Union, TEU); third, it relies on mutual recognition, which means that every Member State recognizes national citizens of other Member States to some extent as its own (the EU freedoms of movements and the non-discrimination principle).

One may wonder whether the EU citizenship has achieved the three prongs associated with supranational citizenship. Since EU citizenship is an ongoing project, its content is still forming. In the next section, in dealing with the evolution, meaning and challenges of EU citizenship, we will be confronted with contradictions and asymmetries in relation to the second and third prongs proposed by Strumia.

3. THE JANUS-FACED EU CITIZENSHIP

The introduction of the status of EU citizen in the Treaty of Maastricht (Article 20 of the Treaty of Functioning of the European Union, TFEU) represents a key milestone in the progressive abandonment of the conception of EU citizens as mere market-citizens – as dubbed by Ros (Ros 2018; Ros 2017) – who use the EU freedoms of circulation to

⁵ On a detailed account of the Constitutional pluralism within the EU level, see the seminal works by Avbelj and Komarek (Avbelj, Komárek 2012).

carry on economic activity. Pursuant to Article 20 and Article 21 of the TFEU, the citizens of the Union have the right to move and reside freely within the territory of the Member States and cannot be discriminated on the grounds of nationality.

Together with the right to move and reside freely, Article 21 codifies the following rights: "[...] b) the right to vote and to stand as candidates in elections to the European Parliament and in municipal elections in their Member State of residence, under the same conditions as nationals of that State; c) the right to enjoy, in the territory of a third country in which the Member State of which they are nationals is not represented, the protection of the diplomatic and consular authorities of any Member State on the same conditions as the nationals of that State; (d) the right to petition the European Parliament, to apply to the European Ombudsman, and to address the institutions and advisory bodies of the Union in any of the Treaty languages and to obtain a reply in the same language."

However, the enshrinement of EU citizenship in TFEU Article 21 - "Every person holding the nationality of a Member State shall be a citizen of the Union. Citizenship of the Union shall be additional to and not replace national citizenship" - still triggers interpretative doubts in the overlapping with the concept of national citizenship. What is the normative content derived from being a European citizen? What is the additional status of being an EU citizen in the overlapping with the domestic nationality? Such normative content is clearly linked to the benefits that either an EU national or third country national can obtain from an EU host country (i.e. social security, residence permit, rejection of expulsion regime in case of criminal cases, etc.). The extent to which these benefits can be granted by the host Member States has experienced an interesting evolution in the case law of the Court of Justice of the EU (CJEU), which has been codified in the Directive 2004/38.6 In our benchmark of three prongs of supranational citizenship, described by Strumia (Strumia 2017), we can identify the enhancement of the principle of mutual recognition (Section 3.1), on one hand, and the progressive introduction of collective goals and values forming the European way of life (Section 3.2), on the other.

⁶ Directive 2004/38/EC of the European Parliament and of the Council of 29 April 2004 on the right of citizens of the Union and their family members to move and reside freely within the territory of the Member States, amending Regulation (EEC) No. 1612/68 and repealing Directives 64/221/EEC, 68/360/EEC, 72/194/EEC, 73/148/EEC, 75/34/EEC, 75/35/EEC, 90/364/EEC, 90/365/EEC, and 93/96/EEC (Text with EEA relevance).

3.1. European Citizenship as Mutual Recognition

3.1.1. Prohibition of Social Tourism: no Economic Burden for the Member States

The Member States have always been reluctant to extend the social assistance benefits to non-nationals. The idea behind was to prevent the migration of individuals to gain access to more favorable social benefits in the host country, under so-called social tourism. Accordingly, the former Directive 90/364 EEC granted the right of residence to nationals of the Member States and family members, provided that they themselves and the members of their families were covered by sickness insurance, in regard to all risks in the host Member State, and had sufficient resources to avoid becoming a burden on the social assistance system of the host Member State during their period of residence.⁷

In Baumbast, the question posed was whether a German citizen who no longer enjoyed a right of residence as a migrant worker in the host Member State (UK) can enjoy a right of residence by direct application of current Article 20 of the TFEU as a citizen of the European Union. The Court ruled that "the Union citizenship is destined to be the fundamental status of nationals of the Member States." The residence in the UK was granted on the following arguments: (1) Mr. Baumbast first, worked and lawfully resided in the host Member State (UK); (2) second, during that period his family also resided in the UK and remained there even after his activities as an employed and self-employed person in the UK came to an end; and (3) he had sufficient economic resources (comprehensive sickness insurance In Germany) so he and his family were no burdens for the public finances of the UK. The requirement of having sufficient resources has been relaxed in successive case law.

In Zhu & Chen, ¹¹ the UK authorities denied the residence permit to a Chinese national and her daughter, who had acquired the Irish nationality because of being born on the island of Ireland. The UK argued that the condition concerning the availability of sufficient resources means that the person concerned (i.e. the daughter) possesses those resources personally and may not use for that purpose those of an accompanying family member. Contrary to this interpretation, the Court held that the minor was covered by the appropriate sickness insurance of the parent who is a third-country national. Therefore, the minor did not become a burden on the public finances of the host Member State (UK), and a right

⁷ See this wording in the current article 7 (1) (b) of Directive 2004/38.

⁸ CJEU, Case C-413/99, Baumbast, ECLI:EU:C:2002:493.

⁹ Baumbast, para. 82.

¹⁰ Baumbast, para. 92.

¹¹ CJEU, Case C-200/02, Zhu & Chen, ECLI:EU:C:2004:639.

to reside for an indefinite period in that State has to be granted. In recent Bajratari case, ¹² the CJEU went a step further to argue that Article 7 (1) (b) of the Directive 2004/38 did not require that the sufficient resources were obtained legally. Such requirement related to the origin of the resources would be disproportionate. ¹³ The fact that the income obtained was derived from the unlawful employment of his father (a third-country national without a residence card or work permit) was sufficient for not being a burden for the Member State. These Court's findings are quite responsive to the difficult situations immigrants face in the host state, usually working without a proper work permit (Haag 2019).

While in the previous cases there were sufficient resources, the question becomes troublesome in cases wherein the EU national does not have sufficient resources to avoid becoming a burden on the social assistance system of the host Member State. This scenario is carved out from the scope of Article 24 (2) of Directive 2004/38/EC, which does not oblige the Member States to grant social assistance benefits to economically inactive citizens. Hence, it directly requires an interpretation of TFEU articles 20 and 21 coupled with TFEU Article 18, which enforces the principle of equal treatment and, eventually, the non-discrimination principle laid down in Article 20 of the EU Charter of Fundamental Rights.

In Martinez Sierra, ¹⁴ Germany's authorities denied a child-raising allowance to a Spanish national who had been lawfully living in Germany since 1984, without interruptions. The Court ruled that "A national of another Member State who is authorised to reside in German territory and who does reside there meets this condition. In that regard, such a person is in the same position as a German national residing in German territory."¹⁵ The same rationale applies in Trojani, ¹⁶ in which a French national residing in Belgium without having sufficient resources could not be excluded from the minimum subsistence allowance, since the principle of nondiscrimination requires equal access to the social benefits available only to nationals. The moment an EU national becomes a lawful resident of another Member States entitles them to ask for social benefits as if they were nationals of the host state, to meet the principle of non-discrimination. 17 Both cases strictly apply the principle of mutual recognition, thereby nationals of Member States should be treated equally to residents of the other Member States.

¹² CJEU, Case C-93/18, Bajratari, ECLI:EU:C:2019:809.

¹³ Bajratari, para. 42

¹⁴ CJEU, Case C-85/96, Martinez Sala, ECLI:EU:C:1998:217.

¹⁵ Martinez Sala, para. 49.

¹⁶ CJEU, Case C-456/02, Trojani, ECLI:EU:C:2004:488, para. 44.

¹⁷ See also 24 (1) of the Directive 2004/38.

The far-reaching consequences of enforcing the mutual recognition principle in Martinez Sierra and Trojani entailed a serious risk of becoming a burden on the social assistance system of the host Member State. Therein, the successive cases limited their scope. For example, the Dano case¹⁸ concerned the denial of social benefits to Ms. Dano and her son, who were Romanian nationals lawfully residing in Germany, but without intention of seeking employment. In the facts of the case, it was stated that Ms. Dano did not work in Germany or Romania, and "there is nothing to indicate that she has looked for a job". 19 While Article 24 (1) of the Directive 2004/38 consolidated the principle of non-discrimination on the grounds of nationality, Article 24 (2) did not oblige the Member States to confer equal treatment to receive social assistance to EU citizens who are not economically active (i.e. seeking employment). Since Ms. Dano was not economically active, Article 24 (1) of the Directive applies. To apply Article 24 (1) of the Directive, Ms. Dano should have had sufficient economic resources in light of Article 7(1)(b) of Directive 2004/38, thereby preventing becoming a burden on the social assistance system of the host Member State. ²⁰ Therefore, the Court concluded that Ms. Dano did not comply with Directive 2004/38.²¹ Economically inactive EU nationals can apply for social benefits in the host Member State on equal footing to nationals if they comply with the requirements of Directive 2004/38. In short, inactive economic citizens cannot claim social benefits on equal footing as nationals of the host state.²² In Alimanovic, the CJEU was quite explicit on asserting that although the assistance awarded to a single applicant "can be scarcely be described as an unreasonable burden for a Member State, [...] the accumulation of all the individual claims which would be submitted to it would be bound to do so". 23

This restrictive approach is confirmed in the denial of social assistance benefits to job-seekers from EU Member States in the host country. In Vatsouras & Koupatanize, ²⁴ a case concerning Greek nationals looking for jobs in Germany, the CJEU held that job-seekers must be compared to national job-seekers under Article 45 of the TFEU (freedom of circulation of workers) in term of the social assistance to be granted by the host State "only after it has been possible to establish a real link

¹⁸ CJEU, Case C-333/12, Dano, ECLI:EU:C:2014:2358.

¹⁹ Dano, para. 39.

²⁰ Dano, para. 64.

²¹ Dano, para, 66.

²² Eric Ros. supra 140, n. 27. This finding has been endorsed in cases like CJEU, Case C-67/14, Alimanovic, ECLI:EU:C:2015:597.

²³ Alimanovic, para. 62.

²⁴ CJEU, Joined cases C-22/08 and C-23/08, Vatsouras & Koupatanize, ECLI:EU:C:2008:201, para. 38.

between the job-seeker and the labour market of that State." In the event that the real link does not exist, due to short period of time looking for a job or the brief period working in the host state, Article 24 (2) of the Directive 2004/38 does not oblige the extension of social assistance benefits to non-national job-seekers.

For those who have economic resources, albeit obtained unlawfully without the work and residence permit, Bajratari does not oblige the Member States to evaluate how those resources were acquired. For those who do not have economic resources (Dano and the successive line of cases), the Court narrowed down the scope of the mutual recognition principle derived from Martinez Sierra and Trojani, which extended social benefits to non-economic actors based purely on the non-discrimination principle in EU law under TFEU Article 18. The need to protect the Member State's budget against social tourism justifies the judicial shift and the return to the territorial argument illustrated in "the real link" of Vatsouras & Koupatanize. The more an individual is integrated into the host Member State, the more the citizen is integrated into a Member State, the more they are entitled to social benefits (Azoulai 2014). Nevertheless, the "real link doctrine" leaves in a difficult situation those EU nationals like Ms. Dano, or Greek job-seekers Vatsouras & Koupatanize, who cannot claim any social assistance due to scarce links with the host EU Member States. Hence, they are compelled to return home or to stay in the host Member State, begging for money and sleeping in homeless shelters (Vonk 2014).

3.1.2. Tax Allowances and Deductions Granted to Economically Active Citizens

In the field of taxation, the Court has exclusively dealt with economically active citizens, inasmuch as they are the ones who work and obviously pay taxes. The questions posed to the CJEU could be summarized in the problems associated with allowances and deductions of EU citizens who reside and work in different EU Member States. The analysis performed is driven under the free movement of workers (TFEU Article 45) in combination with the non-discrimination in articles 18 and 21 of the TFEU. Whilst in the former section, the Court handled the "real link doctrine" in relation to access to social benefits in the host country, similar rationale is followed in the direct tax cases to let nationals of a Member State to benefit from deductions and allowances provided by the host country, where the employment is exercised. In the following cases, the mutual recognition emerges behind the reasoning of the CJEU.

In international taxation, resident and non-resident taxpayers cannot be treated equally in terms of allowances and deductions derived from their personal and family circumstances. Resident taxpayers perform their economic activities and get access to public benefits and services provided by the State of residence. Therefore, only the State of residence is entitled to take into consideration their personal and family circumstances under the ability to pay principle. In case of non-resident taxpayers, they are subject to limited tax liability which is much lower than resident taxpayers on the same amount inasmuch as the source State does not acknowledge their personal and family circumstances. Such principle of international taxation is set aside in confronting the narrative of European integration upheld by the CJEU, thereby still causing perplexity among the tax scholarship (see, Vanistendael 1996).

In Schumacker, 25 the Court allowed Mr. Schumacker, who was living in Belgium with his wife and children but working in Germany, to benefit from the German "splitting tariff" on the grounds that "the nonresident receives no significant income in the State of his residence and obtains the major part of his taxable income from an activity performed in the State of employment, with the result that the State of his residence is not in a position to grant him the benefits resulting from the taking into account of his personal and family circumstances." ²⁶ The principle of mutual recognition recognizes that Mr. Schumacker's situation is substantially equal to that of a German resident, and therefore Germany has to take into consideration his personal and family circumstances. In the successive line of cases, Renneberg, ²⁷ Commission v. Estonia, ²⁸ and X²⁹, the State of source cannot discriminate the non-resident from a different EU Member State when "all or almost all income" is taxed there and the Residence State cannot take into consideration their personal and family circumstances.

Mr. Renneberg, a Dutch national working in Netherlands but residing in Belgium, bought a house subject to a mortgage loan. The Dutch tax authorities denied the deduction of mortgage interest (negative income) since he was a non-resident in the Netherlands. The Court found discriminatory the different treatment between resident and non-resident taxpayers by the Netherlands. The Court rejected the argument put forward by Netherlands, which qualified the dispute as the mere effect of a disparity resulted from the allocation of taxing rights provided under Article 6.1 of Double Tax Convention between the Netherlands and Belgium. While the positive and negative property-related income related to immovable property located in Belgium is attributed to Belgium, the Netherlands is concerned with work-related income.

²⁵ CJEU, Case C-279/93, Finanzamt Köln-Altstadt/Schumacker, ECLI:EU:C: 1995:31.

²⁶ Schumacker, paras. 36 and 41.

²⁷ CJEU, Case C-527/06, Renneberg, ECLI:EU:C:2008:566.

²⁸ CJEU, Case C-39–10, Commission v. Estonia, ECLI:EU:C:2012:282.

²⁹ CJEU, Case C-283/15, X, ECLI:EU:C:2017:102.

In Commission v. Estonia, an Estonian national who was resident in Finland earned two pensions of similar amount, one derived from her work in Finland and the other derived from her work in Estonia. Since the amount of the Finnish pension was very small and not subject to tax, Finland, as the State of residence, could not take into consideration her personal and family circumstances, nor Estonia which required 75% of the income obtained in Estonia by the non-resident to take into account their personal circumstances. The formula "all or almost of the income" was lowered to 50% (income perceived in Finland as State of residence) and consequently the Court compelled Estonia to take into account her personal and family circumstances since Finland could not.

In X judgment, the Court replaced the Schumacker formula "from all or almost all income" with "major part of the income". Mr. X, residing in Spain where he owned a dwelling, with income from Switzerland (40%) and from the Netherlands (60%) requested the deduction of the negative income derived from its dwelling in the Netherlands since his personal and family circumstances could not be taken into account in Spain due to the lack of resources. The Court reproduced the previous findings to rule on the existence of discrimination since Mr. X could not have his personal and family circumstances taken into account by the Netherlands, where he receives 60%, and Spain, where he lived. Accordingly, the Netherlands must enable Mr. X to apply his personal and family circumstances, in proportion to the share of that income received in the Member State of activity. 32

The Schumacher line of cases and especially X judgment disregards the income earned in the source State. It does not matter whether it amounts to 75%, 60% or 50%, because what is really crucial is the fact that the residence State cannot take into account the taxpayer's personal and family circumstances. Ros put it clear: "the X judgment stipulates that it is not decisive whether the taxpaver earns all or almost all his income in one Member State but rather if the Member of State of residence is not in a position to take into account his personal and family situation. In that case it is the Member States of activity that should take into account the personal and family situation of the taxpayer proportionally" (Ros 2018, 158). The pro-rata approach in the X judgment is welcome by the CFE insofar as first, it overturns the outcome of Kieback³³ and second. it supports "an open market economy with free competition, an efficient allocation of production factors, tax neutrality, a level playing field, international tax neutrality, the ability-to-pay principle, the direct benefit principle and origin-based taxation" (CFE ECJ Task Force 2018).

Commission v. Estonia, para. 55.

³¹ X, para. 34.

³² X, paras. 41 and 49.

³³ CJEU, Case C-9/14, Kieback, ECLI:EU:C:2015:406.

In the author's view, this line of cases strengthened the mutual recognition principle within EU citizenship, which basically compels the EU Member State of source to take into consideration the personal and family circumstances of the taxpayers when the State of residence cannot do so due to the lack of taxable income. The mutual recognition stemming from EU citizenship goes beyond the recognition of fundamental economic rights to freely circulate³⁴ to impose a duty on the Member State of source to take into consideration the personal and family circumstances of the taxpayer. Even fierce critics of the Schumacher's rationale, such as Wattel, who do not endorse the discrimination analysis handled by the Court - in a nutshell, the State of residence does not discriminate because it did not exercise its taxing power – solve the issue under the national treatment principle in EU law: "the source state that should (proportionally) extend to the non-resident the same personal allowances it grants to its own residents earning the same income (national treatment)". 35 Either under discrimination analysis or under national treatment, there is a duty of the Member State of source to treat equally the non-resident taxpayers who work there.

Schumacher's line of cases is not at odds with the reasoning followed by the Court in Marks & Spencer³⁶ in which the headquarters country must take into account the final losses of the subsidiaries since the Member State of the residence of the subsidiaries cannot. In both scenarios – corporate tax law (final losses) and personal tax law (personal and family allowances) – the CJEU creates new international allocation rules within the EU polity.³⁷ Likewise, Schumacher is aligned with "the real link" doctrine in cases like Vatsouras & Koupatanize. The more an individual is integrated within the host Member State, in this case by working there, the more they are entitled to the allowances and benefits provided by the host State to its own residents.

The mutual recognition principle does not only operate in the comparison between resident and non-resident taxpayers: the CJEU has stretched its limits to embrace a horizontal comparison of different non-resident taxpayers in Sopora.³⁸ In this case, the Dutch 30% wage tax

³⁴ As noted by Ros (Ros 2018, 159): "[...]the market freedoms are no longer instrumental rights, but are rights granted to EU citizens for their owns sake and can, therefore, be considered as fundamental economic rights".

³⁵ The Court ruled that discrimination arose from the fact that Mr. Schumacher's personal and family circumstances are taken into account neither in the State of residence nor in the State of employment. In Wattel's view, "one cannot define an alleged discrimination by one state by reference to something another state is not doing (Wattel 2015).

³⁶ CJEU, Case C-446/03, Marks & Spencer, ECLI:EU:C:2005:763.

³⁷ Smit observed that the CJEU created a new allocation rule on final losses in relation to Marks & Spencer (Smit 2017, 70).

³⁸ CJEU, Case C-512/13, Sopora, ECLI:EU:C:2015:108.

facility was only applicable to non-residents living outside Netherlands, at a distance of more than 150 kilometers from the border of the given Member State before taking up a job in the Netherlands. Those who did not comply with this requirement prior to taking the job in the Netherlands were required to provide proof of the amount of extra-territorial expenses to be compensated. Mr. Sopora challenged the denial of the beneficial regime, because he was living at a distance less than 150 Km from the Netherlands border. The Court ruled in favor of the horizontal comparison, thereby prohibiting the Netherlands from discriminating the non-resident, provided that the 30% tax wage did not give rise to a net overcompensation in respect of the extraterritorial expenses actually incurred for taxpayers living less than 150 Km from the Dutch border.³⁹

3.2. European Citizenship as Common Values and Ideals ("Union Territory")

While the previous line of cases of EU citizenship operates on the basis of mutual recognition, another stream of cases has enhanced EU citizenship as linked to the "Union territory" beyond the domestic borders. In the landmark Ruiz-Zambrano, 40 the Court dealt with an expulsion order in the field of immigration. The Belgium authorities denied a third country national from Colombia residence in Belgium and his work permit and ordered him to leave the country, despite the fact that his children had already received the Belgian nationality and he made clear efforts to integrate into Belgian society. The Court stated that "A refusal to grant a right of residence to a third country national with dependent minor children in the Member State where those children are nationals and reside, and also a refusal to grant such a person a work permit, [...] would lead to a situation where those children, citizens of the Union, would have to leave the territory of the Union in order to accompany their parents. In those circumstances, those citizens of the Union would, in fact, be unable to exercise the substance of the rights conferred on them by virtue of their status as citizens of the Union". 41

As Azoulai points out, rather than conceiving the territory of the Union as the sum of individual territories of the Member States, the Union territory should be conceived as "a metaphor for a certain conception of the space referred to in Article 2 TEU as 'a [European] society in which pluralism, non-discrimination, tolerance, justice, solidarity and equality between women and men prevail. Following the Court of Justice's reasoning, leaving European territory means not only leaving Europe in the geographical sense; it means leaving a community

On positive appraisal of the Sopora judgement, see Kemmeren (2015).

⁴⁰ CJEU, Case C-34/09, Ruiz Zambrano, ECLI:EU:C:2011:124, para. 44.

⁴¹ Ruiz Zambrano, paras. 43 and 44.

of ideals and values; it means being deprived a certain mode of existence corresponding to the standards of European society. As stated in Ruiz Zambrano, the territory of the Union 'transcends' the 'territorial framework of national communities'. It stands for the mix of material and immaterial things that determines the sustainability of individual existence; what we may call a 'European way of life" (Azoulai 2014, 3).

The Territory of the Union as a space of ideals and values, beyond the Member States' borders, is reaffirmed in Garcia Abello in relation to the surnames. Belgium denied Mr. Garcia Avello and his spouse (Spanish nationals residing in Belgium) the change requested in their patronymic surname of their two children, who were born in Belgium. The justification for the rejection was based on principle of the immutability of surnames as a founding principle of social order to prevent risks of confusion as to identity or parentage of persons. The Court dismissed such justification on the grounds that "parentage cannot necessarily be assessed within the social life of a Member State solely on the basis of the criterion of the system applicable to nationals of that latter State."

The above-mentioned understanding of the EU citizenship beyond domestic borders entitles the Court to make an assessment to what extent the national measure at issue may "restrict the genuine enjoyment of the substance of rights of EU citizen." In other words, the Court wonders whether such domestic measure may jeopardize the enjoyment of a "European way of life," meaning the attachment of the individual to a community of values and ideals promoted by the EU. What is the meaning of "genuine substance of rights"? Azoulai resorts to the image of the "good citizen" illustrated in the Ruiz Zambrano judgment. Mr. Zambrano and his wife made clear efforts to integrate into Belgian society, their children were in school and they paid their taxes. "Such behavior is that of a 'good citizen' for whom public policy is in no way a constraint on the individual, but rather a source of 'enjoyment'. Accordingly, deportation from the Europe would amount to a real 'expatriation.' It would mean displacing an individual and its family from a place they came to occupy and which was assigned to them, a place which they were somehow 'destined' to live in" (Azoulai 2014, 13). To assess whether the genuine enjoyment of EU rights is threatened by domestic norms, the Court searches for a bond between the individual with the community, i.e. "objective traces of social integration" (Azoulai 2014, 13).

This idea of social integration applied to "good citizens" does not simply require an abstract adherence to the values of the Union laid down

⁴² Case C-148/02, Garcia Abello, ECLI:EU:C:2003:539.

⁴³ Garcia Abello, para. 40.

⁴⁴ Garcia Abello, para. 42.

in Article 2 TEU (dignity, equality, rule of law, fundamental rights, etc.). It is the author's understanding that social integration requires that the individual is a full-fledged member of the welfare state of the host Member States. Welfare states, as created in Europe after the Second World War, aim to protect the well-being of their citizens, especially those in financial or social needs, by means of grants, pensions, and other social benefits. To accomplish this protective task assigned to the welfare state, the collection of taxes become essential to redistribute and achieve the well-being of all individuals (Hulten 2019, 33; Heins, Deeming 2015). By paying his taxes to the Belgium State, Mr. Zambrano actually contributed to the welfare state, and therefore he was socially integrated. The "European way of life" – in contrast with the "American way of life" – undoubtedly resorts to the need to achieve the well-being of the individuals through the means of the welfare state.

Until now, only in extreme circumstances such in Ruiz-Zambrano, wherein the children as EU citizens were compelled to leave the territory of the Union, the Court turned down the domestic measure. The third country national's intention to live in Europe or the simple aspiration to keep the family together in Dereci⁴⁵ is not enough for the Court to activate the protection under EU citizenship. Likewise, in the other cases (i.e. McCarthy, ⁴⁶ Alokpa⁴⁷), in which the EU citizen was not economically active and there were no risks of expulsion from the EU, the Court has been more cautious in its assessment.

For example, in McCarthy, British tax authorities denied a residence permit to a Irish national, who was also a UK national living in UK, married to a Jamaican national inasmuch as Mrs. McCarthy was not "a qualified person" (essentially, a worker, self-employed person or selfsufficient person) and, accordingly, that Mr. McCarthy was not the spouse of "a qualified person". Article 3 (1) of the Directive 2004/38 could not apply at the case at stake since Mrs. McCarthy never exercised her right of free movement and has always resided in a Member State of which she is a national. However, TFEU Article 21 is applied to purely internal situations in order to protect the right to move and reside freely within the territory of the Member States, as the Court ruled in Ruiz-Zambrano, and therefore, prevent any damage to the genuine enjoyment of the substance of rights of EU citizen. In the judgment, the CJEU concluded that the denial of residence permit did not affect her right to move and reside freely within the territory of the Union, since she can always move back to Ireland.48

⁴⁵ CJEU, Case C-256/11, Dereci, ECLI:EU:C:2011:734, para. 68.

⁴⁶ CJEU, Case C-434/09, McCarthy, ECLI:EU:C:2011:277, para. 49–50.

⁴⁷ CJEU, Case C-86/12, Alokpa, ECLI:EU:C:2013:645.

⁴⁸ McCarthy, paras. 49–50.

In Alopka, the same CJEU rationale applied to the rejection of residence permit by the Luxembourg authorities to a Togolese national who immigrated to Luxembourg, and gave birth to twins, who were recognized by a French national and received French citizenship. Since their birth, Mrs. Alokpa could benefit of a derived right to live in France, and consequently the refusal by Luxembourg to grant her residence permit did not oblige her and her children to leave the territory of the Union. The shortcoming of Alokpa could be read as disappointing by the commentators. As Ros observes, the conservative approach of the court gave precedence to the nationality of the twins (France), rather that the real link with the host State (Luxembourg) (Ros 2018, 151).

The CJEU's findings in Ruiz-Zambrano, promoting the values and ideals of the Union beyond the frontiers of the Member States, have been challenged in relation to expulsion orders of those committing criminal offences in the host Member State. In this field, a remarkably evolution in the case law has taken place in the direction of embracing a European society of common values. In initial cases PI and MG, ⁵⁰ the Court stressed that crimes reveal the non-compliance by the person with the values expressed by the society of the host Member State in its criminal law. In such cases of "bad citizens", Azoulai noticed that the Court looked to the value system of the host Member State and therefore facilitated expelling Union citizens who breaches its domestic social cohesion (Azoula 2014, 16). Although the Court employs the formula that such behavior shows a "lack of feeling of Union Citizenship," ⁵¹ there is no reference in the judgment to the common values of the Union's public order, which apply to the whole territory of the Union.

However in recent case law – B & Vomero, K & HF ⁵² – the Court has progressively engaged into promoting the values of the Union to protect EU citizens against expulsion orders (Benlolo Carabot 2019; Coutts 2018). In B & Vomero, in measuring the integrative links of the citizen with the host Member State, not only the period of imprisonment counts, but also the reintegration into European society: "the social reintegration of the Union citizen in the State in which he has become genuinely integrated is not only in his interest but also in that of the European Union in general." In K & HF, the Court dealt with expulsion orders of individuals who participated in serious war crimes and remained

⁴⁹ Alopka, para. 34.

 $^{^{50}\,}$ CJEU, Case C-348/09, PI, ECLI:EU:C:2012:300; CJEU, Case C-400/12, MG, ECLI:EU:C:2014:9.

⁵¹ CJEU, Case C-378/12, Onuekwere, ECLI:EU:C:2014:13, para. 24.

⁵² CJEU, Joined Cases C-316/16 & C-424/16, B & Vomero, ECLI:EU:C:2018:256;
CJEU, Joined Cases C-331/16 & C-366/16, K & HF, ECLI:EU:C:2018:296.

⁵³ B & Vomero, para. 75.

in the host Member State (Netherlands) without a legal residence permit but enjoying a family life. Although criminal offences must be assessed in light of the fundamental interests of the host Member State, protected by its criminal code, the Court underlined that war crimes "seriously undermine both fundamental values such as respect for human dignity and human rights, on which, as stated in Article 2 TEU, the European Union is founded, and the peace which it is the Union's aim to promote, under Article 3 TEU."⁵⁴

To sum up, Ruiz-Zambrano and the latest cases regarding criminal offences identify first the territory of the Union as a space to promote certain EU values and ideals beyond the domestic borders of the Member States, and second, citizenship as a driver of social integration. The "good citizens" who pay their taxes and contribute to their welfare state of their host countries cannot be deprived from "the genuine enjoyment of the substance of rights of EU citizens." In other words, the European way of life is anchored in sharing certain fundamental values (Article 2 TEU) and the fact that the State has to provide the well-being of the individuals through the redistributive mechanisms displayed by the welfare state. Such a powerful narrative unfortunately does not apply to economic inactive citizens (i.e. Alopka, McCarthy) who do not contribute to the domestic welfare state of the host country, and therefore are excluded from the benefits derived from being members of "Territory of the Union".

4. TAXING EUROPEAN CITIZENS TO ENFORCE THE EU SOLIDARITY PRINCIPLE

The current deep EU crisis, triggered by austerity measures, migration crisis, anti-European movements in Eastern Europe and Brexit, clearly shows the lack of a feeling of membership of the "peoples of Europe" to the EU polity. In other words, individuals no longer feel as members of the EU polity (Bouza Garcia 2017). The divorce between the "peoples of Europe" and Brussels is quite visible in the low turnout in the European elections. Hence, the EU integration project is doomed to fail in the short term if this feeling of "belonging" is not restored soon. The other way around, as Barroso observed, the peace narrative – the Union has been a space without wars from more than 50 years – is no longer convincing for the "peoples of Europe", and hence, a new narrative based on solidarity and social cohesion must emerge (Barroso 2013).

⁵⁴ K & HF, para. 46.

⁵⁵ It should be noted that the recipients of the EU treaties are not only the Member States but also the peoples of Europe (CJEU, Case 26/62, Van Gend en Loos, ECLI:EU:C:1963:1).

In practical terms, what does "membership" to the EU mean? How do we build up a new narrative, in Barroso's words? How can we reinforce the feeling of membership? In the author's view, two current deficits must be resolved. On one side, it means solving the so-called democratic deficit of the EU (Weiler 1997; Schimitter 2000; Heritier 1999), and on the other enforcing solidarity at the EU level. ⁵⁶ Both channels undoubtedly lead to the concept of EU citizenship. In short, EU citizens, as members of the EU polity, would be citizens who democratically participate in the EU polity and pay taxes to the EU polity. The collection of such an EU tax would be redistributed among the "peoples of Europe" in accordance with the solidarity principle. Such an EU tax would aim to protect those economically inactive citizens, who are completely excluded from the benefits of the European way of life. One may wonder on which economic source of income such the EU tax would be levied. However, rather that linking taxes to a specific economic activity, wealth possession or specific purpose of the taxpayer, the modern constitutionalist doctrine boils down taxes to a mere expropriation. "After all, a tax is a form of expropriation without compensation, where not even the public purpose for which the tax was collected need be given" (Sajo 1999, 159; Menendez 2001, 121). But of course, a legitimate expropriation since it has been agreed on a democratic basis by parliament (De Crouy-Chanel 2006). Taxes aim to prevent inequality in society and therefore comply with a redistribution purpose.⁵⁷ thereby enforcing solidarity.

Malcolm Ross conceives solidarity as a constitutional paradigm in the EU, which aims to transform the EU polity, under the auspices of social justice (Ros 2010). In Ross's perspective (Ros 2010, 42), solidarity as a transformative legal concept emerges across the treaties (i.e. TEU Article 2) and specifically in the case law of the CJEU in dealing with the cumbersome relationships between social and market values (e.g. Viking⁵⁸). However, the last financial crisis has demonstrated the failure

⁵⁶ Solidarity is not only a founding value of the EU, in articles 2 and 3 of the TEU, but also a goal enshrined in the 1950 Schuman Declaration: "Europe will not be made all at once, or according to a single plan. It will be built through concrete achievements, which first create a de facto solidarity" (Ros 2010; De Witte 2015).

⁵⁷ The scope of the theories of distributive justice and taxation goes beyond this contribution. On a detailed account of welfarist approach, the Dworkin's equality of resources and libertarian theories of distributive justice, see for example the following recent contributions (Cappelen, Tungodden 2018; Duff 2017).

⁵⁸ CJEU, Case C-438/05, Viking, ECLI:EU:C:2007:772, para. 79: "Since the Community has thus not only an economic but also a social purpose, the rights under the provisions of the Treaty on the free movement of goods, persons, services and capital must be balanced against the objectives pursued by social policy, which include, as is clear from the first paragraph of Article 136 EC, inter alia, improved living and working conditions, so as to make possible their harmonisation while improvement is being maintained, proper social protection and dialogue between management and labour."

of the EU legal principle of solidarity and conversely, only "reciprocal" solidarity has been visible, "whereby the contributor shares with the recipient in anticipation of a (future) counter-contribution or fair return" (Pantazatou 2015; Nicoli 2015). The transformative EU legal concept of solidarity, coined by Malcom Ross should evolve against these intergovernmental reciprocal responses, derived from the economic crisis, and towards a proper EU redistributive framework among the "peoples of Europe", a fully-fledged fiscal Union, as proposed by Nicoli (Nicoli 2015, 44).

The disaffection of the "peoples of Europe" with the EU integration project shows that the current contributions of the Member States to the EU budget are not enough to reinforce the two deficits mentioned above, namely the demos and the lack of EU solidarity. In the author's view, levying an EU direct tax upon the EU citizens is the concrete to build a democratic and solidary EU polity.

5. CONCLUSION

Both streams of case law by the CJEU – mutual recognition and the Territory of the Union – embrace a resilient concept of EU citizenship in light of the benchmark of EU supranational citizenship developed by Strurmia in Section 2.

However, there are serious drawbacks to achieving a proper solidarity among the "peoples of Europe" put forward in cases in the area of mutual recognition (Dano, Vatsouras & Koupatanize). In these cases, the citizens cannot claim any social assistance due to scarce links with the host EU Member States. The doctrine of a "real link" with the host country jeopardizes the achievement of a truly supranational solidarity. Therein lies the precise criticisms of authors such as Ros, who argues that: "it seems that under the current change in public appetite for EU citizenship, the ECJ finds that some EU citizens are more equal than others. A perception far away from a true fundamental status for EU citizens, economically active or not" (Ros 2018, 159). A radical distinction emerges between economic active citizens who benefit from the mutual recognition principle and those who are not economically active, who are completely abandoned in the EU polity.

In the author's personal opinion, the pessimistic view conveyed by Ros must be reconciled with the other important stream of CJEU case law: the Union Territory as a space of shared values and goals, which includes solidarity. In other words, the Court is constructing a "European way of life" that reflects that the Union is not only an institutional project but also an "existential project" (Editorial Comments CMLR 2017). The

development of the EU citizenship is responsive to the emergence of the EU community beyond the States. For example, Ruiz-Zambrano, Garcia Abello, B & Vomero, K & HF look for integrative bonds of the individual the Union, beyond the borders of the host Member State. EU law becomes a tool for social integration (Azoulai 2018). The Union is not only an space of mobility of individuals under the right to free movement, but also a community that shares values and rights, consumes products or experiences culture from other Member States at home, has relatives in another European country, learns European languages, interacts with Europeans, vindicates consumer or worker rights derived from EU legislation (Editorial Comments CMLR 2017, 358). Accordingly, De Witte refers to the emancipatory power of free movement to liberate individuals from the normative choices imposed by their state of origin, and thereby allowing them to self-realize (De Witte 2016). The Territory of the Union also serves to adhere to welfare state: those "good citizens" who pay their taxes and contribute therefore to the welfare state of the host country are also protected under the EU citizenship. However, in parallel to the cases in relation to the mutual recognition, the Alopka and McCarthy cases show that economic inactive citizens are completely excluded from the protection derived from the "European way of life".

The term "European way of life" has recently crept into de political arena, insofar as the new Ursula von der Leyen Commission has appointed Margaritis Schinas to hold one of the Commission's vice-presidency of Protecting our European way of life. This author does not personally understand the European way of life in a "fascist" manner, as building a fortress. On the contrary, the "European way of life" must refer not only to a space of values such as democracy, protection of fundamental rights, but also as a space of solidarity and social justice. The only possible manner to reconcile both streams of case law is to levy a tax on EU citizens, which would be redistributed according to the premises of the EU solidarity principle. Accordingly, the economically inactive citizens could benefit also from the "European way of life".

REFERENCES

Avbelj, Matej, Jan Komárek. 2012. Constitutional Pluralism in the European Union and Beyond. Oxford: Hart.

Azoulai, Loic. 14/2014. The (Mis)Construction of the European Individual. EUI Working Paper LAW 14. European University Institute, Florence. https://cadmus.eui.eu/bitstream/handle/1814/33

⁵⁹ See EU Press Release 10 September 2019, https://europa.eu/rapid/press-rele ase IP-19-5542 en.htm (last visited 26 November 2019).

- 293/LAW_WP_%202014_14.pdf?sequence=1&isAllowed=y (last visited 26 November 2019).
- Azoulai, Loic. 2018. Le sens du social dans le droit de l'Union européenne. 265–280 in *Les frontières de l'Europe sociale*, edited by Ségolène Barbou des Place. Paris: Pedone.
- Bauböck, Rainer. 2006. Citizenship and Migration Concepts and Controversies. *Migration and Citizenship. Legal Status, Rights and Political Participation*, edited by Rainer Bauböck. Amsterdam: Amsterdam University Press.
- Barroso, Jose Manuel. 2013. A New Narrative for Europe. SPEECH/13/357. Palais des Beaux Arts, Brussels, 23 April 2013. http://europa.eu/rapid/press-release_SPEECH-13-357_en.htm (last visited 26 November 2019).
- Benlolo Carabot, Myriam. 3/2019. Citizenship, integration, and the public policy exception: B and Vomero and K. and H.F. *Common Market Law Review* 56: 771–802.
- Beretta, Giorgio. 7/2018. Mobility of Individuals after BEPS: The Persistent Conflict between Jurisdictions. *Bull. Intl. Taxn* 72: 439–449.
- Beretta, Giorgio. 2/2019. Citizenship and Tax. *World Tax Journal* 11: 227–260.
- Bouza Garcia, Luis. 3/2017. The 'New Narrative Project' and the politicisation of the EU. *Journal of Contemporary European Studies* 25: 340–353.
- Cappelen, Alexander, Bertil Tungodden. 2018. Tax Policy and Fair Inequality. 111–124 in *Philosophical Perspectives on Taxation*, edited by Martin O'Neil and Shepley Orr, Oxford: Oxford University Press.
- CFE ECJ Task Force, 4/2018. Opinion Statement ECJ-TF 4/2017 on the Decision of the Court of Justice of the European Union of 9 February 2017 in X (Case C-283/15) ("Pro-Rata Personal Deductions"), Concerning Personal and Family Tax Benefits in Multi-State Situations. *European Taxation* 58: 149–162.
- Clarke, John. Kathleen Coll, Evelina Dagnino, Catherine Neveu. 2014. *Disputing Citizenship*. Bristol: Bristol University Press.
- Coutts, Stephen. 2018. The Expressive Dimension of the Union Citizenship Expulsion Regime: Joined Cases C-331/16 and C-366/16, K and HF. European Papers 2: 833–841. http://www.europeanpapers.eu/en/europeanforum/expressive-dimension-of-union-citizenship-expulsion-regime (last visited 26 November 2019).

- De Witte, Floris. 2015. *Justice in the EU: The Emergence of Transnational Solidarity.* Oxford: Oxford Univ. Press.
- De Witte, Floris, Rainer Bauböck and Jo Shaw. 2016. Freedom of Movement Under Attack: Is It Worth Defending As the Core of EU Citizenship? https://ssrn.com/abstract=2888363 (last visited 26 November 2019).
- De Crouy-Chanel, Emmanuel. 2006. La Définition juridique de l'impôt. L'exemple de la doctrine française. in *Philosophie de l'impôt* edited by Thomas Berns, Jean-Claude Dupont, Mikhaïl Xifaras and Vincenzo Lavenia. Bruxelles: Bruylant.
- Dobson, Lynn. 2006. *Supranational Citizenship*. Manchester: Manchester University Press.
- Duff, David. 2017. Tax Policy and the Virtuous Sovereign. 167–189 in *Philosophical Foundations of Tax Law*, edited by Monica Bhandari. Oxford: Oxford University Press.
- Editorial Comments CMLR, 2/2017. EU law as a way of life. *Common Market Law Review* 54: 357–368.
- Haag, Maria. 2019. Case C-93/18, Bajratari Unlawful Employment and the Right to Free Movement, *European Law Blog. https://europeanlawblog.eu/2019/10/07/case-c-93-18-bajratari-unlawful-employment-and-the-right-to-free-movement/* (last visited 26 November 2019).
- Heins, Elke, Chris Deeming. 2015. Welfare and well-being inextricably linked. 13–15 in *In Defence of Welfare*, edited by Anne Brunton, Liam Foster, Tina Haux & Chris Deeming. Bristol: Policy Press Bristol. https://www.research.ed.ac.uk/portal/files/20073945/Welfare_and_Wellbeing_inextricably_linked.pdf (last visited 26 November 2019).
- Héritier, Adrianne. 2/1999. Elements of democratic legitimation in Europe: An alternative perspective. *Journal of European Public Policy* 6: 269–82.
- Hulten, Mark Van. 2019. Aiming for Well-Being Through Taxation: A Framework of Caution and Restraint for States. Tilburg: Tilburg University (forthcoming Thesis dissertation),
- Kemmeren, Eric. 4/2015. Sopora: A welcome Landmark Decision on Horizontal Comparison. *EC Tax Review* 24: 178–184.
- Kochenov, Dmitri. 2019. Citizenship. Cambridge: MIT Press.
- Kostíć, Svetislav. 2/2019. In Search of the Digital Nomad Rethinking the Taxation of Employment Income under Tax Treaties. *World Tax Journal* 11: 189–225.

- Linklater, Andrew. 1999. Cosmopolitan Citizenship. 35–59 in *Cosmopolitan Citizenship*, edited by Roland Dannreuther and Kimberly Hutchings. London: Macmillan Press.
- Menendez, Agustin. 2001. Justifying Taxes, Dordrecht: Kluwer.
- Nicoli Franceso. 3/2015. Eurocrisis and the Myth of European Redistribution: Illegitimate, Unsustainable, Inefficient. *Perspectives on Federalism* 7: 19–48.
- Nussbaum, Martha. 1996. Patriotism and Cosmopolitanism. 3–20 in For Love of Country: Debating the Limits of Patriotism, edited by Joshua Cohen and Martha Nussbaum. Boston: Beacon Press.
- Pantazatou, Katherina. 3/2015. Promoting solidarity in crisis times: Building on the EU Budget and the EU Funds. *Perspectives on Federalism*, 7: 49–71.
- Ros, Eric. 2017. EU Citizenship and Direct Taxation. Alphen aan den Rijn; Kluwer Law International.
- Ros, Eric. 3/2018. EU Citizenship and Direct Taxation, The European Court of Justice in the Era of Public Decline for a Citizen's Europe. *EC Tax Review* 27: 147–159.
- Ros, Malcom. Yuri Borgmann-Prebil. 2010. *Promoting Solidarity in the EU*, Oxford: Oxford Univ. Press.
- Sajó, Andras. 1999. Limiting Government. An Introduction to Constitutionalism. Budapest: Central European University Press.
- Schmitter, Philippe. 2000. How to Democratize the European Union... And Why Bother? Lanham, Maryland (US): Rowman & Littlefield.
- Schön, Wolfgang. 2003. General Report. 1–42 in *Tax Competition in Europe* edited by W. Schön. Amsterdam: IBFD.
- Schön, Wolfgang. 2018. Taxation and Democracy, Working Paper. Max Planck Institute for Tax Law and Public Finance No. 2018–13. https://ssrn.com/abstract=3267279 (last visited 26 November 2019).
- Smit, Daniël. 2/2017. International Income Allocation under EU Tax Law: Tinker, Tailor, Soldier, Sailor. *EC Tax Review* 26: 70.
- Strumia, Francesca. 2017. Supranational Citizenship, 669–693 in *The Oxford Handbook of Citizenship*. edited by Ayelet Shachar, Rainer Bauboeck, Irene Bloemraad, and Maarten Vink. Oxford: Oxford Univ. Press.
- Vanistendael, Frans. 1996. The consequences of Schumacker and Wielockx: Two steps forward in the tax procession of Echternach. *Common Market Law Review* 33: 255–269.
- Wattel, Peter. 12/2015. Non-Discrimination à la Cour: The ECJ's (Lack of) Comparability Analysis in Direct Tax Cases. *European Taxation* 55: 542–553.

Weiler, Joseph. 2/1997. The European Union belongs to its citizens: three immodest proposals. *European Law Review* 22: 150–156.

Vonk, Gijsbert. 2014. EU-freedom of movement: No protection for the stranded poor. *European Law Blog. https://europeanlawblog.eu/2014/11/25/eu-freedom-of-movement-no-protection-for-the-stranded-poor/* (last visited 26 November 2019).

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(497.11)

CERIF: S141

DOI: 10.5937/AnaliPFB1904173I

Iva Ivanov, LL.M.*

TAX INCENTIVES FOR KEEPING AND ATTRACTING HIGHLY SKILLED WORKERS: THE CASE OF SERBIA

The recent increased migration of workers has posed a brain drain problem on countries, which lose their citizens to more developed countries offering better working and living conditions. Lowering the tax burden on highly skilled individuals has been one of the most commonly used incentives by both developed and developing countries. The Government of the Republic of Serbia has proposed several tax incentives for providing a more beneficial tax treatment for highly skilled employees with the aim of keeping and attracting them back. The first part of this paper illustrates the problem of emigration, the effects of emigration of highly skilled workers, and the effects of taxation on migration decisions. In the second part, the problem of brain drain and its breadth in the Republic of Serbia is addressed, and a detailed elaboration of newly proposed tax incentives is provided. Furthermore, the author proposes an additional tax incentive.

Key words: Brain drain. – Highly skilled workers. – Migration. – Tax incentive. – The Republic of Serbia.

1. MIGRATION

Migration of people is not a phenomenon inherent to the 21st century. People were emigrating from their home countries in pursuit of a better life from the ancient times. However, the breadth of migration has increased in the past few decades due to the demographic and economic imbalances between countries (OECD 2018, 9). Imbalances have been

^{*} PhD student, University of Belgrade Faculty of Law; Consultant, Ministry of finance of the Republic of Serbia, Sector for Fiscal System.

broadened further with the development of technology and its unequal dispersion across countries, especially when developed and developing countries are compared. Furthermore, the development of technology and science has reduced the costs of migration. Reduction of transportation and living costs has facilitated the migration of the increasing number of people. Moreover, advancement of technologies has provided methods for staying in contact with family members who remain in the source country while not feeling left out and homesick (OECD 2011, 125). As a consequence, there are more people migrating in search of a better life today than there were before.

Especially prone to migration are highly skilled workers: workers with a tertiary education or a specialised skill set. They are more likely to migrate to a country that offers better living and working conditions than settling in the country in which they were born (OECD 2019, 1; Fink, Miguelez 2017, 10). Better economic, working, development and living conditions are mentioned as the most common reasons for migration (OECD 2011, 125).

Increased migration of highly skilled persons is compatible with the rising demand for highly skilled labour. Countries are in need of workers who would induce the development and growth of their economies, given the much more pronounced lack of a satisfying domestic labour supply. As suggested by Liebig, Sousa-Poza (2005, 7), the growing international division of labour and technological progress nowadays requires an increased number of highly skilled individuals internationally. Highly skilled workers are needed to fill in the managerial, professional, and highly technologized job vacancies (Keery 2017, 65; Romer 2000, 222). By offering better financial conditions, developed countries are encouraged and successful in attracting highly skilled workers from outside their borders, while at the same time keeping their own highly skilled workers. As suggested by Brauner (2015, 4), since the wage gap between countries is not closing, further migrations can be expected.

The home grown labour supply has not been able to meet demand since the 1970s. It was estimated that during the 1990s, 2.5 million highly skilled workers residing in the United States of America were immigrants (Docquier, Rapoport 2009, 248). After the 2008 financial crisis, at the level of the European Union (EU), it was recognised that providing better conditions for growth requires promoting a forward-looking and comprehensive labour migration policy that would respond to the needs of labour markets (European Commission 2010, 17). Consequently, the work permit for highly skilled non-EU citizens was introduced in the EU in 2009, called the EU Blue Card, with the aim of easing the conditions and process of hiring highly skilled workers from the non-EU countries. Germany, for example, more recently introduced a new law that further

relaxes the conditions for hiring highly skilled workers from non-EU countries (for more see Taube 2018).

The need to look for highly skilled labour force outside their own borders is also caused by the fact that today's society is an ageing society (Cerna 2018). An ageing population poses the problem to productivity and growth of economies as there are not enough domestic workers to maintain and spur economic growth.

Labour shortage requires attracting foreign highly skilled workers, but it has also highlighted the need for attracting and retaining students who have migrated for study purposes (Burmann *et al.*, 2018, 50; Cerna 2018, 3). Students are future workers, i.e. the pillars of economic growth, making them indispensable for countries. As suggested by Hawthorne (2018, 7–8), they provide a productivity premium to destination countries as they are far younger and with professional carriers likely to span decades, providing fiscal benefits for destination countries for a significant period. Consequently, it does not come as a surprise that the number of countries are offering better conditions for finding a job upon the completion of studies, as well as getting a work permit.¹

Competition between countries in attracting and retaining human capital has emerged as a consequence of policies for attracting highly skilled workers and students (LaRaine Ingram 2016, 225; Brauner 2015, 20; Fink, Miguelez 2017, 2; Docquier, Rapoport 2009, 247). Countries are offering a number of incentives that aim to provide the best economic conditions, infrastructure, scientific institutions and the overall better quality of life for highly skilled workers. Competition is especially burdensome for developing countries as they have to compete with developed countries that are equipped with more and better resources (Liebig, Sousa-Poza 2005, 7). However, understanding the importance of highly skilled individuals, developing countries have started pushing back and providing incentives for keeping their highly skilled workers and students, and even attracting foreign highly skilled workers.

1.1. The Effects of Emigration of Highly Skilled Workers

The most common reason for emigration is a higher salary, i.e. better financial conditions offered in the destination country (Fink, Miguelez 2017, 11). Other reasons such as better working conditions, more and better scientific institutes, more funding for research and development activities, also influence the decision of a highly skilled individual to migrate (Burmann *et al.*, 2018, 42). These are not the only reasons, but they are among the most common ones. Political stability,

Countries like the United States of America, Canada, Australia, New Zealand, the United Kingdom, Germany, France, Italy, the Netherlands, Sweden and many more.

less corruption, better infrastructure, better education institutions, and many other conditions can influence the decision of workers and students to migrate.

The decision to migrate affects two countries: the source country, where the individual was born and educated before deciding to migrate, and the destination country, the country to which the individual moved with the purpose of working and living there. So, while one side loses one highly skilled individual, the other side gains. However, as we will see, the loss for the country of origin can be especially grave as it loses the treasured input for its development which is difficult to compensate.

The positive effect of immigration of highly skilled workers for destination countries has been proven. The studies showed that highly educated immigrants have a positive impact on the growth of innovation in the destination country (Bosetti, Cataneo, Verdolini 2015, 321), especially in highly technologized sectors (Fassio, Montobbio, Venturini 2019, 717). Providing positive effects on innovation, which is an essential part for the growth of the entire economy, immigrants provide the so-called *brain gain* for destination countries (LaRaine Ingram 2016, 225). In addition to inducing innovation growth, brain gain also takes the form of the free increase in the human capital stock and a fiscal gain through taxation of immigrants' income.

However, a different story can be told of the effects of the emigration of the highly skilled workers for the source country. The loss of highly skilled individuals, also known as *brain drain*, is most commonly defined as a migration of highly educated individuals towards countries offering better opportunities to the detriment of the countries of origin, particularly developing ones (OECD 2017a, 198). As suggested by the definition itself, brain drain has a negative effect on the prospects of the source country. Namely, emigration of highly skilled workers slows down the innovation and development of the source country, as there are not enough individuals who can generate development (OECD 2017a, 196–197).

The loss for the source countries can include financial and social loss, as well as a loss in human capital (OECD 2017a, 196). While the source countries have invested in the education of highly educated individuals, incurring costs for their education, after their emigration, such investment becomes a failed investment and a fiscal loss for the source country (OECD 2011, 134). The source country loses the chance to recoup the investment through the taxation of the future income of highly skilled individuals. The loss is always high, if we take into consideration that a tertiary educated individual is more likely to earn higher income compared to other workers, which even with a proportional taxation of personal income, presents a significant loss for the source country.

Developing countries are in danger of an even greater loss, as they lose the necessary human capital to spur the much needed economic growth (Arslan *et al.*, 2014, 4). The lack of highly educated individuals that could perform R&D activities, which are *sine qua non* for innovation and growth of countries, can become detrimental to the development of a developing country (Docquier, Rapoport 2009, 248). The shortage of highly skilled individuals is felt especially negatively in the sectors linked with the wellbeing, such as in the medical field (OECD 2017a, 199). Furthermore, it is argued that brain drain can lead to a reduction of investments in education by governments due to the negative return on the education investments, which can further slowdown the development of a developing country (OECD 2017a, 199–200).

On the other side, there are some arguments in literature about the positive effects of emigration of highly skilled workers for the source country. The most common argument is that emigrants, in most cases, send remittances to their families in the source country which presents an influx of foreign capital (OECD 2017a, 186; Docquiert, Rapoport 2009). It is argued that remittances present a significant source of capital for developing countries, which would not have been obtained otherwise. In order to maximise the benefit from remittances, it is argued that their tax burden in the source country should be reduced (OECD 2017a, 202–203).²

Data gathered on the use and effect of remittances is not very comprehensive. According to existing data, the biggest part of remittances is used for everyday consumption, providing a positive effect on reducing poverty in the country of origin (OECD 2017a, 187; Kostić 2019a, 30). However, the data on the impact of remittances on investment and growth of the economy is unclear. While the positive effects on the economic growth of countries have been advocated for, there is no data that would support it. Available data shows that only a percentage of remittances is used for acquiring land or properties (OECD 2017a, 187) which can be perceived as investing the money acquired from remittances for capital investments. However, there is no data to support the claim that remittances are used for starting a business, an investment that would have a much more significant effect on economic growth. For these reasons, it does not seem that remittances compensate for the loss that the source a country incurs after the emigration of highly skilled workers, so that the net effect for the source country might be, at least, neutral.

Consequently, for a developing country wishing to foster development and growth, it is necessary to introduce adequate policies that will enable it to retain its own highly skilled individuals. However,

² For a different view on the taxation of remittances, which calls for increasing their tax burden, see especially Kostić (2019a, 30–39).

such a policy would have to base on an interdisciplinary approach, given the need to address many issues that encourage highly skilled individuals to emigrate.

1.2. Effects of Taxation on Migration

Providing better financial conditions, such as higher wages is one of the many incentives that could be used in incentivising highly skilled individuals to stay or return to the source country. Lowering tax burden and providing an individual with increased disposable net income can be seen as instrumental in providing better living and working conditions. A lower tax burden can attract highly skilled individuals with the promise of better quality of life. Concurrently, high tax burden can be seen as a disincentive for the highly skilled to migrate. In that regard, tax incentives can be seen as helpful – although not decisive – for fighting brain drain. Reducing the tax burden for the highly skilled, leaving them better off, can be a step in the right direction when it comes to their decision not to emigrate.

The effects of taxation on migration have been studied in literature, mostly concerning the question of the impact of taxes on a decision to migrate (see especially Wilson 2009; Liebig, Sousa-Poza 2005; Egger, Radulescu 2009; Halkyard 2013; Kleven *et al.* 2013; OECD 2017b). Literature suggests that taxation does play a role for migration of workers in a way that lower tax burden attracts more immigrants, especially the highly skilled ones (see especially Liebig, Sousa-Poza for intrastate migration).

According to the findings of Egger, Radulescu (2009, 1377), the progressivity of a tax system on high income brackets has the biggest impact on the decision to migrate, followed by the overall burden of personal income tax borne by an employee. Highly skilled emigrants are concerned with the net amount of their salaries, the amount that would remain for consumption after all taxes and contributions have been paid. If the net amount is not increased in the destination country, the incentive to emigrate is lowered.

As a result, the volume of tax incentives offered by both developed and developing countries for attracting foreign highly skilled workers does not come as a surprise (Burmann *et al.*, 2018; 42, Halkyard 2013, 23). The consequence is the war for talent as countries have to compete in offering better conditions if they wish to attract and retain highly skilled individuals. A long list of countries, such as Australia, China, Belgium, Denmark, Finland, Italy, Israel, the Netherlands, Poland, Portugal, Spain, Sweden, the United Kingdom and New Zealand offer tax incentives that are similar in nature. The most common tax incentives are

the tax exemption for foreign source income, a reduction in personal income tax and additional incentives for employee in R&D (CESifo 2012, 70).

However, most of the tax incentives that are offered are limited in time due to their revenue cost. In Denmark, the tax scheme for attracting highly skilled individuals had a positive effect on the number of highly skilled immigrants (see especially Kleven *et al.* 2013), but only in the short term. The scheme did not motivate immigrants to stay in Denmark in the long term. However, Halkyard (2013, 30–31) advocates for the use of tax incentives even if only short term positive effects are available, due to the sharing of knowledge and the more fluid social and economic environment that is created with the migration of highly skilled workers. This fact provides an additional argument for the use of tax incentives, but not diminishing the importance of creating other positive changes in a country.

2. FIGHTING BRAIN DRAIN IN SERBIA

2.1. Emigration from Serbia in Numbers

The Republic of Serbia is an emigration country (the Government of the Republic of Serbia 2017, 26). There are more people emigrating from the Republic of Serbia, than the ones immigrating to it, especially highly skilled individuals. The vast number of highly skilled workers and students, either after graduating or leaving for studies abroad, decide to leave Serbia in search of a better life. This has been confirmed by the study performed by Gallup (2017), in which it was calculated that Serbia ranks 30th among 152 countries according to the potential net migration index.³ Namely, the study looked at the number of people who would have emigrated had they been able to do so. According to the statistics, 27% of highly skilled workers and 46% of young people (between 15 and 29 years old) would have emigrated from Serbia had they had a chance.

Existence of a brain drain problem in Serbia has also been acknowledged by the European Commission in the Serbia 2019 Report, which states that brain drain remains an economic challenge for Serbia (European Commission 2019, 48). Even though the problem of emigration of highly skilled Serbian citizens to other countries is perceived as a problem in Serbia, there are no official national statistics that would provide precise data on the extent of the emigration. Institutions such as the Organisation for Economic Cooperation and Development (OECD), and Eurostat, the EU Statistical Office, provide some information about

³ For more information see Gallup 2017.

emigration from Serbia in their reports and analyses of international migration.

According to the statistics provided by the OECD (2015, 41) for the period 2000–2010, Serbia was among the countries with the highest increase in the emigration rate, for both the total number of emigrants (third place out of the top 15 countries) and for the number of highly skilled emigrants (fifth place, with only one European country, Moldova, having a higher rate). According to the more recent statistics, Albania and North Macedonia are among the European countries that have a higher percentage of emigrants than Serbia. The most recent data, provided by Eurostat, shows that 4,000 people per month emigrated from Serbia in 2018 and 2019, adding up to 51,000 people annually, and highlighting the extent of the emigration from Serbia.

Even though it does not represent a comprehensive study on the topic of emigration from Serbia, important information has been provided in the study Migration of Students, carried out by the two ministries in the Republic of Serbia (Cabinet of the Minister in Charge for Demography and Population Policy and the Ministry of Education, Science and Technological Development). In the study, 11,013 students in Serbia were interviewed on the subject of emigration. According to the findings, around 31% of all students interviewed plan on leaving Serbia after graduating (Migration of Students 2018, 24), with 50,6% of them not planning to return to Serbia (Migration of Students 2018, 46). The main reasons for emigration are of economic nature, and they concern the prospects of finding a job in the industry for which the students qualified, low salaries in Serbia for jobs for which they qualified and a general low standard of living (Migration of Students 2018, 42). The information that supports the findings of the Gallup study is that 90% of the students that want to emigrate have the full support of their parents (Migration of Students 2018, 41). A negative finding is that only 4.3% of students would change their decision to leave if an adequate loan or other financial help was provided. The main destination countries for emigration are Germany (24%), the United States of America (11.2%), Switzerland (10.7%) and Austria (8.7%).

The data obtained in the study shines some light for the prosperity of Serbia. According to the answers provided by the interviewed students, 51.6% of them would not emigrate if a job for which they qualified were available (Migration of Students 2018, 47). A number of them highlighted the need for better scientific institutes, more respect for every occupation, less corruption, better quality of life and hiring on the basis of merit and

⁴ Gallup 2017.

⁵ More information available at: https://ec.europa.eu/eurostat/data/database (last visited 20 September 2019).

not on the basis of political affiliation. These results show that there is space for improving the conditions for living and working in Serbia. However, it shows that the changes and the effort must involve all of society, including the government, different institutions, the private sector and citizens.

Another reason for working on providing better conditions can be found in the cost that highly educated people pose when they emigrate after finishing their studies in Serbia. Important information is provided in the study The Costs of Youth Emigration, carried out by the Institute for Development and Innovations, in cooperation with the Westminster Foundation for Democracy, in 2019. The study notes that it is estimated that during the 2012–2016 period, around 245,000 people emigrated from Serbia. Especially important findings of the study are the findings on the costs for the Republic of Serbia of each highly educated individual who emigrates. It is calculated that the cost for the budget of the Republic of Serbia of one highly educated person who emigrates amounts to around EUR 34,000, while the cost is increased up to around EUR 55,000 for each PhD student (Institute for Development and Innovations 2019, 21). Moreover, it is calculated that the aggregate cost of emigration of highly educated people for the budget of the Republic of Serbia is somewhere between EUR 960 million and 1.2 billion (Institute for Development and Innovations 2019, 23).

2.2. Proposed Tax Incentives

Targeting the right problems and providing adequate solutions is the path that the Government should follow. The results provided by the mentioned studies on emigration from Serbia could prove to be very helpful for the Government when designing incentives for retaining highly skilled individuals and attracting the return of the ones already abroad.

Tax incentives, as mentioned above, can have an impact on the decision to emigrate. Given that developed countries are offering tax incentives for highly skilled workers so as to attract them, the Republic of Serbia has started the work on introducing tax incentives that can help them decide to stay or to come back to Serbia.

In 2018 the Republic of Serbia introduced an entire set of tax incentives for the knowledge based economy which should help boost the growth of innovation in the Republic of Serbia. These incentives can be expected, at least indirectly, to provide better working and living conditions in Serbia through incentivising business development in Serbia. Tax incentives such as a double recognition of R&D costs, with salaries for individuals performing R&D activities benefiting from the

incentive, is expected to have a positive effect on the desirability of hiring highly educated individuals, and consequently on the increase in income of those individuals. Furthermore, an incentive for investing in newly established innovative companies (start-ups) is provided, as well as the more beneficial tax treatment of employee stock options plan. Another incentive offered is the IP Box which reduces the corporate income tax rate to about 3%, compared to the regular 15%, for income acquired from licensing of intellectual property rights that were developed in the Republic of Serbia.

Given the large number of people who emigrated from developing countries, attracting them to return is of great importance for the country of origin due to the fact that they bring back the financial, social and human capital (OECD 2017a, 192). One of the most important aspects of the financial capital that is brought back is the way that businesses are started upon return and investment in entrepreneurship. According to the studies, returning emigrants are more prone to starting their own businesses upon return (OECD 2017a, 193). For this reason, it can be expected that the newly introduced incentives for knowledge based economy will have at least an indirect effect on the desirability of living and working in Serbia which would also help economic growth.

The return of highly skilled workers can also have a positive effect for the source country through the sharing of knowledge and consequent development. After the return, repatriates can lead to an increase in the human capital stock in the source country due to the sharing of knowledge and skills that they are bringing back (OECD 2017a, 194). For that reason, an additional incentive has been proposed in this paper which aims to ease the conditions for investing in the skills and education of future employees by legal entities in order to support the growth of human capital in Serbia.

Acknowledging the importance of returning emigrants, as well as of attracting highly skilled individuals from other countries, the Government of the Republic of Serbia has proposed the introduction of a set of tax incentives by the end of the 2019, aimed at relaxing the financial conditions for returning highly skilled emigrants and reducing costs of hiring new employees.

2.2.1. Deduction of 70% of Salary Tax for New Residents

One of the proposed tax incentives is directed at providing a better financial position for the highly skilled workers returning or moving to Serbia. Moving to another country involves costs that have to be borne by the emigrant. In order to facilitate the process of moving, countries have started offering incentives that aim to reduce the moving costs or lowering the costs of living and working in the destination country.

Incentive proposed by the Government of the Republic of Serbia follows the logic of the '30% ruling' adopted in the Netherlands. Incentive offered in the Netherlands allows up to 30% reduction of the tax base for salary tax purposes as compensation for moving costs to the Netherlands. To benefit from the incentive, the taxpayer has to have been living outside of the Netherlands for no less than 16 months before starting to work in the country, and to possess specific expertise that is not available or only scarcely available in the Netherlands. Whether the specific expertise condition is fulfilled is proven by a minimum salary that repatriates must earn in the Netherlands, while such a limitation does not exist for individuals performing scientific research at a designated research institution. In the Netherlands, this incentive is offered for five years.⁶

Portugal also offers beneficial tax treatment to new residents. According to the incentive, new residents performing highly qualified activities in the field of science, technology and arts in Portugal, can benefit from a flat income tax rate of 20%, instead of the regular 40%. To benefit from the lower tax rate, the individual must not have been a tax resident of Portugal for five years before moving to the country.

Another interesting tax incentive is provided in Italy for inbound repatriates and foreign highly skilled workers, with the aim of putting a stop to brain drain.⁷ The incentive targets highly skilled individuals and should in turn have a positive effect on the development and progress of the country. This incentive has been offered since 2016 as a 50% reduction of taxable employment and self-employment income, while in 2017 the tax deduction was further increased, leaving only 30% of the employment income taxable.

To benefit from the Italian incentive, the highly skilled individual has to hold a degree, high qualification, specialization or to perform managing roles and to be employed or perform activities for an Italian resident company or a company related to it. Incentive is offered to highly skilled individuals from both EU member states and non-EU countries with which Italy has signed a double tax treaty or an information exchange agreement. Furthermore, the highly skilled worker must not have resided in Italy for five years before moving to the country, while planning to remain in Italy for at least two years after becoming Italian tax resident.

⁶ For more information about the '30% ruling' offered in the Netherlands see at: https://www.belastingdienst.nl/wps/wcm/connect/bldcontenten/belastingdienst/individuals/living_and_working/working_in_another_country_temporarily/you_are_coming_to_work_in_the_netherlands/30_facility_for_incoming_employees/ (last visited 23 September 2019).

⁷ Law No. 232. 2016. Bilancio di previsione dello Stato per l'anno finanziario 2017 e bilancio pluriennale per il triennio 2017–2019 [State budget for the 2017 financial year and multi-year budget for the 2017–2019 three-year period]. *Gazzetta Ufficiale Serie Generale* No. 297, 21 December 2016 – *Suppl. Ordinario* No. 57.

The right to use the tax incentive is allowed for five years, starting from the year in which the highly skilled individual became the tax resident of Italy.

Taking into consideration the abovementioned incentives and their effects, the incentive proposed by the Government of the Republic of Serbia can be said to follow their logic. Incentive is offered through the reduced tax burden for highly skilled individuals that are of great significance for the development of the economy.

The proposed tax incentive requires an amendment to the Law on Personal Income Tax⁸. According to the incentive, the regular salary tax base is reduced for 70%, leaving only 30% of salary income taxable for highly qualified workers. To qualify for the deduction, the highly skilled individual has to obtain a full employment contract for an indeterminate period. The right to use the benefit is limited to five years from the day of the signing of the employment contract.

To be regarded as a highly skilled individual, the individual has to occupy a position for which a specific professional education is required and for which there is a demand that cannot be easily satisfied in the domestic labour market in Serbia. In order to avoid complex definitions that would, most likely, unintentionally preclude individuals from some occupations to benefit from the measure, the solution used in the Netherlands has been adopted. Namely, the salary obtained by the individual is used as a criterion: if the salary of an individual is higher than the three average monthly salaries per employee in Serbia (in the case referred to in paragraph a) below), or higher than the two average monthly salaries in Serbia (in the case referred to in paragraph b) below), that individual will be regarded as a highly qualified individual.

Further explanation of a highly skilled individual is provided in the provisions. In order to benefit from the incentive, the individual who moves to Serbia should fulfil one of the two requirements that: a) 24 months before the day of signing of the contract with a qualified employer, they did not mainly reside in the territory of the Republic of Serbia, or b) 12 months before the signing of the contract with a qualified employer they mainly resided outside the territory of the Republic of Serbia for reasons of further education or advanced training, and at the moment of signing of the contract with a qualified employer, they are younger than 40 years old.

⁸ Zakon o porezu na dohodak građana [Law on Personal Income Tax], *Official Gazette of the RS*, 24/2001, 80/2002, 80/2002, 135/2004, 62/2006, 65/2006, 31/2009, 44/2009, 18/2010, 50/2011, 91/2011, 7/2012, 93/2012, 114/2012, 8/2013, 47/2013, 48/2013, 108/2013, 6/2014, 57/2014, 68/2014, 5/2015, 112/2015, 5/2016, 7/2017, 113/2017, 7/2018, 95/2018 and 4/2019.

As it can be noted, this incentive is open to nationals of all countries, and not just to Serbian citizens. However, individuals moving to Serbia have to be regarded as Serbian tax residents in accordance with the centre of personal and economic interest rule, and in accordance with all Serbian double tax treaties.

To benefit from the incentive, the highly skilled individual has to be employed with a qualified employer. A qualified employer is an employer who is a tax resident of the Republic of Serbia and who is not a related party to the previous employer of the highly skilled individual. Namely, the incentive is not extended to employers who are relocating employees to their related parties in Serbia, as it is assumed that they would have done so even without the incentive, should there be a reason for their relocation.

Nonetheless, there is an exception to this rule. Any employer will be regarded as a qualified employer in the case of a highly skilled individual who, in the 25 years preceding the year in which the contract is signed, has been a tax resident of the Republic of Serbia for at least three years (having a domicile or a centre of personal and economic interests in Serbia). Namely, if an employee was, in any period of their life, regarded as a tax resident of the Republic of Serbia for three years (not necessarily consecutive years), it will be possible to use the incentive even if the employer does not fulfil the abovementioned condition. The main objective of the incentive is to attract Serbian emigrants to return to Serbia, and with this exception, the incentive is open to those who once lived in Serbia, but who emigrated and worked for an employer who is regarded as a related party to the new employer in Serbia.

The issue of the relevant compulsory social security contributions regime is related to the deduction of the tax base for salary tax. In order to truly decrease the costs for employers and employees, it is proposed that the base for social security contributions be reduced by 70% for highly qualified employees. Consequently, the tax base for compulsory social security contributions would be reduced as well, and an incentive would be offered to both employees and employers through the reduction of salary tax and social security contributions.

2.2.2. Special Treatment of Capital Gains of New Residents

The second proposed incentive addresses the widespread international business practice of awarding employees with own securities for free or under preferential regime (Kostić 2019a, 54). The practice serves as a measure for uniting the interests of employers and employees which should provide a positive effect on the growth of the business. By sharing the ownership of the company with employees, employees are motivated to work harder in order to increase the value of the shares so

that they gain from it too, while the employer benefits from more loyal and productive employees.

Even though such practice is not widespread in the Republic of Serbia, by inviting highly skilled emigrants to return to Serbia, there is a need for adjusting the existing legal framework, as some of them might own securities that they do not wish to sell before moving back to Serbia. Moving to Serbia while owning the securities would make the securities subject to the capital gains tax in Serbia upon their sale, even though they were not acquired in Serbia, nor by a Serbian tax resident at the time of their acquisition.

According to Article 72 of the Law on Personal Income Tax, the taxpayer is required to a pay capital gains tax on the sale of securities that they have owned for less than 10 years. In that case, the tax base for capital gains tax is the difference between the purchase price (at the time of the acquisition) and the sale price, with a tax rate of 15% (Articles 72–77 of the Law on Personal Income Tax).

The newly proposed measure adjusts the purchase price with the aim of lowering the burden for capital gains tax for the sale of securities issued by a non-resident company that the taxpayer acquired for consideration. If, at the time of acquisition and six months following the acquisition of those securities, the taxpayer was not a tax resident of the Republic of Serbia, the purchase price will be the market price on the day when the taxpayer became the tax resident of the Republic of Serbia.

The step up in basis allows for taxation of only the part of the increased value of the capital after the taxpayer became a tax resident of the Republic of Serbia. It provides the taxpayer with a beneficial treatment as the difference between the purchase price at the time of the acquisition, and at the time of becoming a tax resident of the Republic of Serbia will be different, i.e. the price will be higher in the latter case, which would reduce the difference which is subject to the capital gains tax. As a consequence, the taxpayer will be subject to a lower tax burden, which should reduce the costs for moving to the Republic of Serbia.

2.2.3. Tax Exemption for Start-up Employees – Founders

In 2018, the Republic of Serbia introduced a set of tax incentives intended to foster the growth of the knowledge based economy. Special attention was paid to tax treatment of newly established companies that perform innovation activities, commonly known as start-ups. Start-ups are set up by a small number of individuals around an innovative business idea. However, a significant amount of capital is required to develop an innovative business idea, which represents the most common obstacle to the development of start-ups. Targeting the capital requirement, incentives were introduced for reducing the costs for start-ups and providing them

with the much needed capital. These incentives introduced the tax credit for investments in start-up capital, a double deduction of R&D costs and a reduced tax burden for royalties.

The Government has proposed an additional incentive for reducing the costs for start-ups even further. It is proposed that the tax burden for salaries of founders of a start-up is reduced. Start-ups would be exempted from paying the calculated and deducted salary tax for salary of founders which are employed in the start-up. The employee-founder has to sign an employment contract and to be registered for compulsory social security insurance. Further, the employee– founder has to own at least 5% of shares or stocks in the start-up. If there are multiple founders which are employed in the start-up, the benefit is provided for each one.

The tax exemption applies to monthly salaries up to 150,000 RSD. If the salary is higher, the tax exemption is provided only for the amount of 150,000 RSD, while the tax exemption does not apply for the remaining part of the salary. If one individual is founder of two or more start-ups, only one start-up has the right to use the benefit for that individual; the other start-ups cannot benefit from the tax exemption for the same individual.

The start-up has to be registered with the competent authority and it cannot be regarded as a related party to any legal person, in accordance with the Law on Corporate Income Tax. Also, it must not derive more than 30% of its total income from other entities that are regarded as related parties to any of its founders. The right to use the benefit is available for 36 months from the incorporation of the start-up, which cannot be later than the 31st December 2020.

This incentive cannot be cumulated with any other incentive available for hiring of the same individual, except for the incentives prescribed in the law governing compulsory social security insurance, even if they are prescribed as a subvention for employment or self-employment. Given that the Law on Personal Income Tax in Article 21dj already allows the tax exemption for salaries of founders of newly established companies, it was necessary to limit their mutual use. However, given that the tax exemption from Article 21dj is allowed only for salaries of employees-founders up to 37,000 RSD, and only for 12 months from the incorporation of the company, it is clear that the newly proposed incentive will be of greater benefit to start-ups.

Contributions for social security insurance impose a higher cost on employers and employees. In order for the tax incentive to be efficient in actually reducing the overall costs, it was necessary to provide a complementary incentive for contributions for social security insurance. For that reason, the tax incentive has been transposed in the field of the compulsory social security insurance. Namely, the incentive proposes to exempt the employer from paying social security contributions, on both

employer's and employee's accounts, for salaries of founders-employees up to 150,000 RSD. If the salary is higher, the part above the limit will not benefit from the exemption. The same conditions have to be fulfilled as for the salary tax incentive. While benefiting from the incentive, contributions for the compulsory social security for founders-employees will be payable from the budget of the Republic of Serbia for the lowest monthly base for contributions.

2.2.4. Tax Exemption for Salaries of New Employees

The proposed tax incentive for individuals moving to Serbia, which provides a 70% salary tax deduction, is open solely to taxpayers who are new tax residents of the Republic of Serbia or repatriates. Namely, current tax residents of the Republic of Serbia are excluded from the scope of that incentive. While not diminishing the value of attracting emigrants to return to Serbia, in order to provide overall better working and living conditions in Serbia, so that other individuals do not emigrate, an additional tax incentive is proposed.

Tax incentive introduces a beneficial tax treatment of salaries for newly employed individuals, mainly targeting younger citizens⁹ and individuals who were working, i.e. performing services, as entrepreneurs. An increasing number of entrepreneurs was noticed in Serbia, with individuals registering as entrepreneurs for providing services to their contractors, as it allowed them to avoid an employment relationship and the increased accompanying costs. The situation has led to the abuse of the entrepreneurship status, i.e. to the bogus self-employment¹⁰. The problem with the practice is that the relationship between the entrepreneur and the contractor is practically the one between an employer and an employee. Namely, in practice, entrepreneur does not act independently from the contractor in performing activities. The contractor is the one who undertakes all the risks and who organizes the work of the 'entrepreneur', and not, as it should be, the entrepreneurs themselves.

Bogus self-employment is not a phenomenon unique for Serbia; it is present in other countries, which started introducing measures for putting an end to it. The most famous decision in the area of bogus self-employment is the decision of the Court of Appeal of the United Kingdom in which the practice of hiring Uber drivers as entrepreneurs was seen as a disguise of an employment contract between Uber and its drivers. ¹¹

⁹ At the moment, unemployment rate of people between 15 and 24 years old in Serbia is high, 22.4%.

For more about bogus self-employment and the ways to tackle it see especially Collins, Horodnic 2017.

¹¹ The decision of the Court of Appeal of the United Kingdom available at: https://www.judiciary.uk/wp-content/uploads/2018/12/uber-bv-ors-v-aslam-ors-judgment-19.12.18.pdf (last visited 29 September 2019).

The current legislative framework in the Republic of Serbia allows individuals to register as entrepreneurs and pay less taxes and contributions than employees and their employers. For that reason, companies will gladly choose the role of a contractor rather than employer, and individuals will rather choose higher net disposable income than incurring higher costs for salary tax and accompanying contributions. Such a disproportionate status of contractors/employers and entrepreneurs/employees, pointed to the need of equalising their statuses and preventing the discrimination against employers and employees, at least for tax law purposes.

Equalising the position of employees and entrepreneurs who have abused the rules, is expected to have an additional positive impact through the increase in labour rights. New rights would become available to new employees (previously entrepreneurs), such as maternity leave, public health insurance (which is still better compared to the private health insurance which does not cover a vast number of more complex medical issues), severance pay, and other rights available to workers in an employment contract, but not to entrepreneurs.

Furthermore, providing a legal solution for bogus self-employment is expected to increase legal certainty in the tax system as well. Taxpayers would have all the necessary information to choose the legal form that best suits the activities that an individual wishes to perform. Taxpayers would know precisely when they are allowed to register as entrepreneurs and when they fall under the normal employee rules, as well as the tax consequences of both statuses.

For the mentioned reasons, a measure has been proposed by the Government which 'tests' the level of independency of registered entrepreneurs in providing services to their contractors. According to the proposed measure, if an entrepreneur 'fails' the test, their income will be regarded as 'other income' for the purposes of the Law on Personal Income Tax which induces an increased tax burden (the tax rate is 20% compared to regular 10%), as well as increased social security contributions.

Introduction of this measure would affect the disposable net income of former entrepreneurs (new employees), as well as the cost for their former contractors, i.e. new employers. In order to provide a transitional solution that would enable entrepreneurs and their contractors to get used to the new circumstances and the new test, the Government has proposed a measure that reduces their (increased) costs during a medium term period of three years.

The new tax incentive provides a tax exemption for employers from paying (in a certain percentage) the calculated and deducted salary tax for qualified individuals. The tax exemption is granted for 36 months,

and exempted percentages depend on the year in which the salary is payable. For the year of 2020, the amount is 70% of calculated and deducted tax, for 2021 that is 65%, and for 2022 it is 60%. By providing this type of incentive, employers are indirectly granted a subvention that would neutralise the latest increased costs.

The incentive is open only for employers who hire qualified individuals. A qualified individual is defined as a newly employed individual who was not registered as an insured employee during the period from 1 January to 31 December 2019, but which gained the status of a registered employee of the qualifying employer or any other employer during the period from 1 January to 30 April 2020. Namely, entrepreneurs were not registered as employees for the purposes of social insurance while providing services as entrepreneurs, which makes this measure available to them, as well as to other individuals who have found their first job.

The right to use the incentive is available to qualifying employers, which are defined as any an employer, legal or physical person, which during the period from 1 January 2020 to 31 December 2022 signs an employment contract with a qualified individual and registers the employee for compulsory social security contributions with the competent authority. Furthermore, the employer has the right to use the incentive if it increases the total number of employees after hiring a qualifying employee, compared to the number of employees on 31 December 2019. The right to use the benefit is also available to any employer which starts to conduct business activities after the 31 December 2019.

If the employer, while using the incentive, decreases the number of employees compared to the number it had on 31 December 2019 (increased for the number of qualified employees), the employer loses the right to use the incentive for an equivalent number of qualified employees. In the case of an employer who started performing business activities after 31 December 2019, the relevant date is 31 December of the year in which it started performing business activities.

This incentive cannot be cumulated with any other incentive offered for hiring individuals, except from the one offered in the law governing compulsory social security contributions, nor can it be used by public authorities.

In order to reduce the overall employment costs, and not just the salary tax, exemption from paying contributions for pension and disability insurance of qualified employees has been provided to employers. Exemption is provided to both employer's and employee's accounts, but only for pension and disability insurance contributions. Given that pension and disability insurance contributions amount to 70% of total social

security contributions, it can be argued that an important reduction has been provided. Additional reduction in contributions for health insurance is avoided as it would have had a negative effect, given that the reduced revenue of the Republic Fund for Health Insurance would have had to be compensated from the budget of the Republic of Serbia, which does not happen normally, unlike the case of pension insurance.

To benefit from the incentive, the same conditions as for the incentive regarding the salary tax must be fulfilled, while the exempted percentages differ. For salaries paid in the year of 2020, 100% of contributions for pension and disability insurance are exempted, for 2021 the figure is 95%, and for 2022 it is 85%. In that period, contributions for compulsory pension and disability insurance for employees are payable from the budget of the Republic of Serbia. This incentive cannot be cumulated with any other incentive (except the complementary incentive for salary tax) intended for hiring the same individuals.

2.2.5. Tax Exemption for Foreign Source Income of Non-residents

Rapid technological advancement has changed the way of doing business and performing services. The need to be in one place physically to perform work tasks does not stand in many cases. The Internet has enabled individuals to work from anywhere in the world using their personal computers, which has posed problems for controlling and taxing the income of those individuals.¹²

Serbia is one of the countries that find it difficult to track the income of persons performing activities while temporarily staying in Serbia, leading to non-taxation even though the right to tax exists. For that reason, it is seen as a good practice to formally exempt the income received by a non-resident individual from abroad, while temporarily staying in the territory of the Republic of Serbia. This measure would equalise the legal framework with the reality and increase tax certainty. Furthermore, it is believed that this measure would provide an incentive for foreign individuals to come and work in Serbia temporarily. Even though they would not pay income taxes, the benefit for the Republic of Serbia would be provided through the increased VAT revenue from their consumption in Serbia.

The proposed incentive introduces exemption from personal income tax for income obtained by a non-resident individual who spends a maximum of 90 days in the Republic of Serbia, in the period of 12 months starting or ending in the respective tax year. To benefit from the exemption, the work has to be performed for a non-resident contractor who does not perform main business activities or other activities in the Republic of Serbia.

¹² For more see: Kostić 2019b.

If the non-resident contractor performs activities in the Republic of Serbia, the income of a non-resident individual will be exempted if the service performed is not used for contractor's main business activity or other activity that it performs in the territory of the Republic of Serbia. Introducing the limitation that the non-resident principal does not perform business activity in the Republic of Serbia aims to safeguard the equal position of resident and non-resident entities by reducing potential for discrimination.

2.3. Proposal of an Additional Tax Incentive: Tax Incentive for Investing in Training and Education of Future Employees

An additional tax incentive is proposed in this paper, with the aim of supporting highly skilled persons remaining in the Republic of Serbia and attracting those who have left.

The incentives proposed by the Government (explained above) aim to reduce costs for employers and indirectly providing more disposable net income to employees. Having in mind that economic conditions are not the only one affecting the decision to emigrate, it is necessary to build an environment where the further development of skills and lifelong learning are encouraged. Employers should be encouraged to invest in further specialisation and education of their employees and future employees as it would provide them with more skilled workers. Employees and future employees, on the other hand, would benefit from the personal development, which would enable them to advance their careers, earn higher income and secure higher quality of life in Serbia, which would reduce the brain drain from the Republic of Serbia.

Incentives offered by employers to their employees as a way of tuition fee assistance have been proven to provide a positive return on investment. According to the study performed by the Lumina Foundation regarding the tuition fee assistance provided by a health care insurance provider to its employees in the United States of America, it was shown that the investment paid off with a 129% return to the employer, while the employees benefited from a 43% salary increase in the three following years (Lumina Foundation 2016, 9-10). Furthermore, a study by the OECD showed that government investment in skills is a sound investment, as every dollar invested is more than fully repaid by the increased future tax revenue (OECD 2017b). Having in mind these facts, there is an incentive for introducing similar measure into the legal framework of the Republic of Serbia as both employees (present and future) and employers benefit from the incentive: the former through the further specialisation and reduced salary tax, and the latter through reduced employment costs and better qualified personnel. Moreover, a gain for the Republic of Serbia would be obtained through the reduced brain drain that would allow its faster economic growth. Individuals would have a secure job in

the field for which they qualified, directly disincentivising them from emigrating.

The current legal framework of the Republic of Serbia does not recognise any specific incentive for investing in the education or specialisation of current and future employees. However, in line with the current provisions, an employer that invests in the work related education or specialisation of its current employees will be allowed to deduct the amount invested as an expenditure if the money was paid directly to the institution or organisation providing the course.^{13¹} Furthermore, the amount invested will not fall into the tax base for salary tax of the employee who benefits from the investment. If, however, the money was paid into the account of an employee, who would pay the course cost himself from the money obtained from the employer, the employer would not have the right to deduct the amount as an expenditure, and the amount obtained would fall into the tax base as an in-kind benefit for salary tax purposes, inducing an increase in the compulsory social security contributions. As a result, only employees whose course cost has been covered directly to the institution providing the course will be able to benefit from the employer's investment in their education or specialisation.

Investment in education or specialisation of individuals who are not current employees of a legal entity, such as students, would have to incur the personal income tax (for the part exceeding the legally prescribed monthly maximum of around EUR 100) according to Article 85 of the Law on Personal Income Tax, as 'other income', with a 20% tax rate. Furthermore, it is unclear whether the legal entity would be able to deduct the invested amount as an expense. It can be argued that if a contract in which the individual commits to working for the investor after completing the course is provided, the employer would have the right to deduct the invested amount. If there is no similar contract, there would be no sufficient relationship with the business activity, and the expenditure would not be permitted. We can notice that there is no actual incentive for legal entities to support the specialisation of individuals who could become an invaluable asset for the business activity.

The situation is not much better even if a legal entity wishes to invest in entities that are registered for providing education services. In that case, legal entity has the right to deduct the sum invested as expenditure, but the maximum deductible amount is capped at 5% of investor's total revenue. ¹⁴ It is noticeable that these rules do not provide

 $^{^{13}\,}$ Opinion of the Ministry of Finance of the Republic of Serbia, No. 414–00–88/2002–04.

¹⁴ Zakon o porezu na dobit pravnih lica [Law on Corporate Income Tax], *Official Gazette of the RS*, 25/2001, 80/2002, 80/2002, 43/2003, 84/2004, 18/2010, 101/2011, 119/2012, 47/2013, 108/2013, 68/2014, 142/2014, 91/2015, 112/2015, 113/2017 and 95/2018, Art. 15.

an actual incentive for companies to engage in direct or indirect investments in human capital.

The desirability of an incentive that would further support investments in human capital in Serbia is supported by the fact that only 16% of people in Serbia have acquired tertiary education (European Commission 2019, 30). Creating a knowledge based society and economy would be impossible without persons who would drive the development. This is even more so taking into consideration the rapid advances in technology that have already changed the way of doing business and introduced new required skills (Fitzpayne, Pollack 2018, 1; York 2019). Tuitions for further studies abroad can be especially very high and unattainable for students, but significantly beneficial due to the resources and opportunities offered by studying the latest achievements in the relevant field. Employers are in demand of such highly skilled workers and only by supporting their development can they both prosper.

The incentive proposed in this paper would require changes to the Law on Personal Income Tax and the Law on Corporate Income Tax.

It is proposed that if an employer covers the cost of a training or a course for a future employees, the total amount invested should be deductible from the employer's corporate income tax base as an expenditure, during the period in which the investment was made. Further, the benefit provided to the future employee would be exempted from personal income tax, i.e. the other tax, and it would not give rise to social security contributions. If the employer does not have enough taxable income from which it could deduct the whole amount invested, the right to carry over the remaining part of the investment would be allowed for up to five years.

To benefit from the incentive, it is necessary that the individual signs a contract with the future employer in which they would commit to working for the legal entity that covered the course cost, for at least two years. If the individual does not sign an employment agreement with the legal entity after the completion of the course, or terminates the employment voluntarily before the two years period has expired, they will be obliged to repay the total cost to the legal entity and to pay the personal income tax, i.e. the other tax, for the whole amount obtained.

The legal entity must be a tax resident of the Republic of Serbia in order to benefit from the incentive. Furthermore, the right of the individuals to apply for funding from their employer has to be available to all individuals who satisfy the requirements posed. The legal entity will decide whether it wishes to fund the specific course according to its business needs.

The qualifying training or education courses have to be work related, i.e. in the interest of the business activity performed by the legal

entity, otherwise, the incentive will not be applicable. The training or education course has to be related to the activities that the legal entity performs as its business activity. If the course is directed at obtaining personal gain for the employee but is not related to legal entity's business needs, the costs will not be deductible (such as obtaining a driving license if the job does not require it). Costs of obtaining another professional vocation (e.g. a secretary taking a course to become an accountant) will not fall into the category of qualified courses if it is of no use to the legal entity.

Costs that can be deducted are:

- the actual course cost (course fee),
- the costs of course materials such as specialised literature,
- daily allowance, if provided by the legal entity,
- travel costs, if provided by the legal entity,
- accommodation costs, if provided by the legal entity.

The maximum amount of the costs that the legal entity can deduct in one fiscal year is not capped. Namely, the incentive provides benefits for both parties, the legal entity and the individual, but it also provides a benefit for the Republic of Serbia. More highly educated and specialised workers will lead to the increase in the number of tertiary educated and highly specialised workers in Serbia, which would help economic development.

As specifically provided in Article 2 para. 4(3) of the Bylaw¹⁵ for applying the double deduction of R&D costs, according to Article 22g of the Law on Corporate Income Tax the costs for training of employees, and consequently of future employees, cannot be deducted in a double amount. This limitation provides a safeguard against eventual abuses of the incentive, while allowing the improvement of the current situation.

3. CONCLUSION

The importance of human capital, especially of the highly skilled individuals, has never been as important for the development of economies as it is in today's rapid technological advancements. Mobility of people, which has been constantly increasing, has speed up in the recent decades.

¹⁵ Pravilnik o uslovima i načinu ostvarivanja prava na priznavanje troškova koji su neposredno povezani sa istraživanjem i razvojem u poreskom bilansu u dvostruko uvećanom iznosu [Bylaw on the conditions and the method for exercising the right of recognition of expenditures directly related to research and development in the tax balance sheet in a double amount], *Official Gazette of the Republic of Serbia*, 50/2019.

People, especially the highly skilled ones, are more inclined to leave their home countries, homes and families in the search of a better life. There are many reasons why individuals leave their homes, but the most common one is the need for better economic conditions.

This trend has posed problems for countries which are losing the talents that could help them grow. Such a problem is especially grave for developing countries when they have to compete with developed countries for talents. However, developing countries have realised that they need to fight back and various incentives have been offered. One type of these incentives are tax incentives, as studies have showed that taxation plays a role in the decision for emigration.

The Republic of Serbia is the talent exporting country facing the problem of brain drain. According to the available statistics, the brain drain is not reducing, which demands a response from the government, in providing better living and working conditions. For that reason, the Government is at the moment proposing a set of tax incentives for reducing the tax burden of highly skilled workers. These incentives have been drafted following existing incentives in other countries, benefiting from their experiences. The incentives that are proposed are a 70% deduction of salary tax for new residents, the step up in basis for capital gains tax for new residents, the tax exemption for salaries of employees-founders of start-ups, the tax exemption for new employees and the tax exemption for the foreign source income of non-residents.

Furthermore, this paper proposes an additional tax incentive, which is proposed in order to promote investing in the education and skills of individuals, i.e. of future employees, willing to work on their professional development. Legal entities are encouraged to invest in the education and specialisation of future employees, by allowing them to deduct the entire amount invested in the training or education of future employees. Incentive is open for courses that are related to the business activity of the legal entity covering the course cost. This incentive is meant to increase the human capital stock in the Republic of Serbia, to increase the number of tertiary educated people who could lead the development of the country.

All of these proposed tax incentives should have a positive impact on the reduction of the brain drain in the Republic of Serbia, which is posing a threat to the country's development. Attracting and retaining highly skilled individuals, which will come at a cost in the form of reduced tax revenues in the short term, should help innovation and economy grow in the long term, through the activities of highly skilled individuals performed in the country. For that reason, steps undertaken by the Government of the Republic of Serbia can be seen as a step in the right direction for improving the country's prospects.

REFERENCES

- Arslan, Cansan, Jean-Christophe Dumont, Zovanga Kone, Yasser Moullan, Caglar Ozden, Christopher Parsons, Theodora Xenogiani. 2014. A New Profile of Migrants in the Aftermath of the Recent Economic Crisis. OECD Social, Employment and Migration Working Papers No. 160. Paris: OECD Publishing.
- Bosetti, Valentina, Cristina Cataneo, Elena Verdolini. 2015. Migration of skilled workers and innovation: A European Perspective. *Journal of International Economics* 96: 311–322.
- Brauner, Yariv. 2015. High-Skilled Migration: A Tax Perspective. Chapter 1 in *Taxation and Migration*, edited by Reuven Avi Yonah, Joel Slemrod. Leiden, The Netherlands: Wolters Kluwer.
- Burmann, Martina, Maria Hofbauer Perez, Vanessa Hoffmann, Carla Rhode, Sebastian Schworm. 2018. Highly Skilled Labour Migration in Europe. *ifo DICE Report* 1: 42–52.
- Cerna, Lucie. 2018. European High-Skilled Migration Policy. Trends and Challenges. Ch. 5 in *High-Skilled Migration: Drivers and Policies*, edited by Mathias Czaika. Oxford: Oxford Scholarship Online.
- CESifo. 2012. DICE Report 1/2012: 69-71.
- Collins, Williams C., Iona Alexandra Horodnic. 2/2017. Tackling bogus self-employment: Some lessons from Romania. *Journal of Developmental Entrepreneurship* 22. https://www.researchgate.net/publication/316184349_Tackling_bogus_self-employment_some_lessons_from_Romania (last visited 29 September 29 2019).
- Docquier, Frederic, Hillel Rapoport. 2009. Skilled Immigration: The Perspective of Developing Countries. 247–284 in *Skilled Migration Today: Prospects, Problems and Policies*, edited by Jagdish Baghwati, Gordon Hanson. New York: Oxford University Press, Inc.
- Egger, Peter, Doina Maria Radulescu. 9/2009. The Influence of Labour Taxes on the Migration of Skilled Workers. *The World Economy* 32: 1365–1379.
- European Commission. 2010. Europe 2020: A strategy for smart, sustainable and inclusive growth. Communication from the Commission COM(2010) 2020. Brussels.
- European Commission. 2019. Serbia 2019 Report. SWD(2019) 219 final. https://ec.europa.eu/neighbourhood-enlargement/sites/near/files/20190529-serbia-report.pdf (last visited 20 September 2019).
- Fassio, Claudio, Fabio Montobbio, Alessandra Venturini. 2019. Skilled Migration and Innovation in European Industries. *Research Policy* 48: 706–719.

- Fink, Carsten, Ernest Miguelez. 2017. The International Mobility of Talent and Innovation: New Evidence and Policy Implications. Introduction. 1–24 in *the International Mobility of Talent and Innovation*, edited by Carsten Fink, Ernest Miguelez. Cambridge: Cambridge University Press.
- Fitzpayne, Alastair, Ethan Pollack. 2018. Worker Training Tax Credit: Promoting Employer Investments in the Workforce. Washington, D.C.: The Aspen Institute.
- Gallup. 2017. Potential Net Migration Index. http://news.gallup.com/migration/interactive.aspx?g_source=link_newsv9&g_campaign=item_245204&g_medium=copy (last visited 20 September 2019).
- Halkyard, Andrew. 1/2013. Tax Incentives to encourage migration of skilled labour: another tax expenditure or a failure to tax residence? *eJournal of Tax Research* 11: 23–39.
- Hawthorne, Lesleyanne. 2018. Attracting and Retaining International Students as Skilled Migrants. Ch. 10 in *High-Skilled Migration: Drivers and Policies*, edited by Mathias Czaika. Oxford: Oxford Scholarship Online.
- Institute for Development and Innovations. 2019. The Costs of Youth Emigration. http://iri.rs/uploads/documents/troskovi-emigracije-mla dih.pdf (last visited 25 September 2019).
- Keery, Angela. 2017. Employee Incentive Structures That Work. *Accountancy Ireland* 65.
- Kleven, Henrik J., Camille Landais, Emmanuel Saez, Esben A. Schultz. 2013. Migration and Wage Effects of Taxing Top Earners: Evidence from the Foreigners' Tax Scheme in Denmark. Working paper 18885. Cambridge: National Bureau of Economic Research.
- Kostić, Svetislav V. 2019a. *Međunarodno oporezivanje u doba anksioznosti*. Belgrade: University of Belgrade Faculty of Law.
- Kostić, Svetislav V. 2/2019b. In Search of the Digital Nomad Rethinking the Taxation of Employment Income Under Tax Treaties. *World Tax Journal* 11.
- La Raine Ingram, Keisha. 2/2016. Attracting and Retaining Talented Professionals in the Baltic States. *Societal Studies* 8: 224–232.
- Liebig, Thomas, Alfonso Sousa-Poza. 2005. Taxation, Ethnic Ties and the Location Choice of Highly-Skilled Immigrants. OECD Social, Employment and Migration Working Papers No. 24. Paris: OECD Publishing.
- Lumina Foundation. 2016. Talent Investments Pay Off White Paper: Cigna Realizes Return on Investment from Tuition Benefits. https://www.luminafoundation.org/files/resources/talent-investments-pay-off-cigna-full.pdf (last visited 31 October 2019)

- OECD. 2011. Taxation and Employment. Paris: OECD Publishing.
- OECD. 2015. A New Profile of Migrants in the Aftermath of the Recent Economic Crisis. Paris: OECD Publishing.
- OECD. 2017a. Perspectives of Global Development, International Migration in a Shifting World. Paris: OECD Publishing.
- OECD. 2017b. Taxation and Skills. How Tax Systems Impact Skills Development in OECD Countries. Paris: OECD Publishing.
- OECD. 2019. Migration Data Brief. Paris: OECD Publishing.
- OECD. International Migration Outlook 2018. Paris: OECD Publishing.
- Romer, Paul. 1/2000. Should the Government Subsidize the Supply or Demand in the Market for Scientists and Engineers? *Innovation Policy and the Economy* 1: 221–252.
- Taube, Friedel. 2018. Germany's new immigration laws open door for skilled labor. *Deutsche Welle*. 2 October. *https://www.dw.com/en/germanys-new-immigration-laws-open-door-for-skilled-labor/a-45734442* (last visited 20 September 2019).
- The Government of the Republic of Serbia. 2017. Migration Profile of the Republic of Serbia for 2017. Belgrade.
- Cabinet of the Minister in Charge for Demography and Population Policy, the Ministry of Education, Science and Technological Development. 2018. Migration of Students. http://www.mdpp.gov.rs/doc/Migracije-studenata.pdf (last visited 29 September 2019).
- Wilson, John Douglas. 2009. Skilled Immigration: The Perspective of Developing Countries. 285–315 in *Skilled Migration Today: Prospects, Problems and Policies*, edited by Jagdish Baghwati, Gordon Hanson. New York: Oxford University Press, Inc.
- York, Erica. 2019. Tax Treatment of Worker Training. Fiscal Fact 644.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(100)

CERIF: S141

DOI: 10.5937/AnaliPFB1904200M

Nikolai Milogolov, PhD*

Azamat Berberov, MS**

BRAIN DRAIN AND TAX COMPETITION: DO WE NEED ANOTHER BEPS?

In the article we argue that the origins of the international tax base erosion in the corporate sector, which are the harmful tax competition for capital and old-fashioned international tax rules, are also relevant for the taxation of income of the high-skilled and mobile workforce. Therefore, a multilateral rethinking of the global tax architecture is proposed in order to conceptually address the problem properly and in a harmonized manner. We point out, based on the examples of Russia and Serbia, several problems of tax base erosion for mobile "talents", with a case study analysis of scenarios of "talent" migration involving sportspersons, researchers and IT specialists. Finally, we propose some ideas for global tax cooperation in order to mitigate the negative tax effects of brain drain based on adapting the existing recommendations of the BEPS Project for the cases of migrating individuals.

Key words: BEPS. – Brain drain. – Tax competition. – OECD Model Convention. – Tax residence.

^{*} ADIT, Principal Researcher, Tax Policy Center, the Financial Research Institute under the Ministry of Finance; Head of the Laboratory of Tax Policy Research, Russian Presidential Academy of National Economy and Public Administration (RANEPA), Moscow, milogolov@iep.ru.

^{**} Researcher (Macroeconomics and Finance), Gaidar Institute for Economic Policy; PhD Candidate, Financial University under the Government of the Russian Federation, Moscow, berberro@yandex.ru.

1. INTRODUCTION

Imagine you are playing Who Wants to Be a Millionaire? on Italian TV. The host asks you the last question (the answer to which can set you up for the rest of your life) – "What makes Italy lose about 14 billion euros annually, which equals approximately 1% of its GDP?" (ANSA Politics 2019) You are given four options, but you don't know the correct answer to the question. You turn to your lifelines: Ask the Audience, 50:50, Phone a Friend. Initially, you call your old friend, but on the end of a telephone they tell you that he moved to another country for a better life. Ok, you take another lifeline, 50:50, but the host apologizes and tells you that the show editor was recently offered higher salaries in the US, and therefore, he quit. Wha-a-at! You turn to Ask the Audience, but there is no one in the studio either – everyone has chosen to leave the country. Sounds like a nightmare for any trivia player!

You probably have already realized what is the correct answer to the question. It is *brain drain*, which does not have an unambiguous definition, but rather a list of definitions:

- emigration of educated or professional people from one country, economic sector, or area for another usually for better pay or living conditions;¹
- migration of health personnel in search of the better standard of living and quality of life, higher salaries, access to advanced technology and more stable political conditions in different places worldwide;²
- the situation in which large numbers of educated and very skilled people leave their own country to live and work in another one where pay and conditions are better.³

The modern world is characterized by a high degree of inequality in the context of the brain drain. We turn to one of the studies of the Gallup Institute (2018), in which the authors tried to determine how much the population of a give country would change if everyone willing to move to another country really moved to where they wanted (Potential Net Migration Index – PNMI). The results are quite shocking – for 109 out of 150 countries negative the values vary from –1% to –70%.

¹ See the definition of *brain drain* in the Merriam-Webster Dictionary. *https://www.merriam-webster.com/dictionary/brain%20drain* (last visited 25 September 2019).

² See Megan Ivy. Why brain drain hurts a developing nation? https://borgenproject. org/brain-drain-hurts-developing-nation/ (last visited 25 September 2019).

³ See the meaning of *brain drain* in the Cambridge English Dictionary. *https://dictionary.cambridge.org/dictionary/english/brain-drain* (last visited 25 September 2019).

However, we presume that the impact of tax burden on the decision to leave is insignificant. In particular, the comparison of the Gallup Institute (2018) data with the national tax rates (NTR) for the jurisdictions of some leaders and outsiders, according to KPMG (2019a), may also support this.

Table 1: Potential Net Migration Index (PNMI) and Nominal Tax Rates (NTR)

Country	PNMI Indicator	NTR Indicator
New Zealand	231%	33%
Singapore	225%	22%
Iceland	208%	46.24%
UAE	204%	0%
Switzerland	187%	40%
Nigeria	-46%	24%
Congo	-50%	35%
Syria	-44%	7%
Senegal	-34%	11%
Sierra Leone	-70%	5%

The results presented in the table show a slight correlation between the PNMI and the NTR. However, in our opinion, the problem of brain drain is still closely intertwined with tax issues both in the original country of residence of the individual and in his new jurisdiction. As Mohapatra (2012, 1) notes "the emigration of workers, especially high-skilled workers, is often perceived to create a fiscal loss — when considering the cost of educating these workers and foregone tax revenues for the home country." Global challenges require global solutions, therefore, this article is focused on the search and development of such global approaches to cooperation in the area of taxation, aimed at solving the abovementioned problem of brain drain and losses of tax revenues, and to streamlining the global approaches to taxation of income of mobile talented professionals.

2. BRAIN DRAIN IN RUSSIA AND SERBIA IN THE CONTEXT OF TAX RULES

2.1. Characteristic of the Brain Drain problem in Russia and Serbia

In addition to the common Slavic roots, the proximity of languages and cultures, the interweaving of historical destinies and, of course, the Orthodox faith, Russia and Serbia are united by the brain drain problem. This issue was recognized by the Serbian President Aleksandar Vučić as a "serious" inhibitory factor in the development of the Serbian economy (B92 2019). In particular, the results of the study conducted by the Institute for Development and Innovations claim that losses stemming form from emigration of Serbia's young people abroad cost the country 1.2 billion euros per year (about 2% of Serbian GDP), which is comparable to the amount of exports of IT services or agricultural products (NI 2019). IMF statistics say that around 50,000 people left Serbia in 2018 (RTS 2019). The results of a study conducted by RANEPA indicate a real increase in skilled emigration from Russia, although this is not about "the annual departure of millions, or even hundreds of thousands of people". According to the calculations, around 100 thousand Russians emigrate to developed countries every year, of which around 40% have higher education (RBC 2018).

Is the tax burden important when the Russian and Serbian "brains" decided whether to emigrate? An analysis conducted by Chernykh (2018) regarding drivers for relocating IT specialists abroad indicates that career factors ("decent wages", "interesting projects, career prospects") have crucial importance for leaking brains, but comfort factors ("climatic conditions", "good ecology") are also important. A tax motivation does not play a significant role in deciding to emigrate. The main motivations of the drain of Serbian "brains" are also classic – according to *RTS* (2015) – it is "high unemployment, low incomes, and insecurity".

2.2. Assessment of the Scale of Tax Losses from Brain Drain in Russia and Serbia

The results of the calculation of tax revenue losses from the brain drain are presented in Table 2. We use presume that emigrating workers earn an average salary at the highest level, so we calculated the potential personal income tax revenues from taxing such wages based on the statutory personal income tax rates of 12% and 13% for Serbia and Russia, respectively.

Index	Russian Federation	Republic of Serbia
Number of emigrants per year	~100,000 (<i>RBC</i> 2018)	~50,000 (<i>RTS</i> 2019)
Average salary (highest)	51,000 RUB (Federal State Statistics Service 2017)	104,000 RSD (Statistical Office 2018)
Amount of tax losses from the brain drain	~7.95 bln. RUB.	~ 6.24 bln. RSD
Amount of collected personal income tax	~ 3301 bln. RUB (Federal Tax Service 2018)	~ 122.9 bln. RSD (Ministry of Finance of Republic of Serbia 2019)
Share	0.24%	5.07%

Table 2. Tax revenue losses from brain drain

These calculations do not take into account indirect benefits for the donor country such as transfer of skills or remittances to family members of the emigrating individual; however, we suggest that even such simple estimation can indicate that the scale of the tax losses from the brain drain is much more significant in Serbia than in Russia. For the Russian Federation the indicator of net tax losses is not significant, not exceeding even 0.5% of personal income tax revenues, while for the Republic of Serbia the brain drain can even seriously affect the collection of personal income tax.

2.3. Comparison of Criteria for Tax Residence of Individuals in Russia and Serbia

Art. 207 of the Tax Code of the Russian Federation (hereinafter – the Tax Code) establishes the general rule for an individual to be considered to be tax resident of Russia, which is their actual physical presence in Russia for at least 183 calendar days of the calendar year⁴. Residents are subject to tax in Russia on their worldwide income and nonresidents are only taxed on incomes derived from the sources in Russia.

⁴ This criterion is not the only one. Among other, secondary criteria, one can single out the recognition by tax residents of the Russian Federation of Russian military personnel serving abroad, as well as employees of state authorities and local self-government bodies sent to work outside of the state, regardless of the actual time spent in the Russian Federation.

Serbian residents are also subject to tax in Serbia on their worldwide income and nonresidents are subject to tax on Serbian-source income only. However, the Russian approach is different from the Serbian in the criteria for determining the residence of individuals. The key difference is that in addition to the objective criteria ("quantitative-day" factor), the Serbian tax legislation also establishes a subjective criteria for the tax residents of individuals, which is having domicile, residence or center of business and life interests in Serbia.

"Individuals are considered to be resident for tax purposes if they have a domicile, residence or center of business and life interests in Serbia or if they spend more than 183 days within a 12-month period, which begins or ends in the tax year (i.e. the calendar year)" (EY 2018a).

In the context of brain drain we can suggest that a stronger personal tax nexus to jurisdiction, which we can see in the Serbian legislation, is probably more beneficial for the donor country because it will allow such country to continue taxing the income of emigrants or remote workers because they are still Serbian tax residents on a worldwide basis. However, as follows from the analysis below, such a taxation possibility can be limited by tax treaties. On the other hand, the simplicity of the Russian tax residence rules for individuals can potentially limit the potential tax base of the state, not allowing it to continue taxing the income of emigrants or remote workers. According to the Ministry of Finance of the Russian Federation (2019), there is currently an intensive discussion in Russia about reforming the individual tax residence criteria in the direction of introducing subjective criteria such as the center of vital and business interests and/or by reducing the required period of physical presence in the country to 90 days. Such reform is intended to tighten the personal income tax residence criteria in Russia and to broaden its tax base.

3. ANALYSIS OF SPECIFIC CASES OF BRAIN DRAIN

Two types of rules that allow countries to tax individual income can be distinguished in the framework of the current approaches to taxation of the income of mobile and qualified individuals. First, these are the rules characterizing the personal nexus of an individual to the country, i.e. the rules of tax residency. Second, these are the rules characterizing the territorial nexus of income, establishing the criteria for income received from sources in the given country. As mentioned above, the current instruments of tax coordination, in relation to the taxation of income of migrant individuals (system of bilateral double tax treaties), mainly focus on the problem of eliminating double taxation by resolving potential conflicts through special rules. Such conflicts, according to

Holmes (2007, 23–24) can be residence-residence, residence-source or source-source. So, tax treaties restrict the rights of countries to tax the income. Such limitations of countries' taxing rights are reflected in the respective bilateral double tax treaties.

We analyze cases of brain drain typical for Russia and Serbia on the examples of an IT specialist, an athlete and a researcher, in the context of the existing international tax architecture and describe possible directions for the development of international tax cooperation in the field

3.1. IT Specialists

3.1.1. Treaty Provisions Related to Income from Employment

The authors' analysis below relates not only to classical IT specialists (such as, for example, software developers) but also to the broader list of professions characterized as the remote workforce. In today's digital world, fewer and fewer professions require physical presence in the office. For example, this statement is almost 100% likely to be related to the activities of beauty-bloggers/vloggers/travel-bloggers, entrepreneurs, foreign language teachers, designers, writers, etc., because such professionals can work from anywhere in the world. This process has even led to the emergence of the concept of the "digital nomadism" and taxes are one of the main issues that both nomads and the state face, according to Kostic (2018, 191). Article 15 of the OECD Model Convention (OECD 2017a), Income from Employment, and Article 15 of the UN Model Convention (United Nations 2017), Dependent Personal Services, 5 set the rules for the elimination of double taxation of cross-border income of such a mobile workforce: 6

- 1. Subject to the provisions of Articles 16, 18 and 19, salaries, wages and other similar remuneration derived by a resident of a Contracting State in respect of an employment shall be taxable only in that State unless the employment is exercised in the other Contracting State. If the employment is so exercised, such remuneration as is derived therefrom may be taxed in that other State.
- 2. Notwithstanding the provisions of paragraph 1, remuneration derived by a resident of a Contracting State in respect of an employment

⁵ Hereinafter, the OECD Model (2017) and UN Model (2017) adopt the OECD and UN Model Tax Conventions. The following abbreviations are used when referring to the UN and OECD Model Conventions: OECD MTC, OECD MC, UN MTC, UN MC.

⁶ Article 15 (paragraphs 1–2) of the OECD Model Convention is given. In our opinion, paragraph 3, which, for example, separately regulates the taxation of income of aircraft crew members, does not apply to the current study.

exercised in the other Contracting State shall be taxable only in the first-mentioned State if:

- (a) The recipient is present in the other State for a period or periods not exceeding in the aggregate 183 days in any twelve-month period commencing or ending in the fiscal year concerned; and
- (b) The remuneration is paid by, or on behalf of, an employer who is not a resident of the other State; and
- (c) The remuneration is not borne by a permanent establishment or a fixed base which the employer has in the other State.

So, in other words the Article 15 of the OECD MC (OECD 2017a) cited above sets the exclusive rights of taxation of the employment income by the state of the employer's residence in two situations:

- if employment is exercised in the state of the employer's residence (Article 15(1));
- if the employment is not exercised in the state of the employer's residence, but all three conditions mentioned in Article 15(2) fulfill: (a) worker does not spent more than 183 days in the country of treaty partner; (b) income is not paid by the employer resident in the country of treaty partner; (c) income is not paid by the employer's PE or fixed base in the country of treaty partner.

In any other situations as prescribed by the Article 15(1) source country can tax such income and its taxing rights are not limited.

What is more, in some situations income of the remote workers can potentially fall in the scope of the Article 14 of the UN Model Convention (United Nations 2017), which covers income from "professional services and other activities of an independent character". The fate of this article is very dramatic. Although it has been removed from the OECD Model Convention for various reasons and is not affected by the Base Erosion and Profit Shifting (BEPS) Multilateral Convention, it is present in various forms in the actual tax treaties including both Russian and Serbian double tax treaties. Here is this article from the UN Model Convention (United Nations 2017):

Article 14 (2) of the OECD MTC (2017a) does not define what "professional services" are, but the commentaries to it note that they "particularly" include "independent scientific, literary, artistic, educational or teaching activities as well as the independent activities of physicians, lawyers, engineers, architects, dentists and accountants." The commentary to the article adds that this list is not exhaustive and that any difficulties in applying this paragraph may be resolved through a mutual agreement procedure.

⁸ In fact, one key reason is that Article 14 of the UN Model Convention (2017) may well be covered by Article 7 Business profits.

⁹ See, for example, double tax treaties that were concluded between the Russian Federation and Germany (1996), or between Bulgaria and Republic of Serbia (1998).

- 1. Income derived by a resident of a Contracting State in respect of professional services or other activities of an independent character shall be taxable only in that State except in the following circumstances, when such income may also be taxed in the other Contracting State:
 - (a) If he has a fixed base regularly available to him in the other Contracting State for the purpose of performing his activities; in that case, only so much of the income as is attributable to that fixed base may be taxed in that other Contracting State; or
 - (b) If his stay in the other Contracting State is for a period or periods amounting to or exceeding in the aggregate 183 days in any twelve-month period commencing or ending in the fiscal year concerned; in that case, only so much of the income as is derived from his activities performed in that other State may be taxed in that other State.

In the other words, Article 14 of the UN MC (UN 2017), cited above, sets the exclusive rights of taxation of the country of residence of the provider of professional services or other activities of an independent character unless there is either (1) a fixed base of such a service provider in the treaty partner state, or (2) time of stay of such a service provider in the treaty partner state is more than 183 days during a calendar year.

3.1.2. Offshorization or Emigration of the Workforce: Is Serbian Tax Base More Protected Against Erosion Than the Russian?

3.1.2.1. Offshorization of the Workforce

Let us consider the following case of the offshorization of the workforce. A Russian citizen and its tax resident, highly-qualified IT specialist, Evgeniy Kaspersky, decided to become a "digital nomad" starting from 1 April 2019. In April 2019, Evgeniy decided to leave Russia and settle in Cyprus, from where he continued performing his job duties for his Russian employer until the end of 2019. Such offshorization can be potentially motivated by corporate tax planning reasons. For example, the management of the Russian-based IT corporation can decide to locate at its subsidiary in Cyprus the development, enhancement, maintenance, protection and exploitation (DEMPE) functions, in relation to important intangibles. This can be done in order to bridge the gap between the corporate and economic structure of the multinational entity (MNE) group and to comply with the post-BEPS transfer pricing requirements (OECD 2015a, 13).

By applying the double tax treaty between Russia and Cyprus¹⁰ to this case we can make observation that Russian taxing rights for such income from employment will be limited under Article 15 of Russia-Cyprus double tax treaty, which is based on Article 15 of the OECD Model, with exception of paragraphs (3) and (4) which relate to building sites, construction, assembly, or installation, to journalists and to employment exercised onboard ships. Therefore, Russia cannot tax such income for two main reasons: first, Evgeny ceases to be Russian tax resident, so both under Article 15(1) and 15(2) Russia cannot tax his employment income as the residence state; second, the employment is physically exercised in Cyprus and not in Russia, so under Article 15(1) Russia cannot tax such income as a source state either.

Let us assume that his Serbian colleague Borislay Diordiević made the same actions – moved to Cyprus and started working remotely for his home-country employer, from February to the end of the calendar year. It should be emphasized that in this situation the rights of the donor country to tax the income from employment are not limited by a tax treaty in case of such a remote worker who still has "vital interests" in the Republic of Serbia (family members and a house in Belgrade) and therefore remains a tax resident of Serbia. We shall also take into account that the residenceresidence conflict can be potentially resolved under Article 4(2)¹¹ in favor of Serbia because such a person can have permanent homes available in both states, and personal ties with Serbia are stronger. Serbia can tax such income as the state of residence even if the employment is physically exercised in Cyprus and not in Serbia. 12 Cyprus also can tax such income from employment because timing condition from Article 15(2(a)) ensuring the exclusive right of the residence state to tax income from employment exercised in the other state is not fulfilled in this case. However, if such income was taxed in Cyprus it is exempt in Serbia, under Article 23(1(a)), 13 because this article uses the exemption method for the elimination of double taxation.

At the same time, if such a remote worker were to cease to be a Serbian tax resident, for example, if he sells house in Belgrade and his family moves to Cyprus, then the rights of Serbia to tax such income would be limited under Article 15 of the tax treaty for the same reasons

Agreement between the Government of the Republic of Cyprus and the Government of the Russian Federation for the avoidance of double taxation with respect to taxes on income and on capital (1998)

¹¹ Convention between the Republic of Cyprus and the Socialist Federative Republic of Yugoslavia for the avoidance of double taxation with respect to taxes on income and on capital (1985)

¹² Ibid.

¹³ Ibid.

as in Russian case above – the employment is not physically exercised in Serbia.

So, we suggest that the Serbian tax base is more protected from the erosion resulting from such offshorization of its workforce than the Russian tax base. However, if such income of the remote employer is taxed in the recipient country, the results are the same for the both states – they lose tax revenue because the recipient country has a priority right to tax such income.

3.1.2.2. Emigration of the Workforce

Let us consider the case of the emigration of the skilled workforce. A Serbian citizen and its tax resident, highly-qualified IT specialist Stefan Ibrahimović, decided to work for UK company starting on 1 April 2019. In April 2019, Stefan leaved Belgrade and settled in London, from where he continued performing his job duties for his UK employer until the end of 2019. Such emigration can be potentially motivated with classical combination of the reasons: quality of life, better career prospects, higher salary, etc. Let us also look at tax consequences of the same scenario performed by his Russian friend Mahomed Burkhanov, who migrated from Moscow to London to work for the same UK company in March 2019 and continued working their until the end of the year.

In this situation, the results of application of the double tax treaty¹⁴ in the case of the Russian emigrant are the same as in the case above for the offshorization of the workforce – Russia loses its taxing rights and its tax base. For Serbian emigrant, the results are also the same as in the case above.¹⁵ Serbia can tax employment income of its resident as long as it considers the emigrant to have strong personal nexus with Serbia. Such nexus can be evidenced by the home available to him in Serbia and the center of business and life interests in Serbia. So, potential residence-residence conflict will be resolved in favor of Serbia under Article 4(2).¹⁶ The only difference from the previous scenario is that if the UK also taxed such employment income, such tax would be credited against the Serbian tax, but not exempted as was the case in the previous example, because the Yugoslavia-UK double tax treaty (DTT)¹⁷ contains the credit method for the elimination of double taxation.

Convention between the Government of the United Kingdom of Great Britain and Northern Ireland and the Government of the Russian Federation for the avoidance of double taxation and the prevention of fiscal evasion with respect to taxes on income and capital gains (1994)

Convention between the United Kingdom of Great Britain and Northern Ireland and the Socialist Federative Republic of Yugoslavia for the avoidance of double taxation with respect to taxes on income (1981)

¹⁶ Ibid.

¹⁷ *Ibid*.

3.1.3. Interim Conclusion

We can conclude from the analysis above that economically the tax consequences of the migration and offshorization of the workforce from Russia and Serbia are almost the same – the donor countries lose their personal income tax base (taxing employment income) and give the recipient countries the priority rights to tax such income. Such tax policy outcomes can potentially result in the losses of tax revenues from personal income tax by the countries suffering from the problem of the brain drain.

3.2. Athletes

The Russian Federation is known all over the world for its athletes. For example, at the 2016 Summer Olympics, the Russian national team was at the 4th place and the Russian national football team reached the quarter-final at the 2018 FIFA World Cup. Serbian representatives of sports are also very famous all over the world. These are the names of Novak Djoković, Ana Ivanović, Jelena Janković, Vlade Divac, Branislav Ivanović, Sinisa Mihajlović and others. As experts note, in individual sports (although, in our opinion, this is also possible in team sports), the change of tax residence for the purpose of tax planning is a very common practice. For example, Gatto (2017) suggest that as of 2017 such well-known tennis players as Novak Djoković, Caroline Wozniacki, Marin Čilić, Petra Kvitová are residents of Monaco, Borna Ćorić, Karen Khachanov and Svetlana Kuznetsova are UAE residents. Each of these countries does not levy taxes on the income of individuals and has very flexible tax residence requirements.

Such tax policy practices can be analyzed in the context of criteria developed in the OECD's work on harmful tax competition (OECD 2015b) because they can potentially lead to the losses of tax base by the other countries which have strong personal nexus with such individuals and can be potentially described as "tax residence for sale". Such practices erode the foundation of tax law because they in fact regard legal rights and tax sovereignty – the concepts that cannot by definition have market value – as market commodities and shift the line between those residents who get residence status by buying it or investing in the country's economy and other persons who get this status resulting from personal inherent nexus with the state (Thirion, Scherrer 2018).

Prevention of double taxation of cross-border income of "entertainers and sportspersons" is regulated by Article 17 of the OECD Model Convention (OECD 2017a), which contains two paragraphs. The first one states that "notwithstanding the provisions of Article 15, income derived by a resident of a Contracting state as an entertainer, such as a theatre, motion picture, radio or television artiste, or a musician, or as a

sportsperson, from that resident's personal activities as such exercised in the other Contracting State, may be taxed in that other State." The second paragraph, states that "where income in respect of personal activities exercised by an entertainer or a sportsperson acting as such accrues not to the entertainer or sportsperson but to another person, that income may, notwithstanding the provisions of Article 15, be taxed in the Contracting State in which the activities of the entertainer or sportsperson are exercised."

However, what income and what share of it may be taxed by the state hosting the sporting event under Article 17(1) of the OECD MTC? The answer to this question is contained in the commentary to Article 17 of the OECD MTC (2017a) relating to the taxation of "entertainers and sportspersons". Paragraph 8 of the commentary states the following: "Paragraph 1 applies to income derived directly and indirectly from a performance by an individual entertainer or sportsperson. In some cases the income will not be paid to the individual, or his impresario or agent, directly with respect to a specific performance. For instance, a member of an orchestra may be paid a salary rather than receive payment for each separate performance: a Contracting State where a performance takes place is entitled, under paragraph 1, to tax the proportion of the musician's salary which corresponds to such a performance." Although this passage mainly refers to artists, it is safe to assume that the same approach can be applied to the salaries of athletes.

As the commentators further note, entertainers and athletes often earn royalties, sponsorships or advertising fees. Where there is a close relationship between the income received and the activities carried out in the country, the host State is also entitled to tax the income. The determination of the level of such relationship can be made on the basis of an analysis of the "timing of the income-generating event," the "nature of the consideration for the payment," and the "contractual arrangements" for participation in such events. Finally, the provisions of Article 17 may also apply to the taxation of remuneration for the time spent on "rehearsal, education or similar training" of athletes in the receiving State.

The distribution of the taxing rights of athletes' income between two countries that have a tax treaty depends on the jurisdiction in which the athlete is a tax resident, as well as on the state in which the sports activities are carried out. Based on this idea, we suggest the borderline scenarios where the donor state loses its tax revenues resulting from athletes' brain drain.

The state drastically loses the right to tax the athlete's income if:

- the athlete ceases to be a tax resident of this State;
- the athlete is not a participant in events taking place in this State.

In our opinion, this risk is more significant for the tax system of the Russian Federation, which uses only a simple "quantitative and daily" approach to determining whether an individual is a tax resident of the state.

3.2.1. "G" Case: How to Lose the Right to Tax the Income of a "Draining Feet"?

Let's consider the relationship between Russian tax legislation and tax treaties in the context of brain drain on an example from professional football. Let us assume that at the end of July 2018, Russian midfielder of FC SKA (Moscow), Alexander G., joined FC Amateur from the Principality of Monaco. Prior to joining the foreign club, Alexander was a tax resident of Russia. Most of the player's matches and training process took place in the Russian Federation. Therefore, it can be stated that regular income from the player's sporting activities was taxed in the Russian Federation, at a rate of 13%.

Let us assume that at the end of August 2018 Alexander G. received his first income from a foreign football club. Probably, the athlete is not yet a tax resident of Monaco, as he has not spent enough time in the Principality. On the other hand, the athlete is likely to be recognized as a tax resident of the Russian Federation at the end of 2018, as he has spent more than 183 days in the country that year. Thus, our hero will have to calculate and pay personal income tax at the rate of 13% on his own, as well as submit a personal income tax return to the tax authority where he resides, no later than April 30 of the following year.

Due to the large number of matches in the French championship (FC Amateur is a rare example of a football club that plays in a foreign league), the athlete begins to visit the Russian Federation less and less often. Let's assume that for the whole year of 2019 the period of his stay in the state was 45 days (vacation and a stay in the national football team). Then, with a high degree of probability, we can conclude that, for the fiscal year 2019 Alexander G. will cease to be a tax resident of the Russian Federation, and so Russia will lose the right to tax his income. There is an example of "draining foot". Additionally, if the athlete becomes a tax resident of Monaco, his income will be taxed at a zero rate.

3.2.2. "Branislav" Case: How Can the Russian Federation Tax the Foreign Player's Income?

As we see from the example above, Russia can easily lose the tax revenue from the taxation of income of its talented sportspersons if they cease to be Russian tax residents. In this section we analyze possible scenarios of taxing a foreign footballer's income in Russia.

The most obvious situation is if any foreign athlete become a tax resident of the Russian Federation. This is very likely scenario, because Russia is not only exporter but also an importer of the football talents. Many of the world stars have been playing for a long time in the Premier League: for example, Branislav Ivanović and André Schürrle.

The second scenario is of a foreign club to arrive to Russia for a tournament match. As mentioned above, the host country may tax the relevant part of the salary, as well as royalties, sponsorships or advertising fees, which are closely related to the activities in the respective country. For the situation in question, such income may include the remuneration for the interview given by the player during the tournament, or the fee for the use of their photo on the posters inviting fans to the match. Let's assume that attacking player of the Italian football club FC Sardinia Bronislav Stanley arrived with his team to play a match against Moscowbased FC Berezovsky. Bronislav is obviously not a tax resident of the Russian Federation, so his income will be taxed at a higher rate of 30%, rather than 13%, which can potentially result in an additional increase in tax revenue for Russia. However, in our opinion, this is the end of the positive aspects of the current mechanism of taxation of athletes arriving solely for sporting events. First, the mechanism of tax collection from a foreign athlete is absolutely incomprehensible. 18 Secondly, the probability of a particular club coming to Russia is not the highest. Third, it is unlikely that the potential amounts can be comparable to the full taxation of the income of an athlete with the status of a tax resident of the Russian Federation.

As noted above, under Serbian national legislation, an individual may also be recognized as a tax resident on the basis of subjective criteria. Probably such an approach protects the tax base of the Balkan country from eroding, but only with the limited extent. Let us turn once again to the case study of tennis player Novak Djoković, in particular to one of his more recent interviews (Gatto 2018) in which he said that the choice in favor of Monaco was made due to the "beauty" of the Principality. He stated that he has a house in which his wife and children live, as well as his current and former coaches in this beautiful country – the presence of such conditions allows Novak Djoković to "fully focus on tennis." If we assume that Novak can be considered a tax resident of Serbia, we would probably be wrong. First, the tennis player does not meet the "quantitative and daily" criterion of at least 183 days of the physical presence in Serbia (which is likely, considering the traveling lifestyle of the athlete). The

[&]quot;Dear Lionel Messi! You have to register for tax purposes in Serbia, because you participated in a friendly match between the Serbia and Argentina. We do not know what part of your income we will tax, but we are still waiting for you with a full set of documents at 5 Save Maskovica Street, Belgrade". In our opinion, this situation absolutely fantastic!

analysis of the subjective criteria also shows that center of his vital interest is more likely in Monaco because his home and family live their indicating his strong personal ties to the Principality.

3.2.3. Athletes in the Digital Age: The Case of Non-Taxation

In the context of the challenges of the digitalization of the economy and brain drain, another potential problem should be mentioned, which is at the intersection of sport and entertainment. Let us turn to one piece of recent news (Grace 2019):

"Electronic violinist Lindsey Stirling is putting on a new kind of interactive virtual concert, performing live to fans in avatar form. The concert, put on in collaboration with streaming platform Wave, will take place at 3 p.m. (EST) on Monday 26 August. Stirling will perform through her avatar, powered by art body motion and face capture technology. Fans will also be able created their own avatars and attend the virtual show by downloading the Wave virtual reality (VR) app, supported by HTC Vive and Oculus Rift. Throughout the concert, Stirling will interact with fans 'in a variety of direct, mysterious and surpris[ing] ways.' Limited edition concert merchandise will be available to buy."

Based on the content of this news we imagine the following case. Suppose that we have two young athletes – a Russian citizen, resident of the Russian Federation, Anton, and a citizen of the Republic of Serbia and its tax resident, Branka. Both are recognized ice skating masters. In 2019 our heroes decided to end their professional sport careers and form a duo to perform colorful ice shows. While devising their marketing strategy, the young athletes turned to the fast-evolving "augmented reality" technology, which allows them to create unique special effects for the audience. This choice also allowed new entrepreneurs not to be physically present in other countries, which greatly facilitated the working conditions for both themselves and the support team.

Based on this logic, the athletes relocated to the small jurisdiction Y, with which both Russia and Serbia have double tax treaties in force. The legislation of state Y offers low rates of personal income tax, as well as simple criteria for determining the residence of individuals (stay in the state Y for 90 days during a calendar year and a "permanent home available"). Finally, country Y is characterized by the availability of cheap and qualified labor and a large number of free ice rinks. In order to replenish the seed capital and to buy real estate in State Y, both Branka and Anton sold their apartments in Belgrade and in Moscow, respectively. Thus, at the beginning of January 2019, our heroes left for jurisdiction Y to settle all the legal issues and practice the "virtual" show. In October

2019, the production was triumphantly launched and broadcast to a large number of viewers, including in the Russian Federation and Serbia. The project is monetized through selling the online advertisement space during the online broadcasting aimed at target groups from Russia and Serbia.

What "piece of the tax pie" can Russia and Serbia claim? Admittedly, it is a small one or even none. It is likely that in 2019 Anton will not be recognized as a tax resident of the Russian Federation and Branka will not be recognized as a tax resident of Serbia due to the "quantitative factor" and the sale of real estate in Belgrade. Physically, our young "athletes" did not travel to the two countries for performances, which severely limits the possibilities for taxation although the fact that the "virtual" value was created in the source countries is obvious.

3.2.4. Interim Conclusion

We can conclude from the analysis above that generally tax treaties do not limit countries' rights to tax the income of the athletes while the athletes are their tax residents. If an athlete ceases to be a tax resident of the country, which is a common scenario in the context of brain drain, then the donor country loses its right to tax the income of such athlete until he physically arrives and take part in sporting events in the country.

The change of tax residence can be also accompanied by harmful tax competition between the jurisdictions, ultimately leading to so-called "tax residence shopping", which is especially relevant for individual star sportspersons who earn very huge amounts of income. If we consider that such sportspersons are often born, raised and trained in developing countries, where they took their first steps as a professional sportspersons, we come to the conclusion that the non-taxed income of such sportspersons is in fact the tax base of such donor countries lost due to brain drain.

Furthermore, our analysis shows that the rule in Article 17(1) allowing the host country to tax part of athlete's income from sports events in this country is impractical and generally useless, because the administrative costs of the implementation of the taxation mechanism for such income and allocation of the appropriate tax base can outweigh the tax benefits received; this rule is also not in line with the process of the digitalization of entertainment and sports content.

3.3. Academics and Scientific Researchers

Besides its achievements in the area of professional sports, the Russian Federation is famous throughout the world for its activities in the field of scientific research. For example, according to the rating of publishing activity by Scimago Journal & Country Rank (2018)¹⁹, the

¹⁹ See: Scimago Journal & Country Rank. https://www.scimagojr.com/countryrank.php?year=2018 (last visited 6 October 2019).

Russian Federation ranks 11th, and according with the rating QS World University Rankings (2019)²⁰ there are currently 27 Russian universities in the world top-1000 list. The rankings of the Republic of Serbia on the same lists are following: the country holds the 54th place in the first nomination and only one Serbian university – the University of Belgrade – is in the list of top-1000.²¹

How do these brain drain statistics in the field of research and education relate to the countries' fiscal interests? The basic tax policy idea can be to expand tax jurisdiction to the widest extent possible in order to try on the one hand to secure the right to tax the income of researchers leaving the country and on the other hand to tax foreigners arriving in the country. The balance between these two tax policy goals can depend on the net balance of export and import in the areas of research and higher education. However, tax policymakers can have also other considerations and use other taxing approaches towards income of researchers. For example, the positive social and economic impact of research and education on the wider society can be considered, therefore countries can exempt from taxation the income of researchers, such as research grants.

3.3.1. Taxation of Academic Researchers' Income Under the OECD and UN Models

Both the OECD MC and the UN MC do not contain separate articles addressing the issues of double taxation of the income of academic researchers (professors, lecturers, etc.). However, we cannot claim that this issue was historically ignored by the academic community. It is presumed that this issue is covered by the provisions of articles 14 (in case of providing independent personal services), 15 (dependent personal services), 19 (if remuneration is paid by the one of the negotiating states) in case of the UN MC (United Nations 2017, 452). However, "it was noted that articles 14 and 15 commonly did not exempt a visiting teacher's compensation from taxation at source because they generally allowed source taxation of service performers who were present in the host country for more than 183 days, and many teaching assignments exceeded that period of time" (United Nations 2017, 452).

At the end of the 20th century experts proposed to amend the UN MC and to add a separate article covering the issues of double taxation of income of "visiting teachers". However, there was no consensus regarding such provisions, so the discussion ended in the compromise decision: not to make amendments into the Convention but to amend the Commentaries

²⁰ See: QS World Universities Rankings. https://www.topuniversities.com/university-rankings/world-university-rankings/2019 (last visited 6 October 2019).

²¹ Ihid.

to the Convention, explaining that double tax treaties can potentially have such special articles and some guidance regarding the contents of such articles in case of having bilateral discussion about the issue (United Nations 2017, 453):

- The purpose of a tax treaty generally is to avoid double taxation, and double exemption of teachers is not desirable,
- It is advisable to limit benefits for visits with a maximum duration (normally two years), and the time limit should be subject to expansion in individual cases by mutual agreement between the competent authorities of the Contracting States,
- Whether the benefits should be limited to teaching services performed at certain institutions "recognized" by the Contracting States where the services are performed,
- Whether, in the case of visiting professors and other teachers who also do research, to limit benefits remuneration for research performed in the public (vs. private) interest,
- Whether an individual may be entitled to the benefits of the article more than once.

If we observe the real networks of bilateral tax treaties of the analyzed countries we can say that the draft provisions proposed by the OECD MC (OECD 2017a) are not accepted as a rule by both the Russian Federation and the Republic of Serbia. There are plenty of double tax treaties of which Russia is a part where taxation of scientific researchers is regulated by a separate article or an article similar to the article contained in the UN MC (United Nations 2017). One example is the double tax treaty between Serbia and Bulgaria²² where the issue of double taxation is regulated by Article 21 Professors and Scientific Researchers, which contains the following provisions:

- 1. An individual who visits a Contracting State for the purpose of teaching or carrying out research at a university, college, school or other recognized educational institution in that State and who is or was immediately before that visit a resident of the other Contracting State, shall be exempt from taxation in the first-mentioned State on remuneration for such teaching or research for a period not exceeding two years from the date of his first visit for that purpose, provided that such remuneration is derived by him from outside that State.
- 2. The provisions of paragraph 1 of this Article shall not apply to income from research if such research is undertaken not in the public

²² See: Convention between the Government of the Republic of Bulgaria and the Federal Government of the Federal Republic of Yugoslavia for the avoidance of double taxation with respect to taxes on income and on capital (2000).

interest but primarily for the private benefit of a specific person or persons.

The provisions of Article 21 of Serbia-Bulgaria DTT precludes the recipient state from taxing the income received by the newly-arrived professor or scientific researcher, but only if such income is derived from sources outside that recipient state. Such an approach protects income of the "drained brain" from being taxed at the place of their migration if such academic migration or travel is financed by a source outside the recipient state. At the same time such an approach does not shield donor state from tax revenue losses in cases when such academic travel or migration is financed by the recipient state. These scenarios and the effects of double tax treaties in the context of brain drain are illustrated by the two cases below.

3.3.2. Case 1: Scientific Researcher Leaves the State and Receives Financing from This State, Donor State Is Temporarily Protected from Loss of Tax Revenues

A teacher of the Serbian language at the local university, Jovana B., a resident of the Republic of Serbia, moves to country Y with the aim of popularizing the Serbian language among the students of the local university. The academic trip is financed from the grant program financed by the Serbian university and lasts for the period of one year. There is a double tax treaty between the Republic of Serbia and country Y with the same provisions as in Article 21 of the Serbia-Bulgaria DTT.

We can conclude that the income of Jovana B. will not be taxed in country Y. If Jovana B. is still a tax resident of the Republic of Serbia, her income will be taxed in Serbia. Therefore, this example illustrates the "normal" scenario of the application of such a provision. The effects in the context of brain drain are: (1) the donor country does not lose its revenues for the period mentioned in the DTT, (2) however, the donor country can still lose its revenue if Jovana's academic trip lasts for more than one year because she will likely not be regarded as tax resident of Serbia, therefore in this case double non-taxation can arise.

3.3.3. Case 2: Scientific Researcher Leaves the Country Financed by the Recipient Country

Let us now assume that, as in the previous case, Jovana moved to the same country Y with the same academic purpose, but in this example her academic visit is financed by country Y. In this case provisions of Article 21 of the DTT, similar to Article 21 of the Serbia-Bulgaria DTT, would not prevent the recipient country from taxing her income because it is sourced outside of the Republic of Serbia. However, such income can

also be taxed in Serbia for as long as Jovana is a Serbian tax resident. So, the issue of double taxation can arise, which will be resolved under the relevant DTT Article on double tax relief. However, the recipient country will have priority right to tax the income and the donor country (Serbia in this example) will be providing relief in the form of credit or exemption.

3.3.4. Interim Conclusion

We can conclude that donor countries can potentially lose the right to tax foreign sourced income of emigrating researchers who are moving to conduct their research and teaching activities abroad, if such relocations are financed by the recipient country.

We also propose the idea for the design of separate article in the UN MC devoted to taxation of researchers and visiting teachers. We believe that in such an article only the country financing the research project or visit should have the exclusive right to tax such income, irrespective from the period of the research visit. We propose such idea because: (1) academic research is usually conducted for public benefit, so we can deliberately create the possibility for individual countries to exempt such public good from taxation if they wish so (Pigouvian subsidy argument) (Pigou 1920); (2) experienced researchers can be very mobile, engaging in projects around the world, so it is more convenient to tax their income based on the source rather than the residence principle.

Application of the tax treaties based on the UN and the OECD models can lead to double non-taxation and inconsistencies between the country of taxation and the country of real economic activity of mobile individuals. Therefore, we put forward the idea of the need to expand areas of global tax coordination aimed at minimizing and mitigating the negative consequences of the migration of the qualified specialists from less-developed countries to more-developed countries, as well as creating global tax rules that would not create additional artificial incentives for migration and would not lead to an unfair loss of tax revenues and resources by human capital donor-countries.

4. BEOGRADSKI EPOHALAN POREZNI SPORAZUM (BEPS 2.0) OR POSSIBLE DIRECTIONS FOR DEVELOPMENT OF INTERNATIONAL TAX COOPERATION ADDRESSING THE BRAIN DRAIN ISSUE

4.1. Limits of International Tax Cooperation in the Area of Brain Drain

In this paragraph we stress the necessity for the rethinking the design of international tax law in the complex, cohesive and multilateral way, with the aim of aligning the place of taxation with the place of talent creation and making some proposals for further advancement. On the one hand, the history of international relations contains a large number of examples of successfully concluded multilateral agreements. On the other hand, they are mostly limited to specific regional unions and are formed where there are economic, cultural or ethnic preconditions for their creation (Bravo 2016, 280). Moreover, according to some experts, multilateral mechanisms, including consensus between countries on certain tax issues, are a "utopian view of international tax law" (Schwartz 2015).

Broekhuijsen, Vording (2016, 43) highlights the following ideas in the context of global governance:

- (neo)realism states are rational players in an anarchist world, and for any development, the initiative of the most developed countries is required;
- (neo)liberalism states are subjects of relations for which economic rationality plays the most important role in participating in multilateral agreements – countries enter into multilateral cooperation only if they see this as an economic benefit that can be estimated.

It seems difficult to unambiguously determine which of the strategies is dominant in the context of tax cooperation for the purpose of combating the brain drain. On the one hand, the tax policies of the BRICs countries (Brazil, Russia, India, and China), especially China and India, whose opinion on tax issues may differ from the position of the OECD, are beginning to gain more weight in the global tax governance debate. On the other hand, the current international tax architecture is "under the umbrella" of the developed OECD countries, which are the main recipients of the "brains" from developing jurisdictions. This, in particular, is again indicated by the Gallup Institute data: out of 41 regions that have potential positive population growth in the scenario of a complete emigration of brains, 22 are developed countries that are members of the OECD:

Table 3: Jurisdiction structure with positive population growth²³

Indicator	OECD countries	Other countries
Amount	23	18

²³ See: compiled by authors based on World Bank Data. https://data.worldbank. org/indicator/ny.gdp.mktp.cdm (last visited 25 September 2019), and Gallup Institute. http://news.gallup.com/migration/interactive.aspx?g_source=link_newsv9&g_campaign=item_245204&g_medium=copy (last visited 25 September 2019).

Share of GDP in global GDP	53.04%	2.57%
Potential Net Migration Index	83%	63%

Based on the results in Table 3, it can be presumed that the role of the developed OECD countries so far remains dominant in global coordination of taxation, which, in particular, is also indicated by the experience of drafting the multilateral agreement under BEPS (Byrne 2016). Therefore, assuming that brain drain is a positive process for recipient countries, it seems unlikely that there will be a radical transformation of the international tax architecture towards a certain redistribution of tax revenues in favor of developing countries suffering from the brain drain. That is why we consider ideas such as the Bhagwati tax (Bhagwati 1976, 34) to be utopian and unrealistic. However, we call for some form of global cooperation and some ideas for such cooperation are presented below.

We take into account the unprecedented scale of tax cooperation observed at the present. Despite the fact that initially researchers from different countries expressed some skepticism regarding the effectiveness of the implementation of the BEPS plan (can the idea of interstate tax cooperation for a more equitable tax collection be a significant incentive for jurisdictions?), as of 2019, it can be stated that the BEPS plan met its expectations at least retarding the rate of expansion of its ideas worldwide.

First, the qualitative and quantitative composition of the participant countries is surprising. As of July 2019, 132 jurisdictions are member states of Inclusive Framework on BEPS, including both developed (e.g. the USA, the UK) and developing countries (e.g., China, India) and typical offshore countries (e.g. British Virgin Islands, Belize, Barbados.). A significant number of participants are also covered by the Multilateral Convention to Implement Tax Treaty Related Measures to Prevent BEPS – "a king treaty" implementing BEPS measures into the actual double tax treaties. Despite the fact that Action 15 of the BEPS plan was not part of the so-called "minimum standard" of BEPS, 89 jurisdictions have already signed it (OECD 2019c).

Second, prior to the implementation of the BEPS plan, many countries aggressively used tax incentives for multinational corporations as part of their international tax policies (for example, having low tax rates for foreign sourced income in conjunction with no exchange of information). However, since the launch of the BEPS plan, the OECD has already analyzed 287 tax regimes, of which 76 were canceled, 11 were recognized as malicious or potentially harmful, 15 are in the process of

²⁴ See: OECD. What is BEPS? http://www.oecd.org/tax/beps/about/ (last visited 26 September 2019).

cancellation/replacement, etc. (KPMG 2019b), which is the result of the BEPS Action 5 on harmful tax competition.

Of course, the BEPS will not stop international tax competition, as its scope does not affect all forms of tax incentives for attracting global capital. However, for the first time, countries have shown such global intention for international tax cooperation on the issue of non-taxation.

4.2. What Can We Pick from the BEPS Project for the Proposed BEPS 2.0?

In our opinion, some points of the BEPS Project are irrelevant for the proposed multilateral tax cooperation in the context of brain drain. These are Action 2 – Neutralizing the Effects of Hybrid Mismatch Arrangements, Action 3 – Controlled Foreign Company, Action 4 – Limitation on Interest Deductions, primarily because these actions relate to corporations and do not relate to individual income. We do not claim that the experience of these actions is generally not applicable to brain drain – rather, we can discuss the need for a more in-depth analysis. We propose some ideas of adapting the approaches developed under the BEPS Project for addressing the brain drain issue, presented in Table 4.

Table 4: Selected BEPS Plan Actions and their applicability to BEPS 2.0

Action	Scope of applicability	
1 – Tax Challenges Arising from Digitalization	Improving the mechanism of income allocation in the context of the migration of talented individuals in the	
8–10 – Transfer Pricing	direction of more alignment between the place of taxation and the place of creation of value, which in	
13 – Country-by-Country Reporting	the case of individuals is the place of "talent creation and skills development" and in the context of the digitalization of the economy is the place of virtual economic activity and economic presence	
5 – Harmful Tax Practices	Assessment of limits of acceptability of "aggressive tax incentives" employed by the countries in the processes of brain drain	
	Exchange of information relating to tax incentives for attracting "brains"	
6 – Prevention of Tax Treaty Abuse	Improving the preamble to bilateral tax treaties in order to show countries' intentions to not create opportunities for non-taxation of migrating individuals	
11 – BEPS Data Analysis	Creating an internationally consistent methodology for assessing tax losses from brain drain	

Action	Scope of applicability
12 – Mandatory Disclosure Rules	Disclosure of information about the level of tax burden on mobile highly-qualified individuals
14 – Mutual Agreement Procedure	Streamlining the mechanism of mutual agreement procedure for addressing the scenarios of "double residence" or "no residence" in the context of brain drain. The development of the concept of "vital interests" in relation to digital nomads
15 – Multilateral Instrument	Multilateral agreement for the synchronized implementation of anti-brain drain measures in bilateral tax treaties

4.2.1. Limits of the Acceptability of "Aggressive Tax Incentives" Employed by Countries in the Processes of Brain Drain and the Launch of a Spontaneous Exchange of Information on Incentives for Attracting "Brains" (Action 5)

Various socio-economic instruments can be used to combat "brain drain". They also include measures to improve national tax policies – here are just a few recent changes:

- a reduction in the tax rate for young Poles with the intention to decrease the level of brain drain in Poland (Voice of America 2019);²⁵
- the elimination of tax incentives for tax residents working abroad (Arendse 2019);²⁶
- lowering the requirements for tax residence to increase win in tax competition for digital nomads (CNews 2018);²⁷

At the end of July 2019, the Polish government abolished the personal income tax for young Poles, imder the age of 26 years old and earning less than PNL 85,500 . As Polish Prime Minister Mateusz Moravecki noted, this decision was made in order to stop the brain drain from the country, the scale of which is very significant. Since 2004 between 1.5 and 2 million citizens left the country. According to experts, about 2 million people will be able to take advantage of this benefit. (Voice of America 2019)

Previously, South African tax residents living and working abroad for more than 183 days (and more than 60 consecutive days) were exempted from paying the national tax on their foreign income, but starting from March 2020, this approach will change. The amendments will require South African specialists who reside and work abroad but are still considered "physically present" (quantitatively daily test) or "usually resident" (subjective assessment of "actions, connections and intentions") in the country to pay tax to the South African state in the amount of up to 45% of their gross foreign income, provided that it exceeds ZAR 1 million. (Arendse 2019)

In 2018, a draft law was discussed in Russia, according to which IT specialists spending more than 90 days a year in Russia can receive tax resident status. The changes are aimed at increasing the country's tax attractiveness for traveling IT professionals,

Despite the different approaches, each of these cases represents the reaction of the states to the global process of the brain drain/brain gain. However, in our opinion, in addition to the cooperative struggle against aggressive corporate tax regimes, in the near future the international tax community may also require a similar "audit" of the provisions of national laws directly or indirectly aimed at attracting and retaining human capital. For example, the reasons for the expansion of tax sovereignty in the case of the Republic of South Africa may indicate its necessity: according to the competent authorities, the cancelled tax exemption was excessively "generous", especially in cases where an individual worked in a jurisdiction with an extremely low or zero personal income tax rate (e.g. UAE) (Arendse 2019).

Thus, the first issue at which such an audit should be directed is the delineation of cross-border situations in which brain drain can lead to non-taxation or reduced taxation, as well as a clear definition of the conditions under which the tax incentives provided by one country harms another country. When developing a methodology for determining the integrity of provisions in domestic tax legislation, attention should also be paid to criteria indicating the potential harmfulness of the preferential regimes outlined in the OECD report Harmful Tax Competition: An Emerging Global Issue (OECD 1998):

Table 5: A list of factors indicating the harmfulness of the preferential regime

Key factors	Other factors
No or low effective tax rates	An artificial definition of the tax base
"Ring-Fencing" of Regimes	Failure to adhere to international transfer pricing principles
Lack of transparency	Foreign source income exempt from residence country tax
Lack of effective exchange of information	Negotiable tax rate or tax base
	Existence of secrecy provisions
	Access to a wide network of tax treaties
	Regimes that are promoted as tax minimization vehicles
	The regime encourages purely tax-driven operations or arrangements

who, according to the new rules, will be able to pay income tax at a rate of 13% – one of the lowest income tax rates in the world. (CNews 2018)

This report includes a sequential set of three questions, the answers to which help determine whether the potentially harmful tax regime is actually harmful (OECD 1998):

- Does the tax regime shift activity from one country to the country providing the preferential tax regime, rather than generate significant new activity?
- Is the presence and level of activities in the host country commensurate with the amount of investment or income?
- Is the preferential tax regime the primary motivation for the location of an activity?

In addition to conducting continuous monitoring and spontaneous exchange of information about such regimes, by analogy with the recommendations of the Report (OECD 1998) and BEPS Action 5, we offer the following: after considering the economic consequences of the existence of preferential provisions in national legislation, such norms can be considered aggressive, and the country will have the opportunity to cancel or modify them by amending national tax legislation. In turn, other countries may take protective measures against the negative impact of such provisions, while also encouraging the possibility of adjusting or even denouncing them.

4.2.2. Unified Methodology for Assessing Tax Losses from Brain Drain (Action 11)

The magnitude of the BEPS problem in the corporate area is between USD 100 bln and USD 240 bln or between 4% and 10% of global corporate income tax (CIT) revenues (OECD 2015c, 15). In addition to significant financial losses, the BEPS process has other economic consequences, including, for example, tilting the playing field in favor of tax-aggressive MNEs, distorting the location of highly-mobile, intangible assets, misdirecting foreign direct investment, etc. (OECD 2015c, 15).

Therefore, in our opinion, monitoring the BEPS magnitude is one of the most important parts of the BEPS plan. As noted by the OECD, "the lack of quality data on corporate taxation has been a major limitation to measuring the fiscal and economic effects of tax avoidance as well as any efforts to measure the impact of the implementation measures agreed as part of the BEPS Project" and "increasing the quality of the data and the analytical tools available, through the ongoing work under Action 11, is crucial in being able measure the impact of tax avoidance and the effect of the implementation of the BEPS measures in curbing these practices." ²⁸

²⁸ See: OECD. Action 11 BEPS data analysis. https://www.oecd.org/tax/beps/beps-actions/action11/ (last visited 24 September 2019).

To date, the Inclusive Framework is actively working on this action of the BEPS Project. In particular, in January 2019, the Corporate Tax Statistics Database was launched, which stores data related to the BEPS process and the taxation of MNEs in general. Additionally, the first Inclusive Framework presented the first summary statistics based on an analysis of the results from the implementation of Action 13 on Country-by-Country Reporting. Finally, the international organization notes that the workflow under Action 11 is too early to stop: the Inclusive Framework is developing "new and enhanced datasets and analytical tools that can assist in measuring and monitoring the fiscal and economic impacts of tax avoidance and the effects of the implementation of the BEPS measures". ³⁰

Unfortunately we cannot say that the international tax community has a similar level of analytical apparatus for assessing tax losses from a brain drain. Humanity does not fully understand the extent of tax losses from a brain drain, although it is intuitively clear that it is not much less than losses of USD 100 bln to USD 240 bln from tax avoidance by the corporations (this, for example, is indicated by the calculation of net tax losses from a brain drain in the Republic of Serbia). As a result of this, we recommend the development of a methodology for assessing tax losses from brain drain, taking into account the best practice of BEPS Action 11,³¹ the use of which could reliably indicate the extent of the problem in the context of jurisdictions. Such a methodology would consider both the direct effects, such as losses of tax revenues in the donor country and their gains in the recipient country, and indirect effects, such as benefits from remittances to members of emigrants' families, transfer of knowledge, etc.

4.2.3. Disclosure of the Tax Burden on Certain Individuals (Action 12)

The authors of the BEPS Plan note: "the lack of timely, comprehensive and relevant information on aggressive tax planning strategies is one of the main challenges faced by tax authorities worldwide" (OECD 2015d, 9). As a result, in order to obtain preventive information, the OECD recommends the development of a set of mandatory rules for the disclosure of information regarding aggressive transactions, taking into account the balance of business and government interests. According to the OECD, the implementation of a tax disclosure mechanism may

²⁹ *Ibid*.

³⁰ Ibid.

³¹ For example, the OECD report on Action 11 of the BEPS Plan provides the possibility of applying six indicators when assessing the extent of tax base erosion. It is also noted that the indicators developed are illustrative because of possible limitations in data availability. Therefore, in our opinion, even the use of the methodology outlined in the first part of the article would make it possible to understand the preliminary scale of tax losses from brain drain.

pursue the simultaneous achievement of the two goals. First, such an instrument allows competent authorities to more effectively respond to changes in tax behavior of the taxpayers, second, it acts as a strong deterrence tool – both taxpayers and promoters of schemes will be more careful in choosing one tax scheme or another, if there is a requirement to disclose it (OECD 2015d, 9). Despite the fact that Action 12 is not included in the so-called BEPS minimum standards, some experts suggest its acknowledgement as a next BEPS minimum standard (Mosquera Valderrama 2018). Moreover, it should be noted that the ideas of BEPS 12 are actively being implemented (Directive 2018/822/EU), however, these changes have far from a positive perception by business representatives (EY 2018b).

So, in our opinion, the basic ideas of the BEPS Action 12 can be applied to cases of brain drain. For example, companies could provide information to the tax authorities about employees:

- 1) who recently moved to work in this country;
- 2) who could potentially be a tax resident of another state (other states);
- 3) whose level of tax burden is zero or close to it;
- 4) whose place of physical location during the performance of employment duties does not coincide with the place of payment of employment income.

In the future, such aggregated data could be included in the scope of the information exchange and form the basis for calculating tax revenue losses. However, it should be noted that the OECD indicates that the "lack of clarity and certainty can lead to inadvertent failure to disclose, which may increase resistance to such rules from taxpayers" or "could result in a tax administration receiving poor quality or irrelevant information" (OECD 2015d, 19). That is why such tax policy measure, if implemented, should be thoroughly designed.

4.2.4. Digitalization of the Mutual Agreement Procedure (Action 14)

Data from the OECD jurisdiction-specific guidance (OECD 2019a) indicates that today subjective criteria are already used by many jurisdictions in determining the tax residence of individuals. This trend will probably continue to grow, since the presence of only objective criteria in the tax legislation does not reflect the tax nexus of an individual with a country properly. A possible overlap of subjective criteria in

³² See: Council Directive (EU) 2018/822 of 25 May 2018 amending Directive 2011/16/EU as regards mandatory automatic exchange of information in the field of taxation in relation to reportable cross-border arrangements

different states inevitably can lead to an increase in the number of crossborder tax disputes.

Almost all double tax agreements contain rules related to the mutual agreement procedure. For example, Article 25 of the OECD MC provides for a mechanism, independent of the usual legal remedies available in domestic law, by which the competent authorities of the Contracting States can resolve disagreements or difficulties related to the interpretation or application of the Convention on a mutually agreed basis. However, as noted by the OECD itself, despite the widespread dissemination of this provision in double tax treaties, the current mutual agreement procedure is still far from ideal and requires reform.³³

The BEPS Action 14 recommendations were aimed at solving some of the mentioned problems but in our opinion it does not contain revolutionary ideas; therefore, a detailed disclosure of the essence of BEPS Action 14 is not the purpose of this article. We applaud the positive developments of the MAP mechanism after the implementation of BEPS Action 14; according to the OECD, as of 2017, the average term for solving tax disputes under the MAP is 30 months, for cases related to transfer pricing, and 17 months for other cases.³⁴ However, we consider these terms "luxurious" for modern international taxation.

In our opinion, it is time for a "real-time" mutual agreement procedure that would reduce the level of transaction costs and time costs for all parties. Moreover, speeding up the MAP process can lead to more rapid accumulation of the MAP experience that states can use to improve their brain drain policies. As a result, we recommend the development of a digital mutual agreement procedure for competent authorities, whose presence would solve the issues of a multiple residence of "drain brains" in "a few clicks".

4.2.5. Multilateral Instrument (Action 15)

In July 2018, the Multilateral Convention to Implement Tax Treaty Related Measures to Prevent Base Erosion and Profit Shifting (Multilateral Instrument or MLI) was signed by more than 85 jurisdictions. This multilateral agreement is the "new word" in international taxation, as it allowed a large number of changes to be introduced in double tax treaties in a synchronized manner, without the need for separate negotiations for each of the double tax treaties.³⁵ We believe that this experience can be

³³ See: OECD. Action 14 Mutual Agreement Procedure. http://www.oecd.org/tax/beps/beps-actions/action14/ (last visited 25 September 2019).

³⁴ See: OECD. Mutual Agreement Procedure Statistics for 2018. https://www.oecd.org/tax/dispute/mutual-agreement-procedure-statistics.htm (last visited 25 September 2019).

³⁵ See: OECD. Action 15 Multilateral Instrument, http://www.oecd.org/tax/beps/beps-actions/action15/ (last visited 25 September 2019).

useful in a coordinated fight against brain drain. Among the priority issues requiring the development of a "single view" in the context of improving the function of double tax treaties, the following can be distinguished:

- Does the current version of the preamble set out in the current versions of the OECD and the UN model conventions (OECD 2017a; United Nations 2017) require clarification that the bilateral tax agreement does not apply in situations leading to "aggressive" attracting of "brains"?
- Does the current version of Article 15 of the OECD and UN MCs related to income from employment reflect the economic nature of such income in cross-border situations?
- Is a separate article related to cross-border taxation of scientists, professors, etc. necessary in the subsequent versions of the OECD and the UN model conventions?
- Does the current version of the Article 14 of the UN MC, about income from independent professional services, need an update?
- Are the provisions of the Article 17 of the OECD and UN MCs, related to income of sportspersons and entertainers, in line with the economic nature of the value creation in this area in the era of digitalization?

4.2.6. Value Creation in the Context of Brain Drain (Actions 1, 8–10, 13)

The current distributive rules and nexus rules that are present in the double tax treaties and domestic legislation based on the concepts of source, residence, place of physical employment and others, discussed in the analysis presented in section 3 above, do not reflect the value creation process of the "talent creating, developing and exploiting". "Talents" can potentially create value for their employers and society at large and can earn high income for themselves. However, "talents" do not come from nowhere; it takes time and effort, nurture, training and education in order to create the "initial talent", which is developed by the different kinds of collaborations and activity. So, if we apply the value-creation approach that is proposed in BEPS Project Actions 8-10 (OECD 2015a) to individuals, we can argue that countries where significant contributions to "talent creation" were made can have potential subsequent rights to tax such an individual's incomes. This line of argument can potentially be reflected in the OECD MC and the UN MC, in order to ensure fairness of sharing the global tax pie consisting of the incomes of such talented individuals. Such a substantial form of nexus may seem complicated, however, in practice it can be a useful proxy for income allocation of highly-skilled workers in the global digitalized and mobile world, where value is increasingly created by intangible assets. Below we offer some conceptual ideas for such an approach.

First, the state where a person obtained their professional education and crucial experience can potentially be regarded as a state which has a right to tax at least some part of the subsequent income of such a person, especially in cases where education was government-funded. This idea can also be developed for the case of sportspersons leaving the state where they were trained and took their first steps in professional sports. Another relevant case is if the researcher who invented a new technology in a given state, based on the research infrastructure in that state, plans to move to a different state for the monetization of his invention. Generally, this idea is just a projection of the approach described in the OECD Transfer Pricing Guidelines International regarding corporate restructurings (OECD 2017b). The basis for such an idea is the result of the analysis, supporting the ease with which the donor state can lose taxing rights for talents in the case of IT specialists, sportspersons and academics, mentioned above.

Second, in determining the place of taxation of the talent's income that was generated distantly by the means of telecommunication or electronic networks, we suggest the allocation of part of the tax base to the county where the economic source of such income is situated – which can be the country of the market audience watching the online broadcast or the country of the employer who pays to the distant IT specialist. This idea is in line with the OECD work on the digitalization of the economy (OECD 2019b).

Third, the approach developed in BEPS Action 13 (OECD 2015e) for developing harmonized global rules of reporting information summarizing the activities of a multinational corporation in all the states where it is present can also be quite useful as an administrative tool, forming basic tax risk assessments for highly mobile individuals. In the same manner as in the case of the country-by-country reporting by Action 13 of the BEPS Project, such an approach can start with the application only in regard to the global economic activities of individuals with a global income exceeding a defined, relatively high threshold. This administrative requirement is critical in the case of taxing the income of the celebrities, including sportspersons and entertainers, but could in the future also be used as the basis for assessment of the income of the "digital nomads".

5. CONCLUSION

The main results of the analysis above are outlined below.

- 1) The domestic tax residence rules of donor countries of qualified specialists make it quite easy to break the personal nexus with the tax jurisdiction of the countries of emigration, which leads to their unfair loss of tax revenues. So, if countries don't want to lose tax revenues they should introduce more strict criteria for tax residence status, as well as special rules aimed at creating tax obstacles for termination of tax residence in the country, such as exit taxes. They should also consider this policy when negotiating tax treaties, which usually limits their rights to taxation of both the migrating and the offshoring workforce.
- 2) The rules for eliminating double taxation provided by both the OECD and the UN model conventions are obsolete and do not reflect the current problems of distribution of taxing rights in the context of the analyzed talent migration strategies. Situations of double taxation, double non-taxation and unfair limitations of taxation rights of the donor states can arise as a result of application of double tax treaties. This problem is exacerbated in the context of several current trends, which include:
 - development of digital marketing strategies for the promotion and distribution of entertainment content, for example, broadcasting sports events over the internet,
 - increasing level of mobility of the skilled workforce, as well as expanding opportunities for working remotely, for example, in regard to IT specialists.
- 3) As for income of the mobile scientific researchers the provisions of the double tax treaties, based on relevant Articles of the OECD and the UN model conventions, ³⁶ can lead to unsatisfactory results: unfair loss by the donor country of the right to tax the income of emigrating researchers and possible double taxation of their income, fragmentation and complexity of regulation.
- 4) In our opinion, a new tax policy ideology is needed for rethinking the global tax architecture, in the context of brain drain issue, which should be based on two general ideas:
 - (1) prevention of the migration of talents obtaining tax benefits, whose incomes are subject to double non-taxation due to the application of a combination of international and national tax rules,

 $^{^{36}\,}$ For example, Article 14, 15 and 20 of the UN MC (2017), Article 15 of the OECD MC (2017)

(2) compensation the unfair loss of tax revenues to the countries which educated and then donated their qualified specialists to other countries.

We believe that the positive experience of the BEPS Project can be transferred to a cooperative approach in addressing the negative tax implications of brain drain. The main objective of such a project would be to reform the existing architecture of international taxation in the context of increasing mobility of qualified specialists and taking into account the interests of developing countries, including, in particular:

- the permissible limits of tax policy in attracting talented migrants to one's jurisdiction,
- approaches to cooperation in administrative matters in this area,
- approaches to developing a methodology for determining the place of creation of added value by skilled migrants, which would be a prerequisite for the country to have the right to tax income created by their activities;
- streamlining the mutual agreement procedure in regard to cases of mobile individuals.

REFERENCES

- ANSA Politics. 2019. Brain drain costs 14 bn a year Tria. 16 July. http://www.ansa.it/english/news/politics/2019/07/16/brain-drain-costs-14-bn-a-year-tria_a699d6fe-17ad-43e7-8ad2-5310065f0434.html (last visited 24 September 2019)
- Arendse, Lameez. 2019. Everything you need to know about the new 'expat tax'. Fin24, 22 June. https://www.fin24.com/Opinion/everything-you-need-to-know-about-the-new-expat-tax-20190621 (last visited 24 September 2019).
- Bhagwati, Jagdish. 3/1976. Taxing the Brain Drain. *Challenge* 19: 34–38. *Blic Biznis*. 2019. Mesečno se iz Srbije iseli 4.250 ljudi! [4,250 people per month emigrate from Serbia!]. 29 August. *https://www.blic.rs/biznis/vesti/mesecno-se-iz-srbije-iseli-4250-ljudi-detaljno-istrazivanje-upalilo-alarme-upozorenje/5wt5y0q* (last visited 6 October 2019).
- B92. 2019. Vučić: Promašio sam za godinu dana, ali biće plate 500 € [Vučić: I missed by a year, but salaries will be 500 €]. 21 February. https://www.b92.net/biz/vesti/srbija.php?yyyy=2019&mm=02&dd=21&nav id=1508468 (last visited 6 October 2019).

- Bravo, Nathalie. 3/2016. The Multilateral Tax Instrument and Its Relationship with Tax Treaties. *World Tax Journal* 8: 279–304.
- Broekhuijsen Dirk., Vording Henk. 1/2016. The Multilateral Tax Instrument: How to Avoid a Stalemate on Distributional Issues? *British Tax Review*: 39–61.
- Byrne, Caroline. 2016. OECD Holds Paris Talks on Multilateral Instrument. *International Tax Review*, 7 July. *https://www.internationaltaxreview.com/article/b1fygdpmlbrpmj/oecd-holds-paris-talks-on-multilateral-instrument* (last visited 24 September 2019).
- Chernykh, Vsevolod. 2018. Migraciya rossijskih IT specialistov za rubezh [Migration of Russian IT specialists abroad]. *Potok*, 28 February. http://miptstream.ru/2018/02/28/it-migration/ (last visited 6 October 2019).
- CNews. 2018. "Zaletnym" ajtishnikam hotyat razreshit' platit' v Rossii 13% nalogov vmesto 30% ("Vagrant" IT specialists want to be allowed to pay 13% of taxes in Russia instead of 30%). 26 July. http://www.cnews.ru/news/top/2018–07–26_puteshestvuyushchie_ajtishniki_smogut_stat_nalogovymi (last visited 24 September 2019).
- EY. 2018a. Worldwide Personal Tax and Immigration Guide 2018–19. https://www.ey.com/Publication/vwLUAssets/ey-2018–19-world wide-personal-tax-and-immigration-guide/\$FILE/ey-2018–19-worldwide-personal-tax-and-immigration-guide.pdf (last visited 6 October 2019).
- EY. 2018b. The European Union's Mandatory Disclosure Regime. A roundtable discussion. https://www.ey.com/Publication/vwLUAsse ts/ey-europes-mandatory-disclosure-regime-a-roundtable-discussion/\$FILE/ey-europes-mandatory-disclosure-regime-a-roundtable-discussion.pdf (last visited 6 October 2019).
- Federal State Statistics Service. 2017. Trud i zanyatost' v Rossiji [Labour and employment in Russia]. https://www.gks.ru/free_doc/doc_2017/trud_2017.pdf (last visited 6 October 2019).
- Federal Tax Service. 2018. Rukovoditel' FNS Rossii Mihail Mishustin podvel itogi raboty nalogovyh organov za 2017 god [Mikhail Mishustin, Head of the Federal Tax Service, summed up the results of the tax authorities' work in 2017]. 28 February. https://www.nalog.ru/rn77/news/activities_fts/7255441/ (last visited 6 October 2019).
- Gallup Institute. 2018. Potential Net Migration Index. http://news.gallup.com/migration/interactive.aspx?g_source=link_newsv9&g_campaign=item_245204&g_medium=copy (last visited 6 October 2019)

- Gatto, Luigi. 2017. Tennis Players and Tax's Heaven: Novak Djokovic, Garbine Muguruza and more. *Tennis World*. 20 September. *https://www.tennisworldusa.org/tennis/news/Novak_Djokovic/47455/tennis-players-and-tax-s-heaven-novak-djokovic-garbine-muguruza-and-more/* (last visited 6 October 2019).
- Gatto, Luigi. 2018. Novak Djokovic explains why he lives in Monte Carlo. *Tennis World*. 16 October. *https://www.tennisworldusa.org/tennis/news/Novak_Djokovic/61759/novak-djokovic-explains-why-he-lives-in-monte-carlo/* (last visited 6 October 2019).
- Grace, Anna. 2019. Lindsey Sterling Performing Live as Avatar. IQ. 21 August. https://www.iq-mag.net/2019/08/lindsey-stirling-performing-live-as-avatar/#.Xbw36lUzaM (last visited 6 October 2019).
- Holmes, Kevin. 2007. International Tax Policy and Double Tax Treaties: an Introduction to Principles and Application. IBFD.
- Kostic, Svetislav. 2/2019. In Search of the Digital Nomad Rethinking the Taxation of Employment Income under Tax Treaties. *World Tax Journal* 11: 189–225.
- KPMG. 2019a. Individual income tax rates table. https://home.kpmg/xx/en/home/services/tax/tax-tools-and-resources/tax-rates-online/individual-income-tax-rates-table.html (last visited 6 October 2019).
- KPMG. 2019b. OECD Continues to Shut Down Harmful Tax Regimes. 30 July. https://home.kpmg/ca/en/home/insights/2019/07/oecd-continues-to-shut-down-harmful-tax-regimes.html (last visited 26 September 2019).
- Ministry of Finance of Republic of Serbia. 2019. Macroeconomic and Fiscal Data. https://www.mfin.gov.rs/pages/issue.php?id=3 (last visited 6 October 2019).
- Ministry of Finance of the Russian Federation. 2019. Osnovnye napravleniya byudzhetnoj, nalogovoj i tamozhenno-tarifnoj politiki na 2020 god i na planovyj period 2021 i 2022 godov [Main directions of budget, tax and customs-tariff policy for 2020 and for the planning period of 2021 and 2022]. http://www.consultant.ru/document/cons doc LAW 308390/ (last visited 1 November 2019).
- Mosquera Valderrama, Irma. 2018. Will BEPS Action 12 (mandatory disclosure) be the next minimum standard? *Globtaxgov*, 17 August. https://globtaxgov.weblog.leidenuniv.nl/2018/08/17/will-beps-action-12-mandatory-disclosure-be-the-next-minimum-standard/ (last visited 24 September 2019).
- Mohapatra, Sanket, Blancka Moreno-Dodson, Dilip Ratha. 5/2012. Migration, Taxation, and Inequality. *Economic Premise* 80: 1–5.
- N1. 2019. Odliv mozgova Srbiju košta i do 1,2 milijarde evra godišnje[Brain drain costs Serbia up to 1.2 billion euros per year]. 27

- March. http://rs.nlinfo.com/Biznis/a471477/Odliv-mozgova-Srbiju-kosta-i-do-1-2-milijarde-evra-godisnje.html (last visited 6 October 2019).
- OECD. 1998. Harmful Tax Competition: An Emerging Global Issue. Paris: OECD Publishing. https://www.oecd-ilibrary.org/taxation/harmful-tax-competition_9789264162945-en (last visited 24 September 2019).
- OECD. 2015a. Aligning Transfer Pricing Outcomes with Value Creation, Actions 8–10 2015 Final Reports. Paris: OECD Publishing.
- OECD. 2015b. Countering Harmful Tax Practices More Effectively, Taking into Account Transparency and Substance, Action 5 2015 Final Report. Paris: OECD Publishing.
- OECD. 2015c. *Measuring and Monitoring BEPS, Action 11 2015 Final Report.* Paris: OECD Publishing.
- OECD. 2015d. *Mandatory Disclosure Rules, Action 12 2015 Final Report.* Paris: OECD Publishing.
- OECD. 2015e. Transfer Pricing Documentation and Country-by-Country Reporting, Action 13 2015 Final Report. Paris: OECD Publishing.
- OECD. 2017a. Model Tax Convention on Income and on Capital: Condensed Version. http://dx.doi.org/10.1787/mtc_cond-2017-en (last visited 7 October 2019).
- OECD. 2017b. Transfer Pricing Guidelines for Multinational Enterprises and Tax Administrations. Paris: OECD Publishing.
- OECD. 2019a. Jurisdiction-specific guidance. https://www.oecd.org/tax/automatic-exchange/crs-implementation-and-assistance/tax-residency/ (last visited 7 October 2019).
- OECD. 2019b. OECD leading multilateral efforts to address tax challenges from digitalisation of the economy. 9 October. https://www.oecd.org/tax/oecd-leading-multilateral-efforts-to-address-tax-challenges-from-digitalisation-of-the-economy.htm (last visited 3 November 2019).
- OECD. 2019c. Signatories And Parties To The Multilateral Convention To Implement Tax Treaty Related Measures To Prevent Base Erosion And Profit Shifting. https://www.oecd.org/tax/treaties/beps-mli-signatories-and-parties.pdf (last visited 26 September 2019).
- Pigou, Arthur Cecil. 1920. *The Economics of Welfare*. London: Macmillan.
- *RBC*. 2018. Eksperty RANHiGS rasskazali ob usilivshejsya utechke mozgov iz Rossii [RANEPA experts spoke about the increased brain drain from Russia]. 23 January. https://www.rbc.ru/society/23/01/2018/5a673f129a794712579d7dbe (last visited 6 October 2019).

- RTS. 2015. Nezaposlenost glavni razlog odlaska mladih iz Srbije [Unemployment is the main reason why young people are leaving Serbia]. 26 July. http://www.rts.rs/page/stories/sr/story/125/drustvo/1929721/nezaposlenost-glavni-razlog-odlaska-mladih-iz-srbije. htm (last visited 24 September 2019).
- RTS. 2019. Srbiju lane napustilo 50.000 ljudi, šta stvarno stoji iza tih brojki [50,000 people left Serbia last year: what is actually behind those numbers]. 30 September. http://www.rts.rs/page/stories/sr/story/125/drustvo/3679307/srbiju-lane-napustilo-50000-ljudi-sta-stvarno-stoji-iza-tih-brojki.html (last visited 6 October 2019).
- Schwartz, Jonathan. 2015. Multilateral Tax Treaty: if we build it, will they come? *Kluwer International Tax Blog*, 5 September. *http://kluwertaxblog.com/2015/09/05/multilateral-tax-treaty-if-we-build-it-will-they-come/* (last visited 24 September 2019).
- Statistical Office of the Republic of Serbia. 2018. *Statistical Yearbook*. http://publikacije.stat.gov.rs/G2018/PdfE/G20182051.pdf (last visited 6 October 2019).
- Thirion, Elodie, Amandine Scherrer. 2018. Citizenship by Investment (CBI) and Residency by Investment (RBI) schemes in the EU. http://www.europarl.europa.eu/RegData/etudes/STUD/2018/627128/EPRS STU(2018)627128 EN.pdf (last visited 7 October 2019).
- Voice of America. 2019. Poland Waives Tax for Young Employees to Counter Brain Drain. 1 August. https://www.voanews.com/europe/poland-waives-tax-young-employees-counter-brain-drain (last visited 24 September 2019).
- United Nations. 2017. United Nations Model Double Taxation Convention between Developed and Developing Countries 2017 Update. https://www.un.org/esa/ffd/wp-content/uploads/2018/05/MDT 2017.pdf (last visited 7 October 2019).

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(495)

CERIF: S141

DOI: 10.5937/AnaliPFB1904238P

Katerina Perrou, PhD*

Katerina Savvaidou, PhD**

BRAIN DRAIN: THE IMPACT ON TAXATION AND MEASURES TO COMBAT THE BRAIN DRAIN

During the debt crisis the number of highly specialized Greek scientists who had migrated abroad surpassed 250,000. The reasons that led to this result include the mismatch of supply and demand for skilled human capital in Greece, the high rates of unemployment, as well as underemployment, and the increased in the tax burden and social security contributions. In order to tackle brain drain, a number of measures have been announced, focusing on reducing the individual income and corporate tax, lowering VAT rates and streamlining tax incentives for investors. Taxrelated measures must take into account the new environment that has been shaped after the implementation of the OECD Base Erosion and Profit Shifting project. In addition to that, they also must respect the fundamental freedoms and general principles of European Union law in general.

Key words: Bhagwati tax. – Free movement of workers. – Harmful tax practices. – Special economic zones. – Tax competition.

^{*} Postdoctoral Researcher, Law School, National Kapodistrian University of Athens, Greece; Legal Counsel to the Governor of the Independent Authority of Public Revenues, Greece, *kperrou@law.uoa.gr*.

^{*} Assistant Professor, Aristotle University of Thessaloniki, Greece; Holder of Jean Monnet Chair "European Tax Policy and Administration"; Former General Secretary of Public Revenues, Greece, savvaidou@law.auth.gr.

1. INTRODUCTION: THE PHENOMENON OF BRAIN DRAIN IN NUMBERS

According to a 2017 report by the European Commission, about 17 million people in the EU moved to another member of the community; 32% of them were up to 34 years old. Germany and Britain were the top destinations. The top countries of origin were Romania, Poland, Italy, and Portugal.

Similarly, according to Eurostat figures, during the debt crisis the number of highly specialized Greek scientists who migrated abroad almost doubled. This has resulted in more than 250,000 Greek scientists working abroad today. The figure corresponds to 12% of Greek university graduates, according to a May 2018 report by the General Secretary of Strategic and Private Investments. Of these, a vast majority are doctors and engineers, as demand in these sectors is huge, while job prospects at home are extremely limited. According to data from the Ministry of Economy and Development's Knowledge Bridges Platform (although, precisely because of freedom of movement, it is virtually impossible to accurately identify how many Greeks work in each EU country) it is clear that the largest number is concentrated in the United Kingdom, followed by the Netherlands, Germany, and Sweden.

According to the latest survey by ICAP People Solutions, 60% moved abroad after having already worked in Greece; 61% left Greece five or more years ago; moreover, over 50% between the ages of 30 and 40 years old, while 29% of Greeks abroad are 41 years or older. In 2015 the latter figure was just 12%. Today 48% of them are single. In 2015, the single persons amounted to 71%. According to the ICAP research, the brain drain population is growing older, living overseas, forming relationships, building a family. At the same time, it is evolving, taking up higher hierarchical positions, and earning more money. But the more that they achieve in their new homes, the more distant the prospect of returning to Greece begins to show.

The reasons that led to brain drain include the mismatch of supply and demand of skilled human capital due to the profile of the Greek economy, the high rates of unemployment as well as underemployment, part-time and related lower-skilled jobs due to the financial crisis, the political unrest, the perception of reduced meritocracy and increased corruption in the country, the general uncertainty, the prospects of personal and professional development abroad, the severe taxation and the enormous increase of social security contributions due to the measures adopted in Greece based to the economic adjustment programs.

2. FREE MOVEMENT OF WORKERS AND BRAIN DRAIN WITHIN THE EUROPEAN UNION

The free movement of workers is one of the fundamental rights enjoyed in the European Union by EU citizens. It is one of the four economic freedoms to which EU citizens are entitled, together with the free movement of goods, services and capital.

Article 45 of the Treaty on the Functioning of the European Union (TFEU) provides that the free movement of workers will be secured within the Union

Free movement of workers entails the right to accept offers of employment actually made and to move freely within the territory of Member States for this purpose. It includes the right to stay in a Member State for the purpose of employment in accordance with the provisions governing the employment of nationals of that State laid down by law, regulation or administrative action and also to remain in the territory of a Member State after having been employed in that State, subject to conditions which is embodied in regulations drawn up by the Commission. ¹

This freedom of movement of workers within the EU has facilitated intra-EU labour mobility. However, in some regions ('sending regions') this freedom has led to a significant out-migration of their highly educated workforce to the advantage of other regions ('receiving regions'). This is determined by the growing competition for talent and the limited capacity of sending regions to create attractive conditions for these workers. Local and regional authorities (LRAs) in sending regions have to cope directly with the socio-economic effects caused by the significant loss of talent or brain drain. Addressing these effects may require the formulation of appropriate policies and/or measures to retain, attract, or regain a highly educated workforce.²

However, the free movement of workers has not only facilitated brain drain, it has also favoured related phenomena such as brain regain, which is the return to a region of the same high skills and/or competencies that were previously lost, and brain circulation, which is the continuous gain-loss of high skills and/or competencies.³

Article 45(3) TFEU.

² See the European Committee of the Regions, Commission for Social Policy, Education, Employment, Research and Culture (2018).

³ *Ibid.*, 5.

3. THE IMPACT OF BRAIN DRAIN ON TAXATION

Academic literature has extensively studied the impact of the brain drain on the sending regions. It has been found that brain drain has a negative impact on the sending regions with severe fiscal consequences as it results in the reduction of taxable income and the unavoidable reduction of income tax as well as a decline in consumption, which also affects state revenue.⁴ At the same time, however, a positive impact is also observed, as sending regions may also experience benefits with regards to brain drain such as return migration, remittances, incentives for investment in education and training and an improvement of governance.⁵ Consequently, a country may experience a loss of tax revenue when the size of workforce shrinks. The labour market is also subject to other changes due to the emigration of highly skilled workers. In particular, when highly skilled workers migrate, the labour market shifts towards workers with lower skills. This in turn may result in an overall reduction of wages, which also affects tax revenue.

Indeed, income tax is closely linked to the constitutional principle of the ability to pay. This principle, enshrined in Article 4 of the Greek Constitution, is linked to the general principle of equality. According to the principle of ability to pay, each person shall contribute to the public revenues according to their capacity. The ability to pay principle imposes limitations on the burden of income taxation that a person can bear. Income tax cannot go beyond the minimum subsistence level; a taxpayer should be left with enough income after tax to provide for their basic needs. Therefore, as a result of the ability to pay principle, the lower the income of a taxpayer is, the lower their ability to pay is and consequently the lower the tax revenue for the state.

Similarly, the reduction of consumption, due to the lower level of income gained, may lead to a reduction of revenue from indirect taxes. Indirect taxes do not have the same limitations as income tax, as far as the ability to pay principle is concerned. Indeed, in the case of VAT, a mechanism of reduced rates for certain categories of goods is used to ensure that the basic needs of taxpayers are met and that they are not overburdened in an inflexible way. Nonetheless, consumption is directly linked to disposable income, i.e. income after tax: when disposable income is reduced, consumption is also negatively affected.

On the other hand, there is also a positive impact of brain drain on sending regions, even in terms of taxation. First of all, the spending power that might be reduced because of the brain drain, can be mitigated by the

⁴ *Ibid.*, 16.

⁵ Ibid.

fact that workers that have migrated keep sending remittances back to their state of origin, therefore, at last partially, closing the created gap. However, remittances could also amplify the negative effect on tax revenues by negatively affecting the decisions to work, for instance by increasing the reservation wage, and de facto further reducing the tax basis. Another effect of emigration is that, as it mostly concerns working age population, the elderly left behind will increases the relative weight of pension and health spending on expenditure.⁶

4. TAX REASONS THAT MAY LEAD TO BRAIN DRAIN

From the diverse causes described, we will focus only on those related to taxation. The following interesting facts come from the OECD report Revenue Statistics 1965–2017.

- 1. In 2007–17, taxes in Greece increased by 8.2 percentage points of GDP, while in the memorandum years tax revenues increased from 32% to 39.4% of GDP. It is noted that although GDP was down 25% during the crisis, tax revenues totalled \$ 95.9 billion in 2010, totalled \$ 71.6 billion in 2015 and jumped to \$ 78.9 billion in 2017. In fact, in the two years 2016–2017 Greece country was seventh in tax increases. In 2017 income and profit taxes amounted to 9% of the GDP and amounted to 22.8% of total government revenue.
- 2. Property taxes increased by 516%. In 2010, real estate taxes accounted for only 0.2% of GDP or close to \in 600 million. Based on recent data, in 2017 taxes amounted to \in 3.7 billion, i.e. 10 times the GDP percentage (2.1%). It should be noted that the situation is the same in Belgium, France and Luxembourg.
- 3. Greece has the lead also in the indirect taxes. Taxes on goods and services reached 15.4% of GDP in 2017, i.e. 39.1% of total taxes. A comparison with the Eurozone 'big ones' illustrates the huge differences: in Germany, indirect taxes accounted for 26.2% of revenue in 2018, 29.2% in Spain, 24.4% in France. Only Portugal exceeds Greece, with indirect taxes accounting for 39.8% of total tax revenue.

5. BRAIN DRAIN AND TAX COMPETITION BETWEEN COUNTRIES

Apparently brain drain affects the tax competition between countries, as they compete to offer a better tax environment in an attempt to reverse brain drain or to attract highly skilled workers (brain gain).

⁶ See Alcidi, Gros (2019).

Workers usually reside in the place (country) where they earn their income and pay taxes. Given differences in (pre-tax) wages, the decision to relocate then must take into account a package of national tax rates combined with local public goods (e.g. infrastructure and other amenities). This can then lead to beneficial competition among jurisdictions regarding the best package of taxes and local public goods.

Overall, while the temptation to lower taxation on labour to prevent further shrinking of the tax base is strong, this may not work as other factors could be much more important when deciding about moving. Satisfaction with standards of living, including opportunities for children, may be much more relevant than a lower marginal tax rate on income. This could especially be the case for high-skilled workers.

Therefore tax competition for mobile labour should not be an important concern. It would anyway be at odds with the idea that the free movement of workers in the EU is a beneficial aspect of the single market and it also seems to contradict the idea that the movement of workers is a channel for absorbing country-specific shocks.⁸

6. MEASURES TO TACKLE BRAIN DRAIN

One of the first measures in order to tackle brain drain introduced by the previous government in Greece was instated on 31 December 2018. Specifically, the Greek parliament approved new provisions that specified the conditions for the application of Article 71D of the Income Tax Code (law 4172/2013), introducing a "super tax deduction" of the gross revenues for employers' social security contributions, for the creation of new full-time jobs⁹.

More specifically, under the above provision, 150% of an employer's social security contributions for the creation of new full-time jobs are deductible from the gross revenues of legal persons and legal entities (including physical persons engaged in a business activity), up to a maximum of 14 times the minimum wage of an unmarried employee over 25 years of age, provided there is an increase of the average number of employees during the year, compared to the average number the previous year, and there is an increase of the total cost of employee wages during the year, compared to the previous year.

Furthermore, administrative decision No. 1244 provides that the tax incentive applies for the recruitment of young people up to 30 years

⁷ Ibid.

⁸ Ibid.

⁹ The incentive applies starting with the 2019 tax year.

old, at the date of recruitment, and long-term unemployed persons registered with the labour employment office (OAED), or other unemployment funds similar in operation to the OAED, at the time of their recruitment. The incentive also applies to the conversion of part-time or rotation contracts, and service or project-based contracts, into full-time employment contracts.

Finally, it is worth mentioning the measures adopted within the framework of the initiative "We Choose Greece - Building Bridges of Knowledge and Cooperation", which aims to help tackle the brain drain phenomenon abroad by 'connecting' all Greek scientists, no matter where in the world they are, by connecting them with Greece, for as long as they remain outside the country. In this context, after an evaluation of the proposals submitted by a special committee, prizes are awarded to transnational scientific-research cooperation networks, involving higher education degree holders 10 or business partnership proposals for the production of a specific product or service in Greece involving higher education holders ¹¹, who are either of Greek or EU origin, or third country nationals who have residence in Greece under specific conditions. In order to ensure that the eligible persons actually receive the prize money. it is provided that the cash prizes cannot be confiscated, they are not subject to any kind of withholding tax and are not offset by any liabilities of the beneficiary to the Greek State¹².

The new Greek government, elected on July 2019, is planning "a comprehensive tax reform that will have a four-year horizon and will accelerate growth". The overhaul will focus on reducing income and corporation tax, cutting VAT, streamlining tax incentives for investors and abolishing emergency levies imposed during the Greek debt crisis to meet conditions set by bailout creditors. Starting 1 July 2020, the insurance contributions will start to decrease gradually. It will only be provided for full-time employees.

The recently voted Law 4621/2019, which was approved by the Greek parliament on 31 July 2019 (several days after the elections),

Specifically, those who are either of Greek or EU origin, or third country nationals who have had residence in Greece for at least three years, and who reside and work in the country they represent for at least one year at the time of application. Each network is represented by at least two (2) countries, one of which must be Greece.

¹¹ Specifically, those who are either of Greek origin or EU or third country nationals who have resided in Greece for at least one year three years, and who have resided and worked abroad for at least one year at the time of the application.

Financing of the total amount of money distributed is financed by the national part of the Public Investment Program (PIP). The total amount of funding may not exceed three million euros per year. The total annual amount of aid is the budget of the number of projects in the Public Investment Program and is determined by decision of the Minister of Finance and Development.

significantly reduces the Annual Real Estate Ownership Tax (ENFIA) paid by individuals. The ENFIA, which is comprised of a main tax and a supplementary tax, is levied on real estate located in Greece, owned by legal entities and individuals. The new legislation also makes improvements to the instalment payment option for tax liabilities that was introduced in May 2019 (law 4611/2019).

Furthermore, the prime minister presented at the 84th Thessaloniki International Fair the main tax measures that his government intends to submit for a vote to Greek parliament, in order to promote economic growth in Greece and also deal with brain drain. He announced the reduction of the corporate income tax rate from 28% to 24%, the reduction from 10% to 5% of the withholding tax on dividends and the reduction of the lower income tax rate for individuals, for income up to 10.000 euros, from 22% to 9%. Furthermore, in order to promote the construction sector, he announced the introduction of a 40% discount on costs related to building renovations. In addition to the above tax incentive for building renovations, he announced the suspension of the payment of VAT on new buildings (for three more years) and the postponement of the imposition of capital tax on the transfer of real estate.

Finally, the prime minister announced further tax measures to be introduced in the next fiscal years, after 2020 and 2021, and specifically the abolishment of the business duty on entrepreneurs and self-employed, the introduction of an accelerated depreciation rate (up to 200%) for specific investments, the progressive reduction of the solidarity contribution on income that was introduced as an urgent measure during the economic crisis, and the income tax reduction on benefits in kind provided by companies to executives and employees, such as vehicles, etc.

The new tax measures announced by the Greek government intend to support the effort for growth and to deal with the brain drain. Lowering tax rates would help the Greek government to achieve the above goals, since higher tax rates influence people choices regarding the location of work and life and generally the mobility of young workers.

Furthermore, at the 84th Thessaloniki International Fair the prime minister announced the creation of a technology park that will host cutting-edge companies, free zones of commerce/free economic zones/special economic zones (SEZs), which can be a model for other free zones of commerce in Greece, especially in some border areas that are very important because of competition to the north. Moreover, the government announced the simplification of the relative procedures specifically for industrial zones.

However, in the course of introducing such free zones of commerce, the Greek government should be cautious, since according to a new report from the OECD and the European Union Intellectual Property Office, titled Trade in Counterfeit Goods and Free Trade Zones the illicit trafficking of products-monkey strengthens, unintentionally, the continuous growth of free trade zones, where economic activity is driven by reduced taxes and customs controls, less regulation and limited supervision. The report notes that exports of counterfeit and pirated products from a given country or economy increase with the number and size of the free trade zones it hosts.

The Greek government has not yet announced the details of these Special Economic Zones. In any case, the government should take into account the Base Erosion and Profit Shifting considerations in order to avoid generating harmful tax practices and competition. In this respect, it should adopt tax measures aimed at avoiding undesirable tax planning structures used by Special Economic Zones, and notably the requirement of substantial activity, controlled foreign company rules, etc. Also, the Greek government should not be overoptimistic, since in addition to tax incentives, other factors also influence the location of incorporation of the companies and corporate decision making in general. Such factors are political and economic stability, legal and tax certainty and transparency, availability of skilled labour, land policy, etc.

Furthermore, until now the Greek government has not introduced measures such as exemption from income tax for young people up to a certain age, which is the case with other countries, such as Croatia and Poland. In any case, analysts have strong doubts about whether such tax reliefs would reverse the brain drain of talented and educated youths to other countries that offer higher wages and other important job opportunities. According to them, tax exemptions, even full exemption, are probably not enough to tackle the brain drain. "They fail to address the root of the problem", explains an Economist Intelligence Unit (EIU) analyst and warns that the flight will be intensified in the long run if there are no effective initiatives to stimulate growth and improve labour market conditions.

It is noted that, in addition to effectiveness considerations, the introduction of such measures, i.e. tax exemptions for young people, may pose issues regarding their conformity with the constitutional principle of tax equality.

If, however, lower tax rates or other tax incentives are not effective in reversing brain drain, can the same be considered responsible for causing brain drain? In 2017 Greece recorded the highest tax rates on labour in the European Union, reaching 43.3%, whereas Bulgaria recorded the lowest rates, at 24.3%.¹³ There is concern that high tax rates might constitute an incentive for emigration, especially in the case of highly skilled persons.

¹³ Alcidi, Gros, CEPS (2019), section 5.1.

However, up to now, there is limited evidence that mobile labour locates on the basis of tax reasons. This seems to be the case even if the semi-elasticity of migration appears higher for higher incomes and some countries had successful preferential tax treatment for high-earning foreigners.¹⁴

7. OTHER PROPOSALS TO TACKLE BRAIN DRAIN: THE EU LAW PERSPECTIVE

Literature has proposed that (usually temporary) restrictions on emigration be imposed on highly skilled or highly educated persons who have received educated at the public expense. Such restrictions, however, on the free movement of workers, even temporary, are not compatible with the freedom of the movement of workers with in the EU, as protected by the TFEU, and therefore do not constitute a realistic proposal.

Another measure that has been proposed is the imposition of what is referred to as a 'Bhagwati tax' on emigrating skilled workers. A Bhagwati tax is, most generally, an 'exit tax' paid by a would-be emigrant with the intent of compensating their country of origin and for the training investment made in their skills. Again, an exit tax that would constitute an obstacle to the free movement of workers within the EU would be very difficult to reconcile with the fundamental freedoms and therefore this measure would not be a viable proposal for addressing brain drain in the EU. The proposal of a form of tax credit would not entail such concerns.

Under this proposal, the receiving country, which is the country of residence of the highly skilled/educated worker, would take on the responsibility to assess the taxpayer with the tax due in their origin country, remit the tax so due to the origin country itself, and give an equivalent tax credit from the tax due in the worker's country of residence. Within the EU there is a very high level of cooperation between tax authorities and this would make this proposal feasible and easy to implement. Such a measure, however, would require political consensus within the EU and with the states competing for tax revenue this seems rather far reaching at this stage of integration of the EU.

¹⁴ Ibid.

¹⁵ See the analysis in Lister (2017, 73 et seq., and n 4), who refers to the proposal by Gillian Brock, Prosperity in Developing Countries, the Effects Departing Individuals Have on Those Left Behind, and Some Policy Options, in *Debating Brain Drain: May Governments Restrict Emigration?*, edited by Gillian Brock and Michael Blake. 36, 37 (2015), aimed at protecting the workforce in developing countries.

¹⁶ See the description and comments in Lister (2019, p. 79, n 26).

¹⁷ See Lister, *ibid*. Lister has loosely based his proposal on the foreign tax credit that is available to US citizens living and working abroad.

8. CONCLUSION

The use of tax policy to address brain drain has been proposed by several authors, institutions and governments around the globe. In this regard, the proposed or introduced tax measures include the co-called 'Bhagwati tax', a form of exit tax, the use of tax credits, some of them modelled on the foreign tax credits US citizens receive in certain situations for taxes paid in other countries, tax incentives for the increase of investments, or even more the suppression of income tax or other tax reliefs for young people up to certain age (e.g. 25 or 30 years old).

Tax policy can contribute to the effort to address drain brain, and specific tax measures can be more efficient than others in stimulating growth in general. Nevertheless, it is strongly proposed that, having in mind fiscal considerations of course, to adopt a larger tax reform, with a reduction of the effective top/marginal rate and the adoption of tax incentives for businesses aimed at the recruitment of young workers, especially the highly skilled. But the brain regain cannot be resolved only by tax policy, but rather through large scale fiscal and social reform, which would deal with the causes of the drain brain phenomenon. Also, if we wish the Greek scientists who moved abroad to return in Greece, or at least to prevent others from making the same decision, we must eliminate the reasons that led them abroad, i.e. to change the country's economic growth pattern in order to match the supply and demand of skilled human capital in Greece.

REFERENCES

- Alcidi, Cinzia, Daniel Gros. 2019. EU Mobile Workers: A challenge to public finances?, https://www.ceps.eu/wp-content/uploads/2019/04/EU%20Mobile%20Workers.pdf (last visited 22 November 2019)
- European Committee of the Regions, Commission for Social Policy, Education, Employment, Research and Culture. Addressing brain drain: The local and regional dimension. https://cor.europa.eu/en/engage/studies/Documents/addressing-brain-drain/addressing-brain-drain.pdf (last visited 22 November 2019)
- Lister, Matthew. 2017, A Tax-Credit Approach to Addressing Brain Drain, *Saint Louis University Law Journal* 62: 73–84.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

UDC 336.22:314.15(100)

CERIF: S141

DOI: 10.5937/AnaliPFB1904249S

Fernando Souza de Man*

TAX MEASURES TO COMBAT BRAIN DRAIN: (IN) COMPATIBILITY ISSUES WITH DOUBLE TAX CONVENTIONS AND A POTENTIAL WAY FORWARD

This article analyzes the interaction between domestic tax legislation applied to avoid or combat a brain drain and the OECD and the UN model tax conventions, the two main models used by states in tax treaty negotiations. After it is demonstrated that brain drain taxes are incompatible with the current tax treaty network, the author presents alternatives that could be included in the model tax conventions, and consequently in tax treaties, to establish the compatibility of the measures, as well as a justification for the adoption of these alternatives in tax treaties involving developing countries.

Key words: Brain drain. – Taxation of emigrants. – Tax treaties. – Right to development. – Tax benefits.

1. INTRODUCTION

Since the migration of highly-skilled labor intensified after the Second World War, a discussion arose regarding the effects that such migration would have on the state of emigration. At first, it was thought that the emigration state would be worse off due to the emigration of highly-educated/highly-skilled citizens, so the term brain drain was coined in literature. It was also argued that this process would have even

^{*} Assistant Professor, International and European Tax Law, Maastricht University, fernando.deman@maastrichtuniversity.nl.

According to Dumitru (2012, 9 n. 3) the term brain drain was created in the 1960s by British tabloids to refer to the emigration of British scientists to the United States.

direr consequences in cases where the highly-skilled/educated individuals moved from a developing country to a developed country, so states and scholars started to analyze how to deter this occurrence or to at least compensate these less developed countries.

Amongst the ideas ventilated, there were proposals for the taxation of the emigrating persons by their former residence states (Bhagwati, Dellalfar 1972, 1-28; Bhagwati, Dellalfar 1973, 94-101) or for the establishment of restrictions on emigration.² Both ideas were met by considerable criticism. Regarding the establishment of a tax on emigrants. it was stated that the taxation would not be feasible and enforceable for developing countries (Oldman, Pomp 1975, 752), that the responsibility for the eventual adverse effects of migrations should not be on the emigrating person (Sager 2014, 573-576), that the migration is the consequence of a problem (Sampson 2013, 162-163) and that the migration can also be beneficial for the emigration state (McAusland, Kuhn 2006, 15–17; Agrawal, Kapur, McHale 2008, 1; Kumar 1967; Commander, Kangasniemi, Winters 2004; Hewitt 2007, 15-39). As for the restriction on the individual's emigration, it is argued that it is a morally questionable choice (Blake 2015, Part II). Ultimately, the proposals were never adopted on a wide scale,³ albeit discussed in considerable detail at the academic level.

The increase in cross-border mobility since the 1990s has renewed the discussion, with an additional idea that states should not punish the individuals for deciding to emigrate; they should rather provide incentives for individuals to stay. Moreover, even though initially the issue had been framed mainly from the perspective of the less developed countries, more attention started being paid also to migrations between and within developed countries, with

legislators taking action to try to discourage migration of the highly-skilled/educated migrants to more developed regions of countries or to other developed countries.⁴

Despite the recent advances in the study of the topic, discussions are focused mainly on whether countries should be able to tax former

² According to Brock (2015, 73–74), states could establish that individuals that complete their studies, with or without a scholarship or loan, in a public or private university, must provide compulsory services to that state for a period of time.

³ Bhagwati, Dellalfar (1972, 26) mention a tax applied by the Soviet Union on Soviet Jews who wished to emigrate to Israel, but by its characteristics it was an exit tax, not a tax like the one they proposed.

⁴ On that issue see the recent tax exemption regime instituted by Poland as well as measures taken by states in the United States to provide tax benefits for graduates that remain in the state. The analysis from the perspective of developed countries is in line with the origin of the discussion during the brain drain of European scientists to the United States after the Second World War.

residents and how this can be done, with little regard to the consequences of potential taxation in the tax relations between states that have signed a convention to avoid the double taxation of income. In that sense, it needs to be assessed whether the potential taxation prescribed in the domestic legislation of the emigrant's country would not be limited by a double tax convention signed by the emigration state. To do so, first it is necessary to define whether such taxes would be under the scope of the convention, the article that would be applicable, and the allocation rule in place. Furthermore, considering that the tax would fall under the scope of the convention and that the taxing rights of the emigration state would be limited, it is important to consider how such taxes could also be applied in a tax treaty situation.

This is the objective of this article, to assess whether the potential domestic prescription of the taxation of income earned abroad by immigrants who used to live in a country would be in line with double tax conventions based on the Organization for Economic Cooperation and Development (OECD) and United Nations (UN) model tax conventions. Moreover, in the event that such taxation is considered in contravention of the current international rules on avoiding double taxation, as prescribed by the model tax conventions, it will be discussed whether it is possible, and feasible, to reconcile these two systems, as well as how to perform this reconciliation.

Therefore, instead of focusing on whether countries should adopt tax measures to avoid brain drain or on analyzing measures that have already been adopted by specific states and their possible effects, this article will consider a scenario in which states have made the decision in their domestic legislation to either establish barriers for emigration, such as the imposition of a tax on future earnings, or to provide a beneficial tax treatment for individuals who decide to remain in their home states. From this starting point, the author will analyze whether these rules are in line with the current international framework of double tax conventions. After the (in)compatibility of the measures has been assessed, the author will look at the possible amendments that could be made to model tax conventions so that states that wish to enforce their domestic rules on curbing the brain drain are not restricted by international tax treaties.

2. THE BRAIN DRAIN CONUNDRUM

As mentioned above, this work will not focus on whether the levying of taxes on highly-skilled/educated individuals who have

⁵ As notable exceptions in this matter, see Brauner (2010) and Stevenson (2016).

emigrated is the right manner in which to deal with brain drain or not;⁶ whether it is better to provide positive reinforcements via tax breaks to individuals who remain or through any other eventual alternatives that exist for their emigration states. The assumption of this article is that the individuals' home state has already analyzed the best manner to deal with the brain drain, and that it is then necessary to verify whether the option adopted in the domestic legislation is in line with its model tax conventions. Before performing this analysis, it is crucial to understand the reasons why states see brain drain as an issue and want to avoid it or at least to guarantee taxing rights over the income of the emigrating person.

2.1 Emigration as a Problem: Brain Drain

One of the first reasons used to support methods to combat brain drain is that brain drain leads to a loss of revenue and welfare in developing countries (Bhagwati, Dellalfar 1972, 1-3; Bhagwati, Dellalfar 1973, 95; Brock 2015, 38). It is also argued that brain drain leads to a shortage of skilled labor in the emigration country (Lister 2017, 78) and this shortage can lead to further problems, especially in case of developing countries, where the number of highly-skilled/educated people is already scarce. It is also assumed that the emigration is a loss of the investment made by the developing country in the individual (Lister 2017, 78; Brauner 2010, 229), so this should be avoided. Furthermore, brain drain may be viewed as a subsidy from developing countries, which financed the education of the individuals, to the state to which the individual will emigrate (Freitas, Levatino, Pécoud 2012, 3; Altbach 2013, 42; Kuehn 2007, 1854), and may hamper the spillover effect and the development of institutionbuilding assets, as studies have shown that higher educated people are more pro-democracy, so when they leave the country the local support for democracy may also diminish (Brock 2015, 40).

2.2 Emigration as Beneficial: Brain Gain/Brain Circulation

On the other hand, it is argued that the migration of highly skilled/educated individuals leads to a brain grain, with the circulation of knowledge, with diaspora effects (OECD 2008; Hewitt 2007, 15–39). Additionally, it is argued that the possibility of leaving provides incentives for individuals to acquire further skills, the income they send back is substantial, and if they return they might bring with them progressive ideas and enhanced human capital (Brock 2015, 40–41; Patterson 2007, 12; McAusland, Kuhn 2006, 19–20; Haupt, Janeba 2004, 21; Agrawal, Kapur, McHale 2008, 1–4). It is also said that countries export citizens,

⁶ For a better understanding of some of the factors that might contribute to the brain drain in developing countries, see Docquier, Lohest, Marfouk (2007).

just like they export goods, and the idea of a loss assumes that the brain would be used at home, which is not always the case (Kumar 1967, 2079).

If it is believed that in the long-term the emigration fosters the development of the emigration state, countries should be drafting legislation to further incentivize the emigration of highly-skilled/educated individuals. However, despite this new line of research, it is still assumed that the probability of a brain drain is more likely (Lien, Wang 2005, 160), so discussions focus on what developing countries can do to restrict or limit this emigration.

2.3. Possible responses to Brain Drain

Based on the assumption that brain drain has a negative impact on a country's economy, proposals have been made on measures that could be adopted to curb this phenomenon. These proposals make use of the tax system in different manners, increasing the cost of a person's move by levying a tax or by providing incentives for the individual to stay in their residence state, and these possibilities will be studied below.

3. TAXES AS A BRAIN DRAIN DETERRENCE

One of the most prominent proposals on how countries should deal with brain drain is the Bhagwati tax, proposed in the 1970s by Jagdish Bhagwati. According to Bhagwati's original proposal, emigrants should have to pay a tax in their new residence state, to compensate the losses of the emigration states. This tax, which was viewed as payment to the developing country for allowing the individual to move abroad, would also reduce the incentive for individuals to move abroad (Bhagwati, Dellalfar 1973, 95).⁷

According to Bhagwati, the tax should be levied after immigration, on the income effectively earned, as opposed to prior to the emigration on expected income, and the tax should be collected by the developed country that received the emigrant. The tax would be a surcharge, in the sense that emigrants would then be subject to a higher tax liability than other residents of this state. Bhagwati favored the idea that the collection should occur during the whole life of the emigrant, but since this would most likely not be accepted by the developed countries, he proposed that

When analyzing the Bhagwati tax, John Douglas Wilson stated that such tax is desirable and that it could be a voluntary tax (Wilson 2011; Wilson 2008; Wilson 2005). Furthermore, it is argued that a brain drain tax can increase the welfare of the remaining residents (Scalera 2012, 447–467).

⁸ Later, recognizing the issues with the collection by developed countries, Bhagwati focused on the tax being collected by developing countries.

the developing countries should be able to tax their former citizens for up to 10 years after emigration (Bhagwati, Dellalfar 1973, 96).

Regarding eventual obstacles to the collection, in the domestic legislation of developed countries, Bhagwati argued that the laws could be changed to accommodate this taxation. Also, he argued that the administration of the money and transfer to the developing countries should be done by the United Nations (Bhagwati, Dellalfar 1973, 95–96).

The proposal has been subject to considerable critics and a tax like the one proposed by Bhagwati has not been widely implemented, despite some of its characteristics being similar to those of an exit tax.

3.1. Brain Drain Taxes and the Model Tax Conventions

Now that we have briefly explained the issue of brain drain/gain and the proposal to limit (by means of income tax) the alleged losses that states suffer when skilled individuals emigrate, it is time to assess whether this tax would be in line with the OECD and the UN model tax conventions on double taxation and the double tax treaties using these models as a reference.

As the brain drain/gain issue focuses on the migration of individuals, in this analysis we will take a closer look at the taxation of individuals in model tax conventions, whether running their own business or working as an employee. Thus, in this section we will assess the compatibility of taxes such as the Bhagwati tax in light of articles 7, 15 and 21 of the OECD and the UN model tax conventions, as well as former Article 14 of the OECD Model Tax Convention and articles 12A and 14 of the UN Model Tax Convention. Articles concerning the receipt of passive income and the alienation of assets will not be covered, as the brain drain/gain debate focuses on the taxation of income that is earned by the emigrant when performing an economic activity. The provisions on pensions will not be dealt with for the same reason, while the article dealing with artistes and sportspersons is beyond the scope of this paper due to its special nature, which already modifies the treatment granted to entertainment activities when compared to other economic activities.

For that matter, we will consider the situation of John Doe, who emigrates from State A to State B, becoming a tax resident of the latter. The tie-breaker rule of Article 4(2) will be applied in the event that John Doe is also considered a tax resident of State A, according to this state's domestic rules.

3.1.1. Brain Drain Tax and its (In)Compatibility with the OECD Model Tax Convention

3.1.1.1. John Doe Earns Business Profits

The OECD Model Tax Convention prescribes, in Article 7, that profits from an enterprise of a contracting state⁹ are taxed solely in that State, unless the enterprise carries on business in the other contracting state through a permanent establishment therein. The term business, as prescribed since the deletion of Article 14 in 2000, also includes the performance of personal services and of other independent activities.¹⁰

Analyzing John Doe's situation considering Article 7, it becomes clear that the profits that he makes by providing services, for instance, as a doctor, will be taxed only in his state of residence. The question then becomes where will John Doe be resident based on Article 4 of a double tax treaty based on the OECD Model Tax Convention.

If following emigration John Doe is a resident solely of State B, State A will not be able to tax any income earned by John Doe, unless he maintains a permanent establishment in his former state of residence and the profits are attributed to the permanent establishment. As the discussions on brain drain/gain focus on taxing the emigrating person on their worldwide income, irrespective of where it was earned, this possibility will not be analyzed in this paper. The focus is, ultimately, on the compatibility of a brain drain tax with the general rule of Article 7, taxation exclusively in the enterprise's state of residence.

Thus, it becomes clear that a domestic tax for emigrants on the income earned after they moved would generally not stand the compatibility test with Article 7, since the emigrant does not commonly maintain resident status in the emigration states. ¹² So, in our example, based on Article 7, only State B would be entitled to tax the profits earned by John Doe after emigration; i.e. taxation of this income by State A,

⁹ The definitions of enterprise and enterprise of a contracting state are given n articles 3(1)(c) and 3(1)(d) of the 2017 OECD Model Tax Convention: "(...) c) the term 'enterprise' applies to the carrying on of any business; d) the terms 'enterprise of a contracting State' and 'enterprise of the other Contracting State' mean respectively an enterprise carried on by a resident of a Contracting State and an enterprise carried on by a resident of the other Contracting State".

¹⁰ As expressed in Article 3(1)(h) of the 2017 OECD Model Tax Convention "(...) h) the term 'business' includes the performance of professional services and of other activities of an independent character".

Doctors are one of the examples normally examined in regard to the brain drain debate, especially the cases of African doctors who emigrate (Kuehn 2007, 1853–1855; Patterson 2007, 9)

¹² In some states, such as Brazil, emigrant can declare that they are not a resident anymore even shortly before he leaves the country.

John Doe's former state of residence, would be restricted by the model conventions and double tax conventions that contain a similar provision on taxation of business profits.

The situation could be different if State A still considered John Doe a resident according to its own domestic law. In line with Article 4 of the OECD Model Tax Convention, John Doe could be viewed, for treaty purposes, as a dual resident taxpayer, and the recourse would have to be made based on the tie-breaker rule of Article 4(2). If John Doe were deemed to be a resident in State B, the situation would have the same outcome as the one explained above: he would not be considered a resident of State A for treaty purposes and this state would only be able to tax his income if this income were linked to a permanent establishment therein.

If, on the other hand, John Doe were deemed a resident in State A. then State A would be entitled to levy income tax on the income earned by John Doe after his emigration. Thus, in the case of a residenceresidence conflict resolved in favor of the former residence state, a brain drain tax instituted by this state would not infringe the provisions of the double tax convention. On this matter, it is also important to stress that if John Doe were deemed to be a resident of one of the contracting states. the other state would not be able to claim residence taxing rights based on the second sentence of Article 4(1). This outcome would not be affected by Article 1(3), the savings clause introduced in the 2017 OECD Model Tax Convention, because for treaty purposes John Doe would no longer be a resident of the other state, despite still being a resident of this state for domestic law purposes. ¹³ A third possibility for a brain drain tax to be in line with a double tax treaty based on the OECD Model Tax Convention would be if the tiebreaker rule did not resolve the residence-residence conflict of the individual, so both states would still be viewed, for treaty purposes, as John Doe's residence state.

Therefore, it remains clear that a brain drain tax can be in line with a tax treaty only in extremely limited situations, i.e. if a residence-residence conflict were resolved in favor of the emigration state or if the residence-residence conflict were not resolved and, as a result, the former resident was still considered, also for treaty purposes, a resident of the emigration state.

Naturally, this domestic brain drain tax could also be applied in cases where the states have not signed a double tax convention, but this does not affect the discussion on the compatibility of a brain drain tax with the model tax conventions and the double tax treaties based on them. Consequently, it can be affirmed that even if a brain drain tax were established by the domestic legislation, it would ultimately be applied in

OECD, 2017 Model Tax Convention, Commentaries on Article 1, para. 21.

restricted situations, residence conflicts where the emigration state is considered John Doe's residence state, unresolved residence-residence conflict situations, and non-treaty situations. Even if it is considered that developing countries are, with a few exceptions, 14 conservative in the signing tax treaties, having a more modest tax treaty network, these cases still seem more an exception than the rule.

On that matter, despite the arguments that can be put forward in favor of the taxation of brain drain, and the fact that such taxation could occur only for a short period of one's lifetime, it remains clear that such taxation, except on the specific cases mentioned above, would be a clear violation of a treaty obligation, and that an eventual valid reasoning for a tax cannot supersede the express wording of a legal obligation as the one assumed under a double tax treaty.

It is interesting to note that in the original proposal Bhagwati suggested that the taxes should be collected by the developed country, but this idea was criticized based on eventual restrictions that the domestic laws of these states might establish on the tax collection on behalf of a foreign tax authority. Although the criticism is valid, as domestic legislations can indeed hamper the intended tax collection, it are incomplete, as the states' international obligations are not considered. As a matter of fact, such structure is the only one in line with Article 7 of the OECD Model Tax Convention since its inception in 1963, if considered that the residence state of the person for treaty purposes is the state to which they emigrated.

Nonetheless, it should be added that even if taxation is done by the developed country, complying in this sense with the wording of Article 7, such taxation could still run foul to treaty obligations, especially if the tax is levied as a surcharge, as originally proposed by Bhagwati. Such a surcharge tax would be in direct conflict with the non-discrimination provision of Article 24(1), because in that case State B would be taxing nationals of State A who are residents in the former, thus in the same circumstance as its own residents, in a more burdensome manner.

Ultimately, the establishment of a brain drain tax based on the proposals made since the 1970s would not produce any effects in treaty situations in which emigrants are carrying on their business as entrepreneurs in the developed country, as such taxation is not allowed based on Article 7 of the OECD Model Tax Convention and double tax treaties modeled after this provision.

3.1.1.2. John Doe Earns Income from Independent Personal Services

If the double tax treaty between State A and State B did not contain provisions on the taxation of income from independent personal services;

¹⁴ e.g. India, which has a broad tax treaty network.

complying with the 2000 OECD Model Tax Convention, this income would be taxed under the scope of Article 7. Hence, the outcome would be the same as prescribed in the previous section, i.e. the treaty restriction on the taxation of income by the emigration state. Nonetheless, one may wonder whether this outcome would be different in treaties that still contain former Article 14 of the OECD Model, such as most treaties signed by developing countries. On that matter, it is important to note that the outcome would not be affected by former Article 14 of the OECD Model Tax Convention.

As expressed by the OECD when arguing for the removal of the provision, it is unclear whether there were any differences between the concepts of permanent establishment and fixed base (OECD 2000). And even if it is argued that the fixed base concept is broader, allowing for the easier establishment of a fixed base when compared with the permanent establishment, this eventual difference would have no bearing on the current situation, as State A is not intending to exercise taxing rights as a source state, but rather as a residence state, based on the fact that the highly educated/skilled individual emigrated.

In the case of John Doe, the emigration state would only be able to assert taxing rights via a brain drain tax if it was deemed to be, for treaty purposes, the residence state of John Doe. On that matter, the possibility for a brain drain tax to be compatible with a double tax convention would be, once again, the situation of a residence-residence conflict resolved in favor of the emigration state or an unresolved residence-residence in which the individual remained, also for treaty purposes, a resident of the emigration state. Hence, the prohibition of brain drain taxation also holds true for double tax conventions containing former Article 14 of the OECD Model Tax Convention.

3.1.1.3. John Doe Earns Income from Employment

The discussions on brain drain usually focus on this specific situation, in which the emigrant is hired by a foreign employer to carry out his activities in a dependent manner, i.e. as an employee of the company. Despite the prevalence of this view, there has hardly been any consistent analysis of the compatibility of the brain drain taxation and Article 15 of the 2017 OECD Model Tax Convention.

According to Article 15, remuneration derived by a resident of a contracting state related to employment is taxable only in that state, unless the employment is exercised in the other contracting state. In that sense, similar to Article 7, the Convention recognizes the primacy of the residence state of the person to levy a tax on their income, unless the

The latter possibility is further clarified by recourse to Article 1(3) introduced in the 2017 Update to the OECD Model Tax Convention.

person is present in the source for a considerable amount of time (more than 183 days) or their employer is a resident in the source state, or the remuneration is borne by a PE in the source state. ¹⁶

The wording of the provision is clear, therefore there is no possibility to construe a theory in which the emigration state, the former residence state of the emigrant, would have taxing rights over the income from employment earned after emigration, unless the employment activities were being conducted therein. In the commentaries there is also no mention of such interpretation by any state, corroborating the idea that apart from the pleas in academic literature for the taxation of brain drain, the issue has not been thoroughly considered by the countries themselves. The situation would naturally be different if the emigration state was, for treaty purposes, still the residence state of the emigrating person, since in that situation Article 15, which focuses on taxation by the residence state of the income earner, would allow for the taxation of the person in the emigration state.

Hence, like the situation involving the potential taxation of business profits and income from independent personal services earned after emigration, the taxation of the income from employment earned after emigration is also not in line with the OECD Model Tax Convention, save in the specific situations already mentioned (residence-residence conflict resolved in favor of emigration state or unresolved conflict which allows the emigration state to still be seen, for treaty matters, as the emigrant's state of residence). Thus, a brain drain tax levied on income earned by John Doe after emigration would have limited applicability in tax treaty situations.

3.1.1.4. John Doe Earns "Other Income"

In cases where the income earned by the emigrating person does not fall under the scope of more specific provisions, such as the ones dealing with the taxation of business profits, independent personal services and employment, it will fall under the scope of Article 21, a catch all clause that also focuses on allocating taxing rights, in an exclusive manner, to the state of residence of the income earner, save in case where the income is earned through a permanent establishment located in the source state.¹⁷

Thus, in the case of John Doe, just as it happens when the income is under the scope of articles 7, 14 or 15, unless he was considered, for treaty purposes, as a resident of the emigration state, this state would not be able to levy a brain drain tax on the income he earned after emigration.

¹⁶ OECD, 2017 Model Tax Convention, Article 15.

¹⁷ OECD, 2017 OECD Model Tax Convention, Article 21.

3.1.1.5. Concluding Remarks on Relation of Brain Drain Taxes and the OECD Model Tax Convention

Taking the aforementioned into consideration, it is clear that irrespective of the existence of a brain drain tax in a country's domestic legislation, this tax would only apply to really limited situations: (i) dual treaty residence with the tie-breaker rule deeming the taxpayer to be a resident solely of the emigration state; (ii) unresolved dual residence conflict, so the taxpayer is still viewed, also for treaty purposes, as a resident of the emigration state; and (iii) non-treaty situations.

It is worthwhile noting that any attempt to justify such taxation by reference to the taxation of unrealized capital gains, which is allegedly permitted by the double tax convention, as described in the commentaries on Article 13,¹⁸ would be vague, as there is an important difference between these cases: while in the taxation of unrealized capital gains there is an actual profit which has been created in the residence state, although it was not yet monetized, in case of the brain drain taxation the tax base will only be created in the future, i.e. at the moment of emigration the taxpayer has not yet earned the income that the residence state wants to tax.

Therefore, it remains clear that the taxation of brain drain, as suggested by Bhagwati and subsequent authors, is not in line with the provisions of the OECD Model Tax Convention, save in very specific circumstances. This means that if the emigration state adopted these provisions in its double tax treaties, the establishment of a brain drain tax in domestic law will barely produce significant effects internationally, which naturally is not the desired result for the implementing country.

3.1.2. Brain Drain Tax and its (In)Compatibility with the UN Model Tax Convention

If, instead of adopting the OECD Model Tax Convention as the basis for its double tax treaties, the states base their negotiations on the UN Model Tax Convention, it remains to be seen whether the outcome would be similar to the one explained above, i.e. brain drain taxes would be severely restricted in face of the double tax treaties.

To reach a conclusion on the issue, we will analyze articles 7, 14 and 15 and 21 of the UN Model Tax Convention, as done in the case of the OECD Model Tax Convention, as well as recently-introduced Article 12A, which deals with the taxation of technical services, since the business carried out by the emigrating person could also fall under the scope of this provision.

 $^{^{18}}$ OECD, 2017 OECD Model Tax Convention, Commentaries on Article 13, paras. $8\!-\!10.$

3.1.2.1. John Doe Earns Business Profits

Article 7 of the UN Model Tax Convention is based on Article 7 of the OECD Model Tax Convention, so it is not surprising that they adopt a similar position regarding the taxation of business profits, i.e. exclusive taxation in the residence state, save if there is a permanent establishment in the source state and the income was earned through a permanent establishment. However, expanding the possibilities for taxation at source, which is exactly the reason why the UN Model Tax Convention was created, Article 7 stipulates that if the income is linked to sales in the other state, of the same or similar goods or merchandise sold by the permanent establishment, or arises from business activities carried on in the other state that are similar to the activities carried out by the permanent establishment, the income will also be taxed at this other state.

Note that, similar to what occurred when analyzing Article 7 of the OECD Model Tax Convention, even though Article 7 of the UN Model Tax Convention would allow the developing country to tax a larger share of the income than Article 7 of the OECD Model Tax Convention, it would still not allow the emigration state to tax income earned by former residents after emigration. That is because the paradigm in international taxation always viewed the developing country as the source state, and the UN Model Tax Convention expands the taxing rights of source states.

In brain drain tax matters, taxation is not based on income being earned in the developing country, but rather by being earned by a person who decided to emigrate. Normally the source of the income will be on the immigration state, which would also be the person's residence state. Thus, as a rule, Article 7 of the UN Model Tax Convention does not allow for the levying of a brain drain tax. The sole exception to this rule occurs, as explained above, if the emigration state is still viewed as the residence state of the emigrating person for treaty purposes, be it by reference to the tiebreaker rule of Article 4(2) resolving the residence-residence conflict in favor of the emigration state or by the lack of a solution by this provision and the continuous view of the emigrating person as a resident, for treaty purposes, of the emigration state. As mentioned above, the latter is further clarified by the existence of a savings clause like Article 1(3) of the 2017 UN Model Tax Convention, which guarantees that the residence state of the individual is not limited on taxing its own residents, save in specific situations which do not include the one at hand, i.e. no solution for dual residence conflict. If this occurs, a brain drain tax would indeed be in line with Article 7 of the UN Model Tax Convention.

The issues concerning whether the new residence state or the former will levy the tax and eventual constitutional restraints to this levying is not affected by the choice for Article 7 of the UN Model Tax Convention, as this is a matter of domestic law. But as mentioned above,

the choice for taxation by the new residence state, which could then transfer the income to the former residence state, would comply with the wording of Article 7 if the new residence state is viewed as the residence state of the person for treaty purposes.

Therefore, in the case of John Doe, State A would still not be able to levy a domestic brain drain tax if State A and State B signed a double tax treaty in line with the UN Model Tax Convention, irrespective of the fact that the UN Model Tax Convention has broader taxing rights than the OECD Model Tax Convention

3.1.2.2. John Doe earns Income from Independent Personal Services

Differently from the OECD, the UN Model Convention maintained Article 14. Similar to the situation concerning Article 7, Article 14 focuses on income being taxed only in the residence state of the income earner, save if specific circumstances occur, i.e. the person has a fixed base regularly available in the other state or spends more than 183-days in such state. In that sense, the presence of the second test broadens the possibility of taxation of income at source when compared to the OECD Model Tax Convention.

The crux of the issue is indeed that this provision allows for taxation at source when economic activities are developed therein. In the case of a brain drain tax, which disregards the place where income was earned and focuses solely on the fact that the person emigrated from the state, Article 14 does not leave any leeway for the emigration state to tax the future income earned by the emigrant.

Hence, as has been a constant regarding the discussions of brain drain taxation and their interaction with double tax treaties, such provision only allows for brain drain taxation if the emigration state is, for treaty purposes, the residence state of the emigrant. As seen before in this article, this would only occur in case there was a residence-residence conflict that were resolved in favor of the emigration state or unsolved.

Considering that normally the emigrant will become a resident of the state which he moved and will have at least most of his personal or economic relations attached to this state, it remains clear that in only a handful of cases a brain drain tax would be allowed, with Article 14 generally restricting the taxation by the emigration state.

In the case of John Doe, if he ceases to be a resident of State A due to the emigration, this state loses the right to levy a tax on his income from independent personal services. But if for some reason he retains residence in State A while also establishing residence in a different state, it would be necessary to check the facts and circumstances to assess where he is a resident for tax treaty purposes. If he is deemed to be a resident of State B, State A would not be entitled to levy a brain drain tax

in treaty situations. On the contrary, if he is deemed to be a treaty resident of State A, either by application of the tiebreaker rule or if there is no decision on the matter so that he remains, for domestic law and treaty purposes, a resident of State A¹⁹, the latter would be entitled to levy a brain drain tax over his income from independent personal services after emigration. Once again, it seems that such taxation would be the exception rather than a rule.

3.1.2.3. John Doe earns Income from Employment

Article 15 of the 2017 UN Model Tax Convention is an exact reproduction of Article 15 of the 2017 OECD Model Tax Convention, so the conclusions presented above are also valid at this point ²⁰. On that sense, taking the specific case of John Doe in consideration, income he earns after emigration shall be taxable solely in State B, save if the activities are done in State A for more than 183 days, or he is paid by an enterprise resident in State A or the payment is borne by a permanent established located in this state. In any case, taxation would occur because State A would be the source state of the income, not due to the levying of a brain drain tax.

As mentioned above, a brain drain tax would only be in line with the taxation of income from employment as prescribed in model tax conventions if State A is still viewed as the residence state of John Doe for treaty purposes. The possibilities of this happening are scarce, i.e. dual residence situation in which State A is still considered to be his residence state, be it because the tie-breaker rule decides in favor of this state or because the dual residence conflict is not resolved.

3.1.2.4. John Doe earns "Other Income"

Article 21 of the 2017 UN Model Tax Convention adopts the same general rule of Article 21 of the 2017 OECD Model Tax Convention, i.e. items of income not dealt with in the distributive rules shall be taxable only at the residence state of the income earner. However, this provision expands on the approach adopted by the OECD, dealing on Article 21(2) also with independent personal services performed from a fixed base and adding a provision that allocates taxing rights to source states as regards income arising in that state.

Despite the broader prescription of source taxing rights, like the situation with Article 7 of the 2017 UN Model Tax Convention, Article 21 has no different bearing on matters of compatibility of brain drain

¹⁹ The latter possibility is reinforced by the introduction of a provision such as Article 1(3) of the UN Model Tax Convention in the double tax treaty between the parties.

²⁰ See section 3.1.1.3.

United Nations, 2017 UN Model Tax Convention, Article 21(1).

taxes with model tax conventions than Article 21 of the 2017 OECD Model Tax Convention. That is why, once again, the focus of the UN model tax convention is to grant more taxing rights to the source state, but in the case of brain drain taxes the states want to assert taxing rights over income earned by former residents, irrespective of where the income was sourced.

Therefore, to assess whether a brain drain tax would be in line with Article 21 of the 2017 UN Model Tax Convention, we need to ascertain where the emigrating person is a resident. In our example, if John Doe is a treaty resident of State A, Article 21 would determine that this state can tax the income earned abroad after emigrating. But if he is a treaty resident of State B, the levying of a brain drain tax in treaty situations would run foul to the international obligations assumed by the signing of the tax treaty. Once again, considering that John Doe moved to State B, the only chance for the levying of a brain drain tax by State A would be if there was a dual residence conflict resolved in favor of State A, or if there was no solution to the conflict and, as a result, John Doe would still be a resident of State A for treaty purposes.

3.1.2.5. John Doe Earns Income from Technical Services

While in the OECD Model Tax Convention technical services fall within the scope of the business profits article, since 2017 there has been a specific provision in the UN Model Tax Convention dealing with the issue. This provision answers the call of developing countries to detach the taxation of technical services from the permanent establishment concept.²² In that matter, the article prescribes that fees for technical services arising in a state and paid to a resident of the other contracting state, may be taxed in the latter.²³ This is a rather unusual wording, as typically model tax conventions use the "may be taxed" formula to establish that the source state may tax.

To eliminate any controversy in regard to where fees for technical services arise, the article determines that fees for technical services arise in the state in which the payer of the fees is resident or in the state in which the payer of the fees has a permanent establishment or fixed base, and the fees are borne by this permanent establishment or fixed base. This sourcing rule is complemented by another one which states that fees for technical services are not deemed to arise in a state in the case where the payer is a resident of that state, and carries on business in the other

²² In practice some developing countries where already avoiding the need for the existence of a permanent establishment by inserting fees for technical services under the scope of the royalties article.

United Nations, 2017 UN Model Tax Convention, Article 12A(1).

United Nations, 2017 UN Model Tax Convention, Article 12A(5).

contracting state through a permanent establishment, or performs independent personal services through a fixed base, and the fees are borne by the permanent establishment or fixed base.²⁵

The provision also clarifies that fees for technical services may be taxed in the contracting state in which they arise, but if the beneficial owner is a resident of the other contracting state the tax levied at the source state will not exceed a certain percentage of the gross amount of the fees paid.²⁶ The beneficial ownership concept is the also adopted on the articles on dividends, interest and royalties. Moreover, this provision establishes the relationship between Article 12A and other provisions of the 2017 UN Model Tax Convention. In that sense, it can be ascertained that in case of a potential conflict between Article 12A and Article 14, the former will apply if there is no fixed base in the source state, while the latter will apply if the income is linked to a fixed base in the source, as expressly provided on Article 12A(4).

The provisions of articles 8, 16 and 17, on the other hand, prevail over Article 12A, which means that even if the income from shipping, director's fees and remuneration of top level managerial officials, the earnings of artistes and sportspersons could be classified as fees for technical services, as defined on Article 12A(3), the rules prescribed on articles 8, 16 and 17 would apply, i.e. there would be no restriction for the taxation of the income at source, which is a different outcome than the one prescribed by Article 12A, which limits taxation to a percentage (to be agreed by the contracting parties) of the gross amount of the fees paid. If, however, the payments are not under the scope of articles 8, 16 and 17, Article 12A still determines the taxation of the income at source, albeit in a limited manner.

The article also provides a treaty definition of the term *fees for technical services*, stating that it entails any payment for services of a managerial, technical or consultancy nature, unless the payment is made: (i) to an employee of the payer; (ii) for teaching in or by an educational institution; or (iii) by an individual for services for the personal use of an individual.²⁷ Unfortunately, there is no definition of the terms managerial, technical or consultancy, with the commentaries recognizing that these terms may overlap.²⁸ On a positive note, the definition of fees for technical services does not make any mention of domestic law of states, which means that the provision intends to establish an autonomous definition of the term, which is a more beneficial approach to avoid conflicts of

United Nations, 2017 UN Model Tax Convention, Article 12A(6).

²⁶ United Nations, 2017 UN Model Tax Convention, Article 12A(2).

United Nations, 2017 UN Model Tax Convention, Article 12A(3).

United Nations, 2017 UN Model Tax Convention, Commentaries on Article 12A, para. 67.

interpretation and subsequent recourse to the domestic law of states based on Article 3(2).

The shared allocation of taxing rights, with a limit on the amount to be taxed at source, is not applicable if the beneficial owner of the fees for technical services, while resident of a contracting state, does business in the other contracting state though a permanent establishment or performs independent personal services in the other state, through a fixed base, and the fees are effectively connected to the fixed base or permanent establishment, or with business activities of the same or similar kind as the ones carried through the permanent establishment or fixed base.²⁹ If this occurs the situation will fall under the scope of Article 7 (business profits) or Article 14 (independent personal services), following the regulation prescribed in these articles. As a shortcoming of this provision, one that can lead to considerable discussion between the states, there is no definition of the expression "effectively connected".

Finally, the article states that when, due to a special relationship between the payer and the beneficial owner of the fees or between both and another person, the amount paid as a fee for technical services is not at arm's length, the article applies only to the arm's length amount, with the excess part remaining taxable according to the laws of each contracting state and considering the provisions of the double tax treaty. On that matter, the commentaries on the provision clarify that the expression "special relationship" covers not only situations of direct and indirect control, but also relationships by blood or marriage. Moreover, it is important to stress that this is the same treatment granted on the articles regarding excessive royalties and excessive interest.

Applying these provisions to the case of John Doe, it can be said that after emigrating from State A he could be taxed on fees for technical services in the state in which the fees arise, i.e. where the payer of the services is a resident or where there is a permanent establishment or fixed base connected to the obligation to pay the fees and which bears the costs of the fees, or in his residence state. Considering this, as expressed above, brain drain taxes are not based on the idea of the business being developed in a state, but rather on the fact that they are earned by a former resident, it remains clear that the state does not intend to ascertain its taxing rights as a source state, reason why this possibility will not be analyzed in the present work.³²

²⁹ United Nations, 2017 UN Model Tax Convention, Article 12A(4).

United Nations, 2017 UN Model Tax Convention, Article 12A(7).

³¹ United Nations, 2017 UN Model Tax Convention, Commentaries on Article 12A, para. 130.

³² It it important to note that if this were the case, State A, as the source state, would be able to levy a tax based on Article 12A.

Therefore, based on Article 12A. State A would only be able to levy a brain drain tax if it were still viewed as John Doe's residence state. As expressed on previous sections of this article, this would only happen if based on State A's and State B's domestic law John Doe is still viewed as a resident, according to the factors referred to in Article 4(1) of the double tax treaty between the states, and if the dual residence issue is subject to the tiebreaker rule, with the matter being resolved in favor of State A being the residence state, or if no decision was reached and John Doe is still viewed, for treaty purposes, as a resident of State A. In the latter case, if the treaty contains a provision like Article 1(3) of the 2017 UN Model Tax Convention it would be clear that State A would be entitled to levy its brain drain tax, but even if the provision were not present, this state would still be able to levy a brain drain tax, as the convention has not established any restriction on the domestic law taxation by State A as this state is still considered to John Doe's residence state.

3.1.2.6. Concluding Remarks on Relation of Brain Drain Taxes and the UN Model Tax Convention

Similar to the situation involving the OECD Model Tax Convention, as a rule, the signing of a double tax treaty based on the UN Model Tax Convention severely hampers the application of brain drain taxes. The differences between the OECD and the UN model tax conventions, with more taxing rights being attributed to source states in the latter, including the addition of a specific provision on fees for technical services, have no bearing on the discussion of the compatibility of brain drain taxes with double tax treaties.

This outcome is not surprising, since the focus of the UN Model Tax Convention is on allowing more taxation at the source and the discussion on brain drain taxes does not focus on taxation based on the source criterion, but rather on taxation by the emigration state. As the emigration state intends to exercise worldwide taxing rights over the income of individuals who have migrated, the only manner in which this can be achieved in a treaty situation is if the emigration state is still viewed, for treaty purposes, as the residence state of the emigrating person.

In the example above, regarding the taxation of John Doe by State A, this would mean that this state would still need to be considered John Doe's residence state to be able to levy a brain drain tax on him. For domestic law purposes that is not a problem, as the state itself determines who is viewed as a resident, but the issue is more complex when the states concerned have signed a double tax convention. This is because the tax treaty purports to determine the sole residence state of the individual,

and chances are that since John Doe has moved to State B and is currently living there, this state will also treat him as a resident under its domestic law, triggering a dual residence conflict. In that case, Article 4(2) would be applied to determine his treaty residence.

As it currently stands, Article 4(2) determines residence first based on where the individual has a permanent home. As John Doe emigrated from State A, chances are that he would only have a permanent home in State B, which would mean that the conflict would be resolved in favor of State B and the brain drain tax instituted by State A would not apply in the situation. If he maintains a permanent home in state A as well, it would be necessary to investigate his center of vital interests, where his personal and economic relations are. Once again, it is more likely that it would be in State B, but here it is possible that his center of vital interests, specially his personal relations, would still be in State A.

In case it was not possible to determine his center of vital interests, or if John Doe does not have a permanent home in either state, then the tiebreaker rule would determine that his residence state is where he has his habitual abode, i.e. where he is regularly. Once more it is more likely that the conflict would be resolved in favor of State B. If he has a habitual abode in both states or in neither of them, he would be deemed a resident only of the state of which he is a national. This is the first test which most likely will favor State A, although this is not certain. Ultimately, if he were a resident of both states or neither of them, states would have to solve the matter by recourse to a mutual agreement procedure. Therefore, as it can be seen, the chances are greater that John Doe would indeed be considered a treaty resident solely of State B, which would bar the levying of a brain drain tax by State A.

In a nutshell, by moving to State B, based on Article 4(1) John Doe would most likely be viewed solely as a resident of this state, which would mean that State A would not be allowed to levy a brain drain tax. Even if, for domestic law purposes, John Doe were still considered a resident of State A, the recourse to the tiebreaker rule of the double tax treaty between State A and State B would probably determine that John Doe is a resident of State B, with the result being, once again, that the levying of a brain drain tax by State A would not be in accordance with the double tax treaty.

If, on the other hand, John Doe were regarded as a resident of State A, the brain drain tax would be compatible with the double tax treaty. There are two possibilities for John Doe to be viewed as a resident of State A: (i) the analysis of the facts and circumstances of the case determine that he should be deemed a resident of State A, which, as demonstrated above, is not the most likely outcome; and (ii) the residence conflict is not solved by recourse to Article 4(2) and John Doe is still

considered a resident of State A and State B for domestic law and treaty purposes.

In the latter case, considering that, as demonstrated above, the articles of the conventions regarding business profits, independent personal services, employment income, fees for technical services and other income favor taxation by the residence state of the income earner, without establishing any restriction on taxation of these types of income by the residence state of the individual, State A would have no problem to levy its brain drain tax. This is corroborated by Article 1(3), the savings clause, recently introduced into the UN Model Tax Convention. But even if the double tax treaty between State A and State B does not contain such provision, State A would still be allowed to levy the brain drain tax, since the treaty would not prohibit such a tax.

Therefore, as it has been argued throughout this article, the possibilities for a brain drain tax to be compatible with double tax treaties are slim, and it cannot be considered that such tax would ultimately be widely used or generate significant revenue for the tax authorities of developing countries.

4. TAXES AS INCENTIVES TO AVOID BRAIN DRAIN

Apart from the institution of a brain drain tax, a state can also reduce the brain drain risk by adopting positive measures, ones that provide incentives for taxpayers to stay or to move to a country. In this section, we will study both possibilities and analyze whether they would be in line with the double tax conventions.

4.1. Taxes as Incentives to Retain Individuals

Just as taxes can act as a deterrent to the migration of individuals, they can also be used to provide individuals a better off situation, by means of credits or a reduced tax burden. One of the approaches is to discard the income tax levied on the population that might migrate, as done by Poland and studied further in another article of this journal, as well as by certain states in the United States, such as Mississippi.

In this sense, the government provides a tax exemption for certain categories of taxpayers, be it young taxpayers as in Poland and the United States, or individuals that earn a certain amount of income, if they remain living in the state. This is a trade-off in which the state ultimately ends up giving up on the tax income that could be collected in exchange for the maintenance of these individuals that are believed to help the state's economy of the state in the long term.

As regards the compatibility of such measures with the double tax conventions signed by states, there is no question that, considering that the benefit is granted to a resident of a contracting state, the situation would be under the scope of such convention as prescribed in Article 1(1) coupled with articles 2(1), 3(1)(a) and 4(1). Furthermore, the tax exemption would not infringe the double tax treaty, as treaties allocate taxing rights between states, they do not stipulate that a state is obliged to tax the income, irrespective of the type of income earned (business profits, independent personal services, employment income, other income). Moreover, it is worth noting that the articles studied above do not put any restriction on taxation by the residence state (apart from Article 23A and B, which prescribe an exemption or credit for taxes paid abroad).

If the resident individual only earned income in their residence state the situation would be even more clear-cut, as in that case this would be the sole state entitled to tax the income.

The fact that the state would not be restricted in its right to tax the resident individual does not mean that there would be no risk that the double tax treaties signed by the states could lead to questions on the tax treatment granted to these individuals. However, in that case the focus would not be on the distributive rules, but rather on the non-discrimination rule contained in Article 24(1). As provided on Article 24(1), "Nationals of a Contracting State shall not be subjected in the other Contracting State to any taxation or any requirement connected therewith which is other or more burdensome than the taxation and connected requirements to which nationals of that other State in the same circumstances, in particular with respect to residence, are or may be subjected. This provision shall, notwithstanding the provisions of Article 1, also apply to persons who are not residents of one or both of the Contracting States."

Considering the example of John Doe, if instead of emigrating to State B he would remain a resident of State A and receive a tax exemption on his income, the nationals of State B that are residents in State A could request the same tax treatment of John Doe on the basis of Article 24(1) of the double tax treaty between State A and State B. Thus, to comply with its treaty obligations, State A might have to extend this beneficial tax treatment to all nationals of the states with which it has signed double tax treaties, leading to a greater loss of tax revenue, i.e. the measure might be counterproductive, since one of the concerns about brain drain was precisely the loss of tax revenue.

There is another issue that might arise in the establishment of preferential tax treatment for certain taxpayers as means to avoid brain drain: inequity in the domestic sphere, as usually benefits to avoid brain drain are granted to individuals who are already better off than the average resident. Despite the importance of this matter, we will not delve further

into its analysis since the focus of this paper is the compatibility of tax measures aimed at avoiding brain drain through double tax treaties, and this issue is a purely domestic one.

There is no issue regarding the compatibility of domestic tax exemptions with double tax treaties, as demonstrated above, save that the state that provides for the exemption might have, based on the non-discrimination clause normally contained in double tax treaties, to extend the exemption to nationals of all states with which it has signed double tax treaties. In that sense, it can already be ascertained that the provision of tax exemptions is a sounder approach to dealing with the brain drain than the institution of a brain drain tax, at least in regard to the state fulfilling its treaty obligations. Nonetheless, the adoption of a tax exemption and extension to all nationals of treaty partners resident in the state might be more detrimental to the state's finances than the restriction imposed by double tax treaties on the establishment of brain drain taxes, so states should consider this carefully.

4.2. Tax as Incentives to Attract Individuals

In addition to providing tax benefits for individuals who opt to remain in the state, it is also possible to offer a tax incentive to attract individuals to move to the state, an option which is becoming increasingly common. As an example of such a measure, the Netherlands has a 30% ruling for highly-skilled labor hired by Dutch employers. According to this system, individuals who move to the Netherlands for their work and did not live within 150kms of the border for at least 16 out of the 24 months prior to moving, are entitled to a tax break on 30% of their income, i.e. 30% of the income will not be taxed for a period of 5 years, which can be renewed once.

Considering the example of John Doe, if such benefits were in place in State B, instead of being taxed by State A after his emigration or receiving a benefit to remain a resident of State A, he would receive a tax break granted by State B, as long as he moved to this state. Thus, once again, the issue would not be whether the emigration state could tax the income he earned after emigration, but rather whether State B "poaching" John Doe would be in line with the double tax treaties based on the OECD and the UN model tax conventions.

Like the incentives provided for individuals to stay in a country, tax breaks to attract individuals do not conflict with double tax treaty rules³⁴ as they merely determine that a state will give up taxing part of the income of certain individuals. In that sense, considering that the state

Wet op de loonbelasting [Wage Tax Act] 1964.

For more on the matter see section 4.1.

is not restricted by a double tax treaty in its taxing rights over this individual, it is still up to the state to determine in its domestic legislation whether the individual will effectively be taxed, so there is no conflict.³⁵ If the double tax treaty prescribes that only the other state can tax the individual, the tax break at hand would also be compatible (on the part that is exempt, as taxation of the remaining taxable income would indeed be restricted in such a situation), because it would also stipulate that no taxes would be levied on the individual.

This conclusion is in line with the fact that double tax treaties allocate taxable income, since it is still up to the person's residence state to determine how it will exercise its taxing rights, i.e. the double tax treaty does not stipulate the manner in which the income should be taxed, but only whether it may be taxed or not. The entire remaining taxation framework stems from domestic rules.

Nonetheless, like tax incentives to maintain individuals in a state, tax incentives to attract individuals may also need to be extended to nationals of a contracting state who decide to take up residence in the state providing the benefit, based on the non-discrimination clause of a double tax treaty (Article 24(1)). Additionally, this measure may generate discussions, from a domestic law perspective, regarding its equity.

The extension of the benefit to foreign nationals is not an issue per se, as these attraction systems, like the Dutch 30% ruling, do not consider the nationality of the person as a distinctive criterion for the receipt of the benefits. The issue concerning the equity of the system from a domestic law perspective, on the other hand, might generate considerable discussions, as the persons that can normally benefit from these attractions schemes are normally highly-skilled individuals, who already earn more than the average resident taxpayer. As mentioned in the previous section, this issue, although extremely important, is merely mentioned as a point of attention because it does not affect the compatibility of the scheme with double tax conventions, and is beyond the scope of this paper.

Overall, it is interesting that the provision of tax benefits is compatible with double tax treaties, but it does not resolve the problem that countries face when opting for a brain drain tax, i.e. it does not burb the loss of tax revenue. On the contrary, the provision of tax breaks for individuals who remain residents or who become residents may lead to an even higher loss of tax revenue than the one caused by the brain drain, as these breaks are also applied to persons who would already remain

Although these measures do not conflict with tax treaties, one can question the behavior of developed countries that adopt such measures, as they can lead to the increasse in brain drain from other states, as argued by Assaf Razin (2017, 13–15). Altbach (2013, 41) is more contentious, stating that developed countries are robbing developing countries of their brains and this will significantly damage the latter.

residents or would move to the state, i.e. these persons' behavior is not affected by the tax break, as intended by the legislators, but they still benefit from the tax incentive.

Therefore, considering the revenue-collecting goal of the brain drain taxes, it is important to assess how these taxes can become compatible with double tax treaties.

5. ESTABLISHING THE COMPATIBILITY OF BRAIN DRAIN TAXES WITH DOUBLE TAX TREATIES

As it has been presented in this paper, the adoption of a brain drain tax presents a challenge to states that have signed double tax treaties, as such a tax cannot be applied in a treaty setting save in specific situations: (i) residence-residence conflict solved in favor of the emigration state, or (ii) unresolved residence-residence conflict. Furthermore, the network effect of tax treaties, normally seen as a positive trait (as it is generally accepted that the more tax treaties a country has signed, the better), leads to greater restriction on the adoption of a brain drain tax in a treaty situation, which is certainly not the idea that countries have when instating brain drain taxes. Thus, the question remains how to make brain drain taxes compatible with double tax treaties.

On that matter, the issue can be tackled by: (i) substituting the residence criterion with the citizenship criterion; (ii) amending the residence article; or (iii) modifying the treaties' distributive rules to allow the emigration state to tax future earnings of the emigrant.

5.1. Substituting the Residence Criterion with the Citizenship Criterion

As mentioned before, double tax treaties modeled after the OECD and the UN model tax conventions are applicable to persons who are residents of one or both contracting states, as prescribed by Article 1(1). In this sense, establishing the residence of the person becomes paramount in determining whether the person falls within the scope of the double tax treaty. A person that decides to emigrate will normally not remain a resident of the emigration state, and even if that occurs, the person will most likely also be a resident of the state to which they have moved and recourse to the tiebreaker rule would probably establish that, for treaty purposes, the person is a resident of the latter state. As a result, considering the distributive rules of the treaty, the emigration state would not be allowed to levy a brain drain tax on this individual.

Considering that a brain drain tax should apply also when the state has a double tax treaty in place, it would be possible to establish, like the United States currently does, that the double tax treaty does not affect the taxation of its citizens by a contracting state.³⁶ The focus on the citizenship status of the individual, rather than only on the individual's residence, would solve any incompatibility between brain drain taxes and the tax convention at hand.

Although the adoption of citizenship-based taxation is possible, one should remind that currently the United States is the only country of the world that taxes on the basis of citizenship, so the adoption of this criterion would entail not only a significant amendment of double tax treaties but also of state's domestic legislation. Furthermore, there is a great risk that developing countries, which are believed to suffer more from the brain drain and would thus be more eager to establish a brain drain tax, would not be able to apply and enforce such a measure. The failed experience of the Philippines related to the adoption of a citizenship criterion for taxation shows that the solution may not be as straightforward as it seems.

Therefore, even though such an amendment would resolve the incompatibility issue between brain drain taxes and double tax treaties, it does not seem to be a feasible option for most states, especially the least developed ones that suffer greatly from brain drain and are in the weaker negotiating position when signing double tax treaties.

5.2. Amendments to the Residence Article

If the shift to taxation based on the citizenship criterion is considered unattainable or politically unfeasible, states still have the option of allowing for the establishing of brain drain taxes by means of amendments to the residence article of their tax treaties. On that matter, a sentence could be added to Article 4(1) determining that citizens are deemed to be perpetual residents of the state to which they are attached or that in case of emigration from one state to another the former would always be deemed to be the residence state of the emigrating person. Naturally, as an individual may live in various places during their lifetime it would be important to establish under what conditions the emigration state would still maintain taxing rights over the emigrating person, and it seems that combining such a rule with the citizenship criterion from the previous section may be a good solution.

Irrespective of which option is chosen, in both cases the amendment of the residence provision would guarantee that the emigration state would not run foul of its treaty obligations by establishing a brain drain tax.

Despite the potential desirability of such proposals, especially if it is considered that states should indeed levy brain drain taxes, it remains

³⁶ United States, United States Model Income Tax Convention, Article 1(4).

clear that they are still a considerable deviation from the current rules prescribed in the OECD and the UN model tax conventions, and that a similar outcome can be achieved by means of a less harsh change. Considering that brain drain taxes are not in line with double tax treaties because the emigration state is not the residence state of the emigrating person, at least for treaty purposes, and that as an exception these taxes are compatible with double tax treaties when a residence-residence conflict is solved in favor of the emigration state, the alignment of brain drain taxes to double tax treaties can be done by means of a reshuffling of the tie-breaker rules.³⁷

As explained above, even in situations in which the emigration state is still viewed, for domestic law purposes, as the residence state of the emigrating individual, its taxing rights would be severely limited by double tax treaties because the tiebreaker rule would most likely be resolved in favor of the individual's new residence state. This is because the order in which the tiebreaker test is set focuses on the individual's physical link to a state (permanent home), economic and personal interests (center of vital interests), physical presence in the state (habitual abode), and ultimately the personal attachment of the individual to the state (nationality), thus the personal attachment to a state is only taken into consideration as a last resort.

However, if the tiebreaker rule were modified and the nationality test were the first criterion to be assessed, the changes that the emigration state would be able to tax its former resident would significantly increase. Note that differently from the switch to taxation based on citizenship or on deeming the emigration state as the perpetual residence of the individual, this would be a less troublesome change, as the nationality criteria is already present in the tie-breaker rule. This amendment would merely bring this criterion to the forefront of the residence-residence conflict for individuals, without establishing any different threshold for taxation. In the event that this criterion is not met, the remaining factors would still be assessed to determine where the individual is resident for treaty purposes.

5.3. New Distributive Rule in Case of Emigration

When analyzing the compatibility of a brain drain tax with the distributive rules of the OECD and the UN model tax conventions, more specifically articles 7, 12A, 14, 15 and 21, it became clear that all provisions allow for taxation of the income effectively earned by the person, giving preference to residence taxation of the income and, if certain conditions are met, allowing for source taxation too. However, a

³⁷ As suggested by Brauner (2010, 250), proposing to consider the individual's center of vital interests before the permanent home criterion.

brain drain tax would be levied on income earned after emigration, i.e. when the taxpayer is already a resident of another state. Consequently, based on the provisions of the model conventions, the emigration state would not have any taxing rights over this income, unless it was sourced therein.

If the option to align brain drain taxes with double tax treaties is chosen, it is necessary to modify the current structure of treaties and prescribe taxing rights to the emigration state over income that is earned after emigration. This could be achieved by adding a paragraph to each article stipulating a specific treatment for emigrating individuals, e.g. Article 7 would also prescribe that profits of an enterprise of an emigrating individual, whenever arising, may also be taxed in the emigration state, irrespective of whether the enterprise carries on business in the emigration state through a permanent establishment. Or a new article could be added to allocate taxing rights relating to emigrating individuals, with emigration serving as the distinctive criterion for special tax treatment, in the same manner as income director's fees are dealt with in Article 16, and income from artistes and sportspersons in Article 17. The allocation rule would then prescribe that income earned by an emigrating individual may also be taxed by the emigration state.

6. JUSTIFICATION FOR THE AMENDMENT OF DOUBLE TAX TREATIES TO MAKE THEM COMPATIBLE WITH DOMESTIC BRAIN DRAIN TAXES

Having established the possible amendments that would make brain drain taxes compatible with double tax conventions, therefore fulfilling the objective of guaranteeing that in case the emigration state opted for a brain drain tax in its domestic law this tax would not be hindered by a double tax treaty, it is also necessary to consider how to justify these changes. As examined previously in this paper, ³⁹ brain drain taxes are normally justified as a means to combat tax revenue losses stemming from the emigration of highly-skilled individuals. However, this is not enough per se to justify the redrafting of the allocation rules, as the same reason can be used by the individual's new residence state to argue that granting taxing rights to the emigration state is the equivalent of restricting their taxing rights over individuals that are residents in their states.

On that matter, considering that brain drain taxes are normally linked to developing countries, which are normally the ones that suffer

 $^{^{38}}$ As the 2017 UN Model Tax Convention has recently done regarding fees for technical services.

³⁹ See section 2.

considerably brain drain (Kuehn 2007, 1853–1855; Patterson 2007, 9), it can be argued that the amendments should be made in order to align the double tax treaties with the right to development (Souza de Man 2017; Silva 2009; Sengupta 2000).⁴⁰

The right to development is a human right recognized by the United Nations as a right of individuals and states to participate in, contribute to and benefit from economic, social, cultural and political development. It is recognized that states should have their right to development respected by other states (Souza de Man 2017, 25) and that more developed countries have the responsibility to support the development of their residents as well as residents of other states (Souza de Man 2017, 25). The Declaration on the Right to Development did not bind the states, not even the ones that adopted it, so it might be argued that the right to development is not actually a right in the strict sense.

Nonetheless, since the Declaration on the Right to Development was adopted more than 30 years ago and that the United Nations has repeatedly stated that the right to development is a human right (UN, 2004, E/CN.4/Sub.2/2004/17; UN, 2001, E/CN.4/2001/WG.18/2 UN, 2000, A/RES/55/2;) with consensus being achieved in the Vienna Declaration and Programme of Action in 1993 (World Conference on Human Rights 1993), it can be argued that this right is already part of customary international law (Mansell, Scott 1994, 174; Villaroman 2001, 8; Have 2013, 3; Kunanayakam 2013, 48) and can indeed be counted on by developing countries in achieving their goal of development through the collecting of taxes (Souza de Man 2017, 31–32).

As a result, states could be considered to have the obligation to facilitate the development of all individuals and states, so developed country should assist developing countries. In fact, such assistance already exists by means of development aids, but this could also be done by allowing developing countries that are suffering brain drain to levy brain drain taxes also in treaty situations. For this to happen, the amendment of the double tax treaty is of utmost importance, as seen above.

Thus, bearing in mind that the right to development of states can be further fostered by the collection of income, if can be affirmed that by agreeing to amendments to double tax treaties with developing countries, to make the levying of brain drain taxes also possible in treaty situations, the developed countries would be respecting the right to development.

⁴⁰ The right to development has also been thoroughly discussed at the UN level (UN, 1999, E/CN.4/1999/WG.18/2).

⁴¹ United Nations, General Assembly, Declaration on the Right to Development, A/RES/41/128, Article 1.

The Declaration on the Right to Development was adopted by 148 countries, with 8 abstentions and one objection (United States).

Moreover, the decision to allow for further collection of income by developing countries is in line with the Millennium Development Goals and the Sustainable Development Goals, as the collected income can be used to further foster development. Finally, if treaty partners believe that the collection of taxes over emigrants would yield considerable funds, they could negotiate a proportional reduction of the aid possibly granted to the developing country. In this scenario the developed countries would be respecting the developing country's right to development, as well as their self-determination on how to collect income and to use it for their development.

The right to development can, thus, provide a conceptual framework for justifying the amendment of double tax treaties necessary to allow for the levying of brain drain taxes by developing countries.

7. CONCLUSION

Since Bhagwati proposed the establishment of a brain drain tax to curb the emigration of highly-skilled individuals to more developed countries in the 1970s, the idea of how tax measures can curb or foster brain drain/gain has been subject to considerable scrutiny in academic circles. Those that side with Bhagwati normally focus on the fairness of a brain drain tax and on the benefits that the collected income could have for the budget of the emigration state, while its detractors point that the emigration of individuals to more developed countries would also be beneficial for the individual's home state. Others focus on providing benefits for individuals to remain put or to attract residents of other states, favoring a preferential tax treatment instead of the levying of a tax as a means of combating brain drain.

Irrespective of the position taken, the focus has mainly been on the benefits and problems of the ideas for each individual state, with almost no attention being paid to its compatibility with the obligations assumed in the signing of double tax treaties. In this article we delve precisely into this issue, to assess whether tax measures focused on combating brain drain can be applied when the states involved have signed a double tax treaty based on the OECD or the UN model tax conventions. The study of this issue has shown that even though in their domestic laws states may resort to brain drain taxes to collect further income from the emigrating person after emigration, this behavior is not in line with the model tax conventions and treaties signed by the states, save in specific circumstances.

The provision of a preferential tax treatment, on the other hand, could deter brain drain without conflicting with double tax treaties, but it would not curb the loss of tax revenue from highly-skilled individuals

and it could lead to questions regarding the equity of the measure. Having the revenue collection goal in mind, if states want to levy a brain drain tax in situations involving treaty partners, it is paramount that they modify their tax treaties. In this article, we present three possible amendments to double tax treaties that could serve to maintain the tax collection rights of the emigration state over the income of the emigrating person, as well as a justification to support these amendments – the right to development.

In brief, this article shows that if states opt to levy a brain drain tax in their domestic legislation, this tax will not serve the purpose of collecting taxes over the income earned, after emigration, by emigrants who move to a state that has a double tax treaty with the emigration state, unless states also modify their tax treaties. Considering that developing countries are the greatest victims of brain drain, and thus more likely to want to introduce such a tax, the necessary amendments to the treaties can be made based on the right to development, a human right duly recognized by states and by the United Nations.

REFERENCES

- Agrawal, Ajay, Devesh Kapur, John McHale. 2008. Brain Drain or Brain Bank? The Impact of Skilled Emigration on Poor Country Innovation. Working Paper No. 14592. National Bureau of Economic Research, Cambridge, Mass.
- Altman, Philip G. 2013. Brain Drain or Brain Exchange?. 41–45 in *The International Imperative in Higher Education*, edited by Philip G. Altbach. Rotterdam: Sense Publishers.
- Bhagwati, Jagdish, Dellalfar William. 1972. The Brain Drain and Income Taxation: A Proposal. Working Paper No. 92. Massachusetts Institute of Technology, Cambridge, Mass.
- Bhagwati, Jagdish, Dellalfar William. 1973. The Brain Drain and Income Taxation. *World Development*, 1(2): 94–101.
- Brauner, Yariv. Fall 2010. Brain Drain Taxation as Development Policy, *Saint Louis University Law Journal* 55: 221–268.
- Brock, Gillian, Michael Blake. 2015. Debating Brain Drain: May Governments Restrict Emigration?. New York, New York: Oxford University Press.
- Commander, Simon, Mari Kangasniemi, L. Alan Winters. 2004. The Brain Drain: Curse or Boon? A Survey of the Literature. 235–272 in *Challenges to Globalization: Analyzing the Economics*, edited by Robert E. Baldwin and L. Alan Winters. Chicago, Ill.: University of Chicago Press.

- Docquier, Frédéric, Olivier Lohest, Abdeslam Marfouk, 2/2007. Brain Drain in Developing Countries, *The World Bank Economic Review* 21: 193–218.
- Dumitru, Speranta. 1/2012. Skilled Migration: Who Should Pay for What: A Critique of the Bhagwati Tax. *Diversities* 14: 9–23.
- Freitas, Any, Antonina Levatino, Antoine Pécoud. 1/2012. Introduction: New Perspectives on Skilled Migration. *Diversities* 14: 1–7
- Haupt, Alexander, Eckhard Janeba. 2004. Education, Redistribution and the Threat of Brain Drain, Working Paper No. 10618. National Bureau of Economic Research, Cambridge, Mass.
- Hewitt, Cynthia Lucas. 2007. Pan-African Brain Circulation. 15–39 in *African Brain Circulation: Beyond the Drain-Gain Debate*, edited by Rubin Patterson. Leiden: Brill.
- Kuehn, Bridget M. 16/2007. Global Shortage of Health Workers, Brain Drain Stress Developing Countries. *Journal of American Medical Association* 298: 1853–1855.
- Kumar, Rajendra. 47/1967. Brain Drain or Brain Gain?. *Economic and Political Weekly* 2: 2079.
- Kunanayakam, Tamara. 2013. The Declaration on the Right to Development in the context of United Nations standard setting. 17–48 in Realizing the Right to Development: Essays in Commemoration of 25 Years of the United Nations Declaration on the Right to Development, New York/Geneva: United Nations Publication.
- Lien, Donald, Yan Wang. 1/2005. Brain drain or brain gain: A revisit. *Journal of Population Economics* 18: 153–163.
- Lister, Matthew J. 1/2017. A Tax-Credit Approach to Addressing Brain Drain, Saint Louis University Law Journal 62: 73–84.
- Mansell, Wade, Joanne Scott. 2/1994. Why Bother about a Right to Development?. *Journal of Law and Society* 21: 171–192.
- McAusland, Carol, Peter J. Kuhn. 2006. The International Migration of Knowledge Workers: When is Brain Drain Beneficial?. Working Paper No. 12761. National Bureau of Economic Research, Cambridge, Mass.
- OECD. 2008. The Global Competition for Talent: Mobility of the Highly Skilled. Paris. France.
- OECD. 2017. Model Tax Convention on Income and on Capital. Paris, France: OECD.
- Oldman, Oliver, Richard Pomp. 10/1975. The Brain Drain: A Tax Analysis of the Bhagwati Proposal. *World Development* 3: 751–763.

- Patterson, Rubin. 2007. Introduction. 1–14 in *African Brain Circulation: Beyond the Drain-Gain Debate*, edited by Rubin Patterson. Leiden: Brill.
- Razin, Assaf. 2017. Globalization Policies and Israel's Brain Drain. Working Paper No. 23251. National Bureau of Economic Research, Cambridge, Mass.
- Sager, Alex. 5/2014. Reframing the Brain Drain, Critical Review of International Social and Political Philosophy 17: 560–579.
- Sampson, Thomas. 1/2013. Brain Drain or Brain Gain? Technology Diffusion and Learning On-the-job, *Journal of International Economics* 90: 162–176.
- Scalera, Domenico. 4/2012. Skilled Migration and Education Policies: Is There Still Scope for a Bhagwati Tax?. *The Manchester School* 80: 447–467.
- Sengupta, Arjun. 4/2002. On the Theory and Practice of the Right to Development. *Human Rights Quarterly* 24: 837–889.
- Silva, Mauro. 2009. Da Competição à Cooperação Tributária Internacional: Aspectos Jurídicos da Promoção do Desenvolvimento Nacional num Cenário Internacionalizado [From competition to international tax cooperation: legal aspects of promoting national development in an internationalized scenario]. Unpublished PhD thesis. University of São Paulo, Faculty of Law.
- Souza de Man, Fernando. 2017. Taxation of Services in Treaties between Developed and Developing Countries: A Proposal for New Guidelines. Amsterdam, The Netherlands: IBFD Doctoral Series Vol. 39.
- Stevenson, Ariel. 1/2016. Recovering Lost Tax Revenue through Taxation of Transnational Households, *Berkeley Journal of International Law* 34: 100–156.
- United Nations. Department of Economic & Social Affairs. 2017. United Nations Model Double Taxation Convention between Developed and Developing Countries: 2017 Update. New York, United States.
- United Nations. Economic and Social Council. Commission on Human Rights. 2004. Mainstreaming the right to development into international trade law and policy at the World Trade Organization. Note by the Secretariat. E/CN.4/Sub.2/2004/17.
- United Nations. Economic and Social Council. Commission on Human Rights. 2001. Third Report of the independent expert on the right to development, Mr. Arjun Sengupta, submitted in accordance with Commission resolution 2000/5. E/CN.4/2001/WG.18/2.
- United Nations. 2000. General Assembly. United Nations Millennium Declaration. A/RES/55/2.

- United Nations. Economic and Social Council. Commission on Human Rights. 1999. Study on the current state of progress in the implementation of the right to development submitted by Mr. Arjun L. Sengupta independent expert, pursuant to Commission resolution 1998/72 and General Assembly Resolution 53/155. E/CN.4/1999/WG.18/2.
- United Nations. World Conference on Human Rights. 1993. Vienna Declaration and Programme of Action.
- United Nations. 1986. General Assembly. Declaration on the Right to Development. A/RES/41/128.
- United States. United States Treasury Department. 2016. United States Income Tax Convention. Washington, D.C.
- van der Have, Nienke Suzanne. 2013. The Right to Development and State Responsibility Can States be Held to Account?, Working Paper No. 2013–23. Amsterdam Center for International Law, Amsterdam, The Netherlands.
- Villaroman, Noel G. 1/2011. Rescuing a Troubled Concept: An Alternative View of the Right to Development. *Netherlands Quarterly of Human Rights* 29: 13–53.
- Wilson, John Douglas. 1/2011. Brain-drain taxes for non-benevolent governments. *Journal of Development Economics* 95: 68–76.
- Wilson, John Douglas. 12/2008. A voluntary brain-drain tax. *Journal of Public Economics* 92: 2385–2391.
- Wilson, John Douglas. 2005. Taxing the Brain Drain: A Reassessment of the Bhagwati Proposal. Paper presented at International Trade and Factor Mobility: Theory and Policy. Columbia University, New York, NY.

Article history: Received: 31. 10. 2019. Accepted: 2. 12. 2019.

GUIDELINES FOR AUTHORS

The BLR languages are English and Serbian.

For papers in English, either British English or American English spelling and terminology should be used, but the one should be followed consistently throughout the paper.

Spelling in references should follow the original source material.

If English is not your first language, before submitting your manuscript you may wish to have it proofread, to ensure that reviewers fully understand your writing. Authors are liable for the costs of such proofreading.

Authors will retain the copyright in their work. It is, however, a condition of publication that authors transfer to the *BLR* (the University of Belgrade Faculty of Law) the non-exclusive right to publish the paper, to be cited as its original publisher in case of reuse, to reproduce, to make available to the public, and to distribute it in all forms and media. The published papers will be distributed under the Creative Commons Attribution 4.0 International (CC BY 4.0) license.

By submitting the paper, the authors declare that the paper has not been published; that the paper has not been accepted for publication nor is under consideration for publication elsewhere; and that the authors are the copyright holders and are informed of, and have fulfilled, any requirements regarding the rights of third parties.

The receipt of all papers will be acknowledged by email. All papers will be screened by the Editorial Board for their suitability for entering the peer review process. Suitable contributions will be subject to a double-blind external peer review.

For further information on *BLR* Editorial Policies, please consult http://ojs.ius.bg.ac.rs/index.php/anali/about/editorialPolicies#openAcces sPolicy

Please abide by the following instructions when submitting a paper to the BLR.

The manuscript should be arranged in the following order:

- 1. Cover page
- 2. Abstract and keywords
- 3. Manuscript and reference list
- 4. Appendices, tables and figures

1. COVER PAGE

The cover page of the submission should include the following information:

- Title of the paper.
- Name, date of birth and affiliation of all contributing authors.
- Full address for correspondence and email address. If the paper is co-authored, please provide the requested information for each of the authors.

2. ABSTRACT AND KEYWORDS

Please provide an abstract that is not longer than 150 words. The abstract should not contain undefined abbreviations or unspecified references.

Please provide five keywords, in alphabetical order, suitable for indexing.

Papers in Serbian should contain an abstract and keywords both in English and in Serbian language, with the abstract and keywords in English positioned following the reference list.

3. MANUSCRIPT AND REFERENCE LIST

As our peer-review process is blind, the names of authors or their affiliations should NOT appear anywhere on the pages of the manuscript.

Papers should be word processed in the following format:

Page format: A4 Margins: 2.5 cm

Font: Times New Roman Line spacing main text: 1.5 Line spacing Footnotes: Easy

Font size main text: 12 pt Font size footnotes: 10 pt

Page numbering: Plain number, bottom of the page, right indent

Refer to persons by first and last name at first mention (John Smith) and only by last name thereafter (Smith). Do not use "Professor", "Mr.", or other titles.

Figures and tables should be mentioned in the text in order of appearance. All figures and tables should be mentioned in the text.

All acronyms and abbreviations should be spelled out in the first reference, with capitals used thereafter.

European Union – EU,

The United Nations Commission on International Trade Law – UNCITRAL

Numbers from one to nine should be written as words; higher numbers should be written as numerals. Dates should be written as follows: 1 January 2012; 2011–2012; or the 1990s.

Footnotes should be substantive and cannot contain purely bibliographic material. Simple citations should be in the text, with the exception of statutes and cases.

BLR uses the following section headings, all centered:

- 1. FULL CAPS
- 1.1. Caps and Lowercase
- 1.1.1. Italic Caps and Lowercase

Citations

All citations, in text and footnotes, should be in author/year/page(s) style.

The preferred form of in-text citation is the citation with locating information: Following Hovenkamp (1994, 366–69);

According to Craswell (2003, 255 n. 13) – where note 13 is on page 255;

As suggested by Craswell (2003, 254 and n. 11) – where note 11 is **not** on page 254.

However, when appropriate and by way of exception, the authors may use in-text citation **without locating information**, with or without a simple signal (*see*, *see especially*, *see for example*, etc.)

(see, for example, Corcoran 2004; Mullen 2000) (see especially Demsetz 1967) (Scott and Coustalin 1995)

One author

T(ext): Following Ely (1980, page), we argue that

R(eference list): Ely, John Hart. 1980. *Democracy and Distrust: A Theory of Judicial Review.* Cambridge, Mass.: Harvard University Press.

Two authors

- T: As demonstrated elsewhere (Daniels, Martin 1995, page),
- R: Daniels, Stephen, Joanne Martin. 1995. Civil Injuries and the Politics of Reform. Evanston, Ill.: Northwestern University Press.

Three authors

- T: As suggested by Cecil, Lind, Bermant (1987, page),
- R: Cecil, Joe S., E. Allan Lind, Gordon Bermant. 1987. Jury Service in Lengthy Civil Trials. Washington, D.C.: Federal Judicial Center.

More than three authors

- T: Following the research design in Turner et al. (2002, page),
- R: Turner, Charles F., Susan M. Rogers, Heather G. Miller, William C. Miller, James N. Gribble, James R. Chromy, Peter A. Leone, Phillip C. Cooley, Thomas C. Quinn, Jonathan M. Zenilman. 2002. Untreated Gonococcal and Chlamydial Infection in a Probability Sample of Adults. *Journal of the American Medical Association* 287: 726–733.

Institutional author

- T: (U.S. Department of Justice 1992, page)
- R: U.S. Department of Justice. Office of Justice Programs. Bureau of Justice Statistics. 1992. *Civil Justice Survey of State Courts*. Washington, D.C.: U.S. Government Printing Office.

No author

T: (Journal of the Assembly 1822, page).

R: Journal of the Assembly of the State of New York at Their Forty-Fifth Session, Begun and Held at the Capitol, in the City of Albany, the First Day of January, 1822. 1822. Albany: Cantine & Leake.

More than one work

Clermont, Eisenberg (1992, page; 1998, page)

More than one work in a year

- *T*: (White 1991a, page)
- R: White, James A. 1991a. Shareholder-Rights Movement Sways a Number of Big Companies. Wall Street Journal, April 4.

Multiple authors and works

(Grogger 1991, page; Witte 1980, page; Levitt 1997, page)

Chapter in a book

- T: Holmes (1988 page) argues that
- R: Holmes, Stephen. 1988. Precommitment and the Paradox of Democracy. 195–240 in *Constitutionalism and Democracy*, edited by John Elster and Rune Slagstad. Cambridge: Cambridge University Press.

Chapter in a multivolume work

- T: Schwartz, Sykes (1998) differ from this view
- R: Schwartz, Warren F., Alan O. Sykes. 1998. Most-Favoured-Nation Obligations in International Trade. 660–64 in vol. 2 of *The New Palgrave Dictionary of Economics and the Law*, edited by Peter Newman. London: MacMillan.

Edition

- T: Using the method of Greene (1997), we constructed a model to show
- R: Greene, William H. 1997. Econometric Analysis. 3d ed. Upper Saddle River, N.J.: Prentice Hall.

Reprint

- T: (Angell, Ames [1832] 1972, 24)
- R: Angell, Joseph Kinniaut, Samuel Ames. [1832] 1972. A Treatise on the Law of Private Corporations Aggregate. Reprint, New York: Arno Press.

Journal article

In the list of references, journal articles should be cited in the following manner: surname and name of the author, number and year of the issue, title of the article, title of the journal, volume number, pages.

- T: The model used in Levine et al. (1999, page)
- R: Levine, Phillip B., Douglas Staiger, Thomas J. Kane, David J. Zimmerman. 2/1999. Roe v. Wade and American Fertility. American Journal of Public Health 89: 199–203.
 - T: According to Podlipnik (2018, page)
- R: Podlipnik, Jernej. 4/2018. The Legal Nature of the Slovenian Special Tax on Undeclared Income. Annals of the Faculty of Law in Belgrade 66: 103–113.

Entire issue of a journal

- T: The fairness or efficiency benefits of bad-faith laws are discussed at length in Texas Law Review (1994)
- R: Texas Law Review. 1994. Symposium: Law of Bad Faith in Contrast and Insurance, special issue. 72: 1203–1702.

Commentary

- T: Smith (1983, page) argues that
- *R:* Smith, John. 1983. Article 175. Unjust Enrichment. 195–240 in *Commentary to the Law on Obligations*, edited by Jane Foster. Cambridge: Cambridge University Press.
 - T: Schmalenbach (2018, page) argues that
- R: Schmalenbach, Kirsten. 2018. Article 2. Use of Terms. 29–55 in Vienna Convention on the Law of Treaties: A Commentary, edited by Oliver Dörr, Kirsten Schmalenbach. Berlin: Springer-Verlag GmbH Germany.

Magazine or newspaper article with no author

- T: had appeared in Newsweek (2000).
- R: Newsweek. 2000. MP3.com Gets Ripped. 18 September.

Magazine or newspaper article with author(s)

- T: (Mathews, DeBaise 2000)
- *R*: Mathews, Anna Wilde, Colleen DeBaise. 2000. MP3.com Deal Ends Lawsuit on Copyrights. *Wall Street Journal*, 11 November.

Unpublished manuscript

- T: (Daughety, Reinganum 2002)
- R: Daughety, Andrew F., and Jennifer F. Reinganum. 2002. Exploiting Future Settlements: A Signaling Model of Most-Favored-Nation Clauses in Settlement Bargaining. Unpublished manuscript. Vanderbilt University, Department of Economics, August.

Working paper

- T: (Eisenberg, Wells 2002)
- R: Eisenberg, Theodore, Martin T. Wells. 2002. Trial Outcomes and Demographics: Is There a Bronx Effect? Working paper. Cornell University Law School, Ithaca, NY.

Numbered working paper

- T: (Glaeser, Sacerdote 2000)
- *R:* Glaeser, Edward L., Bruce Sacerdote. 2000. The Determinants of Punishment: Deterrence, Incapacitation and Vengeance. Working Paper No. 7676. National Bureau of Economic Research, Cambridge, Mass.

Personal correspondence/communication

- T: as asserted by Welch (1998)
- R: Welch, Thomas. 1998. Letter to author, 15 January.

Stable URL

T: According to the Intellectual Property Office (2018),

R: R.S. Intellectual Property Office. 2018. Annual Report for 2017. http://www.zis.gov.rs/about-us/annual-report.106.html (last visited 28 February 2019).

In press

- *T*: (Spier 2003, page)
- *R*: Spier, Kathryn E. 2003. The Use of Most-Favored-Nations Clauses in Settlement of Litigation. *RAND Journal of Economics*, vol. 34, in press.

Forthcoming

T: One study (Joyce, forthcoming) includes the District of Columbia *R:* Joyce, Ted. Forthcoming. Did Legalized Abortion Lower Crime? *Journal of Human Resources*.

Cases

F(ootnote): CJEU, case C-20/12, Giersch and Others, ECLI:EU:C:2013:411, para. 16; Opinion of AG Mengozzi to CJEU, case C-20/12, Giersch and Others, ECLI:EU:C:2013:411, para. 16; Supreme Court of Serbia, Rev. 1354/06, 6. September 2006., Paragraf Lex; Supreme Court of Serbia, Rev. 2331/96, 3. July 1996., *Bulletin of the Supreme Court of Serbia* 4/96, 27.

- *T*: Use abbreviated reference for in-text citations of cases (CJEU C-20/12, or Giersch and Others; Opinion of AG Mengozzi; VSS Rev. 1354/06) consistently throughout the paper.
 - R: Do not include cases in the reference list.

Legislation

- F: Regulation (EU) No. 1052/2013 establishing the European Border Surveillance System (Eurosur), OJ L 295 of 6/11/2013, Art. 2 (3); Directive 2013/32/EU on common procedures for granting and withdrawing international protection (recast), OJ L 180 of 29/6/2013, Art. 6 (3); Zakonik o krivičnom postupku [Code of Criminal Procedure], Official Gazette of the RS, 72/2011, 101/2011, 121/2012, 32/2013, 45/2013, and 55/2014, Art. 2, para. 1, it. 3.
- T: Use abbreviated reference for in-text citations of pieces of legislation (Regulation No. 1052/2013; Directive 2013/32; ZKP, or ZKP of Serbia) consistently throughout the paper.
 - R: Do not include legislation in the reference list.

4. APPENDIXES, TABLES AND FIGURES

Footnotes in appendixes should be numbered consecutively with those in the rest of the text.

Numbering of equations, tables, and figures in appendixes should begin again with 1 (Equation A1, Table A1, Figure A1, etc., for Appendix A; Equation B1, Table B1, Figure B1, etc., for Appendix B).

No more than one table should appear on a page. Tables may run more than one page.

Tables should have brief captions. All explanatory material should be provided in notes at the bottom of the table.

Identify all quantities, units of measurement, and abbreviations for all entries.

Sources should be identified in full at the bottom of each table. Do not give cross-references to footnotes elsewhere in the article.

Figures should be provided in files separate from the text and should be clearly labeled.

Do not use shading or color in graphs. If distinctions need to be made visually, please use hatching and cross hatching or another means of display. Grays are difficult to reproduce and often appear blotchy in the printed journal.

Do not use figure boxes or rules around the figures.

Please use the Times Roman font if there is any lettering or text in your figure. Type should be 7 points or larger.

Graphics files should be black and white.

Figure captions should be placed together on a separate double-spaced page, labeled Figure Legends.

Figures may be no greater than 10 cm x 18 cm. To avoid substantial figure reduction, keys to identifying items in the figure should be set within or beneath the figure.

The publication of this issue was supported by the American people, through the U.S. Agency for International Development (USAID), and the United Nations Development Programme (UNDP). Opinions expressed in this issue are those of the authors. They do not necessarily reflect the views of USAID, the U.S. Government or UNDP.

СІР – Каталогизација у публикацији Народна бибилиотека Србије, Београд 34(497.11)

АНАЛИ Правног факултета у Београду : часопис за правне и друштвене науке = Annals of the Faculty of Law in Belgrade : Belgrade Law review : Journal of Legal and Social Sciences / главни уредник Марија Караникић Мирић. – [Српско изд.]. – Год. 1, бр. 1 (1953)— . – Београд : Правни факултет, 1953— (Београд : Службени гласник). – 24 сm

Тромесечно. — Сваки 4. број у години на енгл. језику. — Drugo izdanje na drugom medijumu: Анали Правног факултета у Београду (Online) = ISSN 2406-2693. — ISSN 0003-2565 = Анали Правног факултета у Београду COBISS.SR-ID 6016514

Annals of the Faculty of Law in Belgrade (Belgrade Law Review) is the oldest Balkan academic law journal, having been founded in 1906 as Archive for Legal and Social Sciences. This law review has been published continuously in Serbian with abstracts in foreign languages. Its Editors-in-Chief have been the most prominent Serbian law professors and lawyers, such as Mihajlo Konstantinović, Milan Bartoš, Vojislav Bakić, Vojislav Simović, Obren Stanković, Dejan Popović, Miodrag Orlić, Danilo Basta, Sima Avramović, Miroljub Labus and Mirko Vasiljević. Since 2006, the law review has added one international edition annually.

The Belgrade Law Review is a peer-reviewed journal that utilizes an international body of editors, advisers and referees to select outstanding contributions. The journal solicits articles, contributions, case law and legislation comments, debates and book reviews on all aspects of law and social sciences, being interdisciplinary oriented. Special emphasis is placed on contemporary legal developments, but the journal's range includes jurisprudence and legal history. The journal serves as a forum for the expression of the legal ideas of foremost scholars, jurists, and societal leaders.

All views expressed in this law review are the opinions of the authors and not necessarily of the Editorial Board or the Editors.

The journal is available online at http://ojs.ius.bg.ac.rs/index.php/anali/

Please send submissions and questions to:

Editor-in-Chief Belgrade Law Review Bulevar Kralja Aleksandra 67 11000 Belgrade, Serbia Email: anali@ius.bg.ac.rs Phone: (381) 11-3027-700

ARTICLES IN THIS ISSUE:

Andrés BÁEZ MORENO, Hugo LÓPEZ LÓPEZ

/Brain Drain Tax vs. Brain Gain Benefits: General Thoughts from a Spanish Perspective

B. Bawono KRISTIAJI

/Tax and Brain Drain: Justification, Policy Options and Prospect for Large Developing Economies

Giorgio BERETTA

/'Fixing' the Social Contract: A Blueprint for Individual Tax Reform

Stjepan GADŽO

/Using Tax Policy to Address Brain Drain and Depopulation: The Case of Croatia

Ricardo GARCÍA ANTÓN

/Restoring the EU Citizenship for Tax Purposes

Iva IVANOV

/Tax Incentives for Keeping and Attracting Highly Skilled Workers: The Case of Serbia

Nikolai MILOGOLOV, Azamat BERBEROV

/Brain Drain and Tax Competition: Do We Need Another BEPS?

Katerina PERROU, Katerina SAVVAIDOU

/Brain Drain: The Impact on Taxation and Measures to Combat the Brain Drain

Fernando SOUZA DE MAN

/Tax Measures to Combat Brain Drain: (In)Compatibility Issues with Double Tax Conventions and a Potential Way Forward

The publication of this issue was supported by:





1SSN 0003-2565